COMMISSION AGENDA – REGULAR MEETING



June 22, 2023, AT 2:00 P.M.

MEETING WILL BE A HYBRID MEETING HELD IN PERSON AND BY TELECONFERENCE

Teleconference will be held for Vice President Loveridge, whom will be teleconferencing from Mammoth Lakes, CA.

Ontario International Airport Authority Administration Offices 1923 East Avion Street, Room 100, Ontario, CA 91761

Live YouTube Streaming for Listening Only: https://www.youtube.com/@flyont/streams

ALAN D. WAPNER President

Chief Executive Officer

ATIF ELKADI

RONALD O. LOVERIDGE Vice President

LORI D. BALLANCE General Counsel JIM W. BOWMAN Secretary JOHN M. SCHUBERT CURT HAGMAN Commissioner

JULIA GOUW Commissioner

NORMA I. ALLEY, MMC Board Clerk/Assistant Secretary

THIS MEETING WILL BE HELD VIA IN PERSON AND TELECONFERENCE

Treasurer

The Ontario International Airport Authority (OIAA) Commission Meeting is being conducted as a hybrid in person and teleconference. Members of the public may utilize alternative measures established by OIAA to view the meetings and/or to address the Commission members. Please see participation instructions below.

To dial in, call: (909) 406-8785

ANY MEMBERS OF THE PUBLIC WISHING TO PROVIDE PUBLIC COMMENT MAY DO SO AS FOLLOWS:

<u>LIVE VIA TELEPHONE</u>: You may call (909) 406-8785 the day of the meeting to speak during the meeting. Comments will be limited to 3 minutes.

<u>E-MAIL</u>: You may submit public comments by e-mail to <u>publiccomment@flyontario.com</u> no later than 4:00 p.m. the day before the meeting. Please identify the Agenda item you wish to address in your comments. All e-mail comments will be included in the meeting record.

MAIL: You may mail in public comments to the OIAA Clerk's Office at 1923 E. Avion St., Ontario, CA 91761. Comments by mail must be received no later than 12:00 p.m. the day before the meeting. Postmarks are not accepted. All written comments will be included in the meeting record.

Any members of the public who require special assistance or reasonable accommodation to participate telephonically in the meeting may contact the Board Clerk at (909) 544-5307 or clerk@flyontario.com. Sign language interpreters, communication access real-time transcription, assistive listening devices, or other auxiliary aids and/or services may be provided upon request. To ensure availability, you are advised to make your request at least 72 hours prior to the meeting you wish to attend. Due to difficulties in securing Sign Language Interpreters, five or more business days' notice is strongly recommended.

WELCOME TO A MEETING OF

THE ONTARIO INTERNATIONAL AIRPORT AUTHORITY

- All documents for public review are on file at the Ontario International Airport Administration Offices located at 1923 E. Avion Street, Ontario, CA 91761.
- This meeting is streamed live from our YouTube channel at https://www.youtube.com/@flyont/streams. Streaming will be for listening only and not participation. Public Comments will be taken by email or in-person only. This is a pilot stream, so errors and bugs may occur. If you have any issues, feel free to email clerk@flyontario.com for resolution after the meeting.
- Anyone wishing to speak during public comment, or on an agenda item, will be required to fill out a Request to Address Card (blue slip). Blue slips must be turned in prior to public comment beginning or before an agenda item is taken up. The Secretary/Assistant Secretary will not accept blue slips after that time.
- You may submit public comments by e-mail to publiccomment@flyontario.com no later than 4:00 p.m. the day before the meeting. Please identify the Agenda item you wish to address in your comments. All e-mail comments will be included in the meeting record.
- Comments will be limited to 3 minutes. Speakers will be alerted when their time is up and no further comments will be permitted. Speakers are then to return to their seats.
- In accordance with State Law, remarks during public comment are to be limited to subjects within the Authority's jurisdiction. Remarks on other agenda items will be limited to those items.
- Remarks from those seated or standing in the back of the Board Room will not be permitted. All those wishing to speak, including Commissioners and Staff, need to be recognized by the Authority President before speaking.
- Sign language interpreters, communication access real-time transcription, assistive listening devices, or other auxiliary aids and/or services may be provided upon request. To ensure availability, you are advised to make your request at least 72 hours prior to the meeting you wish to attend. Due to difficulties in securing Sign Language Interpreters, five or more business days' notice is strongly recommended. Any members of the public who require special assistance or a reasonable accommodation to participate may contact the Board Clerk at (909) 544-5307 or clerk@flyontario.com.

CALL TO ORDER (OPEN SESSION)

ROLL CALL

Loveridge, Bowman, Hagman, Gouw, President Wapner

PLEDGE OF ALLEGIANCE

AGENDA REVIEW/ANNOUNCEMENTS

The Chief Executive Officer will go over all updated materials and correspondence received after the Agenda was distributed to ensure Commissioners have received them.

1. INFORMATION RELATIVE TO POSSIBLE CONFLICT OF INTEREST

Agenda item contractors, subcontractors and agents may require member abstentions due to conflict of interests and financial interests. Commission Member abstentions shall be stated under this item for recordation on the appropriate item.

PUBLIC COMMENTS

The Public Comment portion of the Commission meeting is limited to a maximum of 3 minutes for each Public Comment. Under provisions of the Brown Act, the Commission is prohibited from taking action on oral requests.

CLOSED SESSION

PUBLIC COMMENT ON CLOSED SESSION

CLOSED SESSION

Teleconference will be held for Vice President Loveridge, whom will be teleconferencing from Mammoth Lakes, CA.

• GC § 54957, PUBLIC EMPLOYEE PERFORMANCE EVALUATION Title: Chief Executive Officer

REPORT ON CLOSED SESSION

General Legal Counsel

CONSENT CALENDAR

All matters listed under CONSENT CALENDAR will be enacted by one motion in the form listed below. There will be no separate discussion on these items prior to the time Commission votes on them, unless a member of the Commission requests a specific item be removed from the Consent Calendar for a separate vote.

Each member of the public wishing to address the Commission on items listed on the Consent Calendar will be given a total of 3 minutes.

2. APPROVAL OF MINUTES

Approve minutes for the OIAA meeting on May 25, 2023.

3. CASH DISBURSEMENT REPORT (BILLS/PAYROLL)

Receive and file the Cash Disbursement Report (Bills/Payroll) for the month ended May 31, 2023.

4. PROFESSIONAL SERVICES AGREEMENT WITH GARVER, LLC, FOR SAFETY MANAGEMENT SYSTEM PROGRAM DEVELOPMENT

Authorize the Chief Executive Officer (CEO) to execute an agreement with Garver, LLC, to provide the development and implementation of a Safety Management System (SMS) Program and manual at Ontario International Airport (ONT), in an amount not to exceed \$300,000 for the completion of the project and authorize the CEO to execute budgeted amendments to the contract for additional related services, if needed, up to 30% of the overall contract value.

5. CONTRACT WITH CS&M CONSTRUCTION SERVICES, INC., FOR GLOBAL ENTRY ENROLLMENT CENTER

Authorize the Chief Executive Officer (CEO) to execute a Contract with CS&M Construction Services Inc., in an amount \$200,000 to perform the Construction activities required for the Global Entry program office space at the Federal Inspection Services (FIS) Terminal; and authorize the CEO to execute budgeted amendments to the contract for additional related services, if needed, up to 30% of the overall contract value.

6. CONTRACT EXTENSION WITH PPM GROUP, INC., FOR CONSTRUCTION MANAGEMENT AND RELATED SERVICES

Authorize the Chief Executive Officer (CEO) to approve a contract extension to December 31, 2023, for PPM Group, Inc. for On-Call construction management services related to the Runway 26R ILS Upgrade project at Ontario International Airport (ONT).

7. PAYMENT TO INFORMA MARKETS UK FOR ROUTES WORLD EXHIBITION PARTICIPATION AND SPONSORSHIP

Authorize the Chief Executive Officer (CEO) to pay Informa Markets UK a total of \$113,642 for participation in Routes World Conference.

ADMINISTRATIVE REPORTS/DISCUSSION/ACTION

8. RESOLUTIONS FOR (1) APPROVAL AND CERTIFICATION OF THE FINAL ENVIRONMENTAL IMPACT REPORT, THE MITIGATION MONITORING AND REPORTING PROGRAM, RELATED CEQA FINDINGS, A STATEMENT OF OVERRIDING CONSIDERATIONS, AND FINDINGS REGARDING THE LACK OF ANY SIGNIFICANT HISTORIC RESOURCES ON THE PROJECT SITE, FOR THE SOUTH AIRPORT CARGO CENTER PROJECT AT ONTARIO INTERNATIONAL AIRPORT, AND (2) APPROVAL OF THE SOUTH AIRPORT CARGO CENTER PROJECT AT ONTARIO INTERNATIONAL AIRPORT

The Ontario International Airport Authority Commission adopt:

1. A Resolution approving and certifying the Final Environmental Impact Report, the Mitigation Monitoring and Reporting Program, related CEQA Findings, a Statement of Overriding Considerations, and Findings Regarding the Lack of any Significant Historic Resources on the Project Site, for the South Airport Cargo Center Project at Ontario International Airport

RESOLUTION NO. 2023-10

A RESOLUTION OF THE ONTARIO INTERNATIONAL AIRPORT AUTHORITY TO APPROVE AND CERTIFY THE FINAL ENVIRONMENTAL IMPACT REPORT, THE MITIGATION MONITORING AND REPORTING PROGRAM, RELATED CEQA FINDINGS, A STATEMENT OF OVERRIDING CONSIDERATIONS, AND FINDINGS REGARDING THE LACK OF ANY SIGNIFICANT HISTORIC RESOURCES ON THE PROJECT SITE, FOR THE SOUTH AIRPORT CARGO CENTER PROJECT AT ONTARIO INTERNATIONAL AIRPORT

and

2. A Resolution approving the South Airport Cargo Center Project at Ontario International Airport

RESOLUTION NO. 2023-11

A RESOLUTION OF THE ONTARIO INTERNATIONAL AIRPORT AUTHORITY TO APPROVE THE SOUTH AIRPORT CARGO CENTER PROJECT AT THE ONTARIO INTERNATIONAL AIRPORT AND TO ADOPT FINDINGS REGARDING THE PROJECT

5

9. RESOLUTION TO ADOPT THE FYE 2024 OPERATING BUDGET

Approve a resolution adopting the FYE 2024 Operating Budget.

RESOLUTION NO. 2023-12

A RESOLUTION OF THE ONTARIO INTERNATIONAL AIRPORT AUTHORITY COMMISSION ADOPTING THE ANNUAL OPERATING BUDGET FOR FISCAL YEAR 2023-24

10. RESOLUTION TO ADOPT THE FYE 2024 CAPITAL IMPROVEMENT PROGRAM BUDGET

Approve a resolution adopting the FYE 2024 Capital Improvement Program and Budget.

RESOLUTION NO. 2023-13

A RESOLUTION OF THE ONTARIO INTERNATIONAL AIRPORT AUTHORITY COMMISSION ADOPTING THE ANNUAL CAPITAL BUDGET FOR FISCAL YEAR 2023-24

11. RESOLUTION APPROVING THE ONTARIO INTERNATIONAL AIRPORT SYSTEM RATES AND CHARGES FOR FISCAL YEAR 2023-2024

Approve a resolution approving the Airport System Rates and Charges for Fiscal Year 2023-2024 at Ontario International Airport (ONT).

RESOLUTION NO. 2023-14

A RESOLUTION OF THE ONTARIO INTERNATIONAL AIRPORT AUTHORITY COMMISSION APPROVING THE AIRPORT SYSTEM RATES AND CHARGES FOR FISCAL YEAR 2023-24

MANAGEMENT REPORT

Executive Office

COMMISSION MATTERS

President Wapner

Vice President Loveridge

Secretary Bowman

Commissioner Hagman

Commissioner Gouw

ADJOURNMENT

AFFIDAVIT OF POSTING

I, Norma I. Alley, MMC, Board Clerk of the Ontario International Airport Authority (OIAA), do hereby declare under penalty of perjury that the foregoing agenda has been posted at the administrative office and on the OIAA website in compliance to the Brown Act.

Date Posted: June 19, 2023

Posted Prior To: 12:00 P.M.

Signature:

Norma I. Alley, MMC Ontario International Airport Authority Board Clerk

THIS PAGE INTENTIONALLY LEFT BLANK



MEETING DATE: JUNE 22, 2023

AGENDA SECTION: AGENDA REVIEW/ANNOUNCEMENTS

SUBJECT: RELATIVE POSSIBLE CONFLICT OF INTEREST

RELEVANT STRATEGIC OBJECTIVE: Invest in ONT Invest in ONT

RECOMMENDED ACTION(S): Declare Conflict of Interest pertaining to agenda items and contractors and/or subcontractors, which may require member abstentions due to possible conflicts of interest.

FISCAL IMPACT SUMMARY: N/A

BACKGROUND: In accordance with California Government Code 84308, members of the Ontario International Airport Authority may not participate in any action concerning a contract where they have received a campaign contribution of more than \$250 in the prior twelve (12) months and from an entity or individual if the member knows or has reason to know that the participant has a financial interest, except for the initial award of a competitively bid public works contract. This agenda contains recommendations for action relative to the following contractors:

Item No	Principals & Agents	Subcontractors
04	Garver, LLC	• Faith Group, LLC
05	CS&M Construction Services, Inc.	Zafi Air ConditioningKAS Electric
06	PPM Group Inc.	Kleinfelder Inc.
07	Informa Markets (UK) Limited	

PROCUREMENT: N/A

CEQA COMPLIANCE AND LAND USE APPROVALS: N/A

STAFFING IMPACT (# OF POSITIONS): N/A

IMPACT ON OPERATIONS: N/A

SCHEDULE: N/A

ATTACHMENTS: N/A

STAFF MEMBER PRESENTING: Chief Executive Officer Atif Elkadi

Originator Name:	Norma I. Alley, MMC
Originating Dept.:	Clerk's Office
Director Review:	Norma 1. alley
Chief Review:	$A^{}$
CEO Approval:	$A^{==}$

This Agenda Report has been reviewed by OIAA General Counsel.

The Agenda Report references the terms and conditions of the recommended actions and request for approval. Any document(s) referred to herein, which are not attached or posted online, may be reviewed prior to or following scheduled Commission meetings in the Office of the Clerk of the Board. Office hours are 8:30 a.m. to 5:00 p.m., Monday through Friday, although these hours and review procedures may be modified. In that case, the documents may be requested by email at <u>clerk@flyontario.com</u>.

BOARD DISPOSITION: Approved Denied Continued to _____



MEETING DATE: JUNE 22, 2023

AGENDA SECTION: CONSENT CALENDAR

SUBJECT: APPROVAL OF MINUTES

RELEVANT STRATEGIC OBJECTIVE: 🛛 Invest in ONT 🖾 Master the Basics 🖾 Plan for the Future

RECOMMENDED ACTION(S): Approve minutes for the OIAA Board of Commissioners meeting on May 25, 2023.

FISCAL IMPACT SUMMARY: N/A

BACKGROUND: The OIAA Board of Commission held a public meeting and minutes were recorded in text. In accordance with OIAA's Records Retention Schedule, the OIAA must preserve these historical records in hard copy form for permanent retention.

PROCUREMENT: N/A

CEQA COMPLIANCE AND LAND USE APPROVALS: N/A

STAFFING IMPACT (# OF POSITIONS): N/A

IMPACT ON OPERATIONS: N/A

SCHEDULE: N/A

ATTACHMENTS:

1. Minutes

STAFF MEMBER PRESENTING: Chief Executive Officer Atif Elkadi

Originator Name:	Norma I. Alley, MMC, Board Clerk
Originating Dept.:	Clerk's Office
Director Review:	Norma I. Welly
Chief Review:	A = =
CEO Approval:	$A^{}$

This Agenda Report has been reviewed by OIAA General Counsel.

The Agenda Report references the terms and conditions of the recommended actions and request for approval. Any document(s) referred to herein, which are not attached or posted online, may be reviewed prior to or following scheduled Commission meetings in the Office of the Clerk of the Board. Office hours are 8:30 a.m. to 5:00 p.m., Monday through Friday, although these hours and review procedures may be modified. In that case, the documents may be requested by email at <u>clerk@flyontario.com</u>.

BOARD DISPOSITION: 🗆 Approved	🗆 Denied	Continued to
-------------------------------	----------	--------------

ONTARIO INTERNATIONAL AIRPORT AUTHORITY REGULAR COMMISSION MEETING MINUTES May 25, 2023

CALL TO ORDER

Vice President Loveridge called the Ontario International Airport Authority Commission meeting to order at 2:00 p.m.

ROLL CALL

PRESENT: Commissioners: Jim W. Bowman, Julie Gouw, Curt Hagman, and Ronald O. Loveridge

ABSENT: Commissioners: Alan D. Wapner

A quorum of the Board of Commissioners was present.

Also present were Chief Executive Officer Atif Elkadi; Assistant General Counsel Kevin P. Sullivan; Chief Financial Officer/Treasurer John M. Schubert; and Board Clerk/Assistant Secretary Norma I. Alley, MMC.

PLEDGE OF ALLEGIANCE

The Pledge of Allegiance was performed.

AGENDA REVIEW/ANNOUNCEMENTS

1. INFORMATION RELATIVE TO POSSIBLE CONFLICT OF INTEREST

No conflicts of interests were declared.

CLOSED SESSION

PUBLIC COMMENT ON CLOSED SESSION

Vice President Loveridge called for public comment. Seeing no one come forward, he closed public comment.

CLOSED SESSION

• GC § 54957, PUBLIC EMPLOYEE PERFORMANCE EVALUATION Title: Chief Executive Officer

Assistant General Counsel Kevin P. Sullivan announced the closed session item.

Vice President Loveridge recessed the open session and called the closed session to order at 2:02 p.m. The open session reconvened at 2:14 p.m.

REPORT ON CLOSED SESSION

Vice President Loveridge announced there was no reportable action.

PUBLIC COMMENT

Vice President Loveridge called for public comment. Seeing no one come forward, he closed public comment.

CONSENT CALENDAR

2. APPROVAL OF MINUTES

Approved minutes for the OIAA Board of Commissioners meeting on April 27, 2023.

3. CASH DISBURSEMENT REPORT (BILLS/PAYROLL)

Received and filed the Cash Disbursement Report (Bills/Payroll) for the month ended April 30, 2023.

4. COMMISSIONER'S STIPENDS AS REQUIRED BY AUTHORITY BYLAWS

Approved additional stipends per Article IV, Section 6 of the Authority's Bylaws for President Wapner for the month of April, 2023.

5. RESOLUTION UPDATING THE NON-EXCLUSIVE AIR CARRIER OPERATING PERMIT

Adopted a Resolution updating the Non-Exclusive Air Carrier Operating Permit and authorizing the Chief Executive Officer to make minor amendments to the Permit.

RESOLUTION NO. 2023-09: A RESOLUTION OF THE ONTARIO INTERNATIONAL AIRPORT AUTHORITY AMENDING RESOLUTION NO. 2017-12 APPROVING AN UPDATED NON-EXCLUSIVE AIR CARRIER OPERATING PERMIT AT ONTARIO INTERNATIONAL AIRPORT

6. ENVIROSUITE, INC., SOLE SOURCE AGREEMENT

Authorized the Chief Executive Officer (CEO) to execute a five-year contract with five, one-year extension with Envirosuite, Inc., and authorize the CEO to exercise the five, one-year extensions to support the Airport Noise and Operations Monitoring System (ANOMS) at Ontario International Airport (ONT), in an amount not to exceed \$1,986,209.70 over the ten-year contract.

7. TBI AIRPORT MANAGEMENT CONTRACT AMENDMENT

Authorized the Chief Executive Officer (CEO) to execute the First Contract Amendment to add additional services and increase the contractual authority for the OIAA's existing TBI Airport Management (TBI) agreement, Contract No. SCONT-000082, by 15% of the annual reimbursable expenses related to the additional services not to exceed \$730,000.

8. TRACTOR AND MOWER PURCHASE

Approved an increase of the Capital Improvement Program project budget for fleet vehicles by \$100,000 and authorize the Chief Executive Officer (CEO) to execute the agreements for the purchase of three new fleet vehicles.

- 1. Authorize the Chief Executive Officer (CEO) to execute an agreement with Pape' Machinery, Inc. in the amount of \$334,442.18 for the purchase of three (3) John Deere tractors and three (3) flex wing mowers.
- 2. Approve an increase of the Capital Improvement Program budget of \$100,000 to cover the additional cost for the purchase of these tractors and mowers.

9. FINANCIAL STATEMENTS FOR THE NINE MONTHS ENDED MARCH 31, 2023

Received and filed financial statements for the nine months ended March 31, 2023.

10. INVESTMENT REPORT FOR THE NINE MONTHS ENDED MARCH 31, 2023

Received and filed the Investment Report for the nine months ended March 31, 2023.

11. CAPITAL PROJECT EXPENDITURE INCREASE FOR FLEET VEHICLES

Approved an increase of the Capital Improvement Program project budget for fleet vehicles by \$100,000 and authorize the Chief Executive Officer (CEO) to execute the agreements for the purchase of three new fleet vehicles.

MOTION: Moved by Commissioner Hagman, seconded by Commissioner Bowman, to approve the Consent Calendar, including Resolution No. 2023-09. Motion carried by a 4 Yes/0 No/1 Absent (Wapner) vote.

ADMINISTRATIVE DISCUSSION/ACTION/REPORT

12. HILTACHK MARKETING GROUP CONTRACT AMENDMENT

Authorized the Chief Executive Officer (CEO) to execute the third contract amendment to authorize the media spend and increase by \$1,094,756.00 (for a total contract amount of \$3,094,756 in this fiscal year) in fiscal year 2022-2023 and by \$4,200,000 in fiscal year 2023-2024 (for a total contract amount of \$4,500,000 in fiscal year), for media buys, including but not limited to, targeted digital advertising, out of home, experiential activities, paid social and more.

Chief Executive Officer Elkadi presented the staff report and fielded questions from the Board of Commissioners.

MOTION: Moved by Commissioner Hagman, seconded by Commissioner Bowman. Motion carried by a 4 Yes/0 No/1 Absent (Wapner) vote.

MANAGEMENT REPORT

Chief Executive Officer Elkadi provided updates on passenger traffic numbers, changes to flights, and general airport matters accompanied by a PowerPoint.

COMMISSIONER MATTERS

There were no updates or matters brought forth to the Board of Commissioners.

ADJOURNMENT

Vice President Loveridge adjourned the Ontario International Airport Authority Board of Commissioners Meeting at 2:23 p.m.

RESPECTFULLY SUBMITTED:

APPROVED:

NORMA I. ALLEY, MMC, BOARD CLERK

ALAN D. WAPNER, PRESIDENT



MEETING DATE: JUNE 22, 2023

AGENDA SECTION: CONSENT CALENDAR

SUBJECT: CASH DISBURSEMENT REPORT (BILLS/PAYROLL)

RELEVANT STRATEGIC OBJECTIVE: Invest in ONT Invest in ONT

RECOMMENDED ACTION(S): Receive and file the Cash Disbursement Report (Bills/Payroll) for the month ended May 31, 2023.

FISCAL IMPACT SUMMARY: The funding is approved in the Fiscal Year 2022-2023 budget.

BACKGROUND: In June 2022, the OIAA Board of Commissioners adopted an operating budget that was driven by strong aviation activity and financial performance realized by OIAA in FYE 2022. The operating budget was developed from OIAA goals and objectives and includes significant increases in resources to meet current and expected near term growth. The budget FYE 2023 assumes that there will be no significant impacts associated with any possible COVID outbreaks. As passenger activities continue to increase over pre-pandemic levels, OIAA has continued to resume normal operations, fill vacant positions as well as add new staff positions, and has returned to pre-pandemic initiatives.

PROCUREMENT: N/A

CEQA COMPLIANCE AND LAND USE APPROVALS: N/A

STAFFING IMPACT (# OF POSITIONS): N/A

IMPACT ON OPERATIONS: N/A

SCHEDULE: N/A

ATTACHMENTS:

1. Cash Disbursement Report (Bills/Payroll) for the month ended May 31, 2023.

STAFF MEMBER PRESENTING: Chief Executive Officer Atif Elkadi		
Originator Name:	Kim Snow, Finance Manager	
Originating Dept.:	Finance and Accounting	
Director Review:	<u>N/A</u>	
) ma	
Chief Review:	Chief Financial Officer John Schubert	
CEO Approval:	A	

This Agenda Report has been reviewed by OIAA General Counsel.

The Agenda Report references the terms and conditions of the recommended actions and request for approval. Any document(s) referred to herein, which are not attached or posted online, may be reviewed prior to or following scheduled Commission meetings in the Office of the Clerk of the Board. Office hours are 8:30 a.m. to 5:00 p.m., Monday through Friday, although these hours and review procedures may be modified. In that case, the documents may be requested by email at <u>clerk@flyontario.com</u>.

BOARD DISPOSITION:
Approved Denied Continued to



MEETING DATE: JUNE 22, 2023

AGENDA SECTION: CONSENT CALENDAR

SUBJECT: PROFESSIONAL SERVICES AGREEMENT WITH GARVER, LLC, FOR SAFETY MANAGEMENT SYSTEM PROGRAM DEVELOPMENT

RELEVANT STRATEGIC OBJECTIVE: 🛛 Invest in ONT 🖾 Master the Basics 🖾 Plan for the Future

RECOMMENDED ACTION(S): Authorize the Chief Executive Officer (CEO) to execute an agreement with Garver, LLC, to provide the development and implementation of a Safety Management System (SMS) Program and manual at Ontario International Airport (ONT), in an amount not to exceed \$300,000 for the completion of the project and authorize the CEO to execute budgeted amendments to the contract for additional related services, if needed, up to 30% of the overall contract value.

FISCAL IMPACT SUMMARY: The proposed maximum cost for this project is \$400,000.00, which includes modest contingency and project management and administration costs and is accounted for in the Ontario International Airport Authority (OIAA) FYE 2023 CIP budget.

BACKGROUND: The Federal Aviation Administration's final rule on SMS for certificated airports as defined in Advisory Circular 150/5200-37A, became effective on April 24, 2023. The Implementation Plan for ONT will be required by April 24, 2024. A Safety Management System (SMS) program will help ONT detect and correct safety concerns before they result in accidents or incidents. An SMS enhances safety, ensures compliance with applicable regulatory standards, and can be integrated into all aspects of airport operations including business and management practices. SMS will help ONT develop an explicit, pro-active, and engaged process for identifying and quantifying potential hazards and risks and for managing them in a systematically coherent, logical and reasonable way.

PROCUREMENT: RFP #22000406 Safety Management Program Development and Manual for Ontario International Airport was issued on Friday, April 14, on Planet Bids with the OIAA receiving one qualified proposal. The RFP team reviewed and graded the firm which presented the proposal.

CEQA COMPLIANCE AND LAND USE APPROVALS: Issuance of permits, leases, agreements, and renewals, amendments or extensions thereof, or other entitlements granting use of an existing airport facility involving negligible or no expansion of use and/or alteration or modification of the facilities or its operations beyond that previously existing or permitted is exempt from the requirements of the California Environmental Quality Act (CEQA).

STAFFING IMPACT (# OF POSITIONS): N/A

IMPACT ON OPERATIONS: This action will not impact ONT operations.

SCHEDULE: N/A

ATTACHMENTS: None

STAFF MEMBER PRESENTING: Chief Executive Officer Atif Elkadi

Originator Name:	Dennis Anderson, Director of Operations
Originating Dept.:	Operations
Director Review:	Came Colleron
Chief Review:	1 ml
CEO Approval:	A===

This Agenda Report has been reviewed by OIAA General Counsel.

The Agenda Report references the terms and conditions of the recommended actions and request for approval. Any document(s) referred to herein, which are not attached or posted online, may be reviewed prior to or following scheduled Commission meetings in the Office of the Clerk of the Board. Office hours are 8:30 a.m. to 5:00 p.m., Monday through Friday, although these hours and review procedures may be modified. In that case, the documents may be requested by email at <u>clerk@flyontario.com</u>.

BOARD DISPOSITION: 🗌 Approved 🗌 Denied	Continued to
--	--------------



MEETING DATE: JUNE 22, 2023

AGENDA SECTION: CONSENT CALENDAR

SUBJECT: CONTRACT WITH CS&M CONSTRUCTION SERVICES, INC., FOR GLOBAL ENTRY ENROLLMENT CENTER

RELEVANT STRATEGIC OBJECTIVE: Invest in ONT Invest in ONT

RECOMMENDED ACTION(S): Authorize the Chief Executive Officer (CEO) to execute a Contract with CS&M Construction Services Inc., in an amount \$170,000 to perform the Construction activities required for the Global Entry program office space at the Federal Inspection Services (FIS) Terminal; and authorize the CEO to execute budgeted amendments to the contract for additional related services, if needed, up to 30% of the overall contract value.

FISCAL IMPACT SUMMARY: The proposed maximum cost for this project is \$200,000.00, which includes modest contingency and project management and administration costs and is accounted for in the Ontario International Airport Authority (OIAA) FYE 2024 CIP budget.

BACKGROUND: Global Entry is a trusted traveler program administered by Customs and Border Protection (CBP). Enrollment in this program is a two-part process consisting of a background check and an in-person interview at a Global Entry Enrollment Center. Global Entry Enrollment Centers are operated at CBP designated Ports of Entry. The nearest Enrollment Center is by LAX in Los Angeles and along the southern border at San Ysidro. Appointments may take up to 12-15 months.

Offering an Enrollment Center at ONT at the FIS Facility allows OIAA to provide convenient and world-class amenities to our customers while utilizing the current afternoon and early evening slower periods at our International Arrivals terminal. This will allow the Global Entry interview appointments to be scheduled and occur without adding additional CBP personnel costs to OIAA.

PROCUREMENT: OIAA utilized a Competitive Bidding process to execute this project, received four proposals from four contractors, and proceeded with the lowest responsive and responsible bidder.

CEQA COMPLIANCE AND LAND USE APPROVALS: N/A

STAFFING IMPACT (# OF POSITIONS): N/A

IMPACT ON OPERATIONS: N/A

SCHEDULE: The construction for this project will take two months to complete.

ATTACHMENTS: None

Originator Name:	Michael Bishara, Project Manager
Originating Dept.:	Project Management
Director Review:	Keith Owens Kith & Swinz
Chief Review:	Michelle Brantley Michelle Brantley
CEO Approval:	A = 0

This Agenda Report has been reviewed by OIAA General Counsel.

The Agenda Report references the terms and conditions of the recommended actions and request for approval. Any document(s) referred to herein, which are not attached or posted online, may be reviewed prior to or following scheduled Commission meetings in the Office of the Clerk of the Board. Office hours are 8:30 a.m. to 5:00 p.m., Monday through Friday, although these hours and review procedures may be modified. In that case, the documents may be requested by email at <u>clerk@flyontario.com</u>.

BOARD DISPOSITION: Approved Denied Continued to _____



MEETING DATE: JUNE 22, 2023

AGENDA SECTION: CONSENT CALENDAR

SUBJECT: CONTRACT EXTENSION WITH PPM GROUP, INC., FOR CONSTRUCTION MANAGEMENT AND RELATED SERVICES

RELEVANT STRATEGIC OBJECTIVE: 🛛 Invest in ONT 🖾 Master the Basics 🗌 Plan for the Future

RECOMMENDED ACTION(S): Authorize the Chief Executive Officer (CEO) to approve a contract extension to December 31, 2023, for PPM Group, Inc. for On-Call construction management services related to the Runway 26R ILS Upgrade project at Ontario International Airport (ONT).

FISCAL IMPACT SUMMARY: Funding for the work is included in the Runway 26R ILS Upgrade project which was approved in the FY 2021-22 Capital Budget. No funding or authority increase is required for the extension of the on-call construction management (CM) services through June 30, 2023.

BACKGROUND: The Board first authorized the contract with PPM Group, Inc. on October 23, 2018, to provide on-call construction management services at ONT. The board also previously approved two contract extensions that extended the contract until June 30, 2023. PPM Group, Inc. has been providing construction management services on the Runway 26R ILS Upgrade project since the OIAA authorized the CEO to award a construction contract and enter into a reimbursable agreement with the Federal Aviation Administration (FAA) for the project on October 27, 2022.

The Runway 26R ILS is a critical FAA NAVAID that will be accepted and operated by the FAA at the completion of the project. The upgraded ILS is required to enable closure of Runway 8L-26R over a two-year period for reconstruction. FAA change orders, supply chain issues and delays in scheduling a flight check required to enable procedures to be published have all delayed completion of the project. Extending the PPM Group, Inc. contract to December 31, 2023, allows the OIAA to maintain recommended project coordination and continuity to complete the Runway 26R ILS Upgrade project.

PROCUREMENT: This contract was procured in a competitive bid process in 2018.

CEQA COMPLIANCE AND LAND USE APPROVALS: The proposed staffing services decision is not a "project" under California Environmental Quality Act (CEQA) and does not require environmental review.

STAFFING IMPACT (# OF POSITIONS): N/A

IMPACT ON OPERATIONS: The proposed time extension of staffing services is to ensure completion of the Runway 26R ILS Upgrade project with minimal impacts to airport operations by keeping the current CM in place to the end of the calendar year, December 31, 2023.

SCHEDULE: N/A

ATTACHMENTS: None

STAFF MEMBER PRESENTING: Chief Executive Officer Atif Elkadi	
Originator Name:	Keith Owens, Director of Project Management
Originating Dept.:	Project Management
Director Review:	Keith Owens
Chief Review:	Michelle Brantley
CEO Approval:	A===

This Agenda Report has been reviewed by OIAA General Counsel.

The Agenda Report references the terms and conditions of the recommended actions and request for approval. Any document(s) referred to herein, which are not attached or posted online, may be reviewed prior to or following scheduled Commission meetings in the Office of the Clerk of the Board. Office hours are 8:30 a.m. to 5:00 p.m., Monday through Friday, although these hours and review procedures may be modified. In that case, the documents may be requested by email at <u>clerk@flyontario.com</u>.

BOARD DISPOSITION: Approved Denied Continued to



MEETING DATE: JUNE 22, 2023

AGENDA SECTION: CONSENT CALENDAR

SUBJECT: PAYMENT TO INFORMA MARKETS UK FOR ROUTES WORLD EXHIBITION PARTICIPATION AND SPONSORSHIP

RELEVANT STRATEGIC OBJECTIVE: 🛛 Invest in ONT 🗌 Master the Basics 🖾 Plan for the Future

RECOMMENDED ACTION(S): Authorize the Chief Executive Officer (CEO) to pay Informa Markets UK a total of \$113,642 for participation in Routes World Conference.

FISCAL IMPACT SUMMARY: Funding for this activity was approved under the Fiscal Year 2022-23 budget.

BACKGROUND: Ontario International Airport (ONT) has seen huge success in participating in the largest aviation conference that brings airlines, airports and aviation stakeholders together in support of expanding global air service. ONT participates in one-on-one meetings with a diverse mix of airlines, with the goal of increasing service in and out of ONT. Our mission at this conference is to raise awareness of ONT's product, services, and air carrier incentives with airline partners from all over the world.

PROCUREMENT: N/A

CEQA COMPLIANCE AND LAND USE APPROVALS: The proposed payment is Categorically Exempt (Class 1) from the provisions of the California Environmental Quality Act (CEQA) pursuant to Guidelines Section 15301 because it involves creative Advertising, Marketing and Promotional Services on an as-needed, on-call basis.

STAFFING IMPACT (# OF POSITIONS): N/A

IMPACT ON OPERATIONS: N/A

SCHEDULE: N/A

ATTACHMENTS: None

STAFF MEMBER PRESENTING: Chief Executive Officer Atif Elkadi

Originator Name:	Eren Cello, Chief Marketing and Communications Officer
Originating Dept.:	Marketing and Communications
Director Review:	N/A
Chief Review:	elallo
CEO Approval:	A^{2}

This Agenda Report has been reviewed by OIAA General Counsel.

The Agenda Report references the terms and conditions of the recommended actions and request for approval. Any document(s) referred to herein, which are not attached or posted online, may be reviewed prior to or following scheduled Commission meetings in the Office of the Clerk of the Board. Office hours are 8:30 a.m. to 5:00 p.m., Monday through Friday, although these hours and review procedures may be modified. In that case, the documents may be requested by email at <u>clerk@flyontario.com</u>.

BOARD DISPOSITION: Approved Denied Continued to



MEETING DATE: JUNE 22, 2023

AGENDA SECTION: ADMINISTRATIVE REPORTS/DISCUSSION/ACTION

SUBJECT: RESOLUTIONS FOR (1) APPROVAL AND CERTIFICATION OF THE FINAL ENVIRONMENTAL IMPACT REPORT, THE MITIGATION MONITORING AND REPORTING PROGRAM, RELATED CEQA FINDINGS, A STATEMENT OF OVERRIDING CONSIDERATIONS, AND FINDINGS REGARDING THE LACK OF ANY SIGNIFICANT HISTORIC RESOURCES ON THE PROJECT SITE, FOR THE SOUTH AIRPORT CARGO CENTER PROJECT AT ONTARIO INTERNATIONAL AIRPORT, AND (2) APPROVAL OF THE SOUTH AIRPORT CARGO CENTER PROJECT AT ONTARIO INTERNATIONAL AIRPORT.

RELEVANT STRATEGIC OBJECTIVE: Invest in ONT Invest in ONT

RECOMMENDED ACTION(S): That the Ontario International Airport Authority (OIAA) Commission adopt:

(1) a Resolution approving and certifying the Final Environmental Impact Report, the Mitigation Monitoring and Reporting Program, related CEQA Findings, a Statement of Overriding Considerations, and Findings Regarding the Lack of any Significant Historic Resources on the Project Site, for the South Airport Cargo Center Project at Ontario International Airport, and

(2) a Resolution approving the South Airport Cargo Center Project at Ontario International Airport.

FISCAL IMPACT SUMMARY: A later Ground Lease for the Project operations will generate significant rental and air cargo freight charge revenues for the OIAA. The Project applicant will pay for construction of the Project air cargo center.

BACKGROUND: The underlying purpose of the proposed Project is to develop and operate an air cargo facility at the Airport to meet increased regional air cargo volumes and Project applicant facility requirements. The objectives of the OIAA for the proposed Project include:

A. Allow the Project applicant to accommodate current and projected air cargo volume growth.

B. Integrate the Project applicant's airside, landside, and sorting facilities in a location with access to major surface transportation corridors to improve operational efficiency.

C. Redevelop underutilized Airport property.

D. Maximize revenue generation from Airport property.

E. Provide employment opportunities for residents of the City of Ontario and the Inland Empire.

The Project is an aeronautical development and use that is within the Airport boundaries and is consistent with the Ontario International Airport Layout Plan. The proposed Project would replace existing, underutilized airport related buildings and site improvements with an air cargo center. The proposed Project would include demolition of the existing buildings, site improvements, and the development of a new air cargo center in two phases, as described further below.

The proposed air cargo center includes an Air Cargo Sort Building, truckyard, parking facilities, two aviation support buildings (ground service equipment and aircraft line maintenance buildings), and aircraft apron improvements. The Air Cargo Sort Building, proposed north of East Avion Street, would contain a sorting facility and office spaces. The aircraft parking apron would surround the building to the west, north, and east. A ground-level visitor parking lot and truckyard are proposed on the south side of the cargo building, with access from East Avion Street. A parking structure for employees is proposed south of East Avion Street with a pedestrian bridge connecting the parking structure to the office building.

The proposed Project would be implemented in two phases. Phase 1 would take place on the easternmost 62 acres of the Project site and Phase 2 would occur on the remaining western 35 acres.

PROCUREMENT: N/A

CEQA COMPLIANCE AND LAND USE APPROVALS: A Draft and Final Environmental Impact Report (collectively the FEIR or Final EIR) (SCH#2021100226) (collectively the Final SEIR) (SCH#2021060531) has been prepared by the OIAA, as the Lead Agency, pursuant to the requirements of the California Environmental Quality Act (CEQA) (Public Resources Code Sections 21000 *et seq.*) and the State CEQA Guidelines (14 Cal Code Regs. Sections 15000 *et seq.*) to address the potential environmental impacts associated with the South Airport Cargo Center Project at Ontario International Airport, identified as the proposed Project.

Construction and/or operation of the Project would result in significant, unavoidable impacts to Air Quality (AQ), Greenhouse Gas Emissions (GHG), and Transportation.

There are no feasible mitigation measures or alternatives within the control, authority, and jurisdiction of the OIAA that can be implemented to reduce the identified Project-related significant AQ, GHG and Transportation impacts to a level below significance or to otherwise further reduce the impacts. Some potential mitigation measures or alternatives on these impact items are within the jurisdiction and control of the Federal Aviation Administration or are otherwise infeasible. Therefore, the impacts remain significant and unavoidable.

The Final EIR, at Chapter 4, identifies a Mitigation Monitoring and Reporting Program (MMRP) to address potentially significant environmental impacts from the Project. The MMRP will be enforced as conditions of approval for the Project to ensure the potential Project impacts to the environment, except as to AQ, GHG and Transportation as discussed above, are reduced to a level below significance. The MMRP is also attached as **Exhibit "A"** to the Resolution certifying and approving the Final EIR (Resolution No. 2023-10 attached).

CEQA Findings – Applicable findings can be made under CEQA, and are contained in Sections 3.0 through 10.0 of the CEQA Findings that are attached as **Exhibit "B"** to Resolution No. 2023-10 certifying and approving the Final EIR, which Findings discuss how all feasible mitigation measures have been adopted to either reduce

the significant and unavoidable Project impacts, although not to below a level of significance, or to reduce other potential Project impacts to below a level of significance.

Project Alternatives - Three alternatives (the No Project Alternative, the Reduced Project Size Alternative, and the Different Location on the Airport Alternative) were identified and discussed in the Final EIR. These three alternatives represent a reasonable range of the alternatives that could feasibly attain one or more of the Project objectives and eliminate or reduce some significant impacts. The Final EIR explains that these Alternatives were rejected because they fail to achieve all Project objectives, including due the financial, economic, and practical infeasibility of the three Alternatives. Those three Alternatives, and their rejection as infeasible, are also addressed at Section 6 of the CEQA Findings (**Exhibit "B"** to Resolution No. 2023-10 attached).

Statement of Overriding Considerations – Notwithstanding the significant and unavoidable AQ, GHG and Transportation impacts from the Project discussed above, Section 10.0 of the CEQA Findings (Statement of Overriding Considerations) identifies that the Project would have economic, legal, social, technological, and other overriding benefits. (See **Exhibit "B"** to Resolution No. 2023-10 attached). Such Overring Considerations include (a) improvement of the Airport's air cargo system management, (b) improvement of the Airport's transportation infrastructure, (c) an increase in the Airport's goods movement capacity, (d) maximization of the Airport's revenue to help ensure its required financial self-sustainability, (e) implementation of environmentally sustainable airport development and design features, and (f) increased employment in the region by the creation of about 1,315 jobs. Section 10.0 of the CEQA Findings therefore discusses that many sufficient bases for approving the Project exist, notwithstanding any significant and unavoidable impacts that may occur.

STAFFING IMPACT (# OF POSITIONS): N/A

IMPACT ON OPERATIONS: N/A

SCHEDULE: Phase 1 of the Project is estimated to begin operations in 2025, and Phase 2 of the Project is estimated to begin operations in 2029.

ATTACHMENTS:

- A Resolution approving and certifying the Final Environmental Impact Report, the Mitigation Monitoring and Reporting Program, related CEQA Findings, a Statement of Overriding Considerations, and Findings Regarding the Lack of any Significant Historic Resources on the Project Site, for the South Airport Cargo Center Project at Ontario International Airport.
- 2. A Resolution approving the South Airport Cargo Center Project at Ontario International Airport.
- 3. Link to the Final and Draft EIR materials for the SACC Project is: <u>https://www.flyontario.com/our-neighbors/environment</u>.

STAFF MEMBER PRESENTING: Michelle Brantley

Originator Name:	Kevin P. Sullivan
Originating Dept.:	General Counsel's Office
Director Review:	N/A
Chief Review:	Michelle Brantley Michelle Brantley
	A
CEO Approval:	4

This Agenda Report has been reviewed by OIAA General Counsel.

The Agenda Report references the terms and conditions of the recommended actions and request for approval. Any document(s) referred to herein, which are not attached or posted online, may be reviewed prior to or following scheduled Commission meetings in the Office of the Clerk of the Board. Office hours are 8:30 a.m. to 5:00 p.m., Monday through Friday, although these hours and review procedures may be modified. In that case, the documents may be requested by email at <u>clerk@flyontario.com</u>.

BOARD DISPOSITION:
Approved Denied Continued to _____

A RESOLUTION OF THE ONTARIO INTERNATIONAL AIRPORT AUTHORITY TO APPROVE AND CERTIFY THE FINAL ENVIRONMENTAL IMPACT REPORT. THE MITIGATION MONITORING AND REPORTING PROGRAM. RELATED CEQA FINDINGS. Α STATEMENT OF **OVERRIDING** CONSIDERATIONS, AND FINDINGS **REGARDING THE LACK OF ANY SIGNIFICANT HISTORIC RESOURCES ON THE PROJECT SITE, FOR THE SOUTH** AIRPORT CARGO CENTER PROJECT AT ONTARIO INTERNATIONAL AIRPORT

WHEREAS, the Ontario International Airport Authority (OIAA or Authority) was established under an August 2012 Joint Exercise of Powers Agreement between the City of Ontario and the County of San Bernardino pursuant to the Joint Exercise of Powers Act of the State of California (Government Code Sections 6500.1 *et seq.*), for the purpose of operating, maintaining, managing, developing, and marketing the Ontario International Airport (Airport) and related Airport facilities;

WHEREAS, a Draft and Final Environmental Impact Report (collectively the FEIR or Final EIR) (SCH#2021100226) has been prepared by the OIAA, as the Lead Agency, pursuant to the requirements of the California Environmental Quality Act (CEQA) (Public Resources Code Sections 21000 *et seq.*) and the State CEQA Guidelines (14 Cal Code Regs. Sections 15000 *et seq.*) to address the potential environmental impacts associated with the South Airport Cargo Center (Project) at Ontario International Airport (ONT or Airport);

WHEREAS, the proposed Project is an aeronautical development and use that is within the Airport boundaries and is consistent with the Ontario International Airport Layout Plan approved by the Federal Aviation Administration. The proposed Project would replace existing, underutilized airport related buildings and site improvements with an air cargo center. The proposed Project would include demolition of the existing buildings, site improvements, and the development of a new air cargo center in two phases, as described further below. The proposed air cargo center includes an Air Cargo Sort Building, truckyard, parking facilities, two aviation support buildings (ground service equipment and aircraft line maintenance buildings), and aircraft apron improvements. The Air Cargo Sort Building, proposed north of East Avion Street, would contain a sorting facility and office spaces. The aircraft parking apron would surround the building to the west, north, and east. A ground-level visitor parking lot and truckyard are proposed on the south side of the cargo building, with access from East Avion Street. A parking structure for employees is proposed south of East Avion Street with a pedestrian bridge connecting the parking structure to the office building. The proposed Project would be

implemented in two phases. Phase 1 would take place on the easternmost 62 acres of the Project site and Phase 2 would occur on the remaining western 35 acres;

WHEREAS, on October 14, 2021, OIAA published a Notice of Preparation for the Draft EIR for the proposed Project, together with an Initial Study for the proposed Project;

WHEREAS, the Draft EIR was circulated for a forty five (45) day public review period from March 13, 2023 to April 27, 2023;

WHEREAS, OIAA prepared responses to all written comments on the Draft EIR received during the public review period;

WHEREAS, the Final EIR determined, after significant analysis contained in Section 5.4 (Cultural Resources) and Appendix 5.04-2 (Historic Property Evaluation Report), that none of the structures or features located on the Property qualify as significant historic resources under applicable federal, State, or local regulations or standards;

WHEREAS, a Mitigation Monitoring and Reporting Program (MMRP) is included in the Final EIR to address potentially significant environmental impacts from the proposed Project, which MMRP is attached to this Resolution as **Exhibit "A**";

WHEREAS in accordance with Section 15132 of the State CEQA Guidelines, the Final EIR (SCH#2021100226) consists of:

- a. Draft EIR (SCH#2021100226) for the proposed Project and all appendices and technical reports thereto;
- b. Comments and responses to comments on the Draft EIR (SCH#2021100226) for the proposed Project, including a list of all persons, organizations, and public agencies commenting on the Draft EIR, as well as the MMRP; and

WHEREAS, three alternatives (the No Project Alternative, the Reduced Project Size Alternative, and the Different Location on the Airport Alternative) were identified and discussed in the Final EIR. These three alternatives represent a reasonable range of the alternatives that could feasibly attain one or more of the proposed Project objectives and eliminate or reduce some significant impacts;

WHEREAS, OIAA has rejected the No Project Alternative, the Reduced Project Size Alternative, and the Different Location on the Airport Alternative for reasons set forth in the CEQA Findings, Facts in Support of the Findings, and a Statement of Overriding Considerations (collectively CEQA Findings), which are incorporated by reference, made an express part of this Resolution and attached to this Resolution as **Exhibit "B**";

WHEREAS, the proposed Project best attains a balance between protecting against local environmental impacts and meeting all Project objectives with the least environmental impact;

WHEREAS, Section 21081 of the Public Resources Code and Section 15091 of the State CEQA Guidelines requires the OIAA make one or more of the following findings prior to approving or carrying out a project for which an EIR has been prepared identifying one or more significant effects to project, together with a statement of facts in support of each finding:

(1) Changes or alterations have been required in, or incorporated into, the project, which avoid or substantially lessen the significant effects on the environment.

(2) Changes or alterations are within the responsibility and jurisdiction of another public agency and have been, or can and should be, adopted by that other agency.

(3) Specific economic, legal, social, technological, or other considerations, including provision of employment opportunities for highly trained workers, make infeasible the mitigation measures or project alternatives identified in the Final EIR;

WHEREAS, with respect to significant effects which were subject to a finding under paragraph (3) above, the OIAA must find that specific overriding economic, legal, social, technological, or other benefits of the project outweigh the significant effects on the environment;

WHEREAS, Section 15093(a) of the State CEQA Guidelines requires the OIAA to balance the benefits of a proposed project against its unavoidable environmental risk in determining whether to approve the project;

WHEREAS, Section 15093(b) of the State CEQA Guidelines requires that, where the decision of the OIAA allows the occurrence of significant effects which are identified in an EIR, but are not at least substantially mitigated, OIAA must state in writing the reasons to support its action on the Final EIR or other information in the record;

WHEREAS, Section 15097 of the State CEQA Guidelines requires that a MMRP designed to ensure compliance with mitigation measures imposed on the proposed Project to avoid or substantially lessen the significant effects identified in the Final EIR be prepared;

WHEREAS, on June 12, 2023, consistent with State CEQA Guidelines section 15088(b), the OIAA provided to the each public or government agency that submitted comment letters to the Draft EIR (SCH#2021100226) with detailed responses to such comment letters;

WHEREAS, on June 16, 2023, the OIAA provided to the each Commissioner of the OIAA, as well as to all persons, entities and agencies that submitted comment letters to the Draft EIR (SCH#2021100226), a link to the OIAA website that contained the entire Final EIR materials for the Project;

WHEREAS, on June 19, 2023, the OIAA provided notice of the public meeting on the Final EIR (SCH#2021100226) and the Project scheduled for June 22, 2023, including

to the Commission, in accordance with the Ralph M. Brown Act, Government Code Section 54950, *et seq.*, and provided notice to those persons who specifically requested notice of this meeting;

WHEREAS OIAA staff recommends certification and approval of the Final EIR (SCH#2021100226) for the proposed Project, as well as the MMRP, the CEQA Findings, and the Statement of Overriding Considerations; and

WHEREAS, the Commission has reviewed and considered all such materials for the Final EIR (SCH#2021100226), and for the proposed Project discussed above.

NOW, BE IT RESOLVED BY THE COMMISSION OF THE ONTARIO INTERNATIONAL AIRPORT AUTHORITY, as the airport proprietor of Ontario International Airport, and, in light of the whole record for the Final EIR (SCH#2021100226) and the Project before it, the OIAA Commission finds as follows:

1. All recitals and Whereas clauses above are true and are incorporated in full into these findings;

2. The OIAA Commission has reviewed and considered the Final EIR (SCH#2021100226) for the Project as the Lead Agency pursuant to CEQA and finds that the Final EIR (SCH#2021100226) is adequate, complete and contains all information required by CEQA and the State CEQA Guidelines.

3. The OIAA Commission has provided and will continue to provide the Final EIR (SCH#2021100226), on file with the Ontario International Airport Authority, 1923 East Avion Street, Ontario, California 91761, and available for inspection through the OIAA Clerk.

4. The OIAA Commission determines, based on the significant analysis contained in Final EIR Section 5.4 (Cultural Resources) and Appendix 5.04-2 (Historic Property Evaluation Report), and pursuant to State CEQA Guidelines Section 15064.5(b), that none of the structures or features located on the Property qualify as significant historic resources under applicable federal, State, or local regulations or standards, including Public Resources Code sections 5020.1(j) and 5024.1.

5. The OIAA Commission finds that the MMRP, which is incorporated by reference and attached to this Resolution as **Exhibit "A"**, establishes a mechanism and procedure for implementing and verifying the implementation of, and compliance with, the adopted mitigation measures for the Project pursuant to Public Resources Code Section 21081.6 and CEQA Guidelines Section 15097, and this Commission adopts the MMRP. The MMRP is adopted, is incorporated into the Project as a condition of approval, and shall be fully enforceable.

6. The OIAA Commission finds that the Final EIR (SCH#2021100226) has identified all significant environmental effects of the Project and has analyzed a range of reasonable alternatives to the Project, as set forth in Findings, which are incorporated by reference, and attached to this Resolution as **Exhibit "B"**.

7. The OIAA Commission adopts the appropriate findings set forth in Section 21081 of the Public Resources Code and Section 15091 of the State CEQA Guidelines with respect to each significant environmental effect identified in the Final EIR (SCH#2021100226), and each alternative considered in the Final EIR (SCH#2021100226), and the explanation of its reasoning with respect to each finding as set forth in the CEQA Findings.

The OIAA 8. Commission finds that although the Final EIR (SCH#2021100226) identifies certain significant environmental effects that may occur with implementation of the Project, all significant effects that feasibly can be mitigated or avoided have been reduced to an acceptable level by imposition of mitigation measures, all of which have been identified in Final EIR (SCH#2021100226) and described in the attached CEQA Findings; and all of which are adopted by this Commission to mitigate the environmental effects of the Project.

9. The OIAA Commission finds that the unavoidably significant environmental effects of the Project, as identified in the attached CEQA Findings, have been lessened substantially in their severity, to the extent feasible, by the imposition of mitigation measures identified in the attached CEQA Findings. This Commission also finds that those unavoidable significant environmental effects are outweighed by the economic, social, technological and other benefits of the Project, as set forth in the Statement of Overriding Considerations, as identified at Section 10.0 in the attached CEQA Findings. Such Overring Considerations include (a) improvement of the Airport's air cargo system management, (b) improvement of the Airport's transportation infrastructure, (c) an increase in the Airport's goods movement capacity, (d) maximization of the Airport's revenue to help ensure its required financial self-sustainability, (e) implementation of environmentally sustainable airport development and design features, and (f) increased employment in the region by the creation of about 1,315 jobs.

10. The OIAA Commission adopts the Statement of Overriding Considerations, as identified at Section 10.0 in the attached CEQA Findings, which, based on substantial evidence in the record, supports and justifies approval of the Project, notwithstanding certain unavoidably significant environmental effects that feasibly cannot be mitigated to below a level of significance. The unavoidable significant environmental effects of the Project are deemed acceptable when balanced against the overriding economic, social, technical and other benefits of the Project as set forth at Section 10.0 in the Statement of Overriding Considerations.

11. The OIAA Commission finds that the Final EIR (SCH#2021100226) has described a reasonable range of alternatives to the Project that feasibly could obtain the basic objectives of the project (including the No Project Alternative), even when these alternatives might impede the attainment of Project objectives and might be more costly.

12. The OIAA Commission finds that there is substantial evidence in the entirety of the record to support the conclusions and Findings before this Commission.

13. The OIAA Commission finds that significant new information has not been

added to the Final EIR (SCH#2021100226) since the circulation of the Draft EIR such that recirculation for additional public review is necessary pursuant to State CEQA Guidelines Section 15088.5. The Commission further finds that no information has been presented showing new significant effects and that no feasible alternative that would clearly lessen the significant physical environmental effects identified in the Final EIR has been proposed and rejected by this Commission.

14. The OIAA Commission finds that the Final EIR (SCH#2021100226) reflects the independent review and judgment of the Commission and OIAA.

15. The OIAA Commission finds that the Final EIR (SCH#2021100226) serves as adequate, complete, and appropriate environmental documentation for the Project.

16. The OIAA Commission certifies the Final EIR (SCH#2021100226) as complete and adequate in that Final EIR (SCH#2021100226) addresses all environmental effects of the Project and fully complies with the requirements of CEQA and the State CEQA Guidelines.

17. Effective Date. This Resolution will take effect immediately upon its adoption.

18. Certification. The Secretary/Assistant Secretary shall certify as to the adoption of this Resolution.

PASSED, APPROVED, AND ADOPTED at a Regular Meeting this 22nd day of June, 2023.

ALAN D. WAPNER, OIAA PRESIDENT

ATTEST:

NORMA I. ALLEY, MMC, ASSISTANT SECRETARY

APPROVED AS TO LEGAL FORM:

LORI D. BALLANCE, GENERAL COUNSEL
STATE OF CALIFORNIA COUNTY OF SAN BERNARDINO CITY OF ONTARIO

I, Norma I. Alley, MMC, Board Clerk/Assistant Secretary of the Ontario International Airport Authority, DO HEREBY CERTIFY the foregoing Resolution No. 2023-10 is the original and was duly passed and adopted by the Commission of the Ontario International Airport Authority at their Regular Meeting held on June 22, 2023, by the following roll call vote, to wit:

)

- AYES: COMMISSIONERS:
- NOES: COMMISSIONERS:
- ABSENT: COMMISSIONERS:

NORMA I. ALLEY, MMC BOARD CLERK/ASSISTANT SECRETARY

(SEAL)

EXHIBIT "A"

Mitigation Monitoring and Reporting Program for the Final EIR (SCH#2021100226)

4.1 INTRODUCTION

Public Resources Code Section 21081.6 and Section 15097 of the California Environmental Quality Act (CEQA) Guidelines require public agencies to establish monitoring or reporting programs for projects approved by a public agency whenever approval involves the adoption of either a mitigated negative declaration or specified environmental findings related to environmental impact reports.

This is the Mitigation Monitoring and Reporting Program (MMRP) for the South Airport Cargo Center Project (proposed Project). The intent of the MMRP is to ensure the successful implementation of the mitigation measures identified in the Final Environmental Impact Report (Final EIR) for the proposed Project.

4.2 MITIGATION MEASURES

The mitigation measures in the Final EIR are assigned the same title as in the Draft EIR. Additional new mitigations measures from **Section 2.0: Additions and Corrections** are assigned the same title as in the Final EIR. The MMRP shown in **Table 4-1: Mitigation Monitoring and Reporting Program** describes the actions to be taken to implement each mitigation measure, the timing of these actions, the entities responsible for implementing and monitoring the actions, and, where appropriate, the entities responsible for ensuring that reporting responsibilities are carried out.

In some instances, mitigation measures require a monitor or community liaison to be retained or designated. In those instances, the person or entity designated to perform this task shall be identified to OIAA, and OIAA will review that person or entity's qualifications to confirm that the designated party has the requisite expertise or qualifications.

The MMRP also includes a table identifying "Project Design Features" (PDFs) (**Table 4-2: Project Design Features**). This information is included for convenience and comprehensiveness. These PDFs are not "mitigation measures" as defined by CEQA. PDFs consist of elements or features that have been incorporated into the proposed Project that may reduce the environmental effects of the proposed Project. These PDFs are included in a separate table in order to ensure that these features are implemented.



Page 43

4.3 MMRP COMPONENTS

The components of the attached tables, which contain applicable mitigation measures and project design features, are addressed briefly, below:

- <u>Impact</u>: This column summarizes the impact stated in the Draft EIR, as revised in the Final EIR.
- <u>Mitigation Measure</u>: All mitigation measures identified in the Draft EIR, as revised in the Final EIR, are presented and titled accordingly.
- <u>Implementing Party</u>: The column entitled "Implementing Party" identifies the entity that will undertake the required action. The Implementing Party is most often the Contractor/Applicant, who will be responsible for the design, construction, or operation of each site, phase, or component of the proposed Project. The Contractor/Applicant responsible for undertaking a required action may include the owner or operator of the proposed Project component, as appropriate. In some instances, the required action will or should be undertaken by another party. This column therefore provides clarity regarding the entity that is primarily responsible for carrying out the action.
- <u>Monitoring Party</u>: This column identifies the specific party responsible for monitoring. OIAA is primarily responsible for monitoring that mitigation measures are successfully implemented. Though the proposed Project is within the Airport, the Airport is located in the City of Ontario. Therefore, the City of Ontario would have responsibility for monitoring some aspect of the overall proposed Project.
- <u>Timeframe/Monitoring Milestone</u>: Implementation of the action must occur prior to or during some part of Project approval, Project design or construction, or during ongoing Project operations. The timing and the monitoring milestone for each measure is identified in this column.
- <u>Reporting Requirements</u> (if applicable): Certain measures identify reporting responsibility. In those instances, the MMRP identifies the report or plan that must be prepared so that the monitoring party can confirm that the required action(s) have been implemented. This column also provides additional information that provide clarity concerning how the measure will be carried out.



Page 44

TABLE 4-1 MITIGATION MONITORING AND REPORTING PROGRAM						
Impact		Mitigation Measures	Implementing Party	Monitoring Party	Timeframe/ Monitoring Milestone	Reporting Requirements (if applicable)
		AIR QUALITY				
Threshold AQ-2: Result in a cumulatively considerable net increase of any criteria pollutant for which the project region is nonattainment under an applicable federal or State ambient air quality standard?	MM AQ-1:	The Applicant shall require that construction vendors, contractors, and/or haul truck operators commit to using 2010 model year trucks (e.g., material delivery trucks and soil import/export with a gross vehicle weight rating of at least 14,001 pounds), that meet CARB's 2010 engine emissions standards or newer, cleaner trucks. The OIAA shall confirm that the Applicant includes this requirement in applicable bid documents, purchase orders, and contracts. Operators shall maintain records of all trucks associated with Project construction to document that each truck used meets these emission standards and make the records available for inspection.	OIAA & Applicant	OIAA	Prior to and during construction.	 Maintain records of applicable bid documents, purchase orders, and contracts. Records maintained for all trucks and make them available for inspection.
	MM AQ-2:	The Applicant shall require that construction equipment such as concrete/industrial saws, pumps, aerial lifts, light stands, air compressors, and forklifts be electric or alternative-fueled (i.e., non-diesel), where feasible. Pole power shall be utilized at the earliest feasible point in time and shall be used to the maximum extent feasible in lieu of generators.	Applicant	OIAA	Prior to and during construction.	• Maintain records of applicable bid documents, purchase orders, and contracts.
	MM AQ-3:	The Applicant shall support and encourage ridesharing and transit incentives for the construction crew by providing crews with the resources needed to organize rideshares, such as bulletin boards or email announcements. The Applicant shall also partially subsidize transit fares or passes for the construction crew members who can feasibly use transit. The Applicant shall set a goal to achieve ten percent total construction worker participation in ridesharing programs and transit use.	Applicant	OIAA	During construction.	• Maintain records documenting the ridesharing and transit incentive program offered to the construction crew.

SNT

4.0 Mitigation Monitoring and Reporting Program

ImpactMitigation MeasuresImplementing PartyMM AQ-4:The Applicant shall require, in addition to the GSE noted within PDF AQ-3, all other on-site cargo-handling equipment, such as yard trucks, holsters, yard goats, pallet jacks, and similar equipment, to be electric, with the necessary electrical charging stations provided.Applicant	Monitoring Party OIAA OIAA	Timeframe/ Monitoring Milestone During operation.	Reporting Requirements (if applicable)• Maintain logs documenting engine types for all on-site, cargo-handling equipment.
MM AQ-4: The Applicant shall require, in addition to the GSE noted Applicant within PDF AQ-3, all other on-site cargo-handling equipment, such as yard trucks, holsters, yard goats, pallet jacks, and similar equipment, to be electric, with the necessary electrical charging stations provided.	OIAA OIAA	During operation.	• Maintain logs documenting engine types for all on-site, cargo-handling equipment.
	OIAA	During operation	
MM AQ-5:The Applicant shall require, if and to the extent feasible, the use of zero-emission or near zero-emission on-road heavy duty trucks as part of business operations beginning in 2025 (within at least 25 percent of the Project fleet).Applicant		During operation	• Maintain annual logs documenting delivery truck fleet composition, including engine types, beginning in 2025 and for
The Applicant also shall require, if and to the extent feasible, the use of zero-emission or near zero emission on-road heavy duty trucks as part of the business operations beginning in 2029 (within at least 50 percent of the Project fleet).			all Project operational years subsequent thereto.
MM AQ-6:The Applicant shall include in the design requirements for the Project, cool roof installation to the extent roof space is not occupied by solar panels, in order to reduce energy use and urban heat island effects.Applicant	OIAA	Pre-construction and prior to certificate of occupancy.	• Review and approval of plans.
MM AQ-7: The Applicant shall encourage the use of single engine taxi Applicant operations for Project aircraft.	OIAA	During operation.	• Maintain records documenting encouraged taxi operation protocols for Project aircraft.
MM AQ-8: The Applicant shall utilize Energy Star heating, cooling, and Applicant lighting devices, and appliances (e.g., Heating, Ventilation, and Air Conditioning (HVAC) units in the form of energy efficient commercial heat pumps) within the interior of the Air Cargo Sort Building.	OIAA	Pre-construction and prior to certificate of occupancy.	• Review and approval of plans.
MM AQ-9: In order to reduce trips to and from the Project site during Applicant construction, the Applicant shall provide on-site food trucks during meal times.	OIAA	During construction.	• Maintain records documenting presence of on-site food vendors.



TABLE 4-1 MITIGATION MONITORING AND REPORTING PROGRAM						
Impact	Mitigation Measures	Implementing Party	Monitoring Party	Timeframe/ Monitoring Milestone	Reporting Requirements (if applicable)	
MM AQ	10: Interior- and exterior-facing signs, including signs directed toward all dock and delivery areas, shall be posted by the Applicant to identify contact information to report idling violations to CARB, SCAQMD, and the building manager. These signs also shall inform truck drivers to shut off their engines when not in use.	Applicant	OIAA	During operation.	• Maintain logs documenting signage content, posting locations, and posting dates.	
MM AQ	11: Electric plugs for electric transport refrigeration units shall be provided at dock doors located at the Air Cargo Sort Building.	Applicant	OIAA	Prior to certificate of occupancy.	• Maintain logs documenting installation and location of plugs for electric transport refrigeration units.	
MM AQ	12 : The Applicant shall train operational managers and employees on efficient scheduling and load management to eliminate unnecessary queuing of trucks.	Applicant	OIAA	During operation.	 Maintain logs documenting frequency of trainings and attendee names. 	
MM AQ	13: Signs shall be posted by the Applicant at every truck exit driveway providing directional information to use truck routes as designated by the City of Ontario.	Applicant & OIAA	OIAA	During operation.	• Maintain logs documenting signage content, posting locations, and posting dates.	
MM AQ	14: The Applicant shall require its facility operator(s) to train the staff in charge of keeping vehicle records on diesel technologies and compliance with CARB regulations by attending CARB-approved courses. Also, the Applicant shall require its facility operator(s) to maintain records on-site demonstrating regulatory compliance and make records available for inspection by OIAA, SCAQMD, and State of California upon request.	Applicant	OIAA	During operation.	 Maintain logs documenting frequency of trainings and attendee names. Records on-site demonstrating compliance and to be made available for inspection. 	
MM AQ	15: The Applicant shall include a provision in all operational freight hauling contracts requiring the use of 2010 model year trucks that meet CARB's 2010 engine emission standards, or newer and cleaner trucks, if and to the extent feasible.	Applicant	OIAA	During operation.	• Maintain logs documenting applicable contractual term(s) and model years comprising the trucking fleet.	

TABLE 4-1 MITIGATION MONITORING AND REPORTING PROGRAM						
Impact		Mitigation Measures	Implementing Party	Monitoring Party	Timeframe/ Monitoring Milestone	Reporting Requirements (if applicable)
	MM AQ-16:	During construction, the Applicant shall post interior- and exterior-facing signs to inform construction contractors to shut off truck and equipment engines when not in use.	Applicant	OIAA	During construction.	 Maintain logs documenting signage content, posting locations, and posting dates.
	MM TRANS- TRANS-5.	-1, MM TRANS-2, MM TRANS-3, MM TRANS-4, and MM	See under Threshold TRA- 2.	See under Threshold TRA- 2.	See under Threshold TRA-2.	• See under Threshold TRA-2.
		BIOLOGICAL RESOURCE	CES			
Threshold BIO-1: Have a substantial adverse effect, either directly or through habitat modifications, on any species identified as a candidate, sensitive, or special status species in local or regional plans, policies, or regulations, or by the California Department of Fish and Game or U.S. Fish and Wildlife Service?	MM BIO-1:	 Burrowing Owl. All disturbed areas of the Project site that were determined to have a low potential to provide suitable habitat for burrowing owls, which primarily includes the existing track infield grassy areas of the Project site, require focused preconstruction surveys to be conducted; the first take avoidance survey shall be conducted within 14 days prior to ground disturbance and the second take avoidance survey shall be conducted 24 hours prior to ground disturbance to determine presence of burrowing owls. These surveys shall conform to the survey protocol established by the CDFW Staff Report on Burrowing Owl Mitigation (CDFW 2012) and will be conducted by a qualified biologist across all suitable breeding, wintering, and foraging habitat within the Project and appropriate buffer. Copies of the survey results shall be submitted to CDFW and OIAA. If no burrowing owls are detected, no further mitigation is necessary. If burrowing owls are detected during focused surveys and/or take avoidance surveys, CDFW will be immediately informed of its location and status. The project will avoid all impacts to burrowing owls onsite. If 	Applicant & Qualified biologist	Qualified Biologist, OIAA, USFWS, and CDFW if BUOW found	 Prior to construction activities: 14 days prior to ground disturbance 24 hours prior to ground disturbance 	 Survey results submitted to CDFW and OIAA BUOW Protection and Relocation Plan (if applicable)



	TABLE 4-1 MITIGATION MONITORING AND REPORTING PROGRAM							
Impact	Mitigation Measures	Implementing Party	Monitoring Party	Timefra Monitoring N				
	this is not feasible, a Burrowing Owl Protection Plan will be prepared by a qualified biologist, which must be approved by CDFW prior to initiating the project. The Burrowing Owl Protection Plan will include conserving all nesting, occupied, and satellite burrows and/or burrowing owl habitat such that the habitat acreage, number of burrows, and burrowing owls impacted are maintained and/or replaced. Further coordination with CDFW will occur to identify mitigation for the loss of habitat through the acquisition, conservation, and management of in-kind habitat. Lands conserved will include 1) sufficiently large acreage with fossorial mammals present; 2) permanent protection through a conservation easement for the purpose of conserving burrowing owl habitat and prohibiting activities incompatible with burrowing owl use; 3) development and implementation of a mitigation land management plan to address long-term ecological sustainability and maintenance of the site for burrowing owls; and 4) funding for the maintenance and management of mitigation land through the establishment of a long-term funding mechanism such as an endowment (CDFW, 2012).							
MM BIO-2:	Nesting Birds. Bird nesting season generally extends from February 1 through September 15 in southern California and, specifically, April 15 through August 31 for migratory passoring birds and	Applicant & Qualified biologist	Qualified Biologist, OIAA, and USFWS if active	During nestin season: • Pre-const				

April 15 through August 31 for migratory passerine birds and January 15 to August 31 for raptors. In order to ensure compliance with the Migratory Bird Treaty Act and to avoid impacts to nesting birds (common and special status) during the nesting season, a qualified Avian Biologist must be

nests found

4.0 Mitigation Monitoring and Reporting Program

ame/ Milestone **Reporting Requirements** (if applicable)

iring nesting •	Pre-construction survey report,
ason:	and reports for any necessary
Pre-construction	subsequent surveys
survey no more	
than three days	
prior to the	
commencement of	

TABLE 4-1 MITIGATION MONITORING AND REPORTING PROGRAM

MITIGATION MONITORING AND REPORTING PROGRAM								
Impact	Mitigation Measures	Implementing Party	Monitoring Party	Timeframe/ Monitoring Milestone	Reporting Requirements (if applicable)			
	retained to conduct pre - construction Nesting Bird Surveys (NBS) will occur prior to Project - related disturbance to nestable vegetation to identify any active nests. The NBS shall be performed no more than three days prior to the commencement of construction activities. The survey(s) will occur at the appropriate time of day/night, during appropriate weather conditions. Surveys will encompass all suitable areas, including trees, shrubs, bare ground, burrows, cavities, and structures. Survey duration will take into consideration the acreage of the Project impacts; density, and complexity of the habitat; number of survey participants; survey techniques employed; and will be sufficient to ensure the data collected is complete and accurate. Pre-construction surveys will focus on both direct and indirect evidence of nesting, including nest locations and nesting behavior (i.e., copulation, carrying of food or nest materials, nest building, removal of fecal sacks, flushing suddenly from atypically close range, agitation, aggressive interactions, feigning injury or distraction displays, or other behaviors). The results of the NBS shall be documented by the qualified biologist. If construction is inactive for more than seven days, an additional survey shall be conducted. If no active nests are found, no further action will be required. If a nest is suspected, but not confirmed, the qualified biologist will establish a disturbance-free buffer until additional surveys can be completed, or until the location can be inferred based on observations. The qualified biologist will not risk failure of the nest to determine the exact location or status and will make every effort to limit the nest to potential predation as a result of the survey/monitoring efforts (i.e., limit number of surveyors, limit time spent at/near the nest, scan the site for potential nest predators before approaching, or immediately			construction activities • Additional surveys required if construction is inactive for more than seven days during breeding season				



TABLE 4-1MITIGATION MONITORING AND REPORTING PROGRAM

Impact	Mitigation Measures	Implementing Party	Monitoring Party	Timef Monitoring
	depart nest area if indicators of stress or agitation are			
	displayed). If a nest is observed, but thought to be inactive,			
	the qualified biologist will monitor the nest for 1 hour (4 hours			
	for raptors during the nonbreeding season) prior to			
	approaching the nest to determine status. The qualified			
	biologist will use their best professional judgement regarding			
	the monitoring period and whether approaching the nest is			
	appropriate. If an active nest is found, the biologist will set			
	appropriate no - work buffers (typically 300 feet for passerine			
	and non-special-status species, and 500 feet for hawks and			
	special-status species) around the nest, which will be based			
	upon the nesting species, its sensitivity to disturbance,			
	nesting stage and expected types, intensity, and duration of			
	disturbance – typically 300 feet of a migratory bird and 500			
	feet for raptors. Once the buffer is established, the qualified			
	biologist will document baseline behavior, stage of			
	reproduction, and existing site conditions, including vertical			
	and horizontal distances from proposed work areas, visual or			
	acoustic barriers, and existing level of disturbance. Following			
	documentation of baseline conditions, the qualified biologist			
	may choose to make adjustments to the buffer based on site			
	characteristics, stage of reproduction, and types of Project			
	activities proposed at/near that location. The qualified			
	biologist will monitor the nest at the onset of Project activities,			
	and at the onset of any changes in Project activities (i.e.,			
	increase in number or type of equipment, change in			
	equipment usage, etc.) to determine the efficacy of the buffer.			
	If the qualified biologist determines that Project activities may			
	be causing an adverse reaction, the qualified biologist will			
	adjust the buffer accordingly. The qualified biologist will be			
	onsite daily to monitor all existing nests, the efficacy of			
	established buffers, and to document any new nesting			

4.0 Mitigation Monitoring and Reporting Program

frame/ g Milestone Reporting Requirements (if applicable)



TABLE 4-1
MITIGATION MONITORING AND REPORTING PROGRAM

Impact		Mitigation Measures	Implementing Party	Monitoring Party	Timeframe/ Monitoring Milestone	Reporting Requirements (if applicable)
		occurrences. The qualified biologist will document the status of all existing nests, including the stage of reproduction and the expected fledge date. If a nest is suspected to have been abandoned or failed, the qualified biologist will monitor the nest for a minimum of 1 hour (4 hours for raptors), uninterrupted, during favorable field conditions. If no activity is observed during that time, the qualified biologist may approach the nest to assess the status. The permittee, under the direction of the qualified biologist, may also take steps to discourage nesting on the Project site, including moving equipment and materials daily, covering material with tarps or fabric, and securing all open pipes and construction materials. The qualified biologist will ensure that none of the materials used pose an entanglement risk to birds or other species. The buffer shall remain until the young have fledged the nest and the nest is confirmed to no longer be active, or as determined by the qualified biologist. The nests and buffer zones shall be field checked weekly by a qualified biological monitor. The approved no - work buffer zone shall be clearly marked in the field, within which no disturbance activity shall commence until the qualified biologist has determined the young birds have successfully fledged and the nest is inactive.				
	MM BIO-3:	Roosting Bats. A pre-construction bat roosting survey shall be conducted by a qualified bat biologist on structures and trees being removed or impacted by construction on site that may provide suitable roosting opportunities for local common bat species within 14 days prior to construction. If bats are determined to be present, CDFW shall be consulted on	Applicant & Qualified bat biologist	Qualified Bat Biologist, OIAA, USFWS, and CDFW if bats found	 Prior to construction activities: 14 days prior to ground disturbance 	 Survey results submitted to CDFW and OIAA Bat Mitigation Plan (if applicable)

creating a bat mitigation plan.

TABLE 4-1 MITIGATION MONITORING AND REPORTING PROGRAM							
Impact	Mitigation Measures	Implementing Party	Monitoring Party	Timeframe/ Monitoring Milestone	Reporting Requirements (if applicable)		
	Cultural Resou	JRCES					
Threshold CUL-2: A Cause a substantial adverse change in the significance of an archaeological resource pursuant to § 15064.5?	 MM CUL-1: Archaeological Monitoring of All Ground-Disturbit Activities During Construction of Phase 1 and Phase 2. 1. Prior to the issuance of grading permits by the City Ontario for Phase 1 and Phase 2 of the proposed Projet the OIAA and/or its construction contractor must retain qualified professional archeologist meeting the Secret of Interior's PQS for Archaeology (as defined in the Coof Federal Regulations, 36 CFR Part 61). The qualifier archaeologist will be retained to conduct monitoring rough grading activities conducted during both Projephases. The qualified archaeologist shall have the author to redirect earthmoving activities in the event the suspected cultural resources are unearthed duri construction activities. 2. The qualified archaeologist shall prepare a Cultur Resources. 	ng OIAA of ect, in a ary de ed of ect rity nat ng	OIAA and qualified professional archaeologist	Prior to issuance of grading permits for Phase 1 and Phase 2.	• A Cultural Resources Monitoring and Treatment Plan prepared by a qualified archaeologist.		
	 Resources Monitoring and Treatment Plan that of describe processes for archaeological monitoring and handling incidental discovery of objects, features, a cultural resources for all ground-disturbing construction and preconstruction activities. 3. Prior to the issuance of a grading permit, all construction workers involved with grading and trenching operation shall receive training by the qualified archaeologist recognize unique archaeological resources, including trill cultural resources, should such resources be unearth during ground-disturbing construction activities. T training of all construction workers involved with grading 	will for nd on on ons to oal ed The ng					
	and trenching operations shall explain the importance a legal basis for the protection of significant archaeologi resources. It will include a brief review of the cultu	nd cal ıral					

TABLE 4-1MITIGATION MONITORING AND REPORTING PROGRAM

Impact	Mitigation Measures	Implementing Party	Monitoring Party	Timefr Monitoring
	sensitivity of the construction area and the su area; what resources could potentially be identifi earthmoving activities; the requirements of the m program; the protocols that apply in the event in discoveries of cultural resources are identified, who to contact and appropriate avoidance meas the find(s) can be properly evaluated; and a appropriate protocols. All new construction involved with grading and trenching operations t work following the initial training session must training prior to beginning work; the archaeologist shall be available to provide the tr an as needed basis.	urrounding fied during monitoring hadvertent , including isures until any other personnel that begin at take the qualified training on		
	4. In the event subsurface artifacts or feat encountered during ground-disturbing activ construction supervisor shall be required by his o immediately halt and redirect grading operation 100-foot radius of the discovery and see identific evaluation of the suspected resource by the archaeologist for listing in the NRHP and CI requirement shall be noted on all grading plan construction contractor shall be obligated to co the note.	tures are vities, the contract to ns within a cation and equalified CRHR. This ns and the comply with		
	5. After the qualified archaeologist makes his/l assessment of the nature of the find. The arch shall pursue either protection in place or recovery and treatment of the deposits. Recovery, salv treatment protocols shall be developed in accord applicable provisions of Public Resource Cod	'her initial haeologist y, salvage, vage, and dance with de Section		

21083.2 and State CEQA Guidelines 15064.5 and 15126.4 in consultation with OIAA or with a recognized scientific or

4.0 Mitigation Monitoring and Reporting Program

rame/ | Milestone Reporting Requirements (if applicable)

TABLE 4-1 MITIGATION MONITORING AND REPORTING PROGRAM

Impact	Mitigation Measures	Implementing Party	Monitoring Party	Timeframe/ Monitoring Milestone	Reporting Requirements (if applicable)
	educational repository, including the SCCIC. Preservation in place shall be the preferred means to avoid impacts to archaeological resources qualifying as historical resources, consistent with CEQA Guidelines Section 15126.4(b)(3)(C).				
	Geology and Soi	LS			
Threshold GEO-1:					
 Directly or indirectly cause potential substantial adverse effects, including the risk of loss, injury, or death involving: 					
ii. Strong seismic ground shaking? M	M GEO-5: Geotechnical Investigation Recommendations. The proposed Project shall implement and incorporate the recommendations in the Geotechnical Investigation, Section 5.0 Conclusion and Recommendations (see pages 7 through 24 of Appendix 5.6-1 of this EIR). Prior to contract bidding, site grading and foundation plans shall be reviewed and approved by Cotton, Shires and Associates, Inc. or a certified Geologist, for consistency with the Geotechnical Investigation recommendations.	Applicant & OIAA	OIAA and certified geologist.	Prior to contract bidding and ground disturbance.	 Site grading and foundation plans shall be reviewed and approved by Cotton, Shires and Associates, Inc. or a certified Geologist, for consistency with the Geotechnical Investigation recommendations.
iii. Seismic-related ground failure, Im including liquefaction and lateral spreading?	plementation of MM GEO-5 .	See above under Threshold GEO-1 (ii).	See above under Threshold GEO-1 (ii).	See above under Threshold GEO-1 (ii).	See above under Threshold GEO-1 (ii).
Threshold GEO-2:ImResult in substantial soil erosion or theloss of topsoil?	plementation of MM GEO-5 .	See above under Threshold GEO-1 (ii).	See above under Threshold GEO-1 (ii).	See above under Threshold GEO-1 (ii).	See above under Threshold GEO-1 (ii).



4.0 Mitigation Monitoring and Reporting Program

TABLE 4-1 MITIGATION MONITORING AND REPORTING PROGRAM						
Impact		Mitigation Measures	Implementing Party	Monitoring Party	Timeframe/ Monitoring Milestone	Reporting Requirements (if applicable)
Threshold GEO-3: Be located on a geologic unit or soil that is unstable, or that would become unstable as a result of the project, and potentially result in on- or off-site landslide, lateral spreading, subsidence, liquefaction or collapse?	Implementation of	MM GEO-5.	See above under Threshold GEO-1 (ii).	See above under Threshold GEO-1 (ii).	See above under Threshold GEO-1 (ii).	See above under Threshold GEO-1 (ii).
Threshold GEO-4: Be located on expansive soil, as defined in Table 18-1-B of the Uniform Building Code (1994), creating substantial direct or indirect risks to life or property?	Implementation of	MM GEO-5.	See above under Threshold GEO-1 (ii).	See above under Threshold GEO-1 (ii).	See above under Threshold GEO-1 (ii).	See above under Threshold GEO-1 (ii).
Threshold GEO-6: Directly or indirectly destroy a unique paleontological resource or site or unique geologic feature?	MM GEO-1: Pale (PRN A priest com pale Pale Resc prop requised in area pote conc dete the visua side	eontological Resources Mitigation and Monitoring Plan MMP). Professional paleontologist shall be retained to monitor h-disturbing construction activities. Prior to the mencement of ground-disturbing activities, the qualified eontologist, meeting the Society of Vertebrate eontology (SVP) Standards, must prepare a Paleontological burces Mitigation and Monitoring Plan (PRMMP) for the bosed Project. The PRMMP shall describe the monitoring uired during excavations that extend into Pleistocene iment, at approximately 9 feet bgs, and the location of as deemed to have a high paleontological resource ential. The results of the geotechnical investigation ducted for the proposed Project shall be consulted to ermine the approximate depth of Pleistocene sediment in Project site. Paleontological monitoring shall entail the al inspection of excavated and graded areas and trench ewalls. If the qualified Paleontologist determines full-time	OIAA	OIAA and qualified paleontologist.	Prior to and during earth-disturbing activities.	 Paleontological Resources Mitigation and Monitoring Plan (PRMMP)



	TABLE 4-1 MITIGATION MONITORING AND REPORTING PROGRAM					
Impact	Mitigation Measures	Implementing Party	Monitoring Party	Timefra Monitoring		
	monitoring is no longer warranted, based on the geologic conditions at depth, he or she may recommend that monitoring be reduced or cease entirely.					
MM GEO-2	Workers Environmental Awareness Program (WEAP).	Applicant	OIAA	Prior to the		
	Prior to the start of the proposed Project ground-disturbing activities, all field personnel shall receive a worker's environmental awareness training on paleontological resources. The training must provide a description of the laws and ordinances protecting fossil resources, the types of fossil resources that may be encountered in the proposed Project area, the role of the paleontological monitor, outline steps to follow in the event that a fossil discovery is made and provide contact information for the qualified Paleontologist. The training must be developed by the qualified Paleontologist and can be delivered concurrent with other training.			ground dist		
MM GEO-3	 Fossil Discoveries. In the event that a paleontological resource is discovered, the Paleontological monitor shall have the authority to temporarily divert the construction equipment around the find until it is assessed for scientific significance and, if appropriate, collected. If the resource is determined to be of scientific significance, the Paleontologist shall complete the following: 1. Salvage of Fossils. If fossils are discovered, all work in the immediate vicinity shall be halted to allow the paleontologist to evaluate the discovery and determine if the fossil may be considered significant. If the fossils are determined to be potentially significant, the Project-qualified Paleontologist shall recover them following standard field procedures for collecting paleontological as 	OIAA	OIAA and qualified paleontologist.	During g disturbanc paleonto resource discove		

rame/ Milestone Reporting Requirements (if applicable)

ne start of N/A sturbance.

groundIf fossils are discovered, fossilace, if anyspecimens must be delivered toologicalthe accredited museum orces arerepository no later than 90 daysvered.after all fieldwork is completed



	TABLE 4-1 MITIGATION MONITORING AND REPORTING PROGRAM				
Impact	Mitigation Measures	Implementing Party	Monitoring Party	Timefra Monitoring N	
	outlined in the PRMMP prepared for the project. Typically, fossils can be safely salvaged quickly by a single paleontologist and not disrupt construction activity. In some cases, larger fossils, such as complete skeletons or large mammal fossils, require more extensive excavation and longer salvage periods. In this case the Paleontologist shall have the authority to temporarily direct, divert or halt construction activity to ensure that the fossil(s) can be removed in a safe and timely manner.				
	2. Fossil Preparation and Curation. The PRMMP must identify a museum that has agreed to accept fossils that may be discovered during project-related excavations. Upon completion of fieldwork, all significant fossils collected must be prepared in a properly equipped laboratory to a point ready for curation. Preparation may include the removal of excess matrix from fossil materials and stabilizing or repairing specimens. During preparation and inventory, the fossils specimens must be identified to the lowest taxonomic level practical prior to curation at an accredited museum. The fossil specimens must be delivered to the accredited museum or repository no later than 90 days after all fieldwork is completed. The cost of curation shall be assessed by the repository and shall be the responsibility of the client.				
MM GEO-4:	Final Paleontological Mitigation Report. Upon completion of ground disturbing activity, and curation of fossils if necessary, the qualified Paleontologist shall prepare a final mitigation and monitoring report outlining the results of the mitigation and monitoring program. The report shall include discussion of the location, duration and methods of the monitoring, stratigraphic sections, any recovered fossils, and	OIAA	OIAA and qualified paleontologist.	After comp ground dis activiti	



ame/ Milestone Reporting Requirements (if applicable)

Final mitigation and monitoring report outlining the results of the mitigation and monitoring program

		TABLE 4-1 MITIGATION MONITORING AND RE	Porting prog	iRAM	
Impact		Mitigation Measures	Implementing Party	Monitoring Party	Timefram Monitoring Mi
		the scientific significance of those fossils, and where fossils were curated.			
		GREENHOUSE GAS EMIS	SIONS		
Threshold GHG-1: Generate greenhouse gas emissions, either directly or indirectly, that may have a significant impact on the environment?	Implementati 5, MM AQ-6 MM AQ-12, TRANS-1, M 5.	on of MM AQ-1, MM AQ-2, MM AQ-3, MM AQ-4, MM AQ- , MM AQ-7, MM AQ-8, MM AQ-9, MM AQ-10, MM AQ-11, MM AQ-13, MM AQ-14, MM AQ-15, MM AQ-16, MM M TRANS-2, MM TRANS-3, MM TRANS-4, and MM TRANS-	See under Threshold AQ-2 and Threshold TRA-2.	See under Threshold AQ-2 and Threshold TRA-2.	See under Thr AQ-2 and Thr TRA-2.
		Hazards and Hazardous	MATERIALS		
Threshold HAZ-2: Create a significant hazard to the public or the environment through reasonably foreseeable upset and accident conditions involving the release of hazardous materials into the environment?	MM HAZ-1:	Soil Management Plan. A Soil Management Plan (SMP) containing soil criteria and soil management and construction risk management protocols to be implemented during proposed Project development shall be prepared prior to disturbance of soils on the site by construction activities and implemented during construction to address any soil containing or suspected to contain PFAs on the proposed Project site and any previously undetected contamination encountered during construction. Special attention shall be made to soils disturbed in the Guardian Jet Center, southern hangar and structure previously housing fire prevention equipment due to the known presence of PFAs in these areas. Additional soil sampling shall be conducted as necessary to delineate the extent of PFAs contamination to enable segregation and proper disposal of any contaminated soil during construction.	OIAA	OIAA	Approvals pr disturbance of and impleme during constr
	MM HAZ-2:	Vapor Intrusion Mitigation System. A vapor intrusion mitigation system (VIM system) shall be installed under Phase II of the proposed Air Cargo Building to	OIAA	OIAA	Approvals p disturbance o

frame/ g Milestone		Report (i	ing Require f applicable	ements e)	
r Threshold I Threshold	See Thre	under shold TF	Threshold RA-2.	AQ-2	and

Soil Management Plan (SMP)
 nce of soils
 elemented
 onstruction.

als prior to • Review and approval of plans. nce of soils

	TABLE 4-1 MITIGATION MONITORING AND REPORTING PROGRAM							
Impact	Mitigation Measures	Implementing Party	Monitoring Party	Timefr Monitoring				
	address the potential for vapor intrusion from the subsurface. Alternatively, a soil vapor extraction remediation system could be utilized to reduce trichloroethene (TCE) and chloroform vapor concentrations through removal of volatile organic compounds (VOCs) in Phase II development area.			and imple during con				
Threshold HAZ-5: For a project located within an airport land use plan or, where such a plan has not been adopted, within two miles of a public airport or public use airport, would the project result in a safety hazard or excessive noise for people residing or working in the project area?	Implementation of MM NOI-1 .	See below under Threshold N-1.	See below under Threshold N-1.	See belo Thresho				

		NOISE			
Threshold N-3:	MM NOI-1:	Residential Sound Insulation Program (RSIP).	OIAA	OIAA	Prior to
Located within the vicinity of a private airstrip or an airport land use plan or, where such a plan has not been adopted, within two miles of a public airport or public use airport, if the project would expose people residing or working in the project area to excessive noise levels.		Non-compatible residential land uses within the 65+ decibel (dB) contour with habitable areas inside the home with average noise levels of 45 dB or greater with all windows closed would be eligible for the RSIP. The goal of the Program is to reduce the interior noise level within affected homes by at least five (5) decibels (dB). The results may vary depending upon the existing structural characteristics of the home. In order to achieve this goal, modifications may include the retrofit of exterior doors and windows, installation of a ventilation system, and other miscellaneous treatments. The RISP would include the following: A noise audit will be conducted for each home in the RISP to measure the noise reduction properties of a residence in its			constru



rame/ | Milestone Reporting Requirements (if applicable)

emented nstruction.

ow under See below under Threshold N-1. old N-1.

Project uction.	•	A noise audit conducted for each home in the RISP
	•	Signature of an avigation easement by Property Owners who qualify for the RSIP
	•	Upon completion of the RSIP, disclosure by current owners that the residence was included in the RISP and is subject to an avigation easement.

TABLE 4-1
MITIGATION MONITORING AND REPORTING PROGRAM

Impact	Mitigation Measures	Implementing Party	Monitoring Party	Timefra Monitoring I
	existing condition to confirm that average interior aircraft sound levels are greater than a Community Noise Equivalent Level (CNEL) of 45 decibels (dB), and to provide an indication of the potential effectiveness of noise reducing treatments.			
	The goal of the RISP is to reduce the average interior CNEL of habitable rooms by a minimum of 5 dB (i.e., a clearly detectable reduction), and reduce the average interior CNEL of habitable rooms to below 45 dB.			
	Sound levels will be measured using aircraft as the noise source or simulation methods (loudspeaker(s)).			
	Property owners will be required to sign an avigation easement, guaranteeing the right of flight over a residence, as a requirement to participate in the RISP.			
	Upon completion, current owners will be required to disclose the residence was included in the RISP and is subject to an avigation easement.			
	If housing units do not meet the local building codes required to qualify for sound insulation, a homeowner shall be given the option to sell the property. The residence may be resold to a new owner. The housing unit may or may not be sound insulated and/or upgraded prior to resale but will be subject to an avigation easement.			
	Transportation			
Threshold TRA-2: Would the project conflict or be inconsistent with CEQA Guidelines section 15064.3, subdivision (b)?	MM TRANS-1: Voluntary Commute Reduction Program. The proposed Project shall implement Voluntary Commute Trip Reduction (CTR) programs that discourage single- occupancy vehicle trips and encourage alternative modes of transportation, such as carpooling, taking transit, walking, and	Applicant	OIAA	During P operat

rame/ Milestone Reporting Requirements (if applicable)

Project ation. N/A



Impact	Mitigation Measures	Implementing Party	Monitoring Party	Timef Monitoring		
	biking. Voluntary CTR programs shall include the following elements to apply the VMT reductions reported in literature:					
	 Employer-provided services, infrastructure, and/or incentives for commuting to work using alternative modes (e.g., walking, biking, carpooling/vanpooling, or taking transit). 					
	 Provide information, coordination, and marketing for employee rideshare services, provide onsite infrastructure to support carpools/vanpools, and provide incentives (e.g., free transit passes, monthly bonus for carpooling 3 or more times a week, etc.). 					
	Employer costs may include recurring costs for carpool/vanpool subsidies, capital and maintenance costs for the alternative transportation infrastructure (e.g., showers and lockers), and labor costs for staff to manage the program.					
MM TRA	NS-2: Provide Ridesharing Program.	Applicant	OIAA	During		
	A ridesharing program shall be implemented for employees of the site. The following elements designed to support the Project's ridesharing program:			opera		
	 Provide vanpool parking with designated passenger loading/unloading area near employee entrance. 					
	Create a Carpool Incentive Program.					
	 Provide a minimum of ten (10) carpool parking spaces provided closer to the employee entrance than standard parking spaces. Provide access to a carpool database (Metro rideshare) and/or an on-site matching program for employees. 					





frame/ g Milestone Reporting Requirements (if applicable)

g Project ration.

N/A

	TABLE 4-1 MITIGATION MONITORING AND REPORTING PROGRAM				
Impact	Mitigation Measures	Implementing Party	Monitoring Party	Timefr Monitoring	
	 Provide a monthly incentive for employees that carpool a minimum of three (3) days per week (e.g., \$50 gas card or a \$50 green commuter bonus). 				
	In addition, a staff person would be designated to for provide rideshare information to employees and monitoring the effectiveness of the program.				
	It is assumed all employees are eligible and that additional carpool spaces could be designated if warranted by demand.				
ММ	TRANS-3: Implement Subsidized or Discounted Transit Program.	Applicant	OIAA	During I	
	Subsidized, discounted, or free Omnitrans, Metrolink or Amtrak transit passes shall be provided to employees to encourage use of transit routes/stops located less than a mile from the Project. It is assumed free transit passes are available to all employees.			opera	
	Based on the given shift times of the Project, shifts that start or end at 11:00 PM shall have limited available options as most routes do not provide service that late. This shall limit approximately half the employees from the ability to rely on transit.				
ММ	TRANS-4: Bicycle Facilities.	Applicant	OIAA	During I	
	On-site bicycle parking and end-of-trip facilities shall be provided for employee use. End-of-trip facilities include bike parking, bike lockers, showers, and personal lockers.			opera	
	A bike share program (standard or electric bikes) for employees shall supplement bicycle facilities.				
ММ	TRANS-5: Employer-Sponsored Vanpool Program.	Applicant	OIAA	During I	
	An employer-sponsored vanpool service shall be implemented and be fully funded by the tenant as follows:			opera	

SNT

4.0 Mitigation Monitoring and Reporting Program

ame/
Milestone

Reporting Requirements (if applicable)

Project ation.	N/A	
Project ation.	N/A	
Project ation.	N/A	

	TABLE 4-1 MITIGATION MONITORING AND RE	PORTING PROC	GRAM	
Impact	Mitigation Measures	Implementing Party	Monitoring Party	Timefra Monitoring
	 Provide a minimum of one (1) and up to three (3) vanpool vehicles and associated parking with designated passenger loading/unloading area near employee entrance. 			
	• Pay for the lease of a minimum of one (1) van and up to three (3) vans for the purpose of employee vanpooling3			
	 A ten percent voluntary participation rate is assumed to be the high end of the range for this project. 			
	Tribal Cultural Reso	URCES		
Threshold TRI-1:				
Would the project cause a substantial adverse change in the significance of a tribal cultural resource, defined in Public Resources Code section 21074 as either a site, feature, place, cultural landscape that is geographically defined in terms of the size and scope of the landscape, sacred place, or object with cultural value to a California Native American tribe, and that is:				
 Listed or eligible for listing in the Impleme California Register of Historical Resources, or in a local register of historical resources as defined in Public Resources Code section 5020.1(k), or 	entation of MM CUL-1 .	See above under Threshold CUL-2.	See above under Threshold CUL- 2.	See abov Threshold

ii.	A resource determined by the lead MM TCR-1 :	Retain a Native American Monitor Prior to Commencement	OIAA and	Approved	Prior to
	agency, in its discretion and	of Ground-Disturbing Activities.	appropriate	Native	disturbing
	supported by substantial evidence,				



Page 64

4.0 Mitigation Monitoring and Reporting Program

rame/ Milestone Reporting Requirements (if applicable)

ve under See above under Threshold CUL-2. d CUL-2.

• ground • Executed monitoring agreement g activities.

MITIGATION MONITORING AND REPORTING PROGRAM					
Impact	Mitigation Measures	Implementing Party	Monitoring Party	Timeframe/ Monitoring Milestone	Reporting Requirements (if applicable)
to be significant pursuant to criteria set forth in subdivision (c) of Public Resources Code Section 5024.1. In applying the criteria set forth in subdivision (c) of Public Resource Code Section 5024.1, the lead agency shall consider the significance of the resource to a California Native American tribe.	 a. The project applicant/lead agency shall retain a Native American Monitor from or approved by the appropriate Native American Tribe(s). The monitor shall be retained prior to the commencement of any ground-disturbing activity for the subject project at all project locations (i.e., both onsite and any off-site locations that are included in the project description/definition and/or required in connection with the project, such as public improvement work). "Ground-disturbing activity" shall include, but is not limited to, demolition, pavement removal, potholing, auguring, grubbing, tree removal, boring, grading, excavation, drilling, and trenching. b. A copy of the executed monitoring agreement shall be submitted to the lead agency prior to the commencement of any ground-disturbing activity, or the issuance of any permit necessary to commence a ground-disturbing activity. c. The monitor will complete daily monitoring logs that will provide descriptions of the relevant ground-disturbing activities, the type of construction activities, soil types, cultural-related materials, and any other facts, conditions, materials, or discoveries of significance to the Tribe. Monitor logs will identify and describe any discovered TCRs, including but not limited to, Native American (ancestral) human remains and burial goods. Copies of monitor logs will be provided to the propsed Project 	Native American Tribe(s).	American Monitor.		 Daily monitoring logs performed by the approved Native American Monitor (1) Written confirmation to the appropriate Native American Tribe(s) from a designated point of contact for the proposed Project applicant/lead agency that all ground-disturbing activities and phases that may involve ground-disturbing activities on the project site or in connection with the project are complete; or (2) A determination and written notification by the appropriate Native American Tribe(s) to the proposed Project applicant/lead agency that no future, planned construction activity and/or development/construction phase at the Project site possesses the potential to impact Native American Tribe TCRs.

TABLE 4-1

4.0 Mitigation Monitoring and Reporting Program

Impact	Mitigation Measures	Implementing Party	Monitoring Party	Timefra Monitoring	
	 applicant/lead agency upon written request to the appropriate Native American Tribe(s). d. On-site tribal monitoring shall conclude upon the latter of the following (1) written confirmation to the appropriate Native American Tribe(s) from a designated point of contact for the proposed Project applicant/lead agency that all ground-disturbing activities and phases that may involve ground-disturbing activities on the project site or in connection with the project are complete; or (2) a determination and written notification by the appropriate Native American Tribe(s) to the proposed Project applicant/lead agency that no future, planned construction activity and/or development/construction phase at the Project site possesses the potential to impact Native American Tribe TCRs 				
	e. Upon discovery of any TCRs, all construction activities in the immediate vicinity of the discovery shall cease (i.e., not less than the surrounding 50 feet) and shall not resume until the discovered TCR has been fully assessed by the Native American monitor and/or Native American archaeologist. The appropriate Native American Tribe(s) will recover and retain all discovered TCRs in the form and/or manner the Native American Tribe(s) deem appropriate, in the Tribe's sole discretion, and for any purpose the Native American Tribe(s) deem appropriate, including for educational, cultural and/or historic purposes.				
MM TCR-2	 Unanticipated Discovery of Human Remains and Associated Funerary Objects. a. Native American human remains are defined in PRC 5097.98 (d)(1) as an inhumation or cremation, and in any 	OIAA and appropriate Native American Tribe(s).	Approved Native American Monitor.	During g disturbing a Native Ar	

TABLE 4-1MITIGATION MONITORING AND REPORTING PROGRAM



4.0 Mitigation Monitoring and Reporting Program

rame/ Milestone Reporting Requirements (if applicable)

ground- N/A activities, if merican

	MITIGATION MONITORING AND REPORTING PROGRAM				
Impact	Mitigation N	leasures	Implementing Party	Monitoring Party	Timefra Monitoring N
	state of decompositio objects, called associa Code Section 5097.98 this statute.	n or skeletal completeness. Funerary Ited grave goods in Public Resources 8, are also to be treated according to			human rem found
	b. If Native American h discovered or recogn construction activities Safety Code Section of human skeletal ma to the County Coroner shall immediately hal coroner has determin coroner recognizes th Native American or h American, he or she s hours, the Native Ar Public Resources Cod	uman remains and/or grave goods nized on the project site, then all shall immediately cease. Health and 7050.5 dictates that any discoveries terial shall be immediately reported r and all ground-disturbing activities t and shall remain halted until the ed the nature of the remains. If the ne human remains to be those of a as reason to believe they are Native hall contact, by telephone within 24 merican Heritage Commission, and e Section 5097.98 shall be followed.			
	c. Human remains and alike per California 5097.98(d)(1) and (2).	grave/burial goods shall be treated Public Resources Code section			
	d. Construction activities Project site at a minim human remains and/ Native American Trib that resuming constr acceptable and prov consent of that det mitigation measures Tribe(s) and/or archa Guidelines Section 15	s may resume in other parts of the um of 200 feet away from discovered or burial goods, if the appropriate e(s) determine in its sole discretion uction activities at that distance is ides the project manager express ermination (along with any other the appropriate Native American eologist deems necessary). (CEQA 064.5(f)).			

TABLE 4-1 MITIGATION MONITORING AND REPORTING PROGRAM

4.0 Mitigation Monitoring and Reporting Program

rame/ Milestone Reporting Requirements (if applicable)

mains are nd.



	TABLE 4-1 MITIGATION MONITORING AND REPORTING PROGRAM				
Impact	Mitigation Measures	Implementing Party	Monitoring Party	Timefr Monitoring	
	 e. Preservation in place (i.e., avoidance) is the preferred manner of treatment for discovered human remains and/or burial goods, if feasible. Any historic archaeological material that is not Native American in origin (non-TCR) shall be curated at a public, non-profit institution with a research interest in the materials, such as the Natural History Museum of Los Angeles County or the Fowler Museum, if such an institution agrees to accept the material. If no institution accepts the archaeological material, it shall be offered to a local school or historical society in the area for educational purposes. f. Any discovery of human remains/burial goods shall be kept confidential to prevent further disturbance. 				
MM TCR-3:	 Procedures for Burials and Funerary Remains. a. The appropriate Native American Tribe(s) burial policy shall be implemented. b. If the discovery of human remains includes four or more burials, the discovery location shall be treated as a cemetery and a separate treatment plan shall be created. c. The prepared soil and cremation soils are to be treated in the same manner as bone fragments that remain intact. Associated funerary objects are objects that, as part of the death rite or ceremony of a culture, are reasonably believed to have been placed with individual human remains either at the time of death or later; other items made exclusively for burial purposes or to contain human remains can also be considered as associated funerary objects. Cremations will either be removed in bulk or by means as necessary to ensure complete recovery of all sacred materials 	OIAA and appropriate Native American Tribe(s).	Approved Native American Monitor.	During g disturbing a Native Ai human rer four	



rame/ | Milestone Reporting Requirements (if applicable)

ground-	•	Treatment Plan for	cemetery, if
activities, if		applicable.	
merican	٠	Documentation	including

- mains are nd.
- Documentation including detailed descriptive notes and sketches of the tribal resources discovered.

Impact	TABLE 4-1 MITIGATION MONITORING AND REPORTING PROGRAM					
	Mitigation Measures	Implementing Party	Monitoring Party	Timefra Monitoring		
	d. In the case where discovered human remains canno- fully documented and recovered on the same day, remains will be covered with muslin cloth and a steel p that can be moved by heavy equipment placed ove excavation opening to protect the remains. If this typ steel plate is not available, a 24-hour guard should posted outside of working hours. The Tribe will make e effort to recommend diverting the project and keeping remains in situ and protected. If the project canno diverted, it may be determined that burials wil removed.	ot be , the plate r the be of d be every g the bt be I be				
	e. In the event preservation in place is not possible de good faith efforts by the proposed Pre applicant/developer and/or landowner, before gro disturbing activities may resume on the Project site landowner shall arrange a designated site location w the footprint of the proposed Project for the respe reburial of the human remains and/or ceremonial obje	spite oject ound- , the vithin ectful ects.				
	f. Each occurrence of human remains and association funerary objects will be stored using opaque cloth be All human remains, funerary objects, sacred objects objects of cultural patrimony will be removed to a secontainer on site if possible. These items should retained and reburied within six months of recovery. site of reburial/repatriation shall be on the project site at a location agreed upon between the Tribe and landowner at a site to be protected in perpetuity. T shall be no publicity regarding any cultural mate recovered.	iated pags. and ecure d be The e but d the There erials				
	The proposed Project's qualified archaeologist will closely with the appropriate Native American Tribe(s) to er	work nsure				

rame/ Milestone Reporting Requirements (if applicable)



	TABLE 4-1 MITIGATION MONITORING AND RE	PORTING PROG	RAM	
Impact	Mitigation Measures	Implementing Party	Monitoring Party	Timefra Monitoring
	that the excavation is treated carefully, ethically and respectfully. If data recovery is approved by the appropriate Native American Tribe(s), documentation shall be prepared and shall include (at a minimum) detailed descriptive notes and sketches. All data recovery data recovery-related forms of documentation shall be approved in advance by the appropriate Native American Tribe(s). If any data recovery is performed, once complete, a final report shall be submitted to the appropriate Native American Tribe(s) and the NAHC.			



≥NT

4.0 Mitigation Monitoring and Reporting Program

rame/ | Milestone Reporting Requirements (if applicable)

	TABLE 4-2 PROJECT DESIGN FEATURES			
	Project Design Features	Implementing Party	Monitoring Party	Timeframe/Monitoring Mileston
		A		
PDF AQ-1:	For all phases of construction activity, the Applicant shall require use of off road-construction equipment that is zero emission, if and to the extent available, or diesel-fueled off-road construction equipment that meets the USEPA's Tier 4 emissions standards for offroad diesel-powered construction equipment with 50 horsepower (hp) or greater. To ensure that Tier 4 or the cleanest construction equipment available would be used during the Project's construction, the OIAA shall confirm that the Applicant includes this requirement in applicable bid documents, purchase orders, and contracts. Additionally, the OIAA shall confirm that the Applicant also requires periodic reporting and provision of written construction documents by construction contractor(s) and conducts regular inspections to the maximum extent feasible to ensure and enforce compliance.	Applicant	OIAA	During construction activities.
PDF AQ-2:	The Applicant shall conduct concrete/asphalt demolition on-site to reuse concrete/asphalt generated during construction. During Phase 1, demolition would involve removal of approximately 2,047,320 square feet of asphalt/concrete, which would be recycled within the project site and not require off-site haul truck trips (i.e., avoiding 2,616 haul truck trips). During Phase 2, demolition would involve removal of approximately 1,045,440 square feet of asphalt/concrete, which would be recycled within the project site and not require off-site haul truck trips (i.e., avoiding 910 haul truck trips).	Applicant	OIAA	During construction activities.

Reporting Requirements (if applicable)
 Applicant includes requirement in applicable bid documents, purchase orders, and contracts
 Applicant provides periodic reporting and provision of written construction documents by construction contractor(s)
• OIAA conducts regular inspections to the maximum extent feasible to ensure and enforce compliance

- Applicant provides periodic reporting and provision of written construction documents by construction contractor(s)
- OIAA conducts regular inspections to the maximum extent feasible to ensure and enforce compliance



TABLE 4-2 PROJECT DESIGN FEATURES					
	Project Design Features	Implementing Party	Monitoring Party	Timeframe/Monitoring Milestone	Reporting Requirements (if applicable)
PDF AQ-3:	The Ground Support Equipment (GSE), including (but not limited to) aircraft tugs, baggage tugs, belt loaders, cargo loaders, forklifts, and ground power units, ramp support carts/vans, servicing aircrafts shall be electric by Phase 2.	Applicant & OIAA	OIAA	During construction activities.	 Applicant provides periodic reporting and provision of written construction documents by construction contractor(s) OIAA conducts regular inspections to the maximum extent feasible to ensure and enforce compliance
PDF AQ-4:	A portion of the proposed Project's aircraft fleet shall include electric cargo aircraft. (See Table 3.4 in Section 3.0: Project Description).	Applicant	OIAA	During operation of the Project.	• Maintain log of aircraft fleet associated with Project operations.
PDF AQ-5:	All new aircraft parking positions shall be equipped with ground power and pre-conditioned air, therefore reducing the need to operate auxiliary power units.	Applicant	OIAA	During operation of the Project.	• Maintain records documenting presence of ground power and pre- conditioned air at all aircraft parking positions.
PDF AQ-6:	The Applicant shall conduct maintenance and/or testing on each of the seven standby generators on separate days to limit daily emissions from maintenance/testing activities.	Applicant	OIAA	During operation of the Project.	• Maintain log of maintenance and/or testing activities associated with standby generators that records dates and details.
PDF AQ-7:	The Air Cargo Sort Building shall meet Leadership in Energy and Environmental Design (LEED) certification standards, shall include enhanced building automation systems, and shall utilize advanced low energy HVAC systems.	Applicant & OIAA	OIAA	Prior to certificate of occupancy.	• Review and approval of permits and plans.
PDF AQ-8:	The visitor parking lot shall include 29 parking stalls, 6 of which shall have access to electric charging points. The employee parking structure shall include 932 parking stalls, 300 of which shall have access to electric charging points.	Applicant	OIAA	Prior to certificate of occupancy.	• Review and approval of permits and plans.



TABLE 4-2 PROJECT DESIGN FEATURES					
	Project Design Features	Implementing Party	Monitoring Party	Timeframe/Monitoring Milestone	Reporting Requirements (if applicable)
PDF AQ-9:	 The Air Cargo Sort Building shall incorporate all of the following design specifications and technologies: Building automation Efficient, heat pump HVAC Natural ventilation Purchase of electricity from the SCE 100% Clean Rate Program, if and to the extent feasible Efficient dock seals Rapid rise doors Solar shades Low use water appliances Sustainable, drought-tolerant landscaping featuring a native, non-invasive vegetation palette Submeters with advanced energy monitoring Efficient transformers Battery storage-ready infrastructure 	Applicant	OIAA	Prior to certificate of occupancy.	Review and approval of permits and plans.
	 Building automation by an enhanced building management system Enhanced glazing 				
PDF AQ-10:	The Project shall include electric charging infrastructure in the truckyard that, at a minimum, accords with all applicable requirements of California's Building Energy Efficiency Standards, as set forth within Title 24, Part 6, of the California Code of Regulations.	Applicant	OIAA	During operation of the Project.	• Review and approval of permits and plans.

		PROJEC	TABLE 4-2 T DESIGN FEATURES		
	Project Design Features	Implementing Party	Monitoring Party	Timeframe/Monitoring Milestone	Reporting Requirements (if applicable)
PDF AQ-11:	The storage and maintenance of Project-related delivery trucks shall occur only on site. In the event that overnight parking of delivery trucks is necessary, such trucks shall be parked within the Project site.	Applicant	OIAA	During operation of the Project.	 Review and approval of permits and plans.
			GHG		
PDF GHG-1:	The Air Cargo Sort Building shall be all-electric (no natural gas usage).	Applicant	OIAA	Prior to certificate of occupancy.	• Review and approval of permits and plans.
PDF GHG-2:	The proposed Project shall include a 3.8-Megawatt Solar PV Panel System on the rooftop of the Air Cargo Sort Building and Parking Structure.	Applicant	OIAA	Prior to certificate of occupancy.	 Review and approval of permits and plans.



EXHIBIT "B"

CEQA Findings and Statement of Overriding Considerations for the Final EIR (SCH#2021100226)

CEQA FINDINGS, FACTS IN SUPPORT OF FINDINGS, AND STATEMENT OF OVERRIDING CONSIDERATIONS

FOR THE

ONTARIO INTERNATIONAL AIRPORT SOUTH AIRPORT CARGO CENTER PROJECT (SCH#2021100226)
TABLE OF CONTENTS

1.0	INTRODUCTION			
	1.1 1.2	Statutory Requirements for Findings Organization/Format of Findings	1 2	
2.0	PROJECT DESCRIPTION			
	2.1 2.2 2.3	Location of the Project Project Description Project Objectives	3 3 4	
3.0	EFFI	ECTS DETERMINED TO BE SIGNIFICANT AND UNAVOIDABLE	4	
	3.1 3.2 3.3 3.4 3.5 3.5	Air Quality Air Quality (Cumulative) Greenhous Gas Emissions Greenhous Gas Emissions (Cumulative) Transportation Transportation (Cumulative)	5 11 18 26 27 30	
4.0	POTENTIALLY SIGNIFICANT EFFECTS THAT CAN BE MITIGATED TO BELOW A LEVEL OF SIGNIFICANCE			
	4.1 4.2 4.3 4.4 4.5 4.6 4.7 4.8 4.9 4.10 4.11 4.12	Biological Resources Biological Resources (Cumulative) Cultural Resources (Cumulative) Geology and Soils Geology and Soils (Cumulative) Hazards and Hazardous Materials Hazards and Hazardous Materials (Cumulative) Noise Noise Noise (Cumulative) Tribal Cultural Resource (Cumulative)	34 38 41 44 47 55 57 62 64 66 69 75	
5.0	EFFECTS DETERMINED TO BE NOT SIGNIFICANT OR LESS THAN SIGNIFICANT			
	5.1 5.2 5.3 5.4 5.5 5.6 5.7 5.8 5.9 5.10 5.11 5.12	Aesthetics. Aesthetics (Cumulative) Air Quality Biological Resources Cultural Resources Energy. Energy (Cumulative). Geology and Soils. Hazards and Hazardous Materials Hydrology Hydrology (Cumulative). Noise	81 82 83 86 88 90 92 92 93 95 98 98	
	5.13	Public Services	99	

	5.14	Public Services (Cumulative)	100
	5.15	Transportation	101
	5.16	Utilities	103
	5.17	Utilities (Cumulative)	105
	5.18	Issues Deemed No Impact or Less than Significant Impact in the Initial Study/Notice of Preparation	of
6.0	FIN	DINGS REGARDING PROJECT ALTERNATIVES	109
	61	Alternative 1. No Project/No Development Alternative	110
	6.2	Alternative 7: Reduced Project Size Alternative	111
	6.3	Alternative 3: Different Location on Airport Alternative	113
7.0	OT	HER CEOA CONSIDERATIONS	114
	7.1	Reasons Why the Project is Being Proposed, Notwithstanding Significant Unavoidable	e
		Impacts	114
	7.2	Growth-Inducing Impacts	114
	7.3	Significant Irreversible Environmental Changes	115
	7.4	Potentially Significant Effects from Implementation of Mitigation Measures	115
8.0	GE	NERAL CEQA FINDINGS	115
	8.1	Mitigation Monitoring and Reporting Program	115
	8.2	CEQA Guidelines Sections 15091 and 15092 Findings	116
	8.3	OIAA's Preparation of the EIR Pursuant to CEQA Guidelines Section 15084(d)	116
	8.4	OIAA's Independent Judgment	117
	8.5	Nature of Findings	117
	8.6	Reliance on Record	117
	8.7	Custodian of Records	117
	8.8	Relationship of Findings to EIR	117
	8.9	Recirculation Not Required	117
9.0	CE	RTIFICATION OF THE FINAL ENVIRONMENTAL IMPACT REPORT	,
	CE	QA GUIDELINES, § 15090	118
10.0	STA	ATEMENT OF OVERRIDING CONSIDERATIONS	118
	10.1	Background	118
	10.2	Overriding Considerations	120
	10.3	Conclusion	122

CEQA FINDINGS, FACTS IN SUPPORT OF FINDINGS, AND STATEMENT OF OVERRIDING CONSIDERATIONS FOR THE ONTARIO INTERNATIONAL AIRPORT SOUTH AIRPORT CARGO CENTER PROJECT (SCH#2021100226)

1.0 INTRODUCTION

1.1 Statutory Requirements for Findings

The Commission of the Ontario International Airport Authority (OIAA) certifies that OIAA has reviewed and considered the information contained in the Final Environmental Impact Report (EIR), identified below, for the Ontario International Airport South Airport Cargo Center Project (Project). These Findings Regarding Significant Effects (Findings) have been prepared to support and justify approval of the Project.

OIAA further certifies that the Final EIR has been completed in compliance with the California Environmental Quality Act (CEQA), Public Resources Code sections 21000 et seq., and the CEQA Guidelines (Guidelines), California Code of Regulations, Title 14, sections 15000 et seq. In certifying the Final EIR as adequate under CEQA, OIAA also adopts these CEQA Findings and Statement of Overriding Considerations.

CEQA section 21081 and the CEQA Guidelines section 15091 require that the lead agency, in this case OIAA, prepare written findings for identified significant impacts, accompanied by a brief explanation of the rationale for each finding. Specifically, State CEQA Guidelines section 15091 states, in part, that:

- (a) No public agency shall approve or carry out a project for which an EIR has been certified which identifies one or more significant environmental effects of the project unless the public agency makes one or more written findings for each of those significant effects accompanied by a brief explanation of the rationale for each finding. The possible findings are:
 - (1) Changes or alterations have been required in, or incorporated into, the project which avoid or substantially lessen the significant environmental effects as identified in the Final EIR.
 - (2) Such changes or alterations are within the responsibility and jurisdiction of another public agency and not the agency making the finding. Such changes have been adopted by such other agency or can and should be adopted by such other agency.
 - (3) Specific economic, legal, social, technological, or other considerations, including provision of employment opportunities for trained workers, make infeasible the mitigation measures or project alternatives identified in the final EIR.

For those significant effects that cannot be mitigated to below a level of significance, the public agency is required to find that specific overriding economic, legal, social, technological, or other

benefits of the needed Ontario International Airport South Airport Cargo Center Project outweigh the significant effects on the environment. (CEQA Guidelines, §15093.)

In addition, CEQA requires a public agency to make a finding that the EIR reflects the public agency's independent review and judgment. Therefore, in accordance with the provisions of CEQA and the Guidelines, OIAA, acting in its capacity as the CEQA lead agency, expressly finds that the Final EIR for the Ontario International Airport South Airport Cargo Center Project reflects OIAA's independent review and judgment.

The Final EIR identifies significant or potentially significant environmental effects, prior to and after mitigation, which may occur as a result of OIAA's approval of the Project. However, OIAA finds that the inclusion of certain specified mitigation measures as part of the Project approval will reduce many, but not all, of those effects to less-than-significant levels. Certain impacts not reduced to less-than-significant levels are impacts related to air quality, greenhouse gas (GHG) emissions and transportation (see Section 3.0 – Effects Determined to be Significant and Unavoidable); and are overridden due to specific Project benefits (see, Section 10, Statement of Overriding Considerations).

All of the findings discussed above, and those otherwise required under CEQA and the Guidelines, are collectively referred to as the "Findings." In accordance with CEQA and the Guidelines, OIAA certifies the Final EIR for the Project, adopts these Findings, the statement of overriding considerations, and the Mitigation Monitoring and Reporting Program (MMRP), which have been prepared and are incorporated into the Project's conditions of approval, and approves the Project. Further, based on its independent review and judgment of the complete and entire administrative record, including, but not limited to the Draft EIR, Final EIR, and staff's recommendations, and in accordance with the provisions of CEQA and the Guidelines, OIAA adopts these Findings as part of its certification of the Final EIR for the Ontario International Airport South Airport Cargo Center Project.

1.2 Organization/Format of Findings

In compliance with the statutory requirements, the Findings are organized as follows:

- (1) Significant effects that cannot be mitigated to below the level of significance;
- (2) Effects that can be mitigated to below the level of significance;
- (3) Effects that are less than significant;
- (4) Feasibility and desirability of Project alternatives;
- (5) Other and general CEQA Findings; and
- (7) Statement of Overriding Considerations.

To the extent relevant, the above-enumerated components of these Findings are accompanied by a discussion of significant effects, mitigation measures relevant to the specific effects being considered, findings, and facts in support of those findings.

Ontario International Airport South Airport Cargo Center Project

2.0 PROJECT DESCRIPTION

2.1 Location of the Project

The Project site consists of approximately 97 acres located at Ontario International Airport (Airport) in the City of Ontario, San Bernardino County. Regional access to the Airport and the Project site is via Interstate 10 (I-10), one-mile to the north; State Route 60 (SR-60), approximately 1.25 mile to the south; and I-15, approximately 2.75 miles to the east.

The Project site includes portions of Assessor Parcel Numbers (APN) 11326106, 11326107, 11326108, 11327101, and 11327102, located in the southern half of the Airport, immediately west of the Cucamonga Channel and north of Mission Boulevard. Most of the Project site is located north of East Avion Street with the remainder located between East Avion Street and Mission Boulevard west of South Hellman Avenue.

2.2 **Project Description**

The proposed Project is an aeronautical development and use that is within the Airport boundaries and is consistent with the Ontario International Airport Layout Plan. The proposed Project would replace existing, underutilized airport related buildings and site improvements with an air cargo center. The proposed Project would include demolition of the existing buildings, site improvements, and the development of a new air cargo center in two phases, as described further below.

The proposed air cargo center includes an Air Cargo Sort Building, truckyard, parking facilities, two aviation support buildings (ground service equipment [GSE] and aircraft line maintenance buildings), and aircraft apron improvements. The Air Cargo Sort Building, proposed north of East Avion Street, would contain a sorting facility and office spaces. The aircraft parking apron would surround the building to the west, north, and east. A ground-level visitor parking lot and truckyard are proposed on the south side of the cargo building, with access from East Avion Street. A parking structure for employees is proposed south of East Avion Street with a pedestrian bridge connecting the parking structure to the office building. The proposed Project would be implemented in two phases. Phase 1 would take place on the easternmost 62 acres of the Project site and Phase 2 would occur on the remaining western 35 acres.

Phase 1 construction would include the demolition of existing structures and site improvements in the Phase 1 area, as well as site preparation and construction of all proposed improvements on the eastern 62 acres of the Project site, including the initial phase of the Air Cargo Sort Building, aircraft apron improvements, and parking structure. Phase 2 would occur on the western 35 acres of the Project site and include the demolition of structures and site improvements in the Phase 2 area, site preparation, and construction of the remaining improvements, including the expansion of the Air Cargo Sort Building and aircraft apron improvements.

Landscaping would be proposed along the northern and southern sides of E. Avion Street. Landscaping would include Desert Museum Palo Verde trees with complementary shrub and groundcover species. Some existing Canary Island Pine trees would be retained and incorporated into the landscape areas.

2.3 **Project Objectives**

Section 15124(b) of the CEQA Guidelines states that "the statement of objectives should include the underlying purpose of the project." The underlying purpose of the proposed Project is to develop and operate an air cargo facility at the Airport to meet increased regional air cargo volumes and Project proponent facility requirements. The objectives of the OIAA for the proposed Project include:

- A. Allow the Project proponent to accommodate current and projected air cargo volumes.
- B. Integrate the Project proponent's airside, landside, and sorting facilities in a location with access to major surface transportation corridors to improve operational efficiency.
- C. Redevelop underutilized Airport property.
- D. Maximize revenue generation from Airport property.
- E. Provide employment opportunities for residents of the City of Ontario and the Inland Empire.

3.0 EFFECTS DETERMINED TO BE SIGNIFICANT AND UNAVOIDABLE

Pursuant to CEQA Section 21081, and Guidelines Sections 15091 and 15126.2(b), the following Section sets forth the significant unavoidable effects of the Project, and with respect to each effect, identifies one or more of the required findings, states facts in support of those findings and, as appropriate, refers to the Statement of Overriding Considerations (see Section 10, below). All significant environmental impacts that can be feasibly avoided have been eliminated, or substantially lessened through implementation of the Project Design Features and/or mitigation measures. Based on the Final EIR, the Statement of Overriding Considerations set forth below, and other documents and information in the record with respect to the construction and operation of the Project, all remaining unavoidable significant impacts, as set forth in these findings, are overridden by the benefits of the Project as described in the Statement of Overriding Considerations.

Pursuant to CEQA Section 2108 l(a) and Guidelines Section 1509l(a)(3) OIAA finds that, for each of the following significant effects, specific economic, legal, social, technological, or other considerations, including provisions of employment opportunities for highly trained workers, make infeasible the mitigation measures or alternatives identified in the Final EIR. These findings are explained below and are supported by substantial evidence in the record of proceedings.

Based on the analysis contained in the Final EIR, the following impacts to air quality, greenhouse gas (GHG) emissions, and transportation have been determined to be significant and unavoidable. Such impacts are overridden by the benefits of Project as set forth in the Statement of Overriding Considerations in Section 10 below.

3.1 Air Quality

Result in a cumulatively considerable net increase of any criteria pollutant for which the project region is nonattainment under an applicable federal or State ambient air quality standard.

Impact Summary:

A significant impact could occur if the proposed Project would result in a considerable contribution of federal or State nonattainment pollutants. In regard to determining the significance of the proposed Project's contribution, the South Coast Air Quality Management District's (SCAQMD) significance thresholds determine if a project would contribute a cumulatively considerable increase in emissions for those pollutants for which the Air Basin is in nonattainment.

Significant Effects: AQ-2

The proposed Project's operational emissions during Phase 1 would exceed SCAQMD significance thresholds for CO, VOC, and NOx, primarily due to aircraft emissions, followed by employee vehicles, delivery trucks, and emergency generators. During Phase 2 (buildout of the Project), the proposed Project's operational emissions would exceed SCAQMD significance thresholds for CO, VOC, NO_X, and SO₂. Therefore, the Project would result in a significant operational air quality impact.

Findings:

Pursuant to Public Resources Code Section 21081(a)(1), the OIAA finds that changes or alterations have been required in, or incorporated into, the Project that avoid or substantially lessen the foregoing significant impact as identified in the EIR. However, these effects have not been reduced to a less-than-significant level.

Thus, pursuant to Public Resources Code Section 21081(a)(3), the OIAA finds that specific economic, legal, social, technological, other considerations, including considerations for the provision of employment opportunities for highly trained workers, make infeasible the mitigation measures or alternatives identified in the EIR. There are no feasible mitigation measures or alternatives that would reduce the identified significant impact to a level below significant or otherwise reduce the impact. Therefore, the impact remains significant and unavoidable.

Facts in Support of Findings:

Operational sources of Project-related air pollutant emissions include aircraft, APU, GSE, stationary sources (e.g., emergency generators), motor vehicles (employees and deliveries), area sources (consumer products and landscaping), and energy usage (natural gas and electrical). As discussed above, the proposed Project's operational emissions during Phase 1 and Phase 2 (buildout of the Project) would exceed regional SCAQMD significance thresholds for CO, VOC, NOx and SO₂ (as to Phase 2 (buildout of the Project) only), primarily due to aircraft emissions, followed by employee vehicles, delivery trucks, and emergency generators. Impacts would be potentially significant.

The proposed Project would incorporate Project Design Features PDF AQ-1 through PDF AQ-11 and Mitigation Measures MM AQ-1 through MM AQ-16, as well as MM TRANS-1 through

MM TRANS-5, to reduce operational air quality emissions to the greatest extent feasible. However, neither the SCAQMD nor OIAA have the authority to regulate aircraft operations or emissions from aircraft engines and the majority of the emissions estimated for operation of the Project are from aircraft operations. The federal Clean Air Act exclusively vests the authority to promulgate emission standards for aircraft and aircraft engines with the United States Environmental Protection Agency (USEPA); states and other municipalities are preempted from adopting or enforcing any standard with respect to aircraft engine emissions unless such standard is identical to USEPA standards.

The 2022 Air Quality Management Plan (AQMP) identifies actions that can be taken to address these sources of emissions, including the federal adoption of more stringent criteria pollutant standards for aircraft engines and use of cleaner aviation fuels. It is anticipated that these types of future technology improvements will reduce the aviation emissions associated with the Project over time. As the proposed Project is an air cargo facility serving the region, the operational and economic viability of the proposed Project relies on these aviation operations. For these reasons, there are no additional feasible mitigation measures that would reduce operational emissions to below significance thresholds and operational air quality emissions would remain significant after implementation of all feasible mitigation.

Thus, pursuant to Public Resources Code Section 21081(a)(3), based on the evidence described below in Section 10.0, Statement of Overriding Considerations, the OIAA finds that specific, economic, legal, social, technological, or other considerations, including considerations for the provision of employment opportunities for highly trained worked, makes infeasible the mitigation measures or alternatives identified in the EIR to reduce this impact to a less-than-significant level.

Mitigation Measures:

AQ-1, AQ-2, AQ-3, AQ-4, AQ-5, AQ-6, AQ-7, AQ-8, AQ-9, AQ-10, AQ-11, AQ-12, AQ-13, AQ-14, AQ-15, AQ-16, TRANS-1, TRANS-2, TRANS-3, TRANS-4, and TRANS-5.

- AQ-1 The Applicant shall require that construction vendors, contractors, and/or haul truck operators commit to using 2010 model year trucks (e.g., material delivery trucks and soil import/export with a gross vehicle weight rating of at least 14,001 pounds), that meet CARB's 2010 engine emissions standards or newer, cleaner trucks. The OIAA shall confirm that the Applicant includes this requirement in applicable bid documents, purchase orders, and contracts. Operators shall maintain records of all trucks associated with Project construction to document that each truck used meets these emission standards and make the records available for inspection.
- AQ-2 The Applicant shall require that construction equipment such as concrete/industrial saws, pumps, aerial lifts, light stands, air compressors, and forklifts be electric or alternative-fueled (i.e., non-diesel), where feasible. Pole power shall be utilized at the earliest feasible point in time and shall be used to the maximum extent feasible in lieu of generators.
- AQ-3 The Applicant shall support and encourage ridesharing and transit incentives for the construction crew by providing crews with the resources needed to organize

rideshares, such as bulletin boards or email announcements. The Applicant shall also partially subsidize transit fares or passes for the construction crew members who can feasibly use transit. The Applicant shall set a goal to achieve ten percent total construction worker participation in ridesharing programs and transit use.

- AQ-4 The Applicant shall require, in addition to the GSE noted within PDF AQ-3, all other on-site cargo-handling equipment, such as yard trucks, holsters, yard goats, pallet jacks, and similar equipment, to be electric, with the necessary electrical charging stations provided.
- AQ-5 The Applicant shall require, if and to the extent feasible, the use of zero-emission or near zero-emission on-road heavy duty trucks as part of business operations beginning in 2025 (within at least 25 percent of the Project fleet).

The Applicant also shall require, if and to the extent feasible, the use of zeroemission or near zero-emission on-road heavy duty trucks as part of the business operations beginning in 2029 (within at least 50 percent of the Project fleet).

- AQ-6 The Applicant shall include in the design requirements for the Project, cool roof installation to the extent roof space is not occupied by solar panels, in order to reduce energy use and urban heat island effects.
- AQ-7 The Applicant shall encourage the use of single engine taxi operations for Project aircraft.
- AQ-8 The Applicant shall utilize Energy Star heating, cooling, and lighting devices, and appliances (e.g., Heating, Ventilation, and Air Conditioning (HVAC) units in the form of energy efficient commercial heat pumps) within the interior of the Air Cargo Sort Building.
- AQ-9 In order to reduce trips to and from the Project site during construction, the Applicant shall provide on-site food trucks during meal times.
- AQ-10 Interior- and exterior-facing signs, including signs directed toward all dock and delivery areas, shall be posted by the Applicant to identify contact information to report idling violations to CARB, SCAQMD, and the building manager. These signs also shall inform truck drivers to shut off their engines when not in use.
- AQ-11 Electric plugs for electric transport refrigeration units shall be provided at dock doors located at the Air Cargo Sort Building.
- AQ-12 The Applicant shall train operational managers and employees on efficient scheduling and load management to eliminate unnecessary queuing of trucks.
- AQ-13 Signs shall be posted by the Applicant at every truck exit driveway providing directional information to use truck routes as designated by the City of Ontario.

- AQ-14 The Applicant shall require its facility operator(s) to train the staff in charge of keeping vehicle records on diesel technologies and compliance with CARB regulations by attending CARB-approved courses. Also, the Applicant shall require its facility operator(s) to maintain records on-site demonstrating regulatory compliance and make records available for inspection by OIAA, SCAQMD, and State of California upon request.
- AQ-15 The Applicant shall include a provision in all operational freight hauling contracts requiring the use of 2010 model year trucks that meet CARB's 2010 engine emission standards, or newer and cleaner trucks, if and to the extent feasible.
- AQ-16 During construction, the Applicant shall post interior- and exterior-facing signs to inform construction contractors to shut off truck and equipment engines when not in use.
- **TRANS-1** Voluntary Commute Reduction Program. The proposed Project shall implement Voluntary Commute Trip Reduction (CTR) programs that discourage single-occupancy vehicle trips and encourage alternative modes of transportation, such as carpooling, taking transit, walking, and biking. Voluntary CTR programs shall include the following elements to apply the VMT reductions reported in literature:
 - Employer-provided services, infrastructure, and/or incentives for commuting to work using alternative modes (e.g., walking, biking, carpooling/vanpooling, or taking transit).
 - Provide information, coordination, and marketing for employee rideshare services, provide onsite infrastructure to support carpools/vanpools, and provide incentives (e.g., free transit passes, monthly bonus for carpooling 3 or more times a week, etc.).

Employer costs may include recurring costs for carpool/vanpool subsidies, capital and maintenance costs for the alternative transportation infrastructure (e.g., showers and lockers), and labor costs for staff to manage the program.

- **TRANS-2 Provide Ridesharing Program.** A ridesharing program shall be implemented for employees of the site. The following elements designed to support the Project's ridesharing program:
 - Provide vanpool parking with designated passenger loading/unloading area near employee entrance.
 - Create a Carpool Incentive Program.
 - Provide a minimum of ten (10) carpool parking spaces provided closer to the employee entrance than standard parking spaces.
 - Provide access to a carpool database (Metro rideshare) and/or an on-site matching program for employees.

Ontario International Airport South Airport Cargo Center Project Provide a monthly incentive for employees that carpool a minimum of three (3) days per week (e.g., \$50 gas card or a \$50 green commuter bonus).

In addition, a staff person would be designated to for provide rideshare information to employees and monitoring the effectiveness of the program.

It is assumed all employees are eligible and that additional carpool spaces could be designated if warranted by demand.

TRANS-3 Implement Subsidized or Discounted Transit Program. Subsidized, discounted, or free Omnitrans, Metrolink or Amtrak transit passes shall be provided to employees to encourage use of transit routes/stops located less than a mile from the Project. It is assumed free transit passes are available to all employees.

Based on the given shift times of the Project, shifts that start or end at 11:00 PM shall have limited available options as most routes do not provide service that late. This shall limit approximately half the employees from the ability to rely on transit.

TRANS-4 Bicycle Facilities. On-site bicycle parking and end-of-trip facilities shall be provided for employee use. End-of-trip facilities include bike parking, bike lockers, showers, and personal lockers.

A bike share program (standard or electric bikes) for employees shall supplement bicycle facilities.

- **TRANS-5** Employer-Sponsored Vanpool Program. An employer-sponsored vanpool service shall be implemented and be fully funded by the tenant as follows:
 - Provide a minimum of one (1) and up to three (3) vanpool vehicles and associated parking with designated passenger loading/unloading area near employee entrance.
 - Pay for the lease of a minimum of one (1) van and up to three (3) vans for the purpose of employee vanpooling.
 - A ten percent voluntary participation rate is assumed to be the high end of the range for this project.

Project Design Features:

AQ-1, AQ-2, AQ-3, AQ-4, AQ-5, AQ-6, AQ-7, AQ-8, AQ-9, AQ-10, and AQ-11

AQ-1 For all phases of construction activity, the Applicant shall require use of off-road construction equipment that is zero emission, if and to the extent available, or diesel-fueled off-road construction equipment that meets the USEPA's Tier 4 emissions standards for offroad diesel-powered construction equipment with 50 horsepower (hp) or greater. To ensure that Tier 4 or the cleanest construction

equipment available would be used during the Project's construction, the OIAA shall confirm that the Applicant includes this requirement in applicable bid documents, purchase orders, and contracts. Additionally, the OIAA shall confirm that the Applicant also requires periodic reporting and provision of written construction documents by construction contractor(s) and conducts regular inspections to the maximum extent feasible to ensure and enforce compliance.

- AQ-2 The Applicant shall conduct concrete/asphalt demolition on-site to reuse concrete/asphalt generated during construction. During Phase 1, demolition would involve removal of approximately 2,047,320 square feet of asphalt/concrete, which would be recycled within the project site and not require offsite haul truck trips (i.e., avoiding 2,616 haul truck trips). During Phase 2, demolition would involve removal of approximately 1,045,440 square feet of asphalt/concrete, which would be recycled within the project site and not require offsite haul truck trips (i.e., avoiding 910 haul truck trips).
- AQ-3 The Ground Support Equipment (GSE), including (but not limited to) aircraft tugs, baggage tugs, belt loaders, cargo loaders, forklifts, and ground power units, ramp support carts/vans, servicing aircrafts shall be electric by Phase 2.
- AQ-4 A portion of the proposed Project's aircraft fleet shall include electric cargo aircraft. (See Table 3.4 in Section 3.0: Project Description).
- AQ-5 All new aircraft parking positions shall be equipped with ground power and preconditioned air, therefore reducing the need to operate auxiliary power units.
- AQ-6 The Applicant shall conduct maintenance and/or testing on each of the seven standby generators on separate days to limit daily emissions from maintenance/testing activities.
- AQ-7 The Air Cargo Sort Building shall meet Leadership in Energy and Environmental Design (LEED) certification standards, shall include enhanced building automation systems, and shall utilize advanced low energy HVAC systems.
- AQ-8 The visitor parking lot shall include 29 parking stalls, 6 of which shall have access to electric charging points. The employee parking structure shall include 932 parking stalls, 300 of which shall have access to electric charging points.
- AQ-9 The Air Cargo Sort Building shall incorporate all of the following design specifications and technologies:
 - Building automation
 - Efficient, heat pump HVAC
 - Natural ventilation
 - Purchase of electricity from the SCE 100% Clean Rate Program, if and to the extent feasible
 - Efficient dock seals

Ontario International Airport South Airport Cargo Center Project

- Rapid rise doors
- Solar shades
- Low use water appliances
- Sustainable, drought-tolerant landscaping featuring a native, non-invasive vegetation palette
- Submeters with advanced energy monitoring
- Main meter energy monitoring
- Efficient transformers
- Battery storage-ready infrastructure
- Building automation by an enhanced building management system
- Enhanced glazing
- AQ-10 The Project shall include electric charging infrastructure in the truckyard that, at a minimum, accords with all applicable requirements of California's Building Energy Efficiency Standards, as set forth within Title 24, Part 6, of the California Code of Regulations.
- AQ-11 The storage and maintenance of Project-related delivery trucks shall occur only onsite. In the event that overnight parking of delivery trucks is necessary, such trucks shall be parked within the Project site.

Reference:

For a complete discussion of Project impacts related to air quality, see Section 5.2.3 of the Draft EIR at pages 5.2-70 - 5.2-74, Section 2.0 of the Final EIR at pages 2.0-29 through 2.0-37, and Section 3.0 of the Final EIR (Topical Responses 1, Responses to Comment Letter C, and Responses to Comment Letter H).

3.2 Air Quality (Cumulative)

The Draft EIR discusses the impacts related to air quality in Section 5.2. The following discussion addresses potential cumulative impacts with respect to air quality.

Impact Summary:

During operations, the proposed Project's daily criteria pollutant emissions would exceed SCAQMD significance thresholds for CO, VOC, and NO_x in Phase 1, and CO, VOC, NO_x , and SO_2 in Phase 2 (buildout of the Project). Impacts from the proposed Project would be significant and unavoidable and would result in a cumulatively considerable increase of air emissions during operations.

Significant Effects:

The Project's cumulative contribution to air quality would be cumulatively considerable.

Findings:

Pursuant to Public Resources Code Section 21081(a)(1), the OIAA finds that change or alterations have been required in, or incorporated into, the Project that avoid or substantially lessen the

foregoing significant impact as identified in the EIR. However, these effects have not been reduced to a less-than-significant level.

Thus, pursuant to Public Resources Code Section 21081(a)(3), the OIAA finds that specific economic, legal, social, technological, other considerations, including considerations for the provision of employment opportunities for highly trained workers, make infeasible the mitigation measures or alternatives identified in the EIR. There are no feasible mitigation measures or alternatives that would reduce the identified significant impact to a level below significant or otherwise reduce the impact. Therefore, the impact remains significant and unavoidable.

Facts in Support of Findings:

During operations, the proposed Project's daily criteria pollutant emissions would exceed SCAQMD thresholds for CO, VOC, and NOx in Phase 1, and CO, VOC, NOx, and SO₂ in Phase 2 (buildout of the Project). Operations would not exceed the ambient air quality standards at nearby receptors. Impacts from the proposed Project would be significant and unavoidable and would result in a cumulatively considerable increase of air emissions during operations.

Based on SCAQMD methodology, the proposed Project's operational emissions would represent a cumulatively considerable contribution, and thus the Project would also result in a cumulatively significant impact.

There are no feasible mitigation measures that would reduce the proposed Project's operational impacts to below SCAQMD thresholds because neither SCAQMD or OIAA have the authority to regulate aircraft operations or emissions from aircraft engines, and the majority of the emissions estimated for operation of the Project are from aircraft operations. The federal Clean Air Act exclusively vests the authority to promulgate emission standards for aircraft and aircraft engines with the USEPA; states and other municipalities are preempted from adopting or enforcing any standard with respect to aircraft engine emissions unless such standard is identical to USEPA standards. For these reasons, there are no additional feasible mitigation measures that would reduce operational emissions to below significance thresholds and the proposed Project's operational air quality emissions at the Project-specific and cumulative levels would remain significant and unavoidable.

Thus, pursuant to Public Resources Code Section 21081(a)(3), based on the evidence described below in Section 10.0, Statement of Overriding Considerations, the OIAA finds that specific, economic, legal, social, technological, or other considerations, including considerations for the provision of employment opportunities for highly trained worked, makes infeasible the mitigation measures or alternatives identified in the EIR to reduce this impact to a less-than-significant level.

Mitigation Measures:

AQ-1, AQ-2, AQ-3, AQ-4, AQ-5, AQ-6, AQ-7, AQ-8, AQ-9, AQ-10, AQ-11, AQ-12, AQ-13, AQ-14, AQ-15, AQ-16 TRANS-1, TRANS-2, TRANS-3, TRANS-4, and TRANS-5.

AQ-1 The Applicant shall require that construction vendors, contractors, and/or haul truck operators commit to using 2010 model year trucks (e.g., material delivery trucks and soil import/export with a gross vehicle weight rating of at least 14,001 pounds), that meet CARB's 2010 engine emissions standards or newer, cleaner

Ontario International Airport South Airport Cargo Center Project trucks. The OIAA shall confirm that the Applicant includes this requirement in applicable bid documents, purchase orders, and contracts. Operators shall maintain records of all trucks associated with Project construction to document that each truck used meets these emission standards and make the records available for inspection.

- AQ-2 The Applicant shall require that construction equipment such as concrete/industrial saws, pumps, aerial lifts, light stands, air compressors, and forklifts be electric or alternative-fueled (i.e., non-diesel), where feasible. Pole power shall be utilized at the earliest feasible point in time and shall be used to the maximum extent feasible in lieu of generators.
- AQ-3 The Applicant shall support and encourage ridesharing and transit incentives for the construction crew by providing crews with the resources needed to organize rideshares, such as bulletin boards or email announcements. The Applicant shall also partially subsidize transit fares or passes for the construction crew members who can feasibly use transit. The Applicant shall set a goal to achieve ten percent total construction worker participation in ridesharing programs and transit use.
- AQ-4 The Applicant shall require, in addition to the GSE noted within PDF AQ-3, all other on-site cargo-handling equipment, such as yard trucks, holsters, yard goats, pallet jacks, and similar equipment, to be electric, with the necessary electrical charging stations provided.
- AQ-5 The Applicant shall require, if and to the extent feasible, the use of zero-emission or near zero-emission on-road heavy duty trucks as part of business operations beginning in 2025 (within at least 25 percent of the Project fleet).

The Applicant also shall require, if and to the extent feasible, the use of zeroemission or near zero-emission on-road heavy duty trucks as part of the business operations beginning in 2029 (within at least 50 percent of the Project fleet).

- AQ-6 The Applicant shall include in the design requirements for the Project cool roof installation to the extent roof space is not occupied by solar panels, in order to reduce energy use and urban heat island effects.
- AQ-7 The Applicant shall encourage the use of single engine taxi operations for Project aircraft.
- AQ-8 The Applicant shall utilize Energy Star heating, cooling, and lighting devices, and appliances (e.g., Heating, Ventilation, and Air Conditioning (HVAC) units in the form of energy efficient commercial heat pumps) within the interior of the Air Cargo Sort Building.
- AQ-9 In order to reduce trips to and from the Project site during construction, the Applicant shall provide on-site food trucks during meal times.

- AQ-10 Interior- and exterior-facing signs, including signs directed toward all dock and delivery areas, shall be posted by the Applicant to identify contact information to report idling violations to CARB, SCAQMD, and the building manager. These signs also shall inform truck drivers to shut off their engines when not in use.
- AQ-11 Electric plugs for electric transport refrigeration units shall be provided at dock doors located at the Air Cargo Sort Building.
- AQ-12 The Applicant shall train operational managers and employees on efficient scheduling and load management to eliminate unnecessary queuing of trucks.
- AQ-13 Signs shall be posted by the Applicant at every truck exit driveway providing directional information to use truck routes as designated by the City of Ontario.
- AQ-14 The Applicant shall require its facility operator(s) to train the staff in charge of keeping vehicle records on diesel technologies and compliance with CARB regulations by attending CARB-approved courses. Also, the Applicant shall require its facility operator(s) to maintain records on-site demonstrating regulatory compliance and make records available for inspection by OIAA, SCAQMD, and State of California upon request.
- AQ-15 The Applicant shall include a provision in all operational freight hauling contracts requiring the use of 2010 model year trucks that meet CARB's 2010 engine emission standards, or newer and cleaner trucks, if and to the extent feasible.
- AQ-16 During construction, the Applicant shall post interior- and exterior-facing signs to inform construction contractors to shut off truck and equipment engines when not in use.
- **TRANS-1** Voluntary Commute Reduction Program. The proposed Project shall implement Voluntary Commute Trip Reduction (CTR) programs that discourage single-occupancy vehicle trips and encourage alternative modes of transportation, such as carpooling, taking transit, walking, and biking. Voluntary CTR programs shall include the following elements to apply the VMT reductions reported in literature:
 - Employer-provided services, infrastructure, and/or incentives for commuting to work using alternative modes (e.g., walking, biking, carpooling/vanpooling, or taking transit).
 - Provide information, coordination, and marketing for employee rideshare services, provide onsite infrastructure to support carpools/vanpools, and provide incentives (e.g., free transit passes, monthly bonus for carpooling 3 or more times a week, etc.).

Employer costs may include recurring costs for carpool/vanpool subsidies, capital and maintenance costs for the alternative transportation infrastructure (e.g., showers and lockers), and labor costs for staff to manage the program.

- **TRANS-2 Provide Ridesharing Program.** A ridesharing program shall be implemented for employees of the site. The following elements designed to support the Project's ridesharing program:
 - Provide vanpool parking with designated passenger loading/unloading area near employee entrance.
 - Create a Carpool Incentive Program.
 - Provide a minimum of ten (10) carpool parking spaces provided closer to the employee entrance than standard parking spaces.
 - Provide access to a carpool database (Metro rideshare) and/or an on-site matching program for employees.
 - Provide a monthly incentive for employees that carpool a minimum of three (3) days per week (e.g., \$50 gas card or a \$50 green commuter bonus).

In addition, a staff person would be designated to for provide rideshare information to employees and monitoring the effectiveness of the program.

It is assumed all employees are eligible and that additional carpool spaces could be designated if warranted by demand.

TRANS-3 Implement Subsidized or Discounted Transit Program. Subsidized, discounted, or free Omnitrans, Metrolink or Amtrak transit passes shall be provided to employees to encourage use of transit routes/stops located less than a mile from the Project. It is assumed free transit passes are available to all employees.

Based on the given shift times of the Project, shifts that start or end at 11:00 PM shall have limited available options as most routes do not provide service that late. This shall limit approximately half the employees from the ability to rely on transit.

TRANS-4 Bicycle Facilities. On-site bicycle parking and end-of-trip facilities shall be provided for employee use. End-of-trip facilities include bike parking, bike lockers, showers, and personal lockers.

A bike share program (standard or electric bikes) for employees shall supplement bicycle facilities.

- **TRANS-5** Employer-Sponsored Vanpool Program. An employer-sponsored vanpool service shall be implemented and be fully funded by the tenant as follows:
 - Provide a minimum of one (1) and up to three (3) vanpool vehicles and associated parking with designated passenger loading/unloading area near employee entrance.
 - Pay for the lease of a minimum of one (1) van and up to three (3) vans for the purpose of employee vanpooling.

• A ten percent voluntary participation rate is assumed to be the high end of the range for this project.

Project Design Features:

AQ-1, AQ-2, AQ-3, AQ-4, AQ-5, AQ-6, AQ-7, AQ-8, AQ-9, AQ-10, and AQ-11

- AQ-1 For all phases of construction activity, the Applicant shall require use of off-road construction equipment that is zero emission, if and to the extent available, or diesel-fueled off-road construction equipment that meets the USEPA's Tier 4 emissions standards for off-road diesel-powered construction equipment with 50 horsepower (hp) or greater. To ensure that Tier 4 or the cleanest construction equipment available would be used during the Project's construction, the OIAA shall confirm that the Applicant includes this requirement in applicable bid documents, purchase orders, and contracts. Additionally, the OIAA shall confirm that the Applicant also requires periodic reporting and provision of written construction documents by construction contractor(s) and conducts regular inspections to the maximum extent feasible to ensure and enforce compliance.
- AQ-2 The Applicant shall conduct concrete/asphalt demolition on-site to reuse concrete/asphalt generated during construction. During Phase 1, demolition would involve removal of approximately 2,047,320 square feet of asphalt/concrete, which would be recycled within the project site and not require off-site haul truck trips (i.e., avoiding 2,616 haul truck trips). During Phase 2, demolition would involve removal of approximately 1,045,440 square feet of asphalt/concrete, which would be recycled within the project site and not require offsite haul truck trips (i.e., avoiding 910 haul truck trips).
- AQ-3 The Ground Support Equipment (GSE), including (but not limited to) aircraft tugs, baggage tugs, belt loaders, cargo loaders, forklifts, and ground power units, ramp support carts/vans, servicing aircrafts shall be electric by Phase 2.
- AQ-4 A portion of the proposed Project's aircraft fleet shall include electric cargo aircraft. (See Table 3.4 in Section 3.0: Project Description).
- AQ-5 All new aircraft parking positions shall be equipped with ground power and preconditioned air, therefore reducing the need to operate auxiliary power units.
- AQ-6 The Applicant shall conduct maintenance and/or testing on each of the seven standby generators on separate days to limit daily emissions from maintenance/testing activities.
- AQ-7 The Air Cargo Sort Building shall meet Leadership in Energy and Environmental Design (LEED) certification standards, shall include enhanced building automation systems, and shall utilize advanced low energy HVAC systems.

- AQ-8 The visitor parking lot shall include 29 parking stalls, 6 of which shall have access to electric charging points. The employee parking structure shall include 932 parking stalls, 300 of which shall have access to electric charging points.
- AQ-9 The Air Cargo Sort Building shall incorporate all of the following design specifications and technologies:
 - Building automation
 - Efficient, heat pump HVAC
 - Natural ventilation
 - Purchase of electricity from the SCE 100% Clean Rate Program, if and to the extent feasible
 - Efficient dock seals
 - Rapid rise doors
 - Solar shades
 - Low use water appliances
 - Sustainable, drought-tolerant landscaping featuring a native, non-invasive vegetation palette
 - Submeters with advanced energy monitoring
 - Main meter energy monitoring
 - Efficient transformers
 - Battery storage-ready infrastructure
 - Building automation by an enhanced building management system
 - Enhanced glazing
- AQ-10 The Project shall include electric charging infrastructure in the truckyard that, at a minimum, accords with all applicable requirements of California's Building Energy Efficiency Standards, as set forth within Title 24, Part 6, of the California Code of Regulations.
- AQ-11 The storage and maintenance of Project-related delivery trucks shall occur only on site. In the event that overnight parking of delivery trucks is necessary, such trucks shall be parked within the Project site.

Reference:

For a complete discussion of Project cumulative impacts related to air quality, see Section 5.2.4 of the Draft EIR at pages 5.2-88 - 5.2-90, Section 2.0 of the Final EIR at pages 2.0-29 through 2.0-37, and Section 3.0 of the Final EIR (Topical Responses 1, Responses to Comment Letter C, and Responses to Comment Letter H).

3.3 Greenhous Gas Emissions

Generate greenhouse gas emissions, either directly or indirectly, that may have a significant impact on the environment.

Impact Summary:

The proposed Project's net increase in GHG emissions during Phase 1 and Phase 2 (buildout of the Project) compared to Baseline Conditions is considered a significant impact on the environment. It is the accumulation of GHGs in the atmosphere that may result in global climate change. Climate change impacts are cumulative in nature and, thus, no typical single project would result in emissions of such a magnitude that it, in and of itself, would be significant on a project basis. implementation of the proposed Project would result in a significant and unavoidable impact related to GHG emissions; hence, the proposed Project's incremental contribution of GHG emissions, both before and after mitigation, is considered to be cumulatively considerable.

Significant Effect: GHG-1

The Project would result in a net increase in GHG emissions during Phase 1 and Phase 2 (buildout of the Project) operation over Baseline Conditions, which is considered to be a significant impact on the environment.

Findings:

Pursuant to Public Resources Code Section 21081(a)(1), the OIAA finds that changes or alterations have been required in, or incorporated into, the Project that avoid or substantially lessen the foregoing significant impact as identified in the EIR. However, these effects have not been reduced to a less-than-significant level.

Thus, pursuant to Public Resources Code Section 21081(a)(3), the OIAA finds that specific economic, legal, social, technological, other considerations, including considerations for the provision of employment opportunities for highly trained workers, make infeasible the mitigation measures or alternatives identified in the EIR. There are no feasible mitigation measures or alternatives that would reduce the identified significant impact to a level below significant or otherwise reduce the impact. Therefore, the impact remains significant and unavoidable.

Facts in Support of Findings:

The estimated construction GHG emissions for the proposed Project are 7,248 MTCO₂e. The 30year amortized construction related GHG emissions would be approximately 242 metric tons of MTCO₂e per year. With implementation of the proposed Project, operational annual GHG emissions would be 79,798 MTCO₂e annually for Phase 1 and 128,057 MTCO₂e annually for Phase 2 (buildout of the Project) when compared to emissions under Baseline Conditions. The net increase in GHG emissions over Baseline Conditions is considered to be a significant impact on the environment. As such, impacts would be potentially significant.

The Applicant's approach to building design employs a broad range of green building technologies to achieve carbon neutral design for all of its new buildings (*i.e.*, zero emission buildings) by incorporating a variety of technologies into the building design to reduce energy use, track energy

consumption to support identification of further improvements, generate renewable energy on site and utilize clean energy sources.

Project Design Features **PDF AQ-1** through **PDF AQ-11** and **Mitigation Measures MM AQ-1** through **MM AQ-16**, as well as **MM TRANS-1** through **MM TRANS-5** would serve to reduce GHG emissions. Additionally, the proposed Project includes Project Design Features **PDF GHG-1** and **PDF GHG-2** to reduce GHG emissions to the greatest extent feasible. Neither the SCAQMD nor OIAA have the authority to regulate aircraft operations or emissions from aircraft engines and the majority of the emissions estimated for operation of the Project are from aircraft operations. The federal Clean Air Act exclusively vests the authority to promulgate emission standards for aircraft and aircraft engines with the USEPA; states and other municipalities are preempted from adopting or enforcing any standard with respect to aircraft engine emissions associated with the Project, while it is anticipated future technology improvements are anticipated to reduce Project GHG emissions over time, there are no additional feasible mitigation measures available at this time that would reduce GHG emissions to below significance thresholds and for this reason, operational GHG emissions would remain significant after implementation of all feasible mitigation.

Thus, pursuant to Public Resources Code Section 21081(a)(3), based on the evidence described below in Section 10.0, Statement of Overriding Considerations, the OIAA finds that specific, economic, legal, social, technological, or other considerations, including considerations for the provision of employment opportunities for highly trained worked, makes infeasible the mitigation measures or alternatives identified in the EIR to reduce this impact to a less-than-significant level.

Mitigation Measures:

AQ-1, AQ-2, AQ-3, AQ-4, AQ-5, AQ-6, AQ-7, AQ-8, AQ-9, AQ-10, AQ-11, AQ-12, AQ-13, AQ-14, AQ-15, AQ-16 TRANS-1, TRANS-2, TRANS-3, TRANS-4, and TRANS-5.

- AQ-1 The Applicant shall require that construction vendors, contractors, and/or haul truck operators commit to using 2010 model year trucks (e.g., material delivery trucks and soil import/export with a gross vehicle weight rating of at least 14,001 pounds), that meet CARB's 2010 engine emissions standards or newer, cleaner trucks. The OIAA shall confirm that the Applicant includes this requirement in applicable bid documents, purchase orders, and contracts. Operators shall maintain records of all trucks associated with Project construction to document that each truck used meets these emission standards and make the records available for inspection.
- AQ-2 The Applicant shall require that construction equipment such as concrete/industrial saws, pumps, aerial lifts, light stands, air compressors, and forklifts be electric or alternative-fueled (i.e., non-diesel), where feasible. Pole power shall be utilized at the earliest feasible point in time and shall be used to the maximum extent feasible in lieu of generators.
- AQ-3 The Applicant shall support and encourage ridesharing and transit incentives for the construction crew by providing crews with the resources needed to organize

rideshares, such as bulletin boards or email announcements. The Applicant shall also partially subsidize transit fares or passes for the construction crew members who can feasibly use transit. The Applicant shall set a goal to achieve ten percent total construction worker participation in ridesharing programs and transit use.

- AQ-4 The Applicant shall require, in addition to the GSE noted within PDF AQ-3, all other on-site cargo-handling equipment, such as yard trucks, holsters, yard goats, pallet jacks, and similar equipment, to be electric, with the necessary electrical charging stations provided.
- AQ-5 The Applicant shall require, if and to the extent feasible, the use of zero-emission or near zero-emission on-road heavy duty trucks as part of business operations beginning in 2025 (within at least 25 percent of the Project fleet).

The Applicant also shall require, if and to the extent feasible, the use of zeroemission or near zero-emission on-road heavy duty trucks as part of the business operations beginning in 2029 (within at least 50 percent of the Project fleet).

- AQ-6 The Applicant shall include in the design requirements for the Project cool roof installation to the extent roof space is not occupied by solar panels, in order to reduce energy use and urban heat island effects.
- AQ-7 The Applicant shall encourage the use of single engine taxi operations for Project aircraft.
- AQ-8 The Applicant shall utilize Energy Star heating, cooling, and lighting devices, and appliances (e.g., Heating, Ventilation, and Air Conditioning (HVAC) units in the form of energy efficient commercial heat pumps) within the interior of the Air Cargo Sort Building.
- AQ-9 In order to reduce trips to and from the Project site during construction, the Applicant shall provide on-site food trucks during meal times.
- AQ-10 Interior- and exterior-facing signs, including signs directed toward all dock and delivery areas, shall be posted by the Applicant to identify contact information to report idling violations to CARB, SCAQMD, and the building manager. These signs also shall inform truck drivers to shut off their engines when not in use.
- AQ-11 Electric plugs for electric transport refrigeration units shall be provided at dock doors located at the Air Cargo Sort Building.
- AQ-12 The Applicant shall train operational managers and employees on efficient scheduling and load management to eliminate unnecessary queuing of trucks.
- AQ-13 Signs shall be posted by the Applicant at every truck exit driveway providing directional information to use truck routes as designated by the City of Ontario.

- AQ-14 The Applicant shall require its facility operator(s) to train the staff in charge of keeping vehicle records on diesel technologies and compliance with CARB regulations by attending CARB-approved courses. Also, the Applicant shall require its facility operator(s) to maintain records on-site demonstrating regulatory compliance and make records available for inspection by OIAA, SCAQMD, and State of California upon request.
- AQ-15 The Applicant shall include a provision in all operational freight hauling contracts requiring the use of 2010 model year trucks that meet CARB's 2010 engine emission standards, or newer and cleaner trucks, if and to the extent feasible.
- AQ-16 During construction, the Applicant shall post interior- and exterior-facing signs to inform construction contractors to shut off truck and equipment engines when not in use.
- **TRANS-1** Voluntary Commute Reduction Program. The proposed Project shall implement Voluntary Commute Trip Reduction (CTR) programs that discourage single-occupancy vehicle trips and encourage alternative modes of transportation, such as carpooling, taking transit, walking, and biking. Voluntary CTR programs shall include the following elements to apply the VMT reductions reported in literature:
 - Employer-provided services, infrastructure, and/or incentives for commuting to work using alternative modes (e.g., walking, biking, carpooling/vanpooling, or taking transit).
 - Provide information, coordination, and marketing for employee rideshare services, provide onsite infrastructure to support carpools/vanpools, and provide incentives (e.g., free transit passes, monthly bonus for carpooling 3 or more times a week, etc.).

Employer costs may include recurring costs for carpool/vanpool subsidies, capital and maintenance costs for the alternative transportation infrastructure (e.g., showers and lockers), and labor costs for staff to manage the program.

- **TRANS-2 Provide Ridesharing Program.** A ridesharing program shall be implemented for employees of the site. The following elements designed to support the Project's ridesharing program:
 - Provide vanpool parking with designated passenger loading/unloading area near employee entrance.
 - Create a Carpool Incentive Program.
 - Provide a minimum of ten (10) carpool parking spaces provided closer to the employee entrance than standard parking spaces.
 - Provide access to a carpool database (Metro rideshare) and/or an on-site matching program for employees.

 Provide a monthly incentive for employees that carpool a minimum of three (3) days per week (e.g., \$50 gas card or a \$50 green commuter bonus).

In addition, a staff person would be designated to for provide rideshare information to employees and monitoring the effectiveness of the program.

It is assumed all employees are eligible and that additional carpool spaces could be designated if warranted by demand.

TRANS-3 Implement Subsidized or Discounted Transit Program. Subsidized, discounted, or free Omnitrans, Metrolink or Amtrak transit passes shall be provided to employees to encourage use of transit routes/stops located less than a mile from the Project. It is assumed free transit passes are available to all employees.

Based on the given shift times of the Project, shifts that start or end at 11:00 PM shall have limited available options as most routes do not provide service that late. This shall limit approximately half the employees from the ability to rely on transit.

TRANS-4 Bicycle Facilities. On-site bicycle parking and end-of-trip facilities shall be provided for employee use. End-of-trip facilities include bike parking, bike lockers, showers, and personal lockers.

A bike share program (standard or electric bikes) for employees shall supplement bicycle facilities.

- **TRANS-5** Employer-Sponsored Vanpool Program. An employer-sponsored vanpool service shall be implemented and be fully funded by the tenant as follows:
 - Provide a minimum of one (1) and up to three (3) vanpool vehicles and associated parking with designated passenger loading/unloading area near employee entrance.
 - Pay for the lease of a minimum of one (1) van and up to three (3) vans for the purpose of employee vanpooling.
 - A ten percent voluntary participation rate is assumed to be the high end of the range for this project.

Project Design Features:

AQ-1, AQ-2, AQ-3, AQ-4, AQ-5, AQ-6, AQ-7, AQ-8, AQ-9, AQ-10, and AQ-11, GHG-1, GHG-2

AQ-1 For all phases of construction activity, the Applicant shall require use of off-road construction equipment that is zero emission, if and to the extent available, or diesel-fueled off-road construction equipment that meets the USEPA's Tier 4 emissions standards for off-road diesel-powered construction equipment with 50

Ontario International Airport South Airport Cargo Center Project horsepower (hp) or greater. To ensure that Tier 4 or the cleanest construction equipment available would be used during the Project's construction, the OIAA shall confirm that the Applicant includes this requirement in applicable bid documents, purchase orders, and contracts. Additionally, the OIAA shall confirm that the Applicant also requires periodic reporting and provision of written construction documents by construction contractor(s) and conducts regular inspections to the maximum extent feasible to ensure and enforce compliance.

- AQ-2 The Applicant shall conduct concrete/asphalt demolition on-site to reuse concrete/asphalt generated during construction. During Phase 1, demolition would involve removal of approximately 2,047,320 square feet of asphalt/concrete, which would be recycled within the project site and not require offsite haul truck trips (i.e., avoiding 2,616 haul truck trips). During Phase 2, demolition would involve removal of approximately 1,045,440 square feet of asphalt/concrete, which would be recycled within the project site and not require offsite haul truck trips (i.e., avoiding 910 haul truck trips).
- AQ-3 The Ground Support Equipment (GSE), including (but not limited to) aircraft tugs, baggage tugs, belt loaders, cargo loaders, forklifts, and ground power units, ramp support carts/vans, servicing aircrafts shall be electric by Phase 2.
- AQ-4 A portion of the proposed Project's aircraft fleet shall include electric cargo aircraft. (See Table 3.4 in Section 3.0: Project Description).
- AQ-5 All new aircraft parking positions shall be equipped with ground power and preconditioned air, therefore reducing the need to operate auxiliary power units.
- AQ-6 The Applicant shall conduct maintenance and/or testing on each of the seven standby generators on separate days to limit daily emissions from maintenance/testing activities.
- AQ-7 The Air Cargo Sort Building shall meet Leadership in Energy and Environmental Design (LEED) certification standards, shall include enhanced building automation systems, and shall utilize advanced low energy HVAC systems.
- AQ-8 The visitor parking lot shall include 29 parking stalls, 6 of which shall have access to electric charging points. The employee parking structure shall include 932 parking stalls, 300 of which shall have access to electric charging points.
- AQ-9 The Air Cargo Sort Building shall incorporate all of the following design specifications and technologies:
 - Building automation
 - Efficient, heat pump HVAC
 - Natural ventilation
 - Purchase of electricity from the SCE 100% Clean Rate Program, if and to the extent feasible

- Efficient dock seals
- Rapid rise doors
- Solar shades
- Low use water appliances
- Sustainable, drought-tolerant landscaping featuring a native, non-invasive vegetation palette
- Submeters with advanced energy monitoring
- Main meter energy monitoring
- Efficient transformers
- Battery storage-ready infrastructure
- Building automation by an enhanced building management system
- Enhanced glazing
- AQ-10 The Project shall include electric charging infrastructure in the truckyard that, at a minimum, accords with all applicable requirements of California's Building Energy Efficiency Standards, as set forth within Title 24, Part 6, of the California Code of Regulations.
- AQ-11 The storage and maintenance of Project-related delivery trucks shall occur only onsite. In the event that overnight parking of delivery trucks is necessary, such trucks shall be parked within the Project site.
- GHG-1 The Air Cargo Sort Building shall be all-electric (no natural gas usage).
- **GHG-2** The proposed Project shall include a 3.8-Megawatt Solar PV Panel System on the rooftop of the Air Cargo Sort Building and Parking Structure.

Reference:

For a complete discussion of Project impacts related to greenhouse gas emissions, see Section 5.7. of the Draft EIR at pages 5.7.35 - 5.7-38, Section 2.0 of the Final EIR at pages 2.0-38 through 2.0-44, and Section 3.0 of the Final EIR (Topical Responses 1, Responses to Comment Letter C, and Responses to Comment Letter H).

Conflict with an applicable plan, policy or regulation adopted for the purpose of reducing the emissions of greenhouse gases.

Impact Summary:

Implementation of the proposed Project would not conflict with many of the plans, policies, and regulations that have been adopted for the purpose of reducing GHG emissions. However, the Project may conflict with some plans, policies, and regulations (including Executive Orders S-3-05, B-30-15 and B-55-18, and the 2022 Climate Change Scoping Plan) that are targeting overall reductions in California's emissions profile and carbon neutrality due to its incremental contribution of additional GHG emissions to the atmosphere.

Significant Effect: GHG-2

The Project would conflict with an applicable plan, policy or regulation adopted for the purpose of reducing the emissions of greenhouse gases.

Findings:

Pursuant to Public Resources Code Section 21081(a)(1), the OIAA finds that change or alterations have been required in, or incorporated into, the Project that avoid or substantially lessen the foregoing significant impact as identified in the EIR. However, these effects have not been reduced to a less-than-significant level.

Thus, pursuant to Public Resources Code Section 21081(a)(3), the OIAA finds that specific economic, legal, social, technological, other considerations, including considerations for the provision of employment opportunities for highly trained workers, make infeasible the mitigation measures or alternatives identified in the EIR. There are no feasible mitigation measures or alternatives that would reduce the identified significant impact to a level below significant or otherwise reduce the impact. Therefore, the impact remains significant and unavoidable.

Facts in Support of Findings:

The proposed Project may conflict with some plans, policies, and regulations, including Executive Orders S-3-05, B-30-15, and B-55-18; and the 2022 Climate Change Scoping Plan due to its incremental contribution of additional GHG emissions to the atmosphere. Statewide GHG reduction targets are not directly applicable to individual projects. However, the proposed Project would increase emissions at the Airport when compared to Baseline Conditions. Further, OIAA does not have the authority to regulate aircraft operations or emissions from aircraft engines. The federal Clean Air Act exclusively vests the authority to promulgate emission standards for aircraft and aircraft engines with the USEPA; states and other municipalities are preempted from adopting or enforcing any standard with respect to aircraft engine emissions unless such standard is identical to USEPA standards. In light of the proposed Project's increase in GHG emissions above Baseline Conditions, the proposed Project may conflict with the State's ability to achieve statewide GHG reduction targets. As such, impacts would be potentially significant.

It is anticipated, however, that future aircraft-related GHG emissions will be lower than currently projected based on the continuing trend of improvements in aircraft engine design and lighter, more fuel-efficient aircraft, and use of cleaner aviation fuels, which would serve to reduce GHG emissions, even though these improvements are beyond the scope of the proposed Project and are not within the control of the OIAA.

While it is anticipated future technology improvements are anticipated to reduce Project GHG emissions over time, there are no additional feasible mitigation measures available at this time that would reduce GHG emissions to below significance thresholds and for this reason, operational GHG emissions would remain significant after implementation of all feasible mitigation.

Thus, pursuant to Public Resources Code Section 21081(a)(3), based on the evidence described below in Section 10.0, Statement of Overriding Considerations, the OIAA finds that specific, economic, legal, social, technological, or other considerations, including considerations for the

provision of employment opportunities for highly trained worked, makes infeasible the mitigation measures or alternatives identified in the EIR to reduce this impact to a less-than-significant level.

Mitigation Measures:

None

Project Design Features:

None

Reference:

For a complete discussion of Project impacts related to greenhouse gas emissions, see Section 5.7. of the Draft EIR at pages 5.7.38 - 5.7.44. Section 2.0 of the Final EIR at pages 2.0.38 through 2.0-44, and Section 3.0 of the Final EIR (Topical Responses 1, Responses to Comment Letter C, and Responses to Comment Letter H).

3.4 Greenhous Gas Emissions (Cumulative)

The Draft EIR discusses the impacts related to greenhouse gas emissions in Section 5.7. See 3.3 Greenhouse Gas Emissions, above, for discussion addressing potential cumulative impacts with respect to greenhouse gas emissions.

Impact Summary:

See 3.3 Greenhouse Gas Emissions, above..

Significant Effect:

See 3.3 Greenhouse Gas Emissions, above.

Findings:

See 3.3 Greenhouse Gas Emissions, above.

Facts in Support of Findings:

See 3.3 Greenhouse Gas Emissions, above.

Reference

For a complete discussion of Project cumulative impacts related to greenhouse gas emissions, see Section 5.7.4 of the Draft EIR at pages 5.7-44 - 5.7-45. Section 2.0 of the Final EIR at pages 2.0-38 through 2.0-44, and Section 3.0 of the Final EIR (Topical Responses 1, Responses to Comment Letter C, and Responses to Comment Letter H).

3.5 Transportation

Would the project conflict or be inconsistent with CEQA Guidelines section 15064.3, subdivision (b).

Impact Summary:

The VMT analysis for the Project was conducted in accordance with the VMT guidelines in the City of Ontario's VMT Impact Resolution, adopted in June 2020. For employees, guests, deliveries and empty trucks, trip generation estimates were multiplied by average trip lengths to estimate average daily VMT. Average trip lengths from SBTAM were interpolated between base and future years to estimate Project Opening Years Phase 1 (2025) and Phase 2 (2029) trip lengths for the employee trips. The fixed-route truck trips were each multiplied by the route distance to estimate fixed-route truck VMT. Fuel truck trips were also multiplied by the route distance to estimate fuel truck trip VMT. Implementation of these mitigation measures is not anticipated to reduce the VMT impact of the proposed Project to a less-than significant level.

Significant Effect: TRA-2

The truck, employee and other trips generated by the proposed Project during operation would result in the Project Total vehicle miles traveled (VMT) per service population (employees for this proposed Project) being 22 percent above the City's VMT significance threshold of 29.76 VMT per service population. Therefore, the Project would result in a significant, unavoidable impact due to operation related VMT.

Findings:

Pursuant to Public Resources Code Section 21081(a)(1), the OIAA finds that change or alterations have been required in, or incorporated into, the Project that avoid or substantially lessen the foregoing significant impact as identified in the EIR. However, these effects have not been reduced to a less-than-significant level.

Thus, pursuant to Public Resources Code Section 21081(a)(3), the OIAA finds that specific economic, legal, social, technological, other considerations, including considerations for the provision of employment opportunities for highly trained workers, make infeasible the mitigation measures or alternatives identified in the EIR. There are no feasible mitigation measures or alternatives that would reduce the identified significant impact to a level below significant or otherwise reduce the impact. Therefore, the impact remains significant and unavoidable.

Facts in Support of Findings:

The Total VMT per service population of the Project site is compared to the Ontario Plan Buildout Conditions VMT per service population to determine if it exceeds the City's impact threshold for VMT under for Phase 1 Opening Year (2025), Phase 2 Opening Year (2029), and under Cumulative Year (2040) conditions. Trip generation estimates were multiplied by average trip lengths to estimate average daily VMT. Phase 1 Opening Year (2025) anticipates 2,777 new Project trips and a VMT of 45,411. Phase 2 Opening Year (2029) anticipates 2,824 new Project trips and a VMT of 50,163.

A compilation of substantial evidence that describes why the Project would or would not have a significant impact on VMT, utilized for projects with unique characteristics that cannot be accurately analyzed using SBTAM or the SCAG RTP/SCS model. The proposed Project Total VMT per service population is 23 percent above the City's VMT significance threshold. The majority of the proposed Project VMT would be generated by trucks, as the proposed Project is a logistics facility. The operations and economic viability of the proposed Project relies on trucks picking up and delivering goods. SCAG identifies 900 daily truck trips for Ontario Airport in 2016 and projects 1,725 daily truck trips in 2045 (Table 14 in the Aviation and Ground Access appendix to the RTP/SCS). The Project would generate 672 truck trips per day, an amount that is within, and consistent with, the 2045 truck trip estimate for Ontario Airport. When these truck trips are considered as part of the total Project VMT, the impact of the Project is significant in relation to the City's VMT threshold.

The Cumulative Conditions (2040) anticipated 2,824 new Project trips and 50,465 new proposed Project VMT would increase Citywide VMT on a daily level in the City. The truck VMT is anticipated to be slightly higher compared to more urbanized airports, given the frequency of trips between these airports and other locations. The proposed Project would cause total daily VMT within the City to be higher than the no project alternative under cumulative conditions, based on the qualitative assessment. The proposed Project Total VMT per service population is 23 percent above the City's VMT significance threshold. The majority of the proposed Project VMT would be generated by trucks, as the proposed Project is a logistics facility. When these truck trips are considered as part of the total Project VMT, the impact of the proposed Project is significant in relation to the City's VMT threshold. Mitigation Measures TRANS-1 through TRANS-5 would be implemented to reduce proposed Project VMT to the maximum extent feasible, with maximum effectiveness of 5.10% reduction on total or commute VMT. Note that page 5.12-65 of the Draft EIR, and page 57-58 of the Updated Traffic Study (Appendix 1.0 of the Final EIR), discuss that the voluntary participation rate for the vanpool program for this proposed Project was reduced to 0.00-5.17% because of duplicative dampening, which occurs when multiple Transportation Demand Management (TDM) measures are applied that target the same users, the effectiveness of these mitigation measures is reduced when they are implemented together. Implementation of these mitigation measures is not anticipated to reduce the VMT impact of the proposed Project to a less-than significant level. Therefore, the impact remains significant and unavoidable.

No feasible mitigation is available because the Project Applicant is only responsible for mitigating impacts of their Project to the extent feasible and cannot implement or be responsible for mitigation measures on other development projects that are under construction at the same time as the Project in a manner that would reduce cumulative noise impacts. Thus, pursuant to Public Resources Code Section 21081(a)(3), based on the evidence described below in Section 10.0, Statement of Overriding Considerations, the OIAA finds that specific, economic, legal, social, technological, or other considerations, including considerations for the provision of employment opportunities for highly trained worked, makes infeasible the mitigation measures or alternatives identified in the EIR to reduce this impact to a less-than-significant level.

Mitigation Measures:

TRANS-1, TRANS-2, TRANS-3, TRANS-4, and TRANS-5

Ontario International Airport South Airport Cargo Center Project

- **TRANS-1** Voluntary Commute Reduction Program. The proposed Project shall implement Voluntary Commute Trip Reduction (CTR) programs that discourage single-occupancy vehicle trips and encourage alternative modes of transportation, such as carpooling, taking transit, walking, and biking. Voluntary CTR programs shall include the following elements to apply the VMT reductions reported in literature:
 - Employer-provided services, infrastructure, and/or incentives for commuting to work using alternative modes (e.g., walking, biking, carpooling/vanpooling, or taking transit).
 - Provide information, coordination, and marketing for employee rideshare services, provide onsite infrastructure to support carpools/vanpools, and provide incentives (e.g., free transit passes, monthly bonus for carpooling 3 or more times a week, etc.).

Employer costs may include recurring costs for carpool/vanpool subsidies, capital and maintenance costs for the alternative transportation infrastructure (e.g., showers and lockers), and labor costs for staff to manage the program.

- **TRANS-2 Provide Ridesharing Program.** A ridesharing program shall be implemented for employees of the site. The following elements designed to support the Project's ridesharing program:
 - Provide vanpool parking with designated passenger loading/unloading area near employee entrance.
 - Create a Carpool Incentive Program.
 - Provide a minimum of ten (10) carpool parking spaces provided closer to the employee entrance than standard parking spaces.
 - Provide access to a carpool database (Metro rideshare) and/or an on-site matching program for employees.
 - Provide a monthly incentive for employees that carpool a minimum of three (3) days per week (e.g., \$50 gas card or a \$50 green commuter bonus).

In addition, a staff person would be designated to for provide rideshare information to employees and monitoring the effectiveness of the program.

It is assumed all employees are eligible and that additional carpool spaces could be designated if warranted by demand.

TRANS-3 Implement Subsidized or Discounted Transit Program. Subsidized, discounted, or free Omnitrans, Metrolink or Amtrak transit passes shall be provided to employees to encourage use of transit routes/stops located less than a mile from the Project. It is assumed free transit passes are available to all employees.

Based on the given shift times of the Project, shifts that start or end at 11:00 PM shall have limited available options as most routes do not provide service that late. This shall limit approximately half the employees from the ability to rely on transit.

TRANS-4 Bicycle Facilities. On-site bicycle parking and end-of-trip facilities shall be provided for employee use. End-of-trip facilities include bike parking, bike lockers, showers, and personal lockers.

A bike share program (standard or electric bikes) for employees shall supplement bicycle facilities.

- **TRANS-5** Employer-Sponsored Vanpool Program. An employer-sponsored vanpool service shall be implemented and be fully funded by the tenant as follows:
 - Provide a minimum of one (1) and up to three (3) vanpool vehicles and associated parking with designated passenger loading/unloading area near employee entrance.
 - Pay for the lease of a minimum of one (1) van and up to three (3) vans for the purpose of employee vanpooling.
 - A ten percent voluntary participation rate is assumed to be the high end of the range for this project.

Reference:

For a complete discussion of Project impacts related to transportation, see Section 5.12. of the Draft EIR at pages 5.12-61 - 5.16-65.

3.6 Transportation (Cumulative)

The Draft EIR discusses the impacts related to transportation in Section 5.12. The following discussion addresses potential cumulative impacts with respect to transportation.

Impact Summary:

The City boundary VMT was evaluated, finding under baseline conditions, 5,501,208 daily VMT, and under future general plan buildout conditions, 8,320,682 daily VMT. Project VMT under Year (2040) conditions would be above the citywide average threshold of significance.

Significant Effect:

The Project's cumulative contribution to transportation would be cumulatively considerable.

Findings:

Pursuant to Public Resources Code Section 21081(a)(1), the OIAA finds that change or alterations have been required in, or incorporated into, the Project that avoid or substantially lessen the foregoing significant impact as identified in the EIR. However, these effects have not been reduced to a less-than-significant level.

Thus, pursuant to Public Resources Code Section 21081(a)(3), the OIAA finds that specific economic, legal, social, technological, other considerations, including considerations for the provision of employment opportunities for highly trained workers, make infeasible the mitigation measures or alternatives identified in the EIR. There are no feasible mitigation measures or alternatives that would reduce the identified significant impact to a level below significant or otherwise reduce the impact. Therefore, the impact remains significant and unavoidable.

Facts in Support of Findings:

Construction-related cumulative impacts would be less than significant.

A hybrid approach was used to estimate Project VMT because the available travel demand models for the region (SBTAM and the Southern California Association of Governments (SCAG) Model) are not as accurate at estimating trips, trip length, or VMT associated with the trucking activity associated with an air cargo facility serving a large region. The best way to perform boundary method VMT forecasts consistent with the City's Adopted VMT Resolution would be with a travel demand model. Given the model limitations noted above, the value in the results of the boundary method assessment to understand the Project's effect on VMT would be erroneous.

Given these limitations, a qualitative assessment of the Project's effect on VMT was performed. A qualitative assessment of VMT is a compilation of substantial evidence that describes why a project would or would not have a significant impact on VMT. Qualitative assessments may be used for projects that have unique characteristics that cannot be accurately analyzed using SBTAM or the SCAG RTP/SCS model. Qualitative assessments can include economic or market analysis, socioeconomic or demographic data, or other substantial evidence to support the significance finding.

The City boundary VMT was evaluated, finding under baseline conditions, 5,501,208 daily VMT, and under future general plan buildout conditions, 8,320,682 daily VMT. As shown in **Table 5.12-16**, of the Draft EIR, VMT under Year (2040) conditions would be above the citywide average threshold of significance.

Based on the 2,824 new trips estimated for the Project with an average travel length of approximately five miles within the City boundary, it is estimated that the boundary VMT would increase by approximately 14,120 VMT. This would equate to an increase in Base Year boundary VMT of approximately 0.25 percent in Base Year conditions and 0.17 percent in future buildout conditions. Based on these estimates, it is reasonable to conclude that the Project would increase citywide VMT on a daily level in the City.

Truck VMT was reviewed associated with the Project as proposed based on other available airports in the SCAG region and based on the known Project origins and destinations for Project trucks and known truck routes. The truck VMT is anticipated to be slightly higher as compared to more urbanized airports given the frequency of trips between Long Beach Airport, LAX, and the Airport, and other locations to the north. It is anticipated that commute VMT in Los Angeles would be lower due to higher densities and better access to transit such that the overall VMT would be higher in the City. Based on this qualitative assessment, it was concluded that the Project would cause total daily VMT within the City to increase under cumulative conditions (see **Appendix 5.12-1**, of the Draft EIR). Though development of the proposed Project is expected to occur over an 8-year timeframe, the operational improvement measures would be implemented to ensure the safe and efficient operation of the roadway system. Therefore, the Project's contribution to cumulative VMT impacts would be cumulatively considerable.

The proposed Project is anticipated to increase citywide daily VMT within the City boundary. For these reasons, the Project VMT impacts are significant. Feasible VMT reduction strategies that would be appropriate for the proposed Project are recommended. However, given the maximum reduction potential associated with **Mitigation Measures TRANS-1** through **TRANS-5**, the VMT impacts of the Project would remain significant. Note that page 5.12-65 of the Draft EIR, and page 57-58 of the Updated Traffic Study (Appendix 1.0 of the Final EIR), discuss that the voluntary participation rate for the vanpool program for this proposed Project was reduced to 0.00-5.17% because of duplicative dampening, which occurs when multiple Transportation Demand Management (TDM) measures are applied that target the same users, the effectiveness of these mitigation measures is reduced when they are implemented together.

No feasible mitigation is available because the Project Applicant is only responsible for mitigating impacts of their Project to the extent feasible and cannot implement or be responsible for mitigation measures on other development projects that are under construction at the same time as the Project in a manner that would reduce cumulative noise impacts. Thus, pursuant to Public Resources Code Section 21081(a)(3), based on the evidence described below in Section 10.0, Statement of Overriding Considerations, the OIAA finds that specific, economic, legal, social, technological, or other considerations, including considerations for the provision of employment opportunities for highly trained worked, makes infeasible the mitigation measures or alternatives identified in the EIR to reduce this impact to a less-than-significant level.

Mitigation Measures:

TRANS-1, TRANS-2, TRANS-3, TRANS-4, and TRANS-5

- **TRANS-1** Voluntary Commute Reduction Program. The proposed Project shall implement Voluntary Commute Trip Reduction (CTR) programs that discourage single-occupancy vehicle trips and encourage alternative modes of transportation, such as carpooling, taking transit, walking, and biking. Voluntary CTR programs shall include the following elements to apply the VMT reductions reported in literature:
 - Employer-provided services, infrastructure, and/or incentives for commuting to work using alternative modes (e.g., walking, biking, carpooling/vanpooling, or taking transit).
 - Provide information, coordination, and marketing for employee rideshare services, provide onsite infrastructure to support carpools/vanpools, and provide incentives (e.g., free transit passes, monthly bonus for carpooling 3 or more times a week, etc.).

Employer costs may include recurring costs for carpool/vanpool subsidies, capital and maintenance costs for the alternative transportation infrastructure (e.g., showers and lockers), and labor costs for staff to manage the program.

- **TRANS-2 Provide Ridesharing Program.** A ridesharing program shall be implemented for employees of the site. The following elements designed to support the Project's ridesharing program:
 - Provide vanpool parking with designated passenger loading/unloading area near employee entrance.
 - Create a Carpool Incentive Program.
 - Provide a minimum of ten (10) carpool parking spaces provided closer to the employee entrance than standard parking spaces.
 - Provide access to a carpool database (Metro rideshare) and/or an on-site matching program for employees.
 - Provide a monthly incentive for employees that carpool a minimum of three (3) days per week (e.g., \$50 gas card or a \$50 green commuter bonus).

In addition, a staff person would be designated to for provide rideshare information to employees and monitoring the effectiveness of the program.

It is assumed all employees are eligible and that additional carpool spaces could be designated if warranted by demand.

TRANS-3 Implement Subsidized or Discounted Transit Program. Subsidized, discounted, or free Omnitrans, Metrolink or Amtrak transit passes shall be provided to employees to encourage use of transit routes/stops located less than a mile from the Project. It is assumed free transit passes are available to all employees.

Based on the given shift times of the Project, shifts that start or end at 11:00 PM shall have limited available options as most routes do not provide service that late. This shall limit approximately half the employees from the ability to rely on transit.

TRANS-4 Bicycle Facilities. On-site bicycle parking and end-of-trip facilities shall be provided for employee use. End-of-trip facilities include bike parking, bike lockers, showers, and personal lockers.

A bike share program (standard or electric bikes) for employees shall supplement bicycle facilities.

TRANS-5 Employer-Sponsored Vanpool Program. An employer-sponsored vanpool service shall be implemented and be fully funded by the tenant as follows:

- Provide a minimum of one (1) and up to three (3) vanpool vehicles and associated parking with designated passenger loading/unloading area near employee entrance.
- Pay for the lease of a minimum of one (1) van and up to three (3) vans for the purpose of employee vanpooling.
- A ten percent voluntary participation rate is assumed to be the high end of the range for this project.

Reference

For a complete discussion of Project cumulative impacts related to transportation, see Section 5.12.4 of the Draft EIR at pages 5.12-68 - 5.7-69.

4.0 POTENTIALLY SIGNIFICANT EFFECTS THAT CAN BE MITIGATED TO BELOW A LEVEL OF SIGNIFICANCE

The following section sets forth the potentially significant environmental effects of the Project, determined to be mitigated to below a level of significance, and identifies one or more of the required findings that states facts in support of those findings with respect to each effect.

4.1 **Biological Resources**

Have a substantial adverse effect, either directly or through habitat modifications, on any species identified as a candidate, sensitive, or special status species in local or regional plans, policies, or regulations or by the California Department of Fish and Game or US Fish and Wildlife Service.

Impact Summary:

The Project site could support the Cooper's Hawk, California horned lark, and California gull, which are CDFW Watch List Species. The Project site has a low potential to support burrowing owls and for this reason, focused surveys for burrowing owl were not warranted. Nevertheless, the vacant grassy area in the southeast corner of the Project site may provide suitable foraging and cover habitat for the burrowing owl, and construction activities in this area could potentially impact burrowing owls.

Less than Significant Effects with Mitigation: BIO-2

The Project is not expected to have a substantial adverse effect, either directly or through habitat modifications, on any species identified as a candidate, sensitive, or special status species in local or regional plans, policies, or regulations, or by the California Department of Fish and Game or U.S. Fish and Wildlife Service.

Findings:

Pursuant to Public Resources Code Section 21081(a)(1), the OIAA finds that mitigation measures have been incorporated into the Project which mitigate or avoid the significant effects on the environment to Burrowing Owls and nesting birds. Therefore, impacts would be less than significant with mitigation incorporated.
Facts in Support of Findings:

The Project site is not located within federally designated Critical Habitat. Based on habitat requirements for the identified special-status plant species, the Project site does not have the potential to support any of the special-status plant species known to occur within the vicinity of the site. Additionally, the plant species found in the proposed Project area do not provide suitable long-term roosting or maternity habitat. Of the 57 special-status wildlife species have been recorded as observed in the Guasti and Ontario quadrangles, none of the species were observed during the field survey. No burrowing owls or recent signs (i.e., pellets, feathers, castings, or whitewash) were observed at the Project site. Portions of the Project site-primarily the vacant areas on the eastern end— is unvegetated and/or vegetated with a variety of low-growing plant species that allow for line- of-sight observation favored by burrowing owls. However, no suitable burrows (>4 inches in diameter) capable of providing roosting and nesting opportunities were observed. Additionally, the Project site supports and is surrounded by tall structures, light poles, and fences that offer perching opportunities for larger raptor species (i.e., red-tailed hawk) that prey on burrowing owls. Due to the predominance of vacant land in the immediate vicinity of the Project site, the site is isolated from suitable habitats. Further, the intensity and frequency of routine human disturbance associated with on-site weed abatement activities (i.e., mowing) and Airport-related uses (i.e., cargo storage) precludes burrowing owls from occurring on site.

To avoid potential impacts, **Mitigation Measure BIO-1** would require pre-construction surveys to determine the presence of burrowing owls to ensure that any burrowing owls potentially within this area are protected in accordance with CDFW recommendations. Implementation of **Mitigation Measure BIO-2** would require pre-construction Nesting Bird Surveys and would reduce potential impacts to migratory and nesting birds. **Mitigation Measure BIO-3** would require a pre-construction bat roosting survey shall be conducted by a qualified bat biologist on structures and trees being removed or impacted by construction on site that may provide suitable roosting opportunities for local common bat species within 14 days prior to construction. If bats are determined to be present, CDFW shall be consulted on creating a bat mitigation plan.

Project Design Features:

None

Mitigation Measures:

BIO-1, BIO-2, and BIO-3

BIO-1 Burrowing Owl. All disturbed areas of the Project site that were determined to have a low potential to provide suitable habitat for burrowing owls, which primarily includes the existing track infield grassy areas of the Project site, require focused preconstruction surveys to be conducted; the first take avoidance survey shall be conducted within 14 days prior to ground disturbance and the second take avoidance survey shall be conducted 24 hours prior to ground disturbance to determine presence of burrowing owls. These surveys shall conform to the survey protocol established by the CDFW Staff Report on Burrowing Owl Mitigation (CDFW 2012) and will be conducted by a qualified biologist across all suitable breeding, wintering, and foraging habitat within the

Project and appropriate buffer. Copies of the survey results shall be submitted to CDFW and OIAA.

- If no burrowing owls are detected, no further mitigation is necessary.
- If burrowing owls are detected during focused surveys and/or take avoidance surveys, CDFW will be immediately informed of its location and status. The project will avoid all impacts to burrowing owls onsite. If this is not feasible, a Burrowing Owl Protection Plan will be prepared by a qualified biologist, which must be approved by CDFW prior to initiating the project. The Burrowing Owl Protection Plan will include conserving all nesting, occupied, and satellite burrows and/or burrowing owl habitat such that the habitat acreage, number of burrows, and burrowing owls impacted are maintained and/or replaced. Further coordination with CDFW will occur to identify mitigation for the loss of habitat through the acquisition, conservation, and management of in-kind habitat. Lands conserved will include 1) sufficiently large acreage with fossorial mammals present; 2) permanent protection through a conservation easement for the purpose of conserving burrowing owl habitat and prohibiting activities incompatible with burrowing owl use; 3) development and implementation of a mitigation land management plan to address long-term ecological sustainability and maintenance of the site for burrowing owls; and 4) funding for the maintenance and management of mitigation land through the establishment of a long-term funding mechanism such as an endowment (CDFW, 2012).
- **BIO-2** Nesting Birds. Bird nesting season generally extends from February 1 through September 15 in southern California and, specifically, April 15 through August 31 for migratory passerine birds and January 15 to August 31 for raptors. In order to ensure compliance with the Migratory Bird Treaty Act and to avoid impacts to nesting birds (common and special status) during the nesting season, a qualified Avian Biologist must be retained to conduct pre-construction Nesting Bird Surveys (NBS) will occur prior to Project-related disturbance to nestable vegetation to identify any active nests. The NBS shall be performed no more than three days prior to the commencement of construction activities. The survey(s) will occur at the appropriate time of day/night, during appropriate weather conditions. Surveys will encompass all suitable areas, including trees, shrubs, bare ground, burrows, cavities, and structures. Survey duration will take into consideration the acreage of the Project impacts; density, and complexity of the habitat; number of survey participants; survey techniques employed; and will be sufficient to ensure the data collected is complete and accurate. Pre-construction surveys will focus on both direct and indirect evidence of nesting, including nest locations and nesting behavior (i.e., copulation, carrying of food or nest materials, nest building, removal of fecal sacks, flushing suddenly from atypically close range, agitation, aggressive interactions, feigning injury or distraction displays, or other behaviors). The results of the NBS shall be documented by the qualified biologist. If construction is inactive for more than seven days, an additional survey shall be conducted. If no active nests are found, no further action will be required. If a nest is suspected, but not confirmed, the qualified biologist will

establish a disturbance-free buffer until additional surveys can be completed, or until the location can be inferred based on observations. The qualified biologist will not risk failure of the nest to determine the exact location or status and will make every effort to limit the nest to potential predation as a result of the survey/monitoring efforts (i.e., limit number of surveyors, limit time spent at/near the nest, scan the site for potential nest predators before approaching, or immediately depart nest area if indicators of stress or agitation are displayed). If a nest is observed, but thought to be inactive, the qualified biologist will monitor the nest for 1 hour (4 hours for raptors during the nonbreeding season) prior to approaching the nest to determine status. The qualified biologist will use their best professional judgement regarding the monitoring period and whether approaching the nest is appropriate. If an active nest is found, the biologist will set appropriate no-work buffers (typically 300 feet for passerine and non-special status species, and 500 feet for hawks and special-status species) around the nest, which will be based upon the nesting species, its sensitivity to disturbance, nesting stage and expected types, intensity, and duration of disturbance typically 300 feet of a migratory bird and 500 feet for raptors. Once the buffer is established, the qualified biologist will document baseline behavior, stage of reproduction, and existing site conditions, including vertical and horizontal distances from proposed work areas, visual or acoustic barriers, and existing level of disturbance. Following documentation of baseline conditions, the qualified biologist may choose to make adjustments to the buffer based on site characteristics, stage of reproduction, and types of Project activities proposed at/near that location. The qualified biologist will monitor the nest at the onset of Project activities, and at the onset of any changes in Project activities (i.e., increase in number or type of equipment, change in equipment usage, etc.) to determine the efficacy of the buffer. If the qualified biologist determines that Project activities may be causing an adverse reaction, the qualified biologist will adjust the buffer accordingly. The qualified biologist will be onsite daily to monitor all existing nests, the efficacy of established buffers, and to document any new nesting occurrences. The qualified biologist will document the status of all existing nests, including the stage of reproduction and the expected fledge date. If a nest is suspected to have been abandoned or failed, the qualified biologist will monitor the nest for a minimum of 1 hour (4 hours for raptors), uninterrupted, during favorable field conditions. If no activity is observed during that time, the qualified biologist may approach the nest to assess the status. The permittee, under the direction of the qualified biologist, may also take steps to discourage nesting on the Project site, including moving equipment and materials daily, covering material with tarps or fabric, and securing all open pipes and construction materials. The qualified biologist will ensure that none of the materials used pose an entanglement risk to birds or other species.

The buffer shall remain until the young have fledged the nest and the nest is confirmed to no longer be active, or as determined by the qualified biologist. The nests and buffer zones shall be field checked weekly by a qualified biological monitor. The approved no-work buffer zone shall be clearly marked in the field, within which no disturbance activity shall commence until the qualified biologist has determined the young birds have successfully fledged and the nest is inactive.

BIO-3 Roosting Bats. A pre-construction bat roosting survey shall be conducted by a qualified bat biologist on structures and trees being removed or impacted by construction on site that may provide suitable roosting opportunities for local common bat species within 14 days prior to construction. If bats are determined to be present, CDFW shall be consulted on creating a bat mitigation plan.

Reference

For a complete discussion of Project impacts related to biological resources, see Section 5.3.3 of the Draft EIR at pages 5.3-12 - 5.3-21 and Section 3.0 of the Final EIR (Response to Comment G-3).

4.2 **Biological Resources (Cumulative)**

The Draft EIR discusses the impacts related to biological resources in Section 5.3. The following discussion addresses potential cumulative impacts with respect to biological resources.

Impact Summary:

The geographic context for cumulative impact analysis on biological resources includes the City of Ontario and the surrounding cities and communities within the Guasti and Ontario quadrangles. The Draft EIR analyzed the potential cumulative biological resource impacts to these surrounding cities and communities.

Less than Significant Effects with Mitigation:

The Project's cumulative impacts to biological resources would be less than significant with mitigation.

Findings:

Pursuant to Public Resources Code Section 21081(a)(1), the OIAA finds that mitigation measures have been incorporated into the Project which mitigate or avoid the significant effects on the environment to Burrowing Owls and nesting birds. Therefore, cumulative impacts to biological resources would be less than significant with mitigation incorporated.

Facts in Support of Findings:

The Project site is developed and does not contain any sensitive species or habitat. Given the absence of sensitive species and habitat at the Project site, the proposed redevelopment would not significantly contribute to the cumulative loss and/or degradation of the region's biological resources. The proposed Project's potential effects on burrowing owls, migratory and nesting birds, and common bat species are localized and would be fully mitigated with the implementation of **Mitigation Measures BIO-1**, **BIO-2**, and **BIO-3**. The proposed Project's potential effects on the Cucamonga Channel would not be significant. Additionally, Project implementation would be on Airport property, away from public right-of-way. Separate from the proposed Project, the OIAA approved the East Avion Street Realignment Project, which will realign East Avion Street from its current configuration to connect the western Jurupa Avenue terminus (located east of the Project

site) to the existing segment of East Avion Street, near South Hellman Avenue; the realignment project will also improve the segment of East Avion Street fronting the Project site, westward to South Vineyard Street. The East Avion Street Realignment Project's potential impact to trees on public right-of-way was evaluated under a separate environmental review. If required, the proposed Project would maintain any parkway trees adjacent to the Project site to preserve a neat appearance and non-obstructed use of the realigned East Avion Street and impacts would be less than significant. Therefore, the proposed Project's impacts to biological resources would not be cumulatively considerable.

Mitigation Measures BIO-1, BIO-2, and **BIO-3** would reduce potentially significant impacts on burrowing owls, sensitive wildlife species, including migratory and nesting birds, and bats to a less than significant level. Therefore, no cumulative significant and unavoidable adverse impacts related to biological resources would occur from Project implementation.

Project Design Features:

None

Mitigation Measures:

BIO-1, BIO-2, and BIO-3

- **BIO-1 Burrowing Owl.** All disturbed areas of the Project site that were determined to have a low potential to provide suitable habitat for burrowing owls, which primarily includes the existing track infield grassy areas of the Project site, require focused preconstruction surveys to be conducted; the first take avoidance survey shall be conducted within 14 days prior to ground disturbance and the second take avoidance survey shall be conducted 24 hours prior to ground disturbance to determine presence of burrowing owls. These surveys shall conform to the survey protocol established by the CDFW Staff Report on Burrowing Owl Mitigation (CDFW 2012) and will be conducted by a qualified biologist across all suitable breeding, wintering, and foraging habitat within the Project and appropriate buffer. Copies of the survey results shall be submitted to CDFW and OIAA.
 - If no burrowing owls are detected, no further mitigation is necessary.
 - If burrowing owls are detected during focused surveys and/or take avoidance surveys, CDFW will be immediately informed of its location and status. The project will avoid all impacts to burrowing owls onsite. If this is not feasible, a Burrowing Owl Protection Plan will be prepared by a qualified biologist, which must be approved by CDFW prior to initiating the project. The Burrowing Owl Protection Plan will include conserving all nesting, occupied, and satellite burrows and/or burrowing owl habitat such that the habitat acreage, number of burrows, and burrowing owls impacted are maintained and/or replaced. Further coordination with CDFW will occur to identify mitigation for the loss of habitat through the acquisition, conservation, and management of in-kind habitat. Lands conserved will include 1) sufficiently large acreage with fossorial mammals present; 2) permanent protection

through a conservation easement for the purpose of conserving burrowing owl habitat and prohibiting activities incompatible with burrowing owl use; 3) development and implementation of a mitigation land management plan to address long-term ecological sustainability and maintenance of the site for burrowing owls; and 4) funding for the maintenance and management of mitigation land through the establishment of a long-term funding mechanism such as an endowment (CDFW, 2012).

BIO-2 Nesting Birds. Bird nesting season generally extends from February 1 through September 15 in southern California and, specifically, April 15 through August 31 for migratory passerine birds and January 15 to August 31 for raptors. In order to ensure compliance with the Migratory Bird Treaty Act and to avoid impacts to nesting birds (common and special status) during the nesting season, a qualified Avian Biologist must be retained to conduct pre-construction Nesting Bird Surveys (NBS) will occur prior to Project-related disturbance to nestable vegetation to identify any active nests. The NBS shall be performed no more than three days prior to the commencement of construction activities. The survey(s) will occur at the appropriate time of day/night, during appropriate weather conditions. Surveys will encompass all suitable areas, including trees, shrubs, bare ground, burrows, cavities, and structures. Survey duration will take into consideration the acreage of the Project impacts; density, and complexity of the habitat; number of survey participants; survey techniques employed; and will be sufficient to ensure the data collected is complete and accurate. Pre-construction surveys will focus on both direct and indirect evidence of nesting, including nest locations and nesting behavior (i.e., copulation, carrying of food or nest materials, nest building, removal of fecal sacks, flushing suddenly from atypically close range, agitation, aggressive interactions, feigning injury or distraction displays, or other behaviors). The results of the NBS shall be documented by the qualified biologist. If construction is inactive for more than seven days, an additional survey shall be conducted. If no active nests are found, no further action will be required. If a nest is suspected, but not confirmed, the qualified biologist will establish a disturbance-free buffer until additional surveys can be completed, or until the location can be inferred based on observations. The qualified biologist will not risk failure of the nest to determine the exact location or status and will make every effort to limit the nest to potential predation as a result of the survey/monitoring efforts (i.e., limit number of surveyors, limit time spent at/near the nest, scan the site for potential nest predators before approaching, or immediately depart nest area if indicators of stress or agitation are displayed). If a nest is observed, but thought to be inactive, the qualified biologist will monitor the nest for 1 hour (4 hours for raptors during the nonbreeding season) prior to approaching the nest to determine status. The qualified biologist will use their best professional judgement regarding the monitoring period and whether approaching the nest is appropriate. If an active nest is found, the biologist will set appropriate no-work buffers (typically 300 feet for passerine and non-special status species, and 500 feet for hawks and special-status species) around the nest, which will be based upon the nesting species, its sensitivity to disturbance, nesting stage and expected types, intensity, and duration of disturbance –

typically 300 feet of a migratory bird and 500 feet for raptors. Once the buffer is established, the qualified biologist will document baseline behavior, stage of reproduction, and existing site conditions, including vertical and horizontal distances from proposed work areas, visual or acoustic barriers, and existing level of disturbance. Following documentation of baseline conditions, the qualified biologist may choose to make adjustments to the buffer based on site characteristics, stage of reproduction, and types of Project activities proposed at/near that location. The qualified biologist will monitor the nest at the onset of Project activities, and at the onset of any changes in Project activities (i.e., increase in number or type of equipment, change in equipment usage, etc.) to determine the efficacy of the buffer. If the qualified biologist determines that Project activities may be causing an adverse reaction, the qualified biologist will adjust the buffer accordingly. The qualified biologist will be onsite daily to monitor all existing nests, the efficacy of established buffers, and to document any new nesting occurrences. The qualified biologist will document the status of all existing nests, including the stage of reproduction and the expected fledge date. If a nest is suspected to have been abandoned or failed, the qualified biologist will monitor the nest for a minimum of 1 hour (4 hours for raptors), uninterrupted, during favorable field conditions. If no activity is observed during that time, the qualified biologist may approach the nest to assess the status. The permittee, under the direction of the qualified biologist, may also take steps to discourage nesting on the Project site, including moving equipment and materials daily, covering material with tarps or fabric, and securing all open pipes and construction materials. The qualified biologist will ensure that none of the materials used pose an entanglement risk to birds or other species.

The buffer shall remain until the young have fledged the nest and the nest is confirmed to no longer be active, or as determined by the qualified biologist. The nests and buffer zones shall be field checked weekly by a qualified biological monitor. The approved no-work buffer zone shall be clearly marked in the field, within which no disturbance activity shall commence until the qualified biologist has determined the young birds have successfully fledged and the nest is inactive.

BIO-3 Roosting Bats. A pre-construction bat roosting survey shall be conducted by a qualified bat biologist on structures and trees being removed or impacted by construction on site that may provide suitable roosting opportunities for local common bat species within 14 days prior to construction. If bats are determined to be present, CDFW shall be consulted on creating a bat mitigation plan.

Reference

For a complete discussion of Project cumulative impacts related to biological resources, see Section 5.3.4 of the Draft EIR at pages 5.3-47.

4.3 Cultural Resources

The Draft EIR discusses impacts related to cultural resources in Section 5.4. The following discussion addresses potential impacts with respect to archaeological resources.

Cause a substantial change adverse change in the significance of an archaeological resource pursuant to Section 15064.5

Impact Summary:

The Draft EIR analyzed the possibility that farming in the early twentieth century and potential prehistoric occupation at the Project area may have resulted in surface disturbances and deposition of objects and features at the Project site. Therefore, there is a moderate potential for buried objects in the native soil under the Project site. There is low potential for more deeply buried archaeological deposits associated with the early Holocene and late Pleistocene eras. Because ground disturbing activities for the Project could extend to a depth of up to 20 feet below the existing ground surface, ground disturbing activities during construction may encounter native soils containing potential archeological resources.

Less than Significant Effects with Mitigation: CUL-2

The Project would have a less than significant impact with mitigation on archaeological resources.

Findings:

Pursuant to Public Resources Code Section 21081(a)(1), the OIAA finds that mitigation measures have been incorporated into the Project which mitigate or avoid the significant effects on the environment to archaeological resource.

Facts in Support of Findings:

The Archaeological Resource Assessment, which includes a record search and background research, communication with the Native American Heritage Commission (NAHC), and a reconnaissance pedestrian survey, indicate that subsurface soil has been extensively disturbed. This is additionally evidenced by the built nature of the Project site with pavement, multiple buildings, structures, and landscape, as well as installation of related underground utilities. Archival research indicates the proposed Project area was used for agricultural purposes prior to the construction of the Airport and Cucamonga Channel. Geological mapping indicates artificial fill covers the Project site; however, no indication was seen in the historic aerial imagery of either the emplacement of fill or its potential depth. Additionally, the surface may have been used prehistorically. Ground disturbing activities for the proposed Project could extend to a depth of up to 20 feet below the existing ground surface, therefore, there is a moderate potential for buried objects in the native soil under the Project site. Implementation of **Mitigation Measure CUL-1** during construction activities requires archaeological monitoring during grading or other ground disturbing activities and, if objects are encountered, that work in the immediate area be halted and the resources evaluated.

Project Design Features:

None

Mitigation Measures:

CUL-1

CUL-1 Archaeological Monitoring of All Ground-Disturbing Activities During Construction of Phase 1 and Phase 2.

- a) Prior to the issuance of grading permits by the City of Ontario for Phase 1 and Phase 2 of the proposed Project, the OIAA and/or its construction contractor must retain a qualified professional archeologist meeting the Secretary of Interior's PQS for Archaeology (as defined in the Code of Federal Regulations, 36 CFR Part 61). The qualified archaeologist will be retained to conduct monitoring of rough grading activities conducted during both Project phases. The qualified archaeologist shall have the authority to redirect earthmoving activities in the event that suspected cultural resources are unearthed during construction activities.
- b) The qualified archaeologist shall prepare a Cultural Resources Monitoring and Treatment Plan that will describe processes for archaeological monitoring and for handling incidental discovery of objects, features, and cultural resources for all ground-disturbing construction and preconstruction activities.
- c) Prior to the issuance of a grading permit, all construction workers involved with grading and trenching operations shall receive training by the qualified archaeologist to recognize unique archaeological resources, including tribal cultural resources, should such resources be unearthed during grounddisturbing construction activities. The training of all construction workers involved with grading and trenching operations shall explain the importance and legal basis for the protection of significant archaeological resources. It will include a brief review of the cultural sensitivity of the construction area and the surrounding area; what resources could potentially be identified during earthmoving activities; the requirements of the monitoring program; the protocols that apply in the event inadvertent discoveries of cultural resources are identified, including who to contact and appropriate avoidance measures until the find(s) can be properly evaluated; and any other appropriate protocols. All new construction personnel involved with grading and trenching operations that begin work following the initial training session must take the training prior to beginning work; the qualified archaeologist shall be available to provide the training on an as needed basis.
- d) In the event subsurface artifacts or features are encountered during grounddisturbing activities, the construction supervisor shall be required by his contract to immediately halt and redirect grading operations within a 100-foot radius of the discovery and see identification and evaluation and evaluation of the suspected resource by the qualified archaeologist for listing in the NRHP and CRHR. This requirement shall be noted on all grading plans and the construction contractor shall be obligated to comply with the note.
- e) After the qualified archaeologist makes his/her initial assessment of the nature of the find. The archaeologist shall pursue either protection in place or recovery, salvage, and treatment of the deposits. Recovery, salvage, and

treatment protocols shall be developed in accordance with applicable provisions of Public Resource Code Section 21083.2 and State CEQA Guidelines 15064.5 and 15126.4 in consultation with OIAA or with a recognized scientific or educational repository, including the SCCIC. Preservation in place shall be the preferred means to avoid impacts to archaeological resources qualifying as historical resources, consistent with CEQA Guidelines Section 15126.4(b)(3)(C).

Reference

For a complete discussion of Project impacts related to cultural resources, see Section 5.4.3 of the Draft EIR at pages 5.4-28 - 5.4-36.

4.4 Cultural Resources (Cumulative)

The Draft EIR discusses the impacts related to cultural resources in Section 5.4. The following discussion addresses potential cumulative impacts with respect to cultural resources.

Impact Summary:

The Draft EIR analyzed the cumulative cultural resource impact to Project site and the surrounding area. The proposed Project, like other related development projects, would have the potential to impact archaeological resources that may be present in undisturbed native soils during construction.

Less than Significant Effects with Mitigation:

The Project's cumulative impacts to cultural resources would be less than significant with mitigation.

Findings:

Pursuant to Public Resources Code Section 21081(a)(1), the OIAA finds that mitigation measures have been incorporated into the Project which mitigate or avoid the significant effects on the environment to cultural resources. Therefore, cumulative impacts would be less than significant with mitigation incorporated.

Facts in Support of Findings:

The Project would not directly or indirectly impact any historical resources on the Project site and surrounding area. The Project is not a part of a historical district at the Airport, nor is it a contributor to the significance of historical resources and districts in the City, San Bernardino County, and Southern California (as a region); the growth of aviation in the State or nation; or the prewar efforts related to WWII. None of the built environment resources present on the Project site are eligible for listing in the National Register of Historic Places (NRHP), California Register of Historic Resources (CRHR), or as Ontario Landmarks/Historic Districts, and are not historic properties as defined by CEQA. For this reason, the Project will not contribute to any cumulative impact to historic resources on the Airport or in the City of Ontario.

The proposed Project, like other related development projects, would have the potential to impact archaeological resources that may be present in undisturbed native soils during construction. The

Project would be required to implement **Mitigation Measure CUL-1**, which would require an archaeological monitor to observe all ground disturbing activities associated with the Project. If objects are encountered, work in the immediate area will halt and the resources will be evaluated to mitigate potential impacts to less than significant. For this reason, the Project will not contribute to any cumulative impact to archeological resources. Related projects would be required to comply with PRC Section 21083.2(i), which states a lead agency may make provisions for archaeological sites accidentally discovered during construction. If the find is determined to be a unique archaeological resource, contingency funding and a time allotment sufficient to allow recovering an archaeological sample or to employ one of the avoidance measures may be required, during which construction work may continue on other parts of the site. Compliance with PRC Section 21083.2(i) would ensure that provisions are in place to address accidental discoveries of archaeological resources. For these reasons, no significant cumulative impacts to archeological resources will occur.

Every development project in the State would be required to comply with the provisions of California Health and Safety Code § 7050.5, and Public Resources Code § 5097 et seq., which would ensure that human remains uncovered during construction activities are treated in accordance with prescribed, respectful, and appropriate practices. Therefore, the proposed Project—in combination with related projects—would not result in significant cumulative impacts on human remains.

With implementation of **Mitigation Measure CUL-1** and compliance with the regulatory requirements, Project impacts to cultural resources would be less than significant.

Project Design Features:

None

Mitigation Measures:

CUL-1

CUL-1 Archaeological Monitoring of All Ground-Disturbing Activities During Construction of Phase 1 and Phase 2.

- a) Prior to the issuance of grading permits by the City of Ontario for Phase 1 and Phase 2 of the proposed Project, the OIAA and/or its construction contractor must retain a qualified professional archeologist meeting the Secretary of Interior's PQS for Archaeology (as defined in the Code of Federal Regulations, 36 CFR Part 61). The qualified archaeologist will be retained to conduct monitoring of rough grading activities conducted during both Project phases. The qualified archaeologist shall have the authority to redirect earthmoving activities in the event that suspected cultural resources are unearthed during construction activities.
- b) The qualified archaeologist shall prepare a Cultural Resources Monitoring and Treatment Plan that will describe processes for archaeological monitoring and

for handling incidental discovery of objects, features, and cultural resources for all ground-disturbing construction and preconstruction activities.

- c) Prior to the issuance of a grading permit, all construction workers involved with grading and trenching operations shall receive training by the qualified archaeologist to recognize unique archaeological resources, including tribal cultural resources, should such resources be unearthed during grounddisturbing construction activities. The training of all construction workers involved with grading and trenching operations shall explain the importance and legal basis for the protection of significant archaeological resources. It will include a brief review of the cultural sensitivity of the construction area and the surrounding area; what resources could potentially be identified during earthmoving activities; the requirements of the monitoring program; the protocols that apply in the event inadvertent discoveries of cultural resources are identified, including who to contact and appropriate avoidance measures until the find(s) can be properly evaluated; and any other appropriate protocols. All new construction personnel involved with grading and trenching operations that begin work following the initial training session must take the training prior to beginning work; the qualified archaeologist shall be available to provide the training on an as needed basis.
- d) In the event subsurface artifacts or features are encountered during grounddisturbing activities, the construction supervisor shall be required by his contract to immediately halt and redirect grading operations within a 100-foot radius of the discovery and see identification and evaluation and evaluation of the suspected resource by the qualified archaeologist for listing in the NRHP and CRHR. This requirement shall be noted on all grading plans and the construction contractor shall be obligated to comply with the note.
- e) After the qualified archaeologist makes his/her initial assessment of the nature of the find. The archaeologist shall pursue either protection in place or recovery, salvage, and treatment of the deposits. Recovery, salvage, and treatment protocols shall be developed in accordance with applicable provisions of Public Resource Code Section 21083.2 and State CEQA Guidelines 15064.5 and 15126.4 in consultation with OIAA or with a recognized scientific or educational repository, including the SCCIC. Preservation in place shall be the preferred means to avoid impacts to archaeological resources qualifying as historical resources, consistent with CEQA Guidelines Section 15126.4(b)(3)(C).

Reference

For a complete discussion of Project cumulative impacts related to cultural resources, see Section 5.4.4 of the Draft EIR at pages 5.4-36 - 5.4-39.

4.5 Geology and Soils

The Draft EIR discusses impacts related to geology and soils in Section 5.6. The following discussion addresses potential impacts with respect to geology and soils.

Would the project directly or indirectly cause potential substantial adverse effects, including the risk of loss, injury, or death involving: strong seismic ground shaking.

Impact Summary:

The Project site sits in the Upper Santa Ana River Valley, a highly seismically active area within Southern California. Active or potentially active faults are not known to exist on or trend toward the Project site. There are several active faults surrounding the Project site to the north, east, south, and west, within the Upper Santa Ana River Valley. For these reasons, there is a potential for ground shaking due to an earthquake. Additionally, potential for settlement, foundation, and pavement bearing conditions could occur with the construction of the proposed Project. Therefore, impacts related to strong seismic ground shaking could be potentially significant.

Less than Significant Effects with Mitigation: GEO-1(ii.)

The Project would have a less than significant impact with mitigation from strong seismic ground shaking.

Findings:

Pursuant to Public Resources Code Section 21081(a)(1), the OIAA finds that mitigation measures have been incorporated into the Project which mitigate or avoid the significant effects on the environment from strong seismic ground shaking. Therefore, impacts would be less than significant with mitigation incorporated.

Facts in Support of Findings:

The Project site sits in the Upper Santa Ana River Valley, a highly seismically active area within Southern California. Active or potentially active faults are not known to exist on or trend toward the Project site. There are several active faults surrounding the Project site to the north, east, south, and west. For these reasons, there is a potential for ground shaking due to an earthquake. Recommendations identified in the Geotechnical Study (see **Appendix 5.6-1**, of the Draft EIR) will be incorporated and implemented into the proposed Project through **Mitigation Measure GEO-5**. These recommendations will be incorporated into proposed Project plans and specifications and implemented during construction of the proposed Project. The proposed Project would adhere to the appropriate engineering design measures as required by the latest Greenbook and California Building Code (CBC).

Mitigation Measures:

GEO-5

GEO-5 Geotechnical Investigation Recommendations. The proposed Project shall implement and incorporate the recommendations in the Geotechnical Investigation, Section 5.0 Conclusion and Recommendations (see pages 7 through 24 of Appendix 5.6-1 of this EIR). Prior to contract bidding, site grading

and foundation plans shall be reviewed and approved by Cotton, Shires and Associates, Inc. or a certified Geologist, for consistency with the Geotechnical Investigation recommendations.

Less than Significant Effects with Mitigation: GEO-1(iii.)

The Project would have a less than significant impact with mitigation from seismic-related ground failure, including liquefaction and lateral spreading.

Findings:

Mitigation measures have been incorporated into the Project which mitigate or avoid the significant effects on the environment from seismic-related ground failure, including liquefaction and lateral spreading. Therefore, impacts would be less than significant with mitigation incorporated.

Facts in Support of Findings:

The Project site is located in the Southern California region which is seismically active and commonly experiences strong ground shaking. The Project site surface generally consists of loose to dense silty sand and medium stiff sandy silt to a depth of 4.5 feet. The Project site is not located on a steep slope. Subsurface, the artificial fill (Qaf) encountered to a depth of 4.5 feet generally consists of loose to dense silty sand and medium stiff sandy silt. Seismic settlement for dry sandy soils within the upper 40 feet of alluvium is estimated to be about two (2) to four (4) inches. Recommendations identified in the proposed Project's Geotechnical Study (see Appendix 5.6-1, of the Draft EIR) will be incorporated and implemented into the proposed Project through Mitigation Measure GEO-5. These recommendations will be incorporated into proposed Project plans and specifications and implemented during construction of the proposed Project. These recommendations include, but are not limited to, design standards and requirements pertaining to site preparation, excavation, subgrade stabilization, retaining walls, fill materials and compaction, foundations, site drainage, backdrains, utility trenches, pipe bedding, trench backfilling, corrosivity, pavements, and the infiltration basin. The proposed Project would adhere to the appropriate engineering design measures, as required by the latest Standard Specifications for Public Works Construction (Greenbook) and CBC.

Project Design Features:

None

Mitigation Measures:

GEO-5

GEO-5 Geotechnical Investigation Recommendations. The proposed Project shall implement and incorporate the recommendations in the Geotechnical Investigation, Section 5.0 Conclusion and Recommendations (see pages 7 through 24 of Appendix 5.6-1 of this EIR). Prior to contract bidding, site grading and foundation plans shall be reviewed and approved by Cotton, Shires and Associates, Inc. or a certified Geologist, for consistency with the Geotechnical Investigation recommendations.

Would the project result in substantial soil erosion or loss of topsoil?

Impact Summary:

Construction activities could result in soil erosion or loss of topsoil. Further, there is potential for intermittent areas of exposed graded soil on the Project site to be subject to wind-related erosion. Additionally, potential for settlement, foundation, and pavement bearing conditions could occur with the construction of the proposed Project. During operations, the proposed Project could result in a limited degree of soil erosion from vegetated areas. Therefore, impacts related to soil erosion, or the loss of topsoil, could be potentially significant.

Less than Significant Effects with Mitigation: GEO-2

The Project would have a less than significant impact with mitigation on substantial soil erosion or the loss of topsoil.

Findings:

Pursuant to Public Resources Code Section 21081(a)(1), the OIAA finds that mitigation measures have been incorporated into the Project which mitigate or avoid the significant effects on the environment for substantial soil erosion or the loss of topsoil. Therefore, impacts would be less than significant with mitigation incorporated.

Facts in Support of Findings:

During construction, prior to commencing grading operations, soil materials containing debris, organics, pavement, or other unsuitable materials would be stripped. Demolition would include removal of old foundations, pavements, slabs, abandoned utilities, and soils disturbed during the demolition process. The proposed Project would obtain coverage under the National Pollutant Discharge Elimination System (NPDES) Construction General Permit (CGP). A Storm Water Pollution Prevention Plan (SWPPP) would be developed and implemented prior to the construction, and a Standard Urban Storm Water Mitigation Plan (SUSMP) to be implemented to reduce the level of pollutants in stormwater and urban runoff.

During operations, nonerosive drainage features such as infiltration basins and associated infrastructure, and the maintenance of these structures would be conducted over the long-term operations of the proposed Project. Per CEQA and the San Bernardino County Technical Guidance Document for Water Quality Management Plans (WQMP), a level of low impact design (LID) must be incorporated into all new development projects by implementing Best Management Practices (BMPs). Surface runoff would be directed away from foundations or on-grade improvements. The proposed Project would comply with all applicable City grading permit regulations, plans, and inspections to reduce sedimentation and erosion. The proposed Project would adhere to the appropriate engineering design measures as required by the latest Greenbook and CBC. The potential for adverse impacts as a result of the proposed development from erosion is considered to be low provided with the incorporation of **Mitigation Measure GEO-5**.

Project Design Features:

None

Mitigation Measures:

GEO-5

GEO-5 Geotechnical Investigation Recommendations. The proposed Project shall implement and incorporate the recommendations in the Geotechnical Investigation, Section 5.0 Conclusion and Recommendations (see pages 7 through 24 of Appendix 5.6-1 of this EIR). Prior to contract bidding, site grading and foundation plans shall be reviewed and approved by Cotton, Shires and Associates, Inc. or a certified Geologist, for consistency with the Geotechnical Investigation recommendations.

Would the project be located on a geographic unit or soil that is unstable, or that would become unstable as a result of the project, and potentially result in on- or off-site landslide, lateral spreading, subsidence, liquefaction, or collapse.

Impact Summary:

Existing soils within the Project site are artificial fill and alluvial subsurface materials that are primarily coarse-grained with varying amounts of silt and low levels of clay. Therefore, the Project could potentially result in on- or off-site landslide, lateral spreading, subsidence, liquefaction, or collapse.

Less than Significant Effects with Mitigation: GEO-3

The Project would have a less than significant impact with mitigation on a geologic unit or soil that is unstable, or that would become unstable as a result of the Project, and potentially result in on- or off-site landslide, lateral spreading, subsidence, liquefaction or collapse.

Findings:

Pursuant to Public Resources Code Section 21081(a)(1), the OIAA finds that mitigation measures have been incorporated into the Project which mitigate or avoid the significant effects on the environment for a geologic unit or soil that is unstable, or that would become unstable as a result of the Project, and potentially result in on- or off-site landslide, lateral spreading, subsidence, liquefaction or collapse. Therefore, impacts would be less than significant with mitigation incorporated.

Facts in Support of Findings:

Prior to commencing grading operations, unsuitable soil materials would be stripped. Demolition activities would include removal of soils disturbed during the demolition process. The over-excavation and re-compaction of artificial fill and upper alluvial materials in the airfield apron area would reduce the potential for settlement and provide uniform bearing conditions. Additionally, the artificial fill and upper alluvial materials within the truck area south of the cargo facility, as well as site retaining walls with shallow foundations, would be over-excavated. The proposed Project would comply with all applicable City grading permit regulations, plans, and inspections to reduce sedimentation and erosion.

The California Department of Conservation Landslide Inventory indicates that the Project site is not located in an area that is susceptible to landslides. The Geotechnical Study (see **Appendix 5.6**-

1, of the Draft EIR) indicates the estimation of lateral movements resulting from seismic events is uncertain. There is a potential for ground lurching due to an earthquake. Deep groundwater, and relatively level site grade, the potential for large lateral movements caused by post-seismic residual shear strength reduction is considered to be very low. The proposed Project would adhere to the appropriate engineering design measures, as required by the latest Greenbook and CBC. The risk of subsidence due to water extraction is also low. The Geotechnical Study indicated the Project site is not located within a Liquefaction Hazard Zone as mapped by the State of California. The Ontario Plan Safety Element identifies that the Project site is not located in an area that would be susceptible to liquefaction. The potential for liquefaction to occur on the Project site is very low. Potential for settlement and foundation and pavement bearing conditions could occur with the construction of the proposed Project. Through compliance with the City's construction requirements, implementation of BMPs, compliance with applicable City grading permit regulations, and requirements of the statewide general construction stormwater permit, construction activities would not result in a collapse. Recommendations identified in the Geotechnical Study will be incorporated and implemented into the proposed Project through **Mitigation Measure GEO-5**.

Project Design Features:

None

Mitigation Measures:

GEO-5

GEO-5 Geotechnical Investigation Recommendations. The proposed Project shall implement and incorporate the recommendations in the Geotechnical Investigation, Section 5.0 Conclusion and Recommendations (see pages 7 through 24 of Appendix 5.6-1 of this EIR). Prior to contract bidding, site grading and foundation plans shall be reviewed and approved by Cotton, Shires and Associates, Inc. or a certified Geologist, for consistency with the Geotechnical Investigation recommendations.

Would the Project be located on expansive soils, as defined in Table 18-1-B of the Uniform Building Code (1994), creating substantial direct or indirect risks to life or property.

Impact Summary:

The existing soils within the Project site are artificial fill and alluvial subsurface materials that are primarily coarse-grained with varying amounts of silt and low levels of clay. The potential for soil expansiveness is considered very low due to existing soil conditions. However, water infiltration can cause or exacerbate expansive soil movement. Potential for settlement, foundation, and pavement bearing conditions could occur with construction of the proposed Project.

Less than Significant Effects with Mitigation: GEO-4

The Project would have a less than significant impact with mitigation on expansive soil.

Findings:

Pursuant to Public Resources Code Section 21081(a)(1), the OIAA finds that mitigation measures have been incorporated into the Project which mitigate or avoid the significant effects on the environment for expansive soil. Therefore, impacts would be less than significant with mitigation incorporated.

Facts in Support of Findings:

As discussed in the Geotechnical Study (see **Appendix 5.6-1**, of the Draft EIR) the existing soils within the Project site are artificial fill and alluvial subsurface materials that are primarily coarsegrained with varying amounts of silt and low levels of clay. Consolidation testing performed on near surface sandy soils similar to those encountered within the percolation test holes generally showed less than 0.5 percent collapse upon inundation with water, and at a higher overburden stress than should be experienced by the basin soils. Existing concrete and asphalt demolished at the site may be pulverized and re-used as general compacted fill. The recycled material used as general compacted fill will meet all grading and compaction requirements. Potential for settlement and foundation and pavement bearing conditions could occur with the construction of the proposed Project. The proposed Project would adhere to the appropriate engineering design measures as required by the latest Greenbook and CBC. Recommendations identified in the Geotechnical Study will be incorporated and implemented into the proposed Project through **Mitigation Measure GEO-5**.

Project Design Features:

None

Mitigation Measures:

GEO-5

GEO-5 Geotechnical Investigation Recommendations. The proposed Project shall implement and incorporate the recommendations in the Geotechnical Investigation, Section 5.0 Conclusion and Recommendations (see pages 7 through 24 of Appendix 5.6-1 of this EIR). Prior to contract bidding, site grading and foundation plans shall be reviewed and approved by Cotton, Shires and Associates, Inc. or a certified Geologist, for consistency with the Geotechnical Investigation recommendations.

Would the project directly or indirectly destroy a unique paleontological resource or site or unique geologic feature?

Impact Summary:

Deeper excavations, i.e., beyond nine (9) feet bgs, at the Project site may extend down into older Pleistocene sediments, which are considered to have a high paleontological sensitivity. Accordingly, ground disturbing activities beyond 9 feet bgs could potentially result in significant impacts related to paleontological resource.

Less than Significant Effects with Mitigation: GEO-6

The Project would have a less than significant impact with mitigation on paleontological resources.

Findings:

Pursuant to Public Resources Code Section 21081(a)(1), the OIAA finds that mitigation measures have been incorporated into the Project which mitigate or avoid the significant effects on the environment for paleontological resources. Therefore, impacts would be less than significant with mitigation incorporated.

Facts in Support of Findings:

As indicated in the Paleontological Resource Assessment (see Appendix 5.6-1, of the Draft EIR), the Project site contains artificial fill (Qaf) of the late Holocene epoch, which was deposited on Young alluvial-fan deposits (Qyf1 and Qyf3) of the Pleistocene epoch. Due to the artificial nature and origin off-site of this fill, the Qaf has no paleontological sensitivity. Holocene units typically are considered to have a low paleontological sensitivity. As Holocene units transition with greater depth, they encounter Pleistocene deposits, which have higher sensitivity for findings and the potential to produce the remains of diverse land animals. The proposed Project would require ground disturbance of 20 feet below ground surface (bgs), related to utilities trenching, although most of the ground disturbance would be less than 7 feet bgs. Deeper excavations, beyond nine (9) feet bgs, at the Project site may extend down into older Pleistocene sediments, which are considered to have a high paleontological sensitivity. Accordingly, ground disturbing activities beyond 9 feet bgs could potentially result in significant impacts related to paleontological resources. To reduce potential impacts, monitoring by a qualified paleontological monitor to identify and effectively salvage any recovered resources would be conducted during ground disturbing activities (Mitigation Measure GEO-1 through GEO-4). With the implementation of Mitigation Measure GEO-1 through GEO-4, Project impacts to paleontological resources would be less than significant.

Project Design Features:

None

Mitigation Measures:

GEO-1, GEO-2, GEO-3, and GEO-4

GEO-1 Paleontological Resources Mitigation and Monitoring Plan (PRMMP). A professional paleontologist shall be retained to monitor earth-disturbing construction activities. Prior to the commencement of ground-disturbing activities, the qualified paleontologist, meeting the Society of Vertebrate Paleontology (SVP) Standards, must prepare a Paleontological Resources Mitigation and Monitoring Plan (PRMMP) for the proposed Project. The PRMMP shall describe the monitoring required during excavations that extend into Pleistocene sediment, at approximately 9 feet bgs, and the location of areas deemed to have a high paleontological resource potential. The results of the geotechnical investigation conducted for the proposed Project shall be consulted to determine the approximate depth of Pleistocene sediment in the Project site.

Paleontological monitoring shall entail the visual inspection of excavated and graded areas and trench sidewalls. If the qualified Paleontologist determines full-time monitoring is no longer warranted, based on the geologic conditions at depth, he or she may recommend that monitoring be reduced or cease entirely.

- **GEO-2** Workers Environmental Awareness Program (WEAP). Prior to the start of the proposed Project ground-disturbing activities, all field personnel shall receive a worker's environmental awareness training on paleontological resources. The training must provide a description of the laws and ordinances protecting fossil resources, the types of fossil resources that may be encountered in the proposed Project area, the role of the paleontological monitor, outline steps to follow in the event that a fossil discovery is made and provide contact information for the qualified Paleontologist. The training must be developed by the qualified Paleontologist and can be delivered concurrent with other training.
- **GEO-3 Fossil Discoveries.** In the event that a paleontological resource is discovered, the Paleontological monitor shall have the authority to temporarily divert the construction equipment around the find until it is assessed for scientific significance and, if appropriate, collected. If the resource is determined to be of scientific significance, the Paleontologist shall complete the following:
 - 1. **Salvage of Fossils.** If fossils are discovered, all work in the immediate vicinity shall be halted to allow the paleontological monitor, and/or Project-qualified Paleontologist to evaluate the discovery and determine if the fossil may be considered significant. If the fossils are determined to be potentially significant, the Project-qualified Paleontologist shall recover them following standard field procedures for collecting paleontological as outlined in the PRMMP prepared for the Project. Typically, fossils can be safely salvaged quickly by a single paleontologist and not disrupt construction activity. In some cases, larger fossils, such as complete skeletons or large mammal fossils, require more extensive excavation and longer salvage periods. In this case the Paleontologist shall have the authority to temporarily direct, divert or halt construction activity to ensure that the fossil(s) can be removed in a safe and timely manner.
 - 2. Fossil Preparation and Curation. The PRMMP must identify a museum that has agreed to accept fossils that may be discovered during Project-related excavations. Upon completion of fieldwork, all significant fossils collected must be prepared in a properly equipped laboratory to a point ready for curation. Preparation may include the removal of excess matrix from fossil materials and stabilizing or repairing specimens. During preparation and inventory, the fossils specimens must be identified to the lowest taxonomic level practical prior to curation at an accredited museum. The fossil specimens must be delivered to the accredited museum or repository no later than 90 days after all fieldwork is completed. The cost of curation shall be assessed by the repository and shall be the responsibility of the client.

GEO-4 Final Paleontological Mitigation Report. Upon completion of ground disturbing activity, and curation of fossils if necessary, the qualified Paleontologist shall prepare a final mitigation and monitoring report outlining the results of the mitigation and monitoring program. The report shall include discussion of the location, duration and methods of the monitoring, stratigraphic sections, any recovered fossils, and the scientific significance of those fossils, and where fossils were curated.

Reference

For a complete discussion of Project impacts related to geology and soils, see Section 5.6.3 of the Draft EIR at pages 5.6-12 - 5.3-28.

4.6 Geology and Soils (Cumulative)

The Draft EIR discusses the impacts related to geology and soils in Section 5.6. The following discussion addresses potential cumulative impacts with respect to geology and soils.

Impact Summary:

Geotechnical impacts tend to be site-specific rather than cumulative in nature, and any development occurring within the Airport and the surrounding area would be subject to, at a minimum, site development and construction standards relative to seismic and other geologic conditions that are prevalent within the region. As with the Project site, related projects would be subject to the same local, regional, State, and federal regulations pertaining to geology and soils, as well as to the Greenbook to reduce potentially significant impacts related to geology and paleontological resources.

Less than Significant Effects with Mitigation:

The Project's cumulative impacts to geology and soils would be less than significant with mitigation.

Findings:

Pursuant to Public Resources Code Section 21081(a)(1), the OIAA finds that mitigation measures have been incorporated into the Project which mitigate or avoid the significant effects on the environment to geology and soils. Therefore, impacts would be less than significant with mitigation incorporated.

Facts in Support of Findings:

Geotechnical impacts tend to be site-specific rather than cumulative in nature, and any development occurring within the Airport and the surrounding area would be subject to, at a minimum, site development and construction standards relative to seismic and other geologic conditions that are prevalent within the region. As with the Project site, related projects would be subject to the same local, regional, State, and federal regulations pertaining to geology and soils, as well as to the Standard Specifications for Public Works Construction (Greenbook). Related project impacts would be addressed through imposition of recommendations specific to each project. With conformance to such regulations, cumulative impacts related to geology and soils would be less than significant. Therefore, no significant cumulative impacts will result from the

proposed Project. Related projects, other growth, and the proposed Project's contribution to cumulative impacts will not be cumulatively considerable. **Mitigation Measures GEO-1 through GEO-5** would reduce potentially significant impacts related to geology and paleontological resources to a less than significant level for the Project. Therefore, no significant and unavoidable adverse cumulative impacts related to geology and soils would occur from proposed Project implementation.

Mitigation Measures:

GEO-1, GEO-2, GEO-3, GEO-4, and GEO-5

- GEO-1 Paleontological Resources Mitigation and Monitoring Plan (PRMMP). A professional paleontologist shall be retained to monitor earth-disturbing construction activities. Prior to the commencement of ground-disturbing activities, the qualified paleontologist, meeting the Society of Vertebrate Paleontology (SVP) Standards, must prepare a Paleontological Resources Mitigation and Monitoring Plan (PRMMP) for the proposed Project. The PRMMP shall describe the monitoring required during excavations that extend into Pleistocene sediment, at approximately 9 feet bgs, and the location of areas deemed to have a high paleontological resource potential. The results of the geotechnical investigation conducted for the proposed Project shall be consulted to determine the approximate depth of Pleistocene sediment in the Project site. Paleontological monitoring shall entail the visual inspection of excavated and graded areas and trench sidewalls. If the qualified Paleontologist determines fulltime monitoring is no longer warranted, based on the geologic conditions at depth, he or she may recommend that monitoring be reduced or cease entirely.
- **GEO-2** Workers Environmental Awareness Program (WEAP). Prior to the start of the proposed Project ground-disturbing activities, all field personnel shall receive a worker's environmental awareness training on paleontological resources. The training must provide a description of the laws and ordinances protecting fossil resources, the types of fossil resources that may be encountered in the proposed Project area, the role of the paleontological monitor, outline steps to follow in the event that a fossil discovery is made and provide contact information for the qualified Paleontologist. The training must be developed by the qualified Paleontologist and can be delivered concurrent with other training.
- **GEO-3 Fossil Discoveries.** In the event that a paleontological resource is discovered, the Paleontological monitor shall have the authority to temporarily divert the construction equipment around the find until it is assessed for scientific significance and, if appropriate, collected. If the resource is determined to be of scientific significance, the Paleontologist shall complete the following:
 - 1. **Salvage of Fossils.** If fossils are discovered, all work in the immediate vicinity shall be halted to allow the paleontological monitor, and/or Project-qualified Paleontologist to evaluate the discovery and determine if the fossil may be considered significant. If the fossils are determined to be potentially significant, the Project-qualified Paleontologist shall recover them following

standard field procedures for collecting paleontological as outlined in the PRMMP prepared for the Project. Typically, fossils can be safely salvaged quickly by a single paleontologist and not disrupt construction activity. In some cases, larger fossils, such as complete skeletons or large mammal fossils, require more extensive excavation and longer salvage periods. In this case the Paleontologist shall have the authority to temporarily direct, divert or halt construction activity to ensure that the fossil(s) can be removed in a safe and timely manner.

- 2. Fossil Preparation and Curation. The PRMMP must identify a museum that has agreed to accept fossils that may be discovered during Project-related excavations. Upon completion of fieldwork, all significant fossils collected must be prepared in a properly equipped laboratory to a point ready for curation. Preparation may include the removal of excess matrix from fossil materials and stabilizing or repairing specimens. During preparation and inventory, the fossils specimens must be identified to the lowest taxonomic level practical prior to curation at an accredited museum. The fossil specimens must be delivered to the accredited museum or repository no later than 90 days after all fieldwork is completed. The cost of curation shall be assessed by the repository and shall be the responsibility of the client.
- **GEO-4** Final Paleontological Mitigation Report. Upon completion of ground disturbing activity, and curation of fossils if necessary, the qualified Paleontologist shall prepare a final mitigation and monitoring report outlining the results of the mitigation and monitoring program. The report shall include discussion of the location, duration and methods of the monitoring, stratigraphic sections, any recovered fossils, and the scientific significance of those fossils, and where fossils were curated.
- **GEO-5** Geotechnical Investigation Recommendations. The proposed Project shall implement and incorporate the recommendations in the Geotechnical Investigation, Section 5.0 Conclusion and Recommendations (see pages 7 through 24 of Appendix 5.6-1 of this EIR). Prior to contract bidding, site grading and foundation plans shall be reviewed and approved by Cotton, Shires and Associates, Inc. or a certified Geologist, for consistency with the Geotechnical Investigation recommendations.

Reference

For a complete discussion of Project cumulative impacts related to cultural resources, see Section 5.6.4 of the Draft EIR at page 5.6-28.

4.7 Hazards and Hazardous Materials

The Draft EIR discusses impacts related to hazards and hazardous materials in Section 5.8. The following discussion addresses potential impacts with respect to hazards and hazardous materials.

Create a significant hazard to the public or the environment through reasonably foreseeable upset and accident conditions involving the release of hazardous materials into the environment.

Impact Summary:

The Project's Phase I ESA based on former uses of the Project site, the constituents of concern include: metals, petroleum hydrocarbons, volatile organic compounds (VOCs), semi-volatile compounds (SVOCs), polycyclic aromatic hydrocarbons (PAHs), polychlorinated biphenyls (PCBs), organochlorine pesticides (OCPs), herbicides, per- and polyfluoroalkyl substances (PFAS), asbestos containing materials, and lead-based paint. the Phase II ESA was conducted to assess soil and soil vapor conditions at the Project site, as well as to establish a baseline understanding of the existing subsurface conditions and potential risk to human health by drilling 143 soil borings on Parcels 61, 62, 63, and 68, and collecting soil and soil vapor samples to evaluate subsurface conditions.

Metals, total petroleum hydrocarbons (TPH), VOCs, PAHs, and OCPs, were detected above laboratory reporting limits but below their respective commercial/industrial DTSC SLc and SFBRWQCB ESLs screening levels. Additionally, as the potential human health risks of PFAS are currently under study by regulatory authorities, avoidance of contact with soils containing PFAS during construction is recommended. Based on these results, impacts related to the release of PFAS into the environment are potentially significant.

Less than Significant Effects with Mitigation: HAZ-2

The Project would have a less than significant impact with mitigation on soil management and vapor intrusion.

Findings:

Pursuant to Public Resources Code Section 21081(a)(1), the OIAA finds that mitigation measures have been incorporated into the Project which mitigate or avoid the significant effects on the environment for soil management and vapor intrusion. Therefore, impacts would be less than significant with mitigation incorporated.

Facts in Support of Findings:

No evidence of RECs in connection with the Project site were observed in the Phase I ESA, with the exception of those identified in **Table 5.8-1: RECs Identified**, of the Draft EIR. The Phase II ESA noted the presence of per- and polyfluoroalkyl substances (PFAS) impacted materials where Aqueous Film-Forming Foam (AFFF) is currently stored and in soils within the northern and southern Guardian Jet Center hangars, and former fire house associated with the former ANG facility. Mitigation Measure HAZ-1 includes development, approval, and implementation of a Soil Management Plan (SMP) to reduce the potential for accidental exposure to hazardous materials that may be present in soil that may be disturbed by construction of the proposed Project to a less than significant impact. Based on the results of the additional investigations conducted for the Phase II ESA, **Mitigation Measure HAZ-2** includes installation of a vapor intrusion mitigation system (VIM system) under Phase II of the proposed Air Cargo Sort Building to prevent potential vapor intrusion from the subsurface. Installation of the VIM would reduce the potential for this

exposure to a less than significant impact. With implementation of **Mitigation Measures HAZ-1** and **HAZ-2**, impacts would be less than significant.

Project Design Features:

None

Mitigation Measures:

HAZ-1 and HAZ-2

- **HAZ-1** Soil Management Plan. A Soil Management Plan (SMP) containing soil criteria and soil management and construction risk management protocols to be implemented during proposed Project development shall be prepared prior to disturbance of soils on the site by construction activities and implemented during construction to address any soil containing or suspected to contain PFAs on the proposed Project site and any previously undetected contamination encountered during construction. Special attention shall be made to soils disturbed in the Guardian Jet Center, southern hangar and structure previously housing fire prevention equipment due to the known presence of PFAs in these areas. Additional soil sampling shall be conducted as necessary to delineate the extent of PFAs contamination to enable segregation and proper disposal of any contaminated soil during construction.
- **HAZ-2** Vapor Intrusion Mitigation System. A vapor intrusion mitigation system (VIM system) shall be installed under Phase II of the proposed Air Cargo Building to address the potential for vapor intrusion from the subsurface. Alternatively, a soil vapor extraction remediation system could be utilized to reduce trichloroethene (TCE) and chloroform vapor concentrations through removal of volatile organic compounds (VOCs) in Phase II development area.

For a project located within an airport land use plan or, where such a plan has not been adopted, within two miles of a public airport or public use airport, would the project result in a safety hazard or excessive noise for people residing or working in the project area.

Impact Summary:

The proposed Project is located as a public airport and construction noise could potentially result in a safety hazard or excessive noise for people residing or working in the area. Additionally, the noise impact from aircraft operations a potentially significant impact.

Less than Significant Effects with Mitigation: HAZ-5

The Project would have a less than significant impact with mitigation, being located within an airport land use plan, on safety hazards or excessive noise for people residing or working in the Project area.

Findings:

Pursuant to Public Resources Code Section 21081(a)(1), the OIAA finds that mitigation measures have been incorporated into the Project which mitigate or avoid the significant effects on the environment, being located within an airport land use plan, on safety hazards or excessive noise for people residing or working in the Project area. Therefore, impacts would be less than significant with mitigation incorporated.

Facts in Support of Findings:

During construction, compliance with applicable aviation-related regulations would establish exposure limits for workers, require protective equipment or other protective measures when warranted, and require employers to provide a written health and safety program, worker training, emergency response training, and medical surveillance. Construction noise levels would not exceed the 85 dBA (Leq-1 hour) threshold at nearby sensitive receptors, nor would construction result in excessive noise for people residing or working in the proposed Project area. Therefore, although the proposed Project would be located at a public airport, construction of the proposed Project area.

Operational roadway noise levels would not create a readily perceptible increase of 5 dBA or greater at locations where ambient noise levels are less than 60 dBA; a barely perceptible increase of 3 dBA or greater at locations where ambient noise levels from 60 to 65 dBA; or a community noise level impact increase of 1.5 dBA or greater at locations where ambient noise levels already exceed 65 dBA. Roadway noise levels during Phase 1 and Phase 2 of the proposed Project would not result in excessive noise for people residing or working in the Project area and impacts would be less than significant.

The proposed Project is located within the Ontario Airport Land Use Compatibility Plan (ALUCP). As indicated in the ALUCP, the Project site is also within the Airport's Influence Area (AIA), which contains the safety zones within the Airport. The Safety Zones are determined based upon the generic safety zones provided in the California Airport Land Use Planning Handbook and take into account historic aircraft accidents, existing aircraft flight patterns and aircraft characteristics, as well as the pattern of accidents. The Project site is partially within Safety Zone 5, which is located primarily on airport property, adjacent to the runway, approximately 500 to 1000 feet from the centerline. As a proposed aeronautical development, the proposed Project is consistent with the aeronautical development identified as allowed on the Airport Layout Plan (ALUCP, Chapter 1, Exhibit 1-6) and generally under the ALUCP. The proposed Project is consistent with the applicable safety provisions for Safety Zone 5 in the ALUCP and the proposed Project would not result in a safety hazard for people residing or working in the Project area.

The ALUCP designates the airport influence area, safety zones, noise impact zones, airspace protection zones, and overflight notification zones. Height and noise restrictions for future land uses are established for the airport approach safety zones. All construction and operation of the proposed Project would comply with applicable aviation-related regulations and safeguards. For aircraft noise, implementation of **Mitigation Measure NOI-1** would require a residential sound insulation program (RSIP) for housing units within the future 65-69 dBA which have not been provided with an opportunity to install sound attenuation. With implementation of **Mitigation**

Measure NOI-1 impacts related to aircraft noise would be reduced to less than significant levels. Therefore, implementation of the proposed Project would result in less than significant impacts with mitigation incorporated related to a safety hazard or excessive noise for people residing or working in the proposed Project area.

Project Design Features:

None

Mitigation Measures:

NOI-1

NOI-1 Residential Sound Insulation Program (RSIP). Non-compatible residential land uses within the 65+ decibel (dB) contour with habitable areas inside the home with average noise levels of 45 dB or greater with all windows closed would be eligible for the RSIP.

The goal of the Program is to reduce the interior noise level within affected homes by at least five (5) decibels (dB). The results may vary depending upon the existing structural characteristics of the home. In order to achieve this goal, modifications may include the retrofit of exterior doors and windows, installation of a ventilation system, and other miscellaneous treatments. The RISP would include the following:

A noise audit will be conducted for each home in the RISP to measure the noise reduction properties of a residence in its existing condition to confirm that average interior aircraft sound levels are greater than a Community Noise Equivalent Level (CNEL) of 45 decibels (dB), and to provide an indication of the potential effectiveness of noise reducing treatments.

The goal of the RISP is to reduce the average interior CNEL of habitable rooms by a minimum of 5 dB (i.e., a clearly detectable reduction), and reduce the average interior CNEL of habitable rooms to below 45 dB.

Sound levels will be measured using aircraft as the noise source or simulation methods (loudspeaker(s)).

Property owners will be required to sign an avigation easement, guaranteeing the right of flight over a residence, as a requirement to participate in the RISP.

Upon completion, current owners will be required to disclose the residence was included in the RISP and is subject to an avigation easement.

If housing units do not meet the local building codes required to qualify for sound insulation, a homeowner shall be given the option to sell the property. The residence may be resold to a new owner. The housing unit may or may not be sound insulated and/or upgraded prior to resale but will be subject to an avigation easement.

Reference

For a complete discussion of Project impacts related to hazards and hazardous materials, see Section 5.8.3 of the Draft EIR at pages 5.8-26 - 5.8-44. For a complete discussion of Project impacts related to noise, see Section 5.10.3 of the Draft EIR at pages 5.10-25 - 5.10-55.

4.8 Hazards and Hazardous Materials (Cumulative)

The Draft EIR discusses the cumulative impacts related to hazards and hazardous materials in Section 5.8. The following discussion addresses potential cumulative impacts with respect to hazards and hazardous materials.

Impact Summary:

The Project vicinity is either existing airport uses or is largely urbanized with residential, commercial, and industrial uses. As the area continues to develop, the addition of more development could create a significant hazard to the public or the environment through the routine transport, use, or disposal of hazardous materials.

Less than Significant Effects with Mitigation:

The Project's cumulative impacts to hazards and hazardous materials would be less than significant with mitigation.

Findings:

Pursuant to Public Resources Code Section 21081(a)(1), the OIAA finds that mitigation measures have been incorporated into the Project which mitigate or avoid the significant effects on the environment to hazards and hazardous materials. Therefore, impacts would be less than significant with mitigation incorporated.

Facts in Support of Findings:

Although some of the related projects listed in the Draft EIR also have potential impacts associated with hazardous materials, the environmental concerns associated with hazardous materials are site specific. Each project is required to address any issues related to hazardous material or waste. Like the proposed Project, the related projects would be required to comply with federal, State, and local regulations, and require proven mitigation to remediate or protect against site contamination by hazardous materials. Therefore, the Project's contribution to cumulative impacts related to hazards and hazardous materials would not be cumulatively considerable, and cumulative impacts would be less than significant.

Mitigation Measure HAZ-1 would mitigate potentially significant impacts related to the accidental exposure of PFAS in soil on the Project site to a less than significant level. **Mitigation Measure HAZ-2** would mitigate potentially significant impacts related to accidental vapor intrusion exposure to a less than significant level. Implementation of **Mitigation Measure NOI-1** will mitigate impacts related to aircraft noise to less than significant for the Project. Implementation of the proposed Project would result in less than significant impacts with mitigation incorporation related to a safety hazard or excessive noise for people residing or working in the Project area. Therefore, no significant and unavoidable adverse impacts related to hazards and hazardous materials would occur from Project implementation.

Project Design Features:

None

Mitigation Measures:

HAZ-1, HAZ-2, and NOI-1

- **HAZ-1** Soil Management Plan. A Soil Management Plan (SMP) containing soil criteria and soil management and construction risk management protocols to be implemented during proposed Project development shall be prepared prior to disturbance of soils on the site by construction activities and implemented during construction to address any soil containing or suspected to contain PFAs on the proposed Project site and any previously undetected contamination encountered during construction. Special attention shall be made to soils disturbed in the Guardian Jet Center, southern hangar and structure previously housing fire prevention equipment due to the known presence of PFAs in these areas. Additional soil sampling shall be conducted as necessary to delineate the extent of PFAs contamination to enable segregation and proper disposal of any contaminated soil during construction.
- **HAZ-2** Vapor Intrusion Mitigation System. A vapor intrusion mitigation system (VIM system) shall be installed under Phase II of the proposed Air Cargo Building to address the potential for vapor intrusion from the subsurface. Alternatively, a soil vapor extraction remediation system could be utilized to reduce trichloroethene (TCE) and chloroform vapor concentrations through removal of volatile organic compounds (VOCs) in Phase II development area.
- **NOI-1 Residential Sound Insulation Program (RSIP).** Non-compatible residential land uses within the 65+ decibel (dB) contour with habitable areas inside the home with average noise levels of 45 dB or greater with all windows closed would be eligible for the RSIP.

The goal of the Program is to reduce the interior noise level within affected homes by at least five (5) decibels (dB). The results may vary depending upon the existing structural characteristics of the home. In order to achieve this goal, modifications may include the retrofit of exterior doors and windows, installation of a ventilation system, and other miscellaneous treatments. The RISP would include the following:

A noise audit will be conducted for each home in the RISP to measure the noise reduction properties of a residence in its existing condition to confirm that average interior aircraft sound levels are greater than a Community Noise Equivalent Level (CNEL) of 45 decibels (dB), and to provide an indication of the potential effectiveness of noise reducing treatments.

The goal of the RISP is to reduce the average interior CNEL of habitable rooms by a minimum of 5 dB (i.e., a clearly detectable reduction), and reduce the average interior CNEL of habitable rooms to below 45 dB.

Sound levels will be measured using aircraft as the noise source or simulation methods (loudspeaker(s)).

Property owners will be required to sign an avigation easement, guaranteeing the right of flight over a residence, as a requirement to participate in the RISP.

Upon completion, current owners will be required to disclose the residence was included in the RISP and is subject to an avigation easement.

If housing units do not meet the local building codes required to qualify for sound insulation, a homeowner shall be given the option to sell the property. The residence may be resold to a new owner. The housing unit may or may not be sound insulated and/or upgraded prior to resale but will be subject to an avigation easement.

Reference

For a complete discussion of Project cumulative impacts related to hazards and hazardous materials, see Section 5.8.4 of the Draft EIR at pages 5.8-44 - 5.8-45.

4.9 Noise

The Draft EIR discusses impacts related to noise in Section 5.10. The following discussion addresses potential impacts with respect to noise.

Located within the vicinity of a private airstrip or an airport land use plan or, where such a plan has not been adopted, within two miles of a public airport or public use airport, if the project would expose people residing or working in the project area to excessive noise levels.

Impact Summary:

Unmitigated residences would be exposed to aircraft noise that would be considered significant, the Baseline Condition and Proposed Project condition would result in a potentially significant impact.

Less than Significant Effects with Mitigation: N-3

The Project would have a less than significant impact with mitigation, located within the vicinity of an airport land use plan, on exposing people residing or working in the Project area to excessive noise levels.

Findings:

Pursuant to Public Resources Code Section 21081(a)(1), the OIAA finds that mitigation measures have been incorporated into the Project which mitigate or avoid the significant effects on the environment, for projects located within the vicinity of an airport land use plan, from exposing people residing or working in the Project area to excessive noise levels. Therefore, impacts would be less than significant with mitigation incorporated.

Facts in Support of Findings:

In the year 2025 with the proposed Project, it is estimated that there would be 4 additional unmitigated housing units and 18 persons within the CNEL 65-69 dBA contour and no housing units or persons within the CNEL 70+ dBA contour when compared to the baseline condition. By the year 2029, it is estimated that there would be 12 additional unmitigated housing units and 43 additional persons within the CNEL 65-69 dBA contour and no housing units or persons within the CNEL 65-69 dBA contour and no housing units or persons within the CNEL 65-69 dBA contour and no housing units or persons within the CNEL 65-69 dBA contour and no housing units or persons within the CNEL 70+ dBA contour. When avigation easements are considered for the proposed Project compared to Baseline Conditions, the number of additional unmitigated housing units in 2025 is 3 units with the affected population being 14 additional persons and in 2029 the number of housing units affected is 12 units with the affected population being 51 additional persons. To define the significance of the impact of a proposed project, CEQA regulations require future conditions with a proposed project be compared to existing (i.e., Baseline) conditions. Because such a comparison also includes the potential impact that would occur in the future without a project (i.e., the No Action Alternative), a comparison of the proposed Project and No Action Alternative was completed. Notably, the growth in passenger activity at the Airport, which would occur with or without the proposed Project, would result in increases in aircraft operations and aircraft noise.

When comparing the proposed Project to the No Action Alternative, in the year 2025 with the proposed Project it is estimated that there would be five (5) additional unmitigated housing units and 23 persons within the CNEL 65-69 dBA contour and no housing units or persons within the CNEL 70+ dBA contour. By the year 2029 with the proposed Project, it is estimated that there would be 15 additional unmitigated housing units and 63 additional persons within the CNEL 65-69 dBA contour and no housing units or persons within the CNEL 70+ dBA contour. Because unmitigated residences would be exposed to aircraft noise that would be considered significant, the Baseline Condition and Proposed Project condition would result in a potentially significant impact. Implementation of Mitigation Measure NOI-1 would define a residential noise program for housing units affected by aviation noise generated by the Project. Implementation of Mitigation Measure NOI-1 would include a residential noise program for housing units located within the 65+ dB contour to reduce the interior noise level within affected homes by at least five (5) dB and reduce average interior CNEL of habitable rooms to below 45 dB. Mitigation Measure NOI-1 would reduce impacts related to aircraft noise to less than significant levels.

Project Design Features:

None

Mitigation Measures:

NOI-1

NOI-1 Residential Sound Insulation Program (RSIP). Non-compatible residential land uses within the 65+ decibel (dB) contour with habitable areas inside the home with average noise levels of 45 dB or greater with all windows closed would be eligible for the RSIP.

The goal of the Program is to reduce the interior noise level within affected homes by at least five (5) decibels (dB). The results may vary depending upon the existing structural characteristics of the home. In order to achieve this goal, modifications may include the retrofit of exterior doors and windows, installation of a ventilation system, and other miscellaneous treatments. The RISP would include the following:

A noise audit will be conducted for each home in the RISP to measure the noise reduction properties of a residence in its existing condition to confirm that average interior aircraft sound levels are greater than a Community Noise Equivalent Level (CNEL) of 45 decibels (dB), and to provide an indication of the potential effectiveness of noise reducing treatments.

The goal of the RISP is to reduce the average interior CNEL of habitable rooms by a minimum of 5 dB (i.e., a clearly detectable reduction), and reduce the average interior CNEL of habitable rooms to below 45 dB.

Sound levels will be measured using aircraft as the noise source or simulation methods (loudspeaker(s)).

Property owners will be required to sign an avigation easement, guaranteeing the right of flight over a residence, as a requirement to participate in the RISP.

Upon completion, current owners will be required to disclose the residence was included in the RISP and is subject to an avigation easement.

If housing units do not meet the local building codes required to qualify for sound insulation, a homeowner shall be given the option to sell the property. The residence may be resold to a new owner. The housing unit may or may not be sound insulated and/or upgraded prior to resale but will be subject to an avigation easement.

Reference

For a complete discussion of Project impacts related to noise, see Section 5.10.3 of the Draft EIR at pages 5.10-25 - 5.10-55.

4.10 Noise (Cumulative)

The Draft EIR discusses the impacts related to noise in Section 5.10. The following discussion addresses potential cumulative impacts with respect to noise.

Impact Summary:

Related projects, and growth in the general area of the Project site (within 500 feet), would contribute to cumulative noise impacts. Cumulative construction- noise impacts have the potential to occur when multiple construction projects in the local area generate noise within the same time frame and contribute to the local ambient noise environment. With regard to stationary sources, cumulative significant noise impacts may result from cumulative development.

For operational roadway noise cumulative impacts, the maximum noise level increase along the studied roadway segments would be 0.70 dBA CNEL along Vineyard Avenue between Avion and

Mission. Roadway noise levels along this segment would remain classified with the "Clearly Acceptable" designation. For operational aircraft noise impacts the Runway 8R-26L runway rehabilitation/reconstruction project is scheduled to begin in 2023 and end in 2025, one of the same years for which the proposed Project was evaluated. To evaluate the impact due to the overlap of the proposed Project that is the subject of this report and the Runway 8R-26L runway rehabilitation/reconstruction project, a cumulative aircraft noise analysis was performed.

Less than Significant Effects with Mitigation:

The Project's cumulative impacts to noise would be less than significant with mitigation.

Findings:

Pursuant to Public Resources Code Section 21081(a)(1), the OIAA finds that mitigation measures have been incorporated into the Project which mitigate or avoid the significant effects on the environment to noise. Therefore, cumulative noise impacts would be less than significant with mitigation incorporated.

Facts in Support of Findings:

It is expected that, as with the proposed Project, the related projects would implement noise reduction techniques such as mufflers, shields, and sound barriers, which would minimize any noise-related nuisances during construction. In addition, distance attenuation and intervening structures would further reduce construction noise levels and not result in noticeable increases. Therefore, combined construction-noise impacts of related projects and the proposed Project's contribution would be less than significant.

Stationary sources of noise that could be introduced in the area by cumulative projects could include mechanical equipment, loading docks, and parking lots. Given that these projects would be required to adhere to the City's noise standards, all stationary sources would be required to have shielding or other noise-abatement measures so as not to cause a substantial increase in ambient noise levels. Moreover, due to distance, it is unlikely that noise from multiple cumulative projects would interact to create a significant combined noise impact. As such, it is not anticipated that a significant cumulative increase in permanent ambient noise levels would occur.

Roadway noise levels would not create a readily perceptible increase of 5 dBA or greater at locations where ambient noise levels are less than 60 dBA; a barely perceptible increase of 3 dBA or greater at locations where ambient noise levels from 60 to 65 dBA; and a community noise level impact increase of 1.5 dBA or greater at locations where ambient noise levels already exceed 65 dBA. As such, roadway noise levels during the Future Year (2040) Conditions would not be less than significant.

Implementation of the RSIP in **Mitigation Measure NOI-1** would reduce impacts from the Project to less than significant. the OIAA recently certified a Draft SEIR for the rehabilitation of the Airport's Runway 8R-26L and associated airfield improvements. These improvements are proposed so that the airfield meets current FAA standards, safety is improved, and the efficiency of the airfield is enhanced. To implement the improvements, temporary runway closures would be required and the only change in the use of the airfield would result from suspension of voluntary restrictions on the use of Contra Flow operations during nighttime hours (10:00 PM to 7:00 AM).

Because these voluntary restrictions would not be available when operating only one runway, the Runway 8R-26L Draft SEIR forecasts that there would be a temporary increase in noise exposure to the west of the Airport during nighttime hours.

The Runway 8R-26L runway rehabilitation/reconstruction project is scheduled to begin in 2023 and end in 2025, one of the same years for which the proposed Project was evaluated. The proposed Project would contribute to temporary cumulative noise impacts during construction of the Runway 8R-26L runway rehabilitation/reconstruction project in 2025. With the related projects, it is estimated that there would be 219 additional unmitigated housing units and 991 persons within the CNEL 65-69 dBA contour and no housing units or persons within the CNEL 70+ dBA contour when compared to the baseline condition with the proposed Project and construction of the Runway 8R-26L runway rehabilitation/reconstruction project. Based on these results, in the year proposed Project and construction of the Runway 2025 the 8R-26L runway rehabilitation/reconstruction project would result in a temporary significant cumulative impact. However, these impacts would be temporary only during construction of Runway 8R-26L runway rehabilitation/reconstruction project. No other related project would contribute to cumulative aircraft noise impacts.

The Runway 8R-26L runway rehabilitation/reconstruction project would result in less than significant aircraft noise impacts once operational. Implementation of Mitigation Measure NOI-1 would include a residential noise program for housing units located near the Airport, which, with implementation of **Mitigation Measure NOI-1** proposed Project impacts related to aircraft noise would be reduced to less than significant levels. Therefore, the proposed Project's contribution to temporary cumulative impacts would not be cumulatively considerable.

Project Design Features:

None

Mitigation Measures:

NOI-1

NOI-1 Residential Sound Insulation Program (RSIP). Non-compatible residential land uses within the 65+ decibel (dB) contour with habitable areas inside the home with average noise levels of 45 dB or greater with all windows closed would be eligible for the RSIP.

The goal of the Program is to reduce the interior noise level within affected homes by at least five (5) decibels (dB). The results may vary depending upon the existing structural characteristics of the home. In order to achieve this goal, modifications may include the retrofit of exterior doors and windows, installation of a ventilation system, and other miscellaneous treatments. The RISP would include the following:

A noise audit will be conducted for each home in the RISP to measure the noise reduction properties of a residence in its existing condition to confirm that average interior aircraft sound levels are greater than a Community Noise Equivalent Level (CNEL) of 45 decibels (dB), and to provide an indication of the potential effectiveness of noise reducing treatments.

The goal of the RISP is to reduce the average interior CNEL of habitable rooms by a minimum of 5 dB (i.e., a clearly detectable reduction), and reduce the average interior CNEL of habitable rooms to below 45 dB.

Sound levels will be measured using aircraft as the noise source or simulation methods (loudspeaker(s)).

Property owners will be required to sign an avigation easement, guaranteeing the right of flight over a residence, as a requirement to participate in the RISP.

Upon completion, current owners will be required to disclose the residence was included in the RISP and is subject to an avigation easement.

If housing units do not meet the local building codes required to qualify for sound insulation, a homeowner shall be given the option to sell the property. The residence may be resold to a new owner. The housing unit may or may not be sound insulated and/or upgraded prior to resale but will be subject to an avigation

Reference

For a complete discussion of Project cumulative impacts related to noise, see Section 5.10.4 of the Draft EIR at pages 5.10-55 - 5.10.60.

4.11 Tribal Cultural Resource

The Draft EIR discusses impacts related to tribal cultural resources in Section 5.13. The following discussion addresses potential impacts with respect to tribal cultural resources.

Would the project cause a substantial adverse change in the significance of a tribal cultural resource, defined in Public Resources Code Section 21074 as either a site, feature, place, cultural landscape that is geographically defined in terms of the size and scope of the landscape, sacred place, or object with cultural value to a California Native American tribe, and that is listed or eligible for listing in the California Register of Historic Resources, or in a local register of historical resources as defined in Public Resources Code section 5020.1(k).

Would the project cause a substantial adverse change in the significance of a tribal cultural resource, defined in Public Resources Code Section 21074 as either a site, feature, place, cultural landscape that is geographically defined in terms of the size and scope of the landscape, sacred place, or object with cultural value to a California Native American tribe, and that is a resource determined by the lead agency, in its discretion and supported by substantial evidence, to be significant pursuant to criteria set forth in subdivision (c) of Public Resources Code Section

5024.1, the lead agency shall consider the significance of the resource to a California Native American Tribe.

Impact Summary:

No tribal cultural resources were identified on the Project site. Nevertheless, archival research indicates the Project site in the 1940s contained plowed fields and was adjacent to Cucamonga Channel, which was minimally altered at that time, which flowed into native habitat areas farther east. Accordingly, it is possible that objects and features associated with the prehistoric occupation of local tribes in the Project area are buried in the native soils, underlying the artificial fill at the Project site. Because the Project's ground disturbing activities could extend to a depth of 20 feet below ground surface, there is the potential to encounter native soils and impact any resources that may be present. Impacts related to unidentified tribal cultural resources would be significant.

Less than Significant Effects with Mitigation: TRI-1

The Project would have a less than significant impact with mitigation on tribal cultural resources.

Findings:

Pursuant to Public Resources Code Section 21081(a)(1), the OIAA finds that mitigation measures have been incorporated into the Project which mitigate or avoid the significant effects on the environment for tribal cultural resources.

Facts in Support of Findings:

According to NAHC's Sacred Lands Inventory search, the Project site has not been cataloged as a Native American sacred or cultural place of special religious or social significance, and the NAHC does not have knowledge of Native American cultural resources (e.g., traditional use or gathering area, place of religious or sacred activity, etc.) at and within the immediate vicinity of the Project site. Based on literature review, no tribal cultural resources as defined by PRC Section 5020.1(k) have been identified on the Project site. Observations made during the field survey did not identify any tribal cultural resources. No consultation from the 18 individuals representing 12 Native American tribal groups was requested and no tribes identified any TCRs on site. Two of the 12 tribes-Agua Caliente Band of Cahuilla Indians and San Manuel Band of Mission Indiansresponded that the Project site is not within their ancestral territory. No other tribes have responded to date, so no consultation was required or completed. Therefore, no tribal cultural resources were identified on the Project site. Mitigation Measure CUL-1 would require an archaeological monitor to observe all ground disturbing activities associated with the proposed Project. Mitigation Measure TCR-1 further requires the archaeological monitor to consult local Native American tribes to determine the tribal cultural significance of the object and its treatment, if required. Mitigation Measure TCR-2 and TCR-3 require coordination and procedures with the appropriate Native American Tribe(s) should Native American human remains be discovered or recognized on the Project site. Implementation of Mitigation Measures CUL-1 and TCR-1 through TCR-3 would reduce potentially significant impacts to tribal cultural resources determined by criteria provided Public Resources Code Section 5020.1(k) to less than significant.

Project Design Features:

None
Mitigation Measures:

CUL-1, TCR-1, TCR-2, and TCR-3

CUL-1 Archaeological Monitoring of All Ground-Disturbing Activities During Construction of Phase 1 and Phase 2.

- a) Prior to the issuance of grading permits by the City of Ontario for Phase 1 and Phase 2 of the proposed Project, the OIAA and/or its construction contractor must retain a qualified professional archeologist meeting the Secretary of Interior's PQS for Archaeology (as defined in the Code of Federal Regulations, 36 CFR Part 61). The qualified archaeologist will be retained to conduct monitoring of rough grading activities conducted during both Project phases. The qualified archaeologist shall have the authority to redirect earthmoving activities in the event that suspected cultural resources are unearthed during construction activities.
- b) The qualified archaeologist shall prepare a Cultural Resources Monitoring and Treatment Plan that will describe processes for archaeological monitoring and for handling incidental discovery of objects, features, and cultural resources for all ground-disturbing construction and preconstruction activities.
- c) Prior to the issuance of a grading permit, all construction workers involved with grading and trenching operations shall receive training by the qualified archaeologist to recognize unique archaeological resources, including tribal cultural resources, should such resources be unearthed during grounddisturbing construction activities. The training of all construction workers involved with grading and trenching operations shall explain the importance and legal basis for the protection of significant archaeological resources. It will include a brief review of the cultural sensitivity of the construction area and the surrounding area; what resources could potentially be identified during earthmoving activities; the requirements of the monitoring program; the protocols that apply in the event inadvertent discoveries of cultural resources are identified, including who to contact and appropriate avoidance measures until the find(s) can be properly evaluated; and any other appropriate protocols. All new construction personnel involved with grading and trenching operations that begin work following the initial training session must take the training prior to beginning work; the qualified archaeologist shall be available to provide the training on an as needed basis.
- d) In the event subsurface artifacts or features are encountered during grounddisturbing activities, the construction supervisor shall be required by his contract to immediately halt and redirect grading operations within a 100-foot radius of the discovery and see identification and evaluation and evaluation of the suspected resource by the qualified archaeologist for listing in the NRHP and CRHR. This requirement shall be noted on all grading plans and the construction contractor shall be obligated to comply with the note.

e) After the qualified archaeologist makes his/her initial assessment of the nature of the find. The archaeologist shall pursue either protection in place or recovery, salvage, and treatment of the deposits. Recovery, salvage, and treatment protocols shall be developed in accordance with applicable provisions of Public Resource Code Section 21083.2 and State CEQA Guidelines 15064.5 and 15126.4 in consultation with OIAA or with a recognized scientific or educational repository, including the SCCIC. Preservation in place shall be the preferred means to avoid impacts to archaeological resources qualifying as historical resources, consistent with CEQA Guidelines Section 15126.4(b)(3)(C).

TCR-1 Retain a Native American Monitor Prior to Commencement of Ground-Disturbing Activities.

- a) The project applicant/lead agency shall retain a Native American Monitor from or approved by the appropriate Native American Tribe(s). The monitor shall be retained prior to the commencement of any ground-disturbing activity for the subject project at all project locations (i.e., both onsite and any off-site locations that are included in the project description/definition and/or required in connection with the project, such as public improvement work). "Grounddisturbing activity" shall include, but is not limited to, demolition, pavement removal, potholing, auguring, grubbing, tree removal, boring, grading, excavation, drilling, and trenching.
- b) A copy of the executed monitoring agreement shall be submitted to the lead agency prior to the commencement of any ground-disturbing activity, or the issuance of any permit necessary to commence a ground-disturbing activity.
- c) The monitor will complete daily monitoring logs that will provide descriptions of the relevant ground-disturbing activities, the type of construction activities performed, locations of ground-disturbing activities, soil types, cultural-related materials, and any other facts, conditions, materials, or discoveries of significance to the Tribe. Monitor logs will identify and describe any discovered TCRs, including but not limited to, Native American cultural and historical artifacts, remains, places of significance, etc., (collectively, tribal cultural resources, or "TCR"), as well as any discovered Native American (ancestral) human remains and burial goods. Copies of monitor logs will be provided to the proposed Project applicant/lead agency upon written request to the appropriate Native American Tribe(s).
- d) On-site tribal monitoring shall conclude upon the latter of the following (1) written confirmation to the appropriate Native American Tribe(s) from a designated point of contact for the proposed Project applicant/lead agency that all ground-disturbing activities and phases that may involve ground-disturbing activities on the project site or in connection with the project are complete; or (2) a determination and written notification by the appropriate Native

American Tribe(s) to the proposed Project applicant/lead agency that no future, planned construction activity and/or development/construction phase at the Project site possesses the potential to impact Native American Tribe TCRs.

e) Upon discovery of any TCRs, all construction activities in the immediate vicinity of the discovery shall cease (i.e., not less than the surrounding 50 feet) and shall not resume until the discovered TCR has been fully assessed by the Native American monitor and/or Native American archaeologist. The appropriate Native American Tribe(s) will recover and retain all discovered TCRs in the form and/or manner the Native American Tribe(s) deem appropriate, in the Tribe's sole discretion, and for any purpose the Native American Tribe(s) deem appropriate, including for educational, cultural and/or historic purposes.

TCR-2 Unanticipated Discovery of Human Remains and Associated Funerary Objects.

- a) Native American human remains are defined in PRC 5097.98 (d)(1) as an inhumation or cremation, and in any state of decomposition or skeletal completeness. Funerary objects, called associated grave goods in Public Resources Code Section 5097.98, are also to be treated according to this statute.
- b) If Native American human remains and/or grave goods discovered or recognized on the project site, then all construction activities shall immediately cease. Health and Safety Code Section 7050.5 dictates that any discoveries of human skeletal material shall be immediately reported to the County Coroner and all ground-disturbing activities shall immediately halt and shall remain halted until the coroner has determined the nature of the remains. If the coroner recognizes the human remains to be those of a Native American or has reason to believe they are Native American, he or she shall contact, by telephone within 24 hours, the Native American Heritage Commission, and Public Resources Code Section 5097.98 shall be followed.
- c) Human remains and grave/burial goods shall be treated alike per California Public Resources Code section 5097.98(d)(1) and (2).
- d) Construction activities may resume in other parts of the Project site at a minimum of 200 feet away from discovered human remains and/or burial goods, if the appropriate Native American Tribe(s) determine in its sole discretion that resuming construction activities at that distance is acceptable and provides the project manager express consent of that determination (along with any other mitigation measures the appropriate Native American Tribe(s) and/or archaeologist deems necessary). (CEQA Guidelines Section 15064.5(f)).

- e) Preservation in place (i.e., avoidance) is the preferred manner of treatment for discovered human remains and/or burial goods, if feasible. Any historic archaeological material that is not Native American in origin (non-TCR) shall be curated at a public, non-profit institution with a research interest in the materials, such as the Natural History Museum of Los Angeles County or the Fowler Museum, if such an institution agrees to accept the material. If no institution accepts the archaeological material, it shall be offered to a local school or historical society in the area for educational purposes.
- f) Any discovery of human remains/burial goods shall be kept confidential to prevent further disturbance.

TCR-3 Procedures for Burials and Funerary Remains.

- a) The appropriate Native American Tribe(s) burial policy shall be implemented.
- b) If the discovery of human remains includes four or more burials, the discovery location shall be treated as a cemetery and a separate treatment plan shall be created.
- c) The prepared soil and cremation soils are to be treated in the same manner as bone fragments that remain intact. Associated funerary objects are objects that, as part of the death rite or ceremony of a culture, are reasonably believed to have been placed with individual human remains either at the time of death or later; other items made exclusively for burial purposes or to contain human remains can also be considered as associated funerary objects. Cremations will either be removed in bulk or by means as necessary to ensure complete recovery of all sacred materials.
- d) In the case where discovered human remains cannot be fully documented and recovered on the same day, the remains will be covered with muslin cloth and a steel plate that can be moved by heavy equipment placed over the excavation opening to protect the remains. If this type of steel plate is not available, a 24-hour guard should be posted outside of working hours. The Tribe will make every effort to recommend diverting the project and keeping the remains in situ and protected. If the project cannot be diverted, it may be determined that burials will be removed.
- e) In the event preservation in place is not possible despite good faith efforts by the proposed Project applicant/developer and/or landowner, before ground-disturbing activities may resume on the Project site, the landowner shall arrange a designated site location within the footprint of the proposed Project for the respectful reburial of the human remains and/or ceremonial objects.
- f) Each occurrence of human remains and associated funerary objects will be stored using opaque cloth bags. All human remains, funerary objects, sacred objects and objects of cultural patrimony will be removed to a secure container

Ontario International Airport South Airport Cargo Center Project on site if possible. These items should be retained and reburied within six months of recovery. The site of reburial/repatriation shall be on the project site but at a location agreed upon between the Tribe and the landowner at a site to be protected in perpetuity. There shall be no publicity regarding any cultural materials recovered.

g) The proposed Project's qualified archaeologist will work closely with the appropriate Native American Tribe(s) to ensure that the excavation is treated carefully, ethically and respectfully. If data recovery is approved by the appropriate Native American Tribe(s), documentation shall be prepared and shall include (at a minimum) detailed descriptive notes and sketches. All data recovery data recovery-related forms of documentation shall be approved in advance by the appropriate Native American Tribe(s). If any data recovery is performed, once complete, a final report shall be submitted to the appropriate Native American Tribe(s) and the NAHC.

Reference

For a complete discussion of Project impacts related to tribal cultural resources, see Section 5.13.3 of the Draft EIR at pages 5.13-6 - 5.13-10.

4.12 Tribal Cultural Resource (Cumulative)

The Draft EIR discusses the impacts related to tribal cultural resources in Section 5.13. The following discussion addresses potential cumulative impacts with respect to tribal cultural resources.

Impact Summary:

The potential for the Project to result in a cumulatively considerable contribution to a significant cumulative impact to tribal cultural resources was analyzed in conjunction with other developments located in the influence areas of the tribes that occupied the region. As discussed, construction of the Project may disturb native soils containing prehistoric objects and features that may be determined to be a tribal cultural resource.

Less than Significant Effects with Mitigation:

The Project's cumulative impacts to tribal cultural resources would be less than significant with mitigation.

Findings:

Pursuant to Public Resources Code Section 21081(a)(1), the OIAA finds that mitigation measures have been incorporated into the Project which mitigate or avoid the significant effects on the environment to tribal cultural resources. Therefore, impacts would be less than significant with mitigation incorporated.

Facts in Support of Findings:

No consultation from the 18 individuals representing 12 Native American tribal groups was requested and no tribes identified any TCRs on site. Two of the 12 tribes—Agua Caliente Band of

Cahuilla Indians and San Manuel Band of Mission Indians—responded that the Project site is not within their ancestral territory. No other tribes have responded to date, so no consultation was required or completed. Therefore, no tribal cultural resources were identified on the Project site. The proposed Project would require **Mitigation Measures CUL-1** and **TCR-1 through TCR-3** to reduce Project-level impacts to less than significant. Related projects in the region would also be required to mitigate potential inadvertent discoveries of subsurface resources, including tribal cultural resources, and would comply with PRC 21083.2, which allows lead agencies to make provisions for accidentally discovering archaeological resources, including tribal cultural resources during construction. Furthermore, the proposed Project and related projects would also be required to comply with California Health and Safety Code Section 7050.5, Public Resources Code Section 5097.98, and CEQA Guidelines Section 15064.5, which provides guidance on the discovery of human remains and its treatment or disposition with appropriate dignity. Therefore, compliance with existing policies and regulations, and implementation of Project mitigation, would result in the Project's contribution to impacts on tribal cultural resources being less than cumulatively considerable.

With implementation of **Mitigation Measures CUL-1**, **TCR-1**, **TCR-2**, and **TCR-3** and compliance with regulatory requirements, Project impacts to tribal cultural resources would be less than significant.

Mitigation Measures:

CUL-1, TCR-1, TCR-2, and TCR-3

CUL-1 Archaeological Monitoring of All Ground-Disturbing Activities During Construction of Phase 1 and Phase 2.

- a) Prior to the issuance of grading permits by the City of Ontario for Phase 1 and Phase 2 of the proposed Project, the OIAA and/or its construction contractor must retain a qualified professional archeologist meeting the Secretary of Interior's PQS for Archaeology (as defined in the Code of Federal Regulations, 36 CFR Part 61). The qualified archaeologist will be retained to conduct monitoring of rough grading activities conducted during both Project phases. The qualified archaeologist shall have the authority to redirect earthmoving activities in the event that suspected cultural resources are unearthed during construction activities.
- b) The qualified archaeologist shall prepare a Cultural Resources Monitoring and Treatment Plan that will describe processes for archaeological monitoring and for handling incidental discovery of objects, features, and cultural resources for all ground-disturbing construction and preconstruction activities.
- c) Prior to the issuance of a grading permit, all construction workers involved with grading and trenching operations shall receive training by the qualified archaeologist to recognize unique archaeological resources, including tribal cultural resources, should such resources be unearthed during grounddisturbing construction activities. The training of all construction workers

involved with grading and trenching operations shall explain the importance and legal basis for the protection of significant archaeological resources. It will include a brief review of the cultural sensitivity of the construction area and the surrounding area; what resources could potentially be identified during earthmoving activities; the requirements of the monitoring program; the protocols that apply in the event inadvertent discoveries of cultural resources are identified, including who to contact and appropriate avoidance measures until the find(s) can be properly evaluated; and any other appropriate protocols. All new construction personnel involved with grading and trenching operations that begin work following the initial training session must take the training prior to beginning work; the qualified archaeologist shall be available to provide the training on an as needed basis.

- d) In the event subsurface artifacts or features are encountered during grounddisturbing activities, the construction supervisor shall be required by his contract to immediately halt and redirect grading operations within a 100-foot radius of the discovery and see identification and evaluation and evaluation of the suspected resource by the qualified archaeologist for listing in the NRHP and CRHR. This requirement shall be noted on all grading plans and the construction contractor shall be obligated to comply with the note.
- e) After the qualified archaeologist makes his/her initial assessment of the nature of the find. The archaeologist shall pursue either protection in place or recovery, salvage, and treatment of the deposits. Recovery, salvage, and treatment protocols shall be developed in accordance with applicable provisions of Public Resource Code Section 21083.2 and State CEQA Guidelines 15064.5 and 15126.4 in consultation with OIAA or with a recognized scientific or educational repository, including the SCCIC. Preservation in place shall be the preferred means to avoid impacts to archaeological resources qualifying as historical resources, consistent with CEQA Guidelines Section 15126.4(b)(3)(C).

TCR-1 Retain a Native American Monitor Prior to Commencement of Ground-Disturbing Activities.

a) The project applicant/lead agency shall retain a Native American Monitor from or approved by the appropriate Native American Tribe(s). The monitor shall be retained prior to the commencement of any ground-disturbing activity for the subject project at all project locations (i.e., both onsite and any off-site locations that are included in the project description/definition and/or required in connection with the project, such as public improvement work). "Grounddisturbing activity" shall include, but is not limited to, demolition, pavement removal, potholing, auguring, grubbing, tree removal, boring, grading, excavation, drilling, and trenching.

- b) A copy of the executed monitoring agreement shall be submitted to the lead agency prior to the commencement of any ground-disturbing activity, or the issuance of any permit necessary to commence a ground-disturbing activity.
- c) The monitor will complete daily monitoring logs that will provide descriptions of the relevant ground-disturbing activities, the type of construction activities performed, locations of ground-disturbing activities, soil types, cultural-related materials, and any other facts, conditions, materials, or discoveries of significance to the Tribe. Monitor logs will identify and describe any discovered TCRs, including but not limited to, Native American cultural and historical artifacts, remains, places of significance, etc., (collectively, tribal cultural resources, or "TCR"), as well as any discovered Native American (ancestral) human remains and burial goods. Copies of monitor logs will be provided to the proposed Project applicant/lead agency upon written request to the appropriate Native American Tribe(s).
- d) On-site tribal monitoring shall conclude upon the latter of the following (1) written confirmation to the appropriate Native American Tribe(s) from a designated point of contact for the proposed Project applicant/lead agency that all ground-disturbing activities and phases that may involve ground-disturbing activities on the project site or in connection with the project are complete; or (2) a determination and written notification by the appropriate Native American Tribe(s) to the proposed Project applicant/lead agency that no future, planned construction activity and/or development/construction phase at the Project site possesses the potential to impact Native American Tribe TCRs.
- e) Upon discovery of any TCRs, all construction activities in the immediate vicinity of the discovery shall cease (i.e., not less than the surrounding 50 feet) and shall not resume until the discovered TCR has been fully assessed by the Native American monitor and/or Native American archaeologist. The appropriate Native American Tribe(s) will recover and retain all discovered TCRs in the form and/or manner the Native American Tribe(s) deem appropriate, in the Tribe's sole discretion, and for any purpose the Native American Tribe(s) deem appropriate, including for educational, cultural and/or historic purposes.

TCR-2 Unanticipated Discovery of Human Remains and Associated Funerary Objects.

a) Native American human remains are defined in PRC 5097.98 (d)(1) as an inhumation or cremation, and in any state of decomposition or skeletal completeness. Funerary objects, called associated grave goods in Public Resources Code Section 5097.98, are also to be treated according to this statute.

- b) If Native American human remains and/or grave goods discovered or recognized on the project site, then all construction activities shall immediately cease. Health and Safety Code Section 7050.5 dictates that any discoveries of human skeletal material shall be immediately reported to the County Coroner and all ground-disturbing activities shall immediately halt and shall remain halted until the coroner has determined the nature of the remains. If the coroner recognizes the human remains to be those of a Native American or has reason to believe they are Native American, he or she shall contact, by telephone within 24 hours, the Native American Heritage Commission, and Public Resources Code Section 5097.98 shall be followed.
- c) Human remains and grave/burial goods shall be treated alike per California Public Resources Code section 5097.98(d)(1) and (2).
- d) Construction activities may resume in other parts of the Project site at a minimum of 200 feet away from discovered human remains and/or burial goods, if the appropriate Native American Tribe(s) determine in its sole discretion that resuming construction activities at that distance is acceptable and provides the project manager express consent of that determination (along with any other mitigation measures the appropriate Native American Tribe(s) and/or archaeologist deems necessary). (CEQA Guidelines Section 15064.5(f)).
- e) Preservation in place (i.e., avoidance) is the preferred manner of treatment for discovered human remains and/or burial goods, if feasible. Any historic archaeological material that is not Native American in origin (non-TCR) shall be curated at a public, non-profit institution with a research interest in the materials, such as the Natural History Museum of Los Angeles County or the Fowler Museum, if such an institution agrees to accept the material. If no institution accepts the archaeological material, it shall be offered to a local school or historical society in the area for educational purposes.
- f) Any discovery of human remains/burial goods shall be kept confidential to prevent further disturbance.

TCR-3 Procedures for Burials and Funerary Remains.

- a) The appropriate Native American Tribe(s) burial policy shall be implemented.
- b) If the discovery of human remains includes four or more burials, the discovery location shall be treated as a cemetery and a separate treatment plan shall be created.
- c) The prepared soil and cremation soils are to be treated in the same manner as bone fragments that remain intact. Associated funerary objects are objects that, as part of the death rite or ceremony of a culture, are reasonably believed to have been placed with individual human remains either at the time of death

or later; other items made exclusively for burial purposes or to contain human remains can also be considered as associated funerary objects. Cremations will either be removed in bulk or by means as necessary to ensure complete recovery of all sacred materials.

- d) In the case where discovered human remains cannot be fully documented and recovered on the same day, the remains will be covered with muslin cloth and a steel plate that can be moved by heavy equipment placed over the excavation opening to protect the remains. If this type of steel plate is not available, a 24-hour guard should be posted outside of working hours. The Tribe will make every effort to recommend diverting the project and keeping the remains in situ and protected. If the project cannot be diverted, it may be determined that burials will be removed.
- e) In the event preservation in place is not possible despite good faith efforts by the proposed Project applicant/developer and/or landowner, before ground-disturbing activities may resume on the Project site, the landowner shall arrange a designated site location within the footprint of the proposed Project for the respectful reburial of the human remains and/or ceremonial objects.
- f) Each occurrence of human remains and associated funerary objects will be stored using opaque cloth bags. All human remains, funerary objects, sacred objects and objects of cultural patrimony will be removed to a secure container on site if possible. These items should be retained and reburied within six months of recovery. The site of reburial/repatriation shall be on the project site but at a location agreed upon between the Tribe and the landowner at a site to be protected in perpetuity. There shall be no publicity regarding any cultural materials recovered.
- g) The proposed Project's qualified archaeologist will work closely with the appropriate Native American Tribe(s) to ensure that the excavation is treated carefully, ethically and respectfully. If data recovery is approved by the appropriate Native American Tribe(s), documentation shall be prepared and shall include (at a minimum) detailed descriptive notes and sketches. All data recovery data recovery-related forms of documentation shall be approved in advance by the appropriate Native American Tribe(s). If any data recovery is performed, once complete, a final report shall be submitted to the appropriate Native American Tribe(s) and the NAHC.

Reference

For a complete discussion of Project cumulative impacts related to tribal cultural resources, see Section 5.13.4 of the Draft EIR at page 5.13.10.

5.0 EFFECTS DETERMINED TO BE NOT SIGNIFICANT OR LESS THAN SIGNIFICANT

OIAA finds that, based upon substantial evidence in the record, the following impacts associated with the Project, and other effects identified as less than significant in the Final EIR, are less than significant and no mitigation is required pursuant to CEQA Section 21081(a) and CEQA Guidelines Section 15091(a).

5.1 Aesthetics

Less than Significant Effects: Impact AES-1

The Project is not expected to have a substantial adverse effect on a scenic vista. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

Views of the Santa Ana Mountains and Chino Hills to the east and south would not be affected with implementation of the proposed Project. During construction of Phase 1 and Phase 2 of the proposed Project, equipment would be staged on-site, which would have a minimal impact on scenic views from East Mission Boulevard looking north during proposed Project development. Development within this area of the Airport would not substantially alter the scenic views provided along Mission Boulevard of the San Gabriel Mountains backdrop because the peaks rise to 7,000 feet above mean sea level (amsl). For these reasons, the development of the proposed Project would not have a substantial adverse effect on a scenic vista. Therefore, impacts would be less than significant, and no mitigation is required. (Draft EIR at pages 5.1-19 - 5.1-29).

No Effects: AES-2

The Project is not expected to impact scenic resources, including, but not limited to trees, rock outcroppings, and historic buildings within a state scenic highway. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

The Project site is not located in the vicinity of a State Scenic Highway. Due to the distance and intervening land uses, no portion of the Project site or surrounding area is viewable from the officially designated SR-91 or the eligible portion of the SR-142, which are approximately 16 miles southwest and 9.5 miles southwest of the Project site, respectively. Additionally, the Project site does not contain any scenic resources, such as rock outcroppings or trees, or historic buildings that would be damaged by the proposed Project. As such, the Project would not result in impacts related to the substantial damage of scenic resources within a State Scenic Highway. Therefore, impacts would be less than significant, and no mitigation is required. (Draft EIR at page 5.1-29).

Less than Significant Effects: AES-3

The Project is not expected to conflict with applicable zoning and other regulations governing scenic quality. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

The Project site is designated "Airport" in the Ontario Plan and zoned "ONT" – Ontario Airport zone. Use of the Project site is subject to regulatory oversight by OIAA and the FAA through the approved Ontario International Airport Layout Plan (ALP) and Airport Land Use Compatibility Plan (ALUCP). The proposed Project would be consistent with applicable policies in The Ontario Plan to the Airport and regulations in the ALUCP. Therefore, the proposed Project would not conflict with applicable zoning and other regulations governing scenic quality, and impacts would be less than significant. (Draft EIR at pages 5.1-29 - 5.1-30).

Less than Significant Effects: AES-4

The Project is not expected to create a new source of substantial light or glare which would adversely affect day or nighttime views in the area. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

Construction activities would occur during daylight hours to the extent feasible. Additional lighting during the construction period would be placed within and along the exterior of the Project site and would be available during night-time for on-site security and pedestrian safety purposes. The proposed Project would not introduce a substantial source of light which would affect day or nighttime views in the area. Outdoor lights would be designed and constructed to reflect light away from East Avion Street and adjacent properties. Additionally, lighting would be installed such that light would not shine directly at or cause reflections on the Airport's taxiways or runways. All new lighting would comply with applicable regulations of the 2019 State Building Energy Efficiency Standards (Title 24). Glare associated with the proposed Project design would be minimal and site efforts would be taken to reduce as much glare as possible. Impacts would be less than significant. (Draft EIR at pages 5.1.30 - 5.1-32).

5.2 Aesthetics (Cumulative)

Less than Significant Effects:

The Project's cumulative contribution to aesthetics would not be cumulatively considerable.

Facts in Support of Findings:

Implementation of the proposed Project would not result in a significant impact related to aesthetics. The proposed Project and all related projects are required to adhere to Airport, City, and State regulations designed to reduce and/or avoid impacts related to aesthetics. Additionally, projects within the Airport and the proposed Project would be subject to FAA and OIAA approval to avoid impacts related to aesthetics and aviation. With compliance with these regulations, no significant cumulative impacts related to aesthetics would result from the proposed Project, related projects, and other growth; and the proposed Project's contribution to cumulative impacts would not be cumulatively considerable. (Draft EIR at pages 5.1-32 - 5.1-33).

5.3 Air Quality

Less Than Significant Effects: AQ-1

The Project is not expected to conflict with or obstruct implementation of the applicable air quality plan. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

During construction, the proposed Project would comply with CARB's requirements to minimize short-term emissions from on-road and off-road diesel equipment, including limiting heavy duty diesel motor vehicle idling to no more than 5 minutes at any given time, and with SCAQMD's regulations, such as Rule 403 for controlling fugitive dust and Rule 1113 for controlling VOC emissions from architectural coatings. Furthermore, the proposed Project would use vehicles from vendors that comply with fleet rules to reduce on-road truck emissions under CARB's Truck and Bus regulation. Compliance with these measures and requirements would be consistent with and meet or exceed the 2016 AQMP and 2022 AQMP requirements for control strategies intended to reduce emissions from construction equipment and activities. Additionally, the proposed Project would comply with the measures included in the Airport's Air Quality Improvement Plan (AQIP) such as using Tier 4 construction equipment. The proposed Project would result in short-term employment growth and would not conflict with employment or housing projections within the AQMP. Impacts related to construction are not expected to conflict with or obstruct implementation of the applicable air quality plan and would be less than significant.

For operations, the AQIP includes a GSE Policy (RM1), which requires the Airport to implement a GSE policy that promotes the use of newer, cleaner equipment for ground operations. The proposed Project would include the use and operation of electric-powered equipment, including forklifts, loaders, tugs, ground power units, and ramp support (vans/carts) that would be stored and charged in designated areas in the cargo building and aircraft apron.

The region is projected to see a 12 percent growth in population, 17 percent growth in housing units, 11 percent growth in employment, and 5 percent growth in vehicle miles traveled between 2018 and 2037. As such, aircraft operations within the region are also expected to increase.

The SCAQMD does not have regulatory authority over the aircraft or aircraft operations, which are the primary sources of air emissions associated with airports. Operation of the proposed Project would be consistent with the measures in the Airport's AQIP as it would include all-electric GSE, which is consistent with control strategy MOB:4 Emissions Reductions at Commercial Airports in the 2022 AQMP. The RTP/SCS notes that SCAG has no authority over airports or airport activity and that the FAA has this authority. SCAG is interested in how traffic going and coming from airports affects the roads, highways, and transit systems in the region. The Aviation and Ground Access appendix to the RTP/SCS has air cargo forecasts and SCAG modeling estimates truck trips for the 5 busiest airports in the region and Ontario is one of these airports. As shown in Table 14 in the Aviation and Ground Access appendix, SCAG identifies 900 daily truck trips for Ontario Airport in 2016 and projects 1,725 daily truck trips in 2045. The proposed Project would generate 450 additional truck trips per day, an amount that is within, and consistent with, the 2045 truck trip estimate for Ontario Airport. As such, the proposed Project would accommodate the regional

movement of goods per SCAG projections. Impacts related to operation would be less than significant. (Draft EIR at pages 5.2-65 - 5.2-70).

Less Than Significant Effects: AQ-2 (Construction)

Construction activities also would not exceed the ambient air quality standards at nearby receptors and cumulative impacts would be less than significant. Implementation of mitigation measures would further reduce potential daily emissions from construction activities.

Facts in Support of Findings:

During construction, the proposed Project's daily criteria pollutant emissions would not exceed SCAQMD thresholds. Implementation of mitigation measures would further reduce potential daily emissions from construction activities. As such, air quality impacts from the construction of the proposed Project would be less than significant and would not result in a cumulatively considerable increase of air emissions during the construction period.

Less Than Significant Effects: AQ-3

The Project is not expected to expose sensitive receptors to substantial pollutant concentrations. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

An air dispersion analysis was conducted to determine the ambient concentrations at nearby receptors which would result from Project construction and operation. Air pollution concentrations during construction and operation of Phase 1 and Phase 2 would be below the significant thresholds for NO₂, PM₁₀, PM_{2.5}, SO₂, and CO.

The maximum impacts due to construction activities occur near the Project site along the Airport boundary and dissipate moderately within 1,000 feet of the Project site. Phase 1 construction activities would occur between 2023 and 2025, while Phase 2 construction activities would occur from 2025 through 2029. For the air quality receptors during Project construction of Phase 1, the incremental 1-hour NO₂ impacts, including background concentrations, would be a maximum of 0.10 ppm, which is below the State threshold of 0.18 ppm. The maximum Project construction incremental annual NO₂ impacts, including background concentrations, would be 0.02 ppm, which is below the thresholds of 0.03 ppm (state) and 0.053 ppm (federal). Moreover, the maximum incremental 24-hour and annual PM₁₀ impacts would be 3.67 μ g/m³ and 0.09 μ g/m³, respectively; impacts would be below the 24-hour PM₁₀ threshold of 10.4 μ g/m³ and below the annual PM₁₀ maximum incremental 24-hour PM_{2.5} impacts would be 0.97 μ g/m³, which would be below the 24-hour PM_{2.5} threshold of 10.4 μ g/m³. Additionally, the maximum incremental SO₂ and CO impacts including background concentrations would be well below the significance thresholds.

For the air quality receptors during Project construction during Phase 2, the incremental 1-hour NO₂ impacts, including background concentrations, would be a maximum of 0.10 ppm, which is below the State threshold of 0.18 ppm. The maximum Project construction incremental annual NO₂ impacts, including background concentrations, would be 0.02 ppm, which is below the thresholds of 0.03 ppm (state) and 0.053 ppm (federal). Moreover, the maximum incremental 24-hour and annual PM₁₀ impacts would be 3.66 μ g/m³ and 0.11 μ g/m³, respectively; impacts would be below the 24-hour PM₁₀ threshold of 10.4 μ g/m³ and below the annual PM₁₀ threshold

of 1.0 μ g/m³. The Project construction maximum incremental 24-hour PM_{2.5} impacts would be 0.65 μ g/m³, which would be below the 24-hour PM_{2.5} threshold of 10.4 μ g/m³. Additionally, the maximum incremental SO₂ and CO impacts including background concentrations would be well below the significance thresholds.

For the air quality receptors during Phase 1 of Project operations, the incremental 1-hour NO₂ impacts, including background concentrations, would be a maximum of 0.13 ppm, which is below the State threshold of 0.18 ppm. The maximum 1-hour NO₂ concentrations (i.e., 236 μ g/m³ or 0.13 ppm compared to the threshold of 0.18 ppm) occur to the southeast of the Airport and are mostly a result of aircraft departures. The maximum Project operation incremental annual NO₂ impacts, including background concentrations, would be 0.02 ppm, which is below the thresholds of 0.03 ppm (state) and 0.053 ppm (federal). Moreover, the maximum incremental 24-hour and annual PM₁₀ impacts would be 0.64 μ g/m³ and 0.20 μ g/m³, respectively; impacts would be below the 24-hour PM₁₀ threshold of 2.5 μ g/m³ and below the annual PM₁₀ threshold of 1.0 μ g/m³. The Project operation maximum incremental 24-hour PM_{2.5} impacts would be 0.39 μ g/m³, which would be below the 24-hour PM_{2.5} threshold of 2.5 μ g/m³.

For the air quality receptors during Phase 2 of Project (buildout of the Project) operations, the incremental 1-hour NO₂ impacts, including background concentrations, would be a maximum of 0.16 ppm, which is below the State threshold of 0.18 ppm. The maximum 1-hour NO₂ concentrations (291 μ g/m³ or 0.16 ppm compared to the threshold of 0.18 ppm) occur to the southeast of the Airport and are mostly a result of aircraft departures. The maximum Project operation incremental annual NO₂ impacts, including background concentrations, would be 0.02 ppm, which is below the thresholds of 0.03 ppm (state) and 0.053 ppm (federal). Moreover, the maximum incremental 24- hour and annual PM10 impacts would be 0.98 μ g/m3 and 0.22 μ g/m3, respectively; impacts would be below the 24-hour PM₁₀ threshold of 2.5 μ g/m3 and below the annual PM₁₀ threshold of 1.0 μ g/m3. The Project operation maximum incremental 24-hour PM_{2.5} impacts would be 0.83 μ g/m3, which would be below the 24-hour PM_{2.5} threshold of 2.5 μ g/m3. Additionally, the maximum incremental SO₂ and CO impacts including background concentrations would be well below the significance thresholds.

Localized CO concentration levels were forecasted at the proposed Project's three most potentially impacted intersections using the CALINE-4 dispersion model developed by Caltrans, peak-hour traffic volumes, and conservative meteorological assumptions. Project generated traffic volumes are forecasted to have a negligible effect on the projected 1-hour and 8-hour CO concentrations at each of the three intersection locations analyzed. None of the proposed Project's studied intersections would exceed 400,000 vehicles per day. Additionally, none of the proposed Project's studied intersections worsen from C to D or increase the V/C ratio at any intersection rated D or worse by two percent or more.

An HRA was conducted for the proposed Project to address the potential for human health impacts associated with construction and operation of the proposed Project. The cancer risk for off-site worker receptors due to construction activities would be below the SCAQMD threshold of 10 per one million persons. Additionally, the chronic health impact due to construction activities at all off-site worker receptors would be below the Project-level threshold of a hazard index greater than 1. The cancer risk for residence, off-site worker receptor (such as office buildings, retail centers, hotels, hospitals), on-site worker terminal receptor, and on-site non-terminal worker receptor due

to operational activities of the proposed Project would be below the SCAQMD threshold of 10 per one million persons. Finally, the acute and chronic health impact due to operational activities at all sensitive receptors would be below the Project-level threshold of 1. As such, impacts would be less than significant. (Draft EIR at pages 5.2-74 - 5.2-87).

Less Than Significant Effects: AQ-4

The Project is not expected to result in other emissions, such as those leading to odors, adversely affecting a substantial number of people. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

The proposed Project does not contain land uses typically associated with emitting objectionable odors. Potential odor sources associated with the proposed Project may result from construction equipment exhaust and the application of asphalt and architectural coatings during construction activities and the temporary storage of typical solid waste (refuse) associated with the proposed Project's (long-term operational) uses. Standard construction requirements would minimize odor impacts. The construction odor emissions would be temporary, short-term, and intermittent in nature and would cease upon completion of the respective phase of construction and is thus considered less than significant. It is expected that Project-generated refuse would be stored in covered containers and removed at regular intervals in compliance with the City's solid waste regulations. The proposed Project would also be required to comply with SCAQMD Rule 402 to prevent occurrences of public nuisances. As such, impacts would be less than significant. (Draft EIR at page 5.2-88).

5.4 Biological Resources

Less Than Significant Effects: BIO-2 -

The Project is not expected to have a substantial adverse effect on any riparian habitat or other sensitive natural community identified in local or regional plans, policies, regulations or by the California Department of Fish and Game or U.S. Fish and Wildlife Service. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

There are no blue-line streams within the Project site. The Project site is developed with airport and cargo operations and does not support any identifiable drainage courses, inundated areas, wetland features, hydric soils, or hydrogeomorphic features such as perennial creeks. There are no riparian corridors, creeks, or natural areas existing within or connecting the Project site to natural, undeveloped areas. The Cucamonga Channel adjacent to the eastern boundary of the Project site is identified as a riverine resource. However, the Cucamonga Channel is an open concrete box culvert and does not support riparian habitat or other sensitive natural plant communities. Therefore, impacts would be less than significant. (Draft EIR at pages 5.3-21 - 5.3-42).

Less Than Significant Effects: BIO-3

The Project is not expected to have a substantial adverse effect on state or federally protected wetlands (including, but not limited to, marsh, vernal pool, coastal, etc.) through direct removal,

Ontario International Airport South Airport Cargo Center Project filling, hydrological interruption, or other means. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

No inundated areas, wetland features, or wetland plant species that would be considered wetlands as defined by Section 404 of the Clean Water Act occur within the proposed Project footprint. As the proposed Project would utilize the existing drainage outlet points and implement BMPs to release stormwater at a controlled rate into the Cucamonga Channel, the proposed Project would not significantly impact the Cucamonga Channel. Should a new outlet into the Cucamonga Channel be needed for the proposed Project, the Cucamonga Channel is an open concrete box culvert and does not support riparian habitat, vegetation, other sensitive natural plant communities, or protected wetland. The proposed Project, utilizing the existing outlet points or a new outlet into the Cucamonga Channel, would not have a substantial adverse effect on state or federally protected wetlands. Therefore, impacts would be less than significant. (Draft EIR at pages 5.3-43 - 5.3-44).

Less Than Significant Effects: BIO-4

The Project is not expected to interfere substantially with the movement of any native resident or migratory fish or wildlife species or with established native resident or migratory wildlife corridors, or impede the use of native wildlife nursery sites. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

According to the San Bernardino Countywide Plan, the Project site is not within a wildlife corridor or linkage. Additionally, according to the Ontario General Plan EIR, no regional wildlife movement corridors have been identified in the City. The Project site is developed with airport-related improvements and does not contain any wildlife corridors or linkages. Project implementation would be confined to developed areas on the Project site, which is away from regional wildlife corridors and linkages, such as the Santa Ana River. Project implementation would not directly impact existing wildlife movement opportunities. The segment of the Cucamonga Channel adjacent to the Project site is an open concrete box culvert surrounded by airport operations. It does not support plant communities suitable for use as a wildlife corridor nor connect two comparatively undisturbed habitat fragments. The proposed Project would not change the designated use of the channel as Open Space – Non-Recreation. The proposed Project would not substantially impair the Cucamonga Channel, wildlife movement opportunities, nor prevent local wildlife movement through the area. Therefore, impacts would be less than significant. (Draft EIR at pages 5.3-44 – 5.3-55).

Less Than Significant Effects: BIO-5

The Project is not expected to conflict with any local policies or ordinances protecting biological resources, such as a tree preservation policy or ordinance. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

The proposed Project's design would comply with the ONT Wildlife Hazard Management Plan, the ONT Rules and Regulations, and FAA Policy. The landscape trees for the proposed Project

would be Desert Museum Palo Verde, approved by OIAA in consultation with the United States Department of Agriculture (USDA) Wildlife Biologist. These trees would not cause more wildlife to occupy the Project site. Should birds or other wildlife be observed to be a hazard to flight operations, ONT Airside Operations staff shall report to FAA ONT Air Traffic Control Tower. Additionally, the proposed Project would require the removal of vegetation, including trees. As such, Project implementation would comply with all requirements specified in the City of Ontario Parkway Tree Regulations. If required, the proposed Project would maintain any parkway trees adjacent to the Project site to preserve a neat appearance and non-obstructed use of the realigned East Avion Street. Therefore, impacts would be less than significant. (Draft EIR at pages 5.3-45 - 5.3-46).

Less Than Significant Effects: BIO-6

The Project is not expected to conflict with the provisions of an adopted Habitat Conservation Plan, Natural Community Conservation Plan, or other approved local, regional, or state habitat conservation plan. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

The Project site is not located within an adopted Habitat Conservation Plan, Natural Community Conservation Plan, or other approved local, regional, or state Habitat Conservation Plan. Therefore, impacts would be less than significant. Draft EIR at page 5.3-47).

5.5 Cultural Resources

Less Than Significant Effects: CUL-1-

The Project is not expected to cause a substantial adverse change in the significance of a historical resource pursuant to § 15064.5. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

The 1980s-era private jet center is not of sufficient age to be eligible for listing in the National Historic Preservation Act (NRHP), California Register of Historical Resources (CRHR), or as local Ontario Landmarks/Historic Districts, based on the records search, research, field survey, and applicable cultural resource codes and regulations. the Ontario ANG hangar and the GE maintenance facility are not eligible for either the NRHP or CRHR. The Ontario ANG hangar is not eligible for listing as an Ontario Historic Landmark. It is not considered a significant historic resource under any of the NRHP criterion analyzed in the Draft EIR, nor within the CRHR. The Ontario Historic Landmark criteria is nearly identical to the NRHP and CRHR criteria, with Criteria 1 and 2 directly relating to meeting the NRHP or CRHR criteria, respectively. Therefore, the Ontario ANG hangar is not eligible for listing in either the NRHP or CRHR.

Further, the GE maintenance facility is not considered a significant historic resource under any of the NRHP criterion analyzed in the Draft EIR, nor within the CRHR. The Ontario Historic Landmark criteria is nearly identical to the NRHP and CRHR criteria, with Criteria 1 and 2 directly relating to meeting the NRHP or CRHR criteria, respectively. As previously discussed, the GE

maintenance facility is not eligible for listing in either the NRHP or CRHR. Therefore, the GE maintenance facility is not eligible for listing as an Ontario Landmark/Historic District.

As such, they are not historical resources as defined by CEQA and the Project would not directly or indirectly impact any historical resources on the Project site and surrounding area. Therefore, impacts to historical resources during construction and operation of the proposed Project would be less than significant. (Draft EIR at pages 5.4-28 - 5.4-34).

Less Than Significant Effects: CUL-3

The Project is not expected to disturb any human remains, including those interred outside of formal cemeteries. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

The Project site is not a dedicated graveyard or cemetery. Additionally, according to the Native American Heritage Commission (NAHC) Sacred Lands Inventory search, the Project site is not cataloged as a Native American sacred or cultural place of special religious or social significance, which would include graves and cemeteries. Based on the developed condition of the Project site and its historic use as farmland, it is very unlikely that human remains would be discovered at the Project site. In the event human remains were discovered during construction ground disturbance activities, the proposed Project would be required to comply with California Health and Safety Code Section 7050.5, Public Resources Code Section 5097.98, and CEQA Guidelines Section 15064.5, which provide guidance on the discovery of human remains and its treatment or disposition, with appropriate dignity. Therefore, impacts would be less than significant. (Draft EIR at pages 5.4-35 - 5.4-36).

5.6 Energy

Less Than Significant Effects: ENE-1

The Project is not expected to result in potentially significant environmental impact due to wasteful, inefficient, or unnecessary consumption of energy resources, during Project construction or operation. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

During construction, energy would be consumed in the form of electricity associated with the conveyance of water used for dust control, and on a limited basis, powering lights, electronic equipment, or other construction activities necessitating electrical power. An on-site asphalt/concrete recycling operation is proposed on the south side of East Avion Street on a partially paved and flat parcel that is flanked by East Mission Boulevard (and railroad tracks) to the south and industrial abandoned (industrial) uses on either side (which is within the Project site). The recycling operations would reduce the total vehicle miles traveled needed for asphalt/concrete delivery trucks. Moreover, PDF AQ-1 requires the use of Tier 4 off-road equipment during construction, which is more fuel efficient than lower tiered equipment. Due to the temporary nature of the construction process, and the fact that the extent of energy consumption is inherent to construction projects of this size and nature, the proposed Project would not result in inefficient or unnecessary consumption of energy resources during construction.

The proposed Project incorporates sustainable project design features and technology in both design and operation. The Air Cargo Sort Building would meet LEED certification standards and would be all-electric (no natural gas usage). The Project would install a 3.8-Megawatt Solar PV Panel system on the rooftops of the Air Cargo Sort Building and the parking structure. The proposed Project, at full buildout, would include the use and operation of electric-powered equipment, including forklifts, loaders, tugs, ground power units, and ramp support (vans/carts) that would be stored and charged in designated areas in the cargo building and aircraft apron. (see PDF AQ-3). Moreover, a portion of the proposed Project's aviation operations would include electric cargo planes (see Table 3.4 in Section 3.0: Project Description, of the Draft EIR and PDF AQ-4), for which charging stations would be provided in the southeast corner of the Project site. The civil operation of Alice Electric cargo aircraft is subject to FAA certification. For that reason, a footnote has been added in Table 3.4 in Section 3.0: Project Description in the Draft EIR to clarify that the inclusion of Alice Electric cargo aircraft as part of the proposed Project fleet is subject to its certification by the FAA. A new substation proposed by Southern California Edison (SCE) for the proposed Project would be located to the west of the parking structure. As such, the proposed Project would not result in inefficient or unnecessary consumption of energy resources during operation. Therefore, impacts would be less than significant. (Draft EIR at pages 5.5-23 -5.5-28).

Less Than Significant Effects: ENE-2

The Project is not expected to conflict with or obstruct a State or local plan for renewable energy or energy efficiency. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

The proposed Project incorporates sustainable project design features and technology in both design and operation. The Aviation and Ground Access appendix to the RTP/SCS has air cargo forecasts and SCAG modeling estimates truck trips for the 5 busiest airports in the region and Ontario is one of these airports. As shown in Table 14 in the Aviation and Ground Access appendix, SCAG identifies 900 daily truck trips for Ontario Airport in 2016 and projects 1,725 daily truck trips in 2045. The proposed Project would generate 450 additional truck trips per day, an amount that is within, and consistent with, the 2045 truck trip estimate for Ontario Airport. As such, the proposed Project would accommodate the regional movement of goods per SCAG projections. Additionally, the proposed Project would not conflict with the energy policies within the City's general plan and the Environmental Resources Element of the Ontario Plan. As such, the proposed Project would not conflict with or obstruct a State or local plan for renewable energy or energy efficiency. Therefore, impacts would be less than significant. (Draft EIR at pages 5.5-28 - 5.5-30).

5.7 Energy (Cumulative)

Less Than Significant Effects:

The Project's cumulative contribution to energy would not be cumulatively considerable. Therefore, impacts related to cumulative energy impacts would be less than significant.

Facts in Support of Findings:

Electricity within the City is supplied by SCE. The geographic scope for cumulative electricity impacts is SCE's electricity service area. Development of the proposed Project and related projects could cumulatively increase demands on the existing electricity supply. However, each project will require a site-specific assessment to determine any impacts to existing and forecasted electricity supply. Specifically, all related projects would be required to assess construction and operational electricity usage and coordinate with SCE prior to project approval.

The proposed Project would include a 3.8-Megawatt Solar PV Panel system on the rooftops of the Cargo Sort Building and the Parking Structure. The proposed Project would also include the use and operation of electric-powered equipment. (see PDF AQ-3). Moreover, a portion of the proposed Project's aviation operations would include electric cargo planes (see Table 3.4 in Section 3.0: Project Description, of the Draft EIR and PDF AQ-4), for which charging stations would be provided in the southeast corner of the Project site. The civil operation of Alice Electric cargo aircraft is subject to FAA certification. For that reason, a footnote has been added in Table 3.4 in Section 3.0: Project Description in the Draft EIR to clarify that the inclusion of Alice Electric cargo aircraft as part of the proposed Project fleet is subject to its certification by the FAA. Electric charging stations would also be provided in the employee and visitor parking lots, and truckyard. As discussed above, the proposed Air Cargo Sort Building would meet LEED certification standards and would be all-electric (no natural gas usage). Further, like the proposed Project, other related projects would be required to incorporate energy conservation features in order to comply with applicable mandatory regulations including CALGreen and State energy standards in Title 24, and incorporate mitigation measures, as necessary. Therefore, the proposed Project's impacts related to the consumption of electricity would not be cumulatively considerable and its cumulative impact would be less than significant.

The geographic scope for cumulative transportation fuel impacts is the SCAG region. As discussed previously, the proposed Project would generate 450 additional truck trips per day, an amount that is within, and consistent with, the 2045 truck trip estimate for Ontario Airport under SCAG's RTP/SCS. As such, the proposed Project would accommodate the regional movement of goods per SCAG projections. The proposed Project also would incorporate various design elements to enhance the efficiency of fuel consumption. (see PDF AQ-2 through PDF AQ-5 and PDF AQ-8). In addition, during the operational lifetime of the proposed Project and related projects, newer vehicles sold on the market would be required to comply with the latest engine efficiency and fuel economy standards, which are reasonably expected and projected to incrementally take effect. Accordingly, fuel consumption is anticipated to decrease each year through implementation of regulation that require higher energy efficiencies and higher efficiency, alternative-fueled vehicles. Similarly, efforts made by the FAA and SFO to increase usage of alternative jet fuels are expected to occur during the lifetime of the proposed Project. Therefore, the proposed Project's impacts related to the consumption of transportation fuels would not be cumulatively considerable and its cumulative impact would be less than significant. (Draft EIR at pages 5.5-30 - 5.5-31).

5.8 Geology and Soils

Less Than Significant Effects: GEO-1(i.)

The Project is not expected to directly or indirectly cause potential substantial adverse effects, including the risk of loss, injury, or death involving rupture of a known earthquake fault, as delineated on the most recent Alquist-Priolo Earthquake Fault Zoning Map issued by the State Geologist for the area or based on other substantial evidence of a known fault. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

As indicated in the Geotechnical Study (see **Appendix 5.6-1**, of the Draft EIR), active or potentially active faults are not known to exist on or trend toward the Project site. There are several active faults surrounding the Project site to the north, east, south, and west, within the Upper Santa Ana River Valley. The Project site is not located within a designated Alquist-Priolo Earthquake Hazard Zone. The proposed Project would adhere to the appropriate engineering design measures as required by the latest Standard Specifications for Public Works Construction (Greenbook) and CBC. Therefore, impacts would be less than significant. (Draft EIR at pages 5.6-14 - 5.6-15).

Less Than Significant Effects: GEO-1(iv.)

The Project is not expected to directly or indirectly cause potential substantial adverse effects, including the risk of loss, injury, or death involving landslides. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

There is a potential for ground lurching due to an earthquake. Based on the California Department of Conservation Landslide Inventory, the Project site is not located in an area that is susceptible to landslides. As such, the potential for landslides at the Project site is very low. Therefore, impacts would be less than significant. (Draft EIR at pages 5.6-18 - 5.6-19).

No Effects: GEO-5

The Project is not expected to have soils incapable of adequately supporting the use of septic tanks or alternative wastewater disposal systems where sewers are not available for the disposal of wastewater. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

The proposed Project will connect to the City's sewer system and will not require the use of septic tanks or alternative wastewater disposal systems. Therefore, the proposed Project will have no construction or operational impacts with respect to site soils incapable of adequately supporting the use of septic tanks or alternative wastewater disposal systems. Therefore, impacts would be less than significant. (Draft EIR at page 5.6-27).

5.9 Hazards and Hazardous Materials

Less Than Significant Effects: HAZ-1

The Project is not expected to create a significant hazard to the public or the environment through the routine transport, use, or disposal of hazardous materials. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

The use, storage, transport, and disposal of construction and operation related hazardous materials would be required to conform to existing laws and regulations. Furthermore, strict adherence to all emergency response plan requirements set forth by San Bernardino County Fire Protection District (SBCFPD) and the Ontario Fire Department would be required through the duration of the proposed Project construction phase. The materials used during construction would not be in such quantities or stored in such a manner as to pose a significant safety hazard. Activities requiring use of hazardous materials would also be short term in nature or single-use instances and would cease upon completion of the proposed Project's construction phase. Project construction workers would also be trained in safe handling and hazardous materials use.

Project operations activities associated with the proposed Project would not involve the use of unusually high quantities of any materials identified as hazardous. Consistent with existing and former Airport operations for the delivery of fuel for aeronautical activities, Project operation would involve ground transport of fuels and other materials related to air cargo transport. Project operation would involve ground transport of fuels and other materials related to air cargo transport. For Phase 1, while the underground fuel pipeline and fueling hydrants are being completed, fuel trucks would be utilized to transport aircraft fuel from the existing Airport fuel farm to the proposed Project site. Aircraft fuel trucks would operate in compliance with the fueling operations and fuel spills rules set forth in the Ontario International Airport Rules and Regulations to minimize the risk of fuel release. Therefore, implementation of the proposed Project would result in less than significant impacts related to the routine transport, use, or disposal of hazardous materials; no mitigation is required. (Draft EIR pages 5.8-30 - 5.8-32).

Less Than Significant Effects: HAZ-3

The Project is not expected to emit hazardous emissions or handle hazardous or acutely hazardous materials, substances, or waste within one-quarter mile of an existing or proposed school. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

There are no schools located within 0.25 miles of the Project site. The nearest school to the Project site is Bon View Elementary School located approximately two (2) miles southwest. The proposed Project would not pose a significant risk of hazardous emissions or significant handling of hazardous materials or substances within one-quarter mile of an existing or proposed school. Therefore, impacts would be less than significant. (Draft EIR at page 5.8-39).

Less Than Significant Effects: HAZ-4

The Project is not expected to be located on a site which is included on a list of hazardous materials sites compiled pursuant to Government Code Section 65962.5 and, as a result, would it create a significant hazard to the public or the environment. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

The Project site is not included on a list of hazardous materials sites compiled pursuant to Government Code Section 65962.5 and would also not be affected or impacted by contamination identified in the general vicinity of the proposed Project site. For these reasons, the proposed Project would not create a significant hazard to the public or the environment. Impacts would be less than significant. (Draft EIR at pages 5.8-39 - 5.8-40).

Less Than Significant Effects: HAZ-6

The Project is not expected to impair implementation of or physically interfere with an adopted emergency response plan or emergency evacuation plan. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

The proposed Project is not located along interstates within the City that would serve as major emergency response and evacuation routes. During construction and long-term operation of the proposed Project, adequate emergency access for emergency vehicles would be maintained along public streets that abut the Project site. The proposed Project would not, therefore, impair implementation of or physically interfere with an adopted emergency response plan or emergency evacuation plan. Therefore, impacts would be less than significant. (Draft EIR at pages 5.8-43 - 5.8-44).

No Effects: HAZ-7

The Project is not expected to expose people or structures, either directly or indirectly, to a significant risk of loss, injury or death involving wildland fires. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

The Project site is in a Local Responsibility Area and classified by CAL FIRE as non-VHFHSZ (non-very high fire hazard severity zone). The site and surrounding areas are flat and developed with urban uses that would not contribute to the uncontrolled spread of wildfire or exacerbate potential wildfire risks, including downslope flooding and landslides caused by runoff, slope instability, or drainage changes from wildfire. Furthermore, as further discussed above, the proposed Project would not impair adopted emergency response and evaluation plans. Therefore, the proposed Project would not result in, or be subject to, significant effects related to wildfire risk. Therefore, no impacts would occur. (Draft EIR at page 5.8-44).

5.10 Hydrology

Less Than Significant Effects: HYD-1

The Project is not expected to violate any water quality standards or waste discharge requirements or otherwise substantially degrade surface or ground water quality. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

The proposed stormwater treatment system for the proposed Project would target and reduce pollutants of concern in runoff from the proposed Project site in compliance with the San Bernardino County MS4 permit requirements. Submittal and implementation of the SWPPP, and the erosion control plan prior to the construction phase of the proposed Project would address the potential for construction of the Project to affect water quality. The proposed Project would comply with all applicable regional and local water quality standards and waste discharge requirements as stated above in the Regulatory Setting, including the MS4 permit and NPDES permit. Compliance with the regulatory requirements and conditions of the San Bernardino County MS4 Permit as well as the Construction General Permit, including incorporation of operational BMPs to target pollutants of concern, would ensure that water quality impacts, degradation of water quality, increased pollutant discharge, alteration of receiving water quality, or impacts on surface water quality to marine, fresh, or wetland waters during Project operation would be less than significant. (Draft EIR at pages 5.9-21 - 5.9-27).

Less Than Significant Effects: HYD-2

The Project is not expected to substantially decrease groundwater supplies or interfere substantially with groundwater recharge such that the Project may impede sustainable groundwater management of the basin. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

A Water Supply Assessment (WSA) was prepared for the Project site to determine if the water demand during operation of the proposed Project would be sufficiently accommodated by the existing system within the City.4 The WSA concluded that the City would have sufficient water supplies available during normal, single dry, and multiple dry years through the year 2045 to meet all projected water demands associated with its existing and future customers, including the proposed Project. Additionally, there are no existing wells on the Project site and construction and operation of the proposed Project would not include groundwater extraction. For these reasons, the proposed Project will not impede sustainable groundwater management of the Chino Basin and Project impacts related to a decrease in groundwater supplies or interference with groundwater recharge would be less than significant. (Draft EIR at pages 5.9-27 - 5.9-29).

Less Than Significant Effects: HYD-3(i.)

The Project is not expected to substantially alter the existing drainage pattern of the site or area, including through the alteration of the course of a stream or river or through the addition of impervious surfaces, in a manner which would result in substantial erosion or siltation on- or off-site. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

The Construction General Permit requires preparation of a SWPPP. The SWPPP would detail erosion control and sediment control BMPs to be implemented during construction to minimize erosion and retain sediment on site. With compliance with the regulatory requirements and conditions of the Construction General Permit, and with implementation of the construction BMPs, construction impacts related to on-site, off-site, or downstream erosion or siltation would be less than significant. Furthermore, the collection, treatment, and controlled release of stormwater runoff in the proposed Project's planned underground water treatment facility to the drainage channels would ensure that runoff from the site does not remove significant amounts of sediment into the drainage channels and result in substantial erosion or siltation on the site. Impacts would be less than significant. (Draft EIR at pages 5.9-29 - 5.9-31).

Less Than Significant Effects: HYD-3(ii.)

The Project is not expected to substantially alter the existing drainage pattern of the site or area, including through the alteration of the course of a stream or river or through the addition of impervious surfaces, in a manner which would Substantially increase the rate or amount of surface runoff in a manner which would result in flooding on- or off-site. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

With the implementation of specified BMPs and detention features, the proposed Project would not substantially increase the rate or amount of surface runoff in a manner that would result in onor off-site flooding. Also, the site design LID features and on-site detention facilities would ensure that stormwater runoff does not exceed the capacity of the City's storm drain system, which includes the Airport. As the runoff from the Project site would be collected by the new Avion Street drainage facilities, the proposed Project would not result in or contribute to flooding. For these reasons, impacts to related to increase in runoff resulting in flooding would be less than significant. (Draft EIR at pages 5.9-31 - 5.9-33).

Less Than Significant Effects: HYD-3(iii.)

The Project is not expected to substantially alter the existing drainage pattern of the site or area, including through the alteration of the course of a stream or river or through the addition of impervious surfaces, in a manner which would create or contribute runoff water which would exceed the capacity of existing or planned stormwater drainage systems or provide substantial additional sources of polluted runoff. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

Project construction would comply with the requirements of the Construction General Permit and would include the preparation and implementation of a SWPPP and applicable BMPs. The incorporation of the proposed operational BMPs would allow the proposed Project to comply with San Bernardino County drainage requirements. Furthermore, on-site stormwater detention facilities including underground storage would be included in the proposed Project to reduce the amount of additional runoff into the new Avion Street drainage facilities. Operational impacts related to creation or contribution of runoff water that would exceed the capacity of existing, or

Ontario International Airport South Airport Cargo Center Project planned stormwater drainage systems or provide substantial additional sources of polluted runoff, would be less than significant. (Draft EIR at pages 5.9-33 - 5.9-34).

Less Than Significant Effects: HYD-3(iv.)

The Project is not expected to substantially alter the existing drainage pattern of the site or area, including through the alteration of the course of a stream or river or through the addition of impervious surfaces, in a manner which would Substantially increase the rate or amount of surface runoff in a manner which would Impede or redirect flood flows. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

The entire Project site would potentially be subject to inundation by 100-year storm floodwaters at depths of one foot or less. The proposed Project would be required to address these potential flood hazards as stated in Ontario Municipal Code Section 8-13.501: Standards of construction. Additionally, the proposed Project would include an underground stormwater detention and infiltration which would discharge stormwater at a controlled rate not greater than 24 cfs for the main portion of the Project site and 9 cfs for the portion of the Project site for the proposed parking garage (for the 100-year storm) into a new East Avion Street drainage system that will be completed prior to the opening of the proposed Project and into Cucamonga Channel. Based on these design conditions, the proposed Project impacts related to impeding or redirecting flood flows would be less than significant. (Draft EIR at pages 5.9-34 - 5.9-35).

Less Than Significant Effects: HYD-4

The Project is not expected to risk release of pollutants due to Project inundation. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

There are no open bodies of water in the vicinity of the Project site and the proposed Project is therefore not located within an inundation zone of a seiche. The Project site is located approximately 36 miles east of the Pacific Ocean and is not located within a tsunami inundation zone, according to the California Department of Water Resources. The proposed Project would also keep the storage of potentially hazardous materials on-site to a minimum, which would reduce the potential for hazardous materials to be released into surface water during flooding. With implementation of existing regulations to reduce flood hazards, risk of release of pollutants due to Project inundation would be less than significant. (Draft EIR at pages 5.9-35 - 5.9-36).

Less Than Significant Effects: HYD-5

The Project is not expected to conflict with or obstruct implementation of a water quality control plan or sustainable groundwater management plan. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

Adherence to the regulatory requirements and conditions of the State General Construction Permit, implementation of the SWPPP, and adherence to the City's Erosion and Sediment Control Plan requirements, would ensure that surface and groundwater quality are not adversely impacted

during construction. In addition, implementation of the LID and BMP measures at the site, including catch basins, underground detention, and sediment filtration chambers, would ensure that water quality would not be impacted during the operation of the proposed Project. As a result, site development would not obstruct or conflict with the implementation of the Santa Ana River Basin Water Quality Control Plan. The Project would not obstruct or conflict with the Chino Basin Optimum Basin Management Program (OBMP), applicable water quality control plans, or applicable sustainable groundwater management plans Therefore, impacts would be less than significant. (Draft EIR at page 5.9-36).

5.11 Hydrology (Cumulative)

Less Than Significant Effects:

The Project's cumulative contribution to hydrology would not be cumulatively considerable.

Facts in Support of Findings:

Cumulative projects within the Chino Basin Watershed could increase impervious areas and increase stormwater runoff rates. However, all projects within the watershed would be required to prepare and implement WQMPs that include provisions for the capture and infiltration of runoff or the temporary detention of stormwater runoff so that post-development runoff discharges do not exceed pre-development runoff rates, in accordance with the NPDES and MS4 permits. The Project would increase the area of impervious surface on site and increase the amount of localized runoff during a storm event. However, the peak flow rate would not substantially increase due to the proposed underground storage and infiltration chamber, which would reduce the peak flow rate to a maximum of 24 cfs for the main site and 9 cfs for the parking garage. With implementation of the required BMPs such as underground storage and filters, impacts related to a substantial increase in the rate or amount of surface runoff, flow, and volume that would result in flooding or reduced surface water quality, would be less than significant. Related projects have the potential to generate pollutants during project construction and operation. All construction projects that disturb one acre or more of land would be required to prepare and implement project-specific SWPPPs in order to obtain coverage under the Statewide GCP. All projects within the watershed would also be required to prepare and implement WQMPs specifying BMPs, including LID measures, which would be applied during project design and project operation to minimize water pollution from project operation. Compliance with these existing regulatory requirements will ensure that no significant cumulative impacts would result from the Project, related projects and other growth, and the Project's contribution to cumulative impacts will not be cumulatively considerable. (Draft EIR at page 5.9-37).

5.12 Noise

Less Than Significant Effects: N-1

The Project is not expected to generate a substantial temporary or permanent increase in ambient noise levels in the vicinity of the Project in excess of standards established in the local general plan or noise ordinance, or applicable standards of other agencies. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

Construction noise levels would not exceed the significance threshold at the nearby sensitive receptors. Additionally, roadway noise levels would not create a readily perceptible increase of 5 dBA or greater at locations where ambient noise levels are less than 60 dBA; a barely perceptible increase of 3 dBA or greater at locations where ambient noise levels range from 60 to 65 dBA; and community noise level impact increase of 1.5 dBA or greater at locations where ambient noise levels range from 60 to 65 dBA; and community noise level impact increase of 1.5 dBA or greater at locations where ambient noise levels are able to 65 dBA. (Draft EIR at pages 5.10-35 - 5.10-44).

Less Than Significant Effects: N-2

The Project is not expected to generate excessive groundborne vibration or groundborne noise levels. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

The forecasted vibration levels due to on-site construction activities would not exceed the strictest building damage significance threshold of 0.12 PPV ips for all Project-identified sensitive receptors due to distance, changes in elevations, and intervening structures. Based on FTA published vibration data, the anticipated ground vibration environment in the Project vicinity would be below the perceptible levels. As such, impacts related to building damage from operational groundborne vibration would be less than significant. (Draft EIR at pages 5.10-44 - 5.10-45).

5.13 **Public Services**

Less Than Significant Effects: PUB-1(i.)

The Project is not expected to result in substantial adverse physical impacts associated with the provision of new or physically altered governmental facilities, need for new or physically altered governmental facilities, the construction of which could cause significant environmental impacts, in order to maintain acceptable service ratios, response times or other performance objectives for fire protection. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

The existing Ontario Fire Department (OFD) facilities, Station 10, meet current and future needs for fire protection services, including the needs of the proposed Project. Due to Station 10's proximity to the Project site, a potential response to the Project site would be less than three minutes. The existing fire protection equipment and services offered at Station 10 are sufficient to accommodate the proposed Project. Demolition and construction activities would comply with all applicable California Fire Code requirements.

During operation, the primary need for fire services at the Project site would relate to fires and potential incidents involving hazardous materials by aircraft ground operations, aircraft fueling, the storage of cleaning and maintenance materials, and the handling of cargo within the facility. The proposed structures would be built to current fire codes and standards, and would have fire extinguishers, wet and dry sprinkler systems, pre-action sprinkler systems, fire alarm systems, fire pumps, backflow devices, and clean agent waterless fire suppression systems pursuant to the

California Fire Code, CBC, City of Ontario Fire Code, OIAA, and other applicable regulations regarding fire safety. (Draft EIR at pages 5.11-14 - 5.11-17).

Less Than Significant Effects: PUB-1(ii.)

The Project is not expected to result in substantial adverse physical impacts associated with the provision of new or physically altered governmental facilities, need for new or physically altered governmental facilities, the construction of which could cause significant environmental impacts, in order to maintain acceptable service ratios, response times or other performance objectives for police protection. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

The Ontario Police Department (OPD) currently patrols the Project site for suspicious persons and trespassing. The Airport Operations Bureau (AOB) would respond to calls for service requiring a police response. The response time to the Project site would vary by type of call and location of OPD officers. Emergency calls would have officers at the site within one to ten minutes. Non-emergency calls are immediately responded to if there are available officers. During construction, the entire construction area would be fenced off. No access would be allowed into the airfield and other secured Airport areas from the construction site, and access in and out would be limited to one to two access points that would be gated and secured by a security guard. Once constructed, the property would be fully secured, with limited access into the Air Cargo Sort Building. The Air Cargo Sort Building would also include areas for facility security, administered by TSA, FAA, and OIAA. Fencing would be installed along the perimeter of the property in accordance with airport standards. The entire Project site, including the interior and exterior of the cargo building and parking garage on the south side of East Avion Street would be installed with security cameras, alarm systems, and adequate lighting for operations during the day and nighttime security.

Construction of Phase 2 of the proposed Project would require the relocation of the AOB K-9 Substation, currently located in the OIAA administrative offices on East Avion Street, to a vacant hangar on the north side of the Airport prior to the start of Phase 2. The relocation of the K9 substation would not impact response times, which would remain between 1 and 10 minutes. The relocation to the vacant hangar would not result in a substantial adverse physical impact. Therefore, impacts would be less than significant, and no mitigation is required. (Draft EIR at pages 5.11-17 – 5.11-19).

5.14 Public Services (Cumulative)

Less Than Significant Effects:

The Project's cumulative contribution to public services would not be cumulatively considerable.

Facts in Support of Findings:

The geographic area for cumulative impact analysis is the service areas of OFD and OPD. As discussed above, the proposed Project would not significantly impact OFD and OPD facilities or reduce their existing service ratios, staffing levels, or performance objectives, which could result in the need for new facilities or the expansion of existing facilities for which environmental impact analysis would be required. If the City determines that new facilities are necessary at some point in the future, such facilities (1) would occur where allowed under the designated land use, (2)

Ontario International Airport South Airport Cargo Center Project would be expected to be located on parcels that are infill opportunities on lots that are typically between approximately 0.5 to 2 acres in size, and (3) would likely qualify for a Categorical Exemption under CEQA Guidelines Section 15301 or 15332, Negative Declaration or Mitigated Negative Declaration, and would not be expected to result in significant impacts. Accordingly, the potential need for additional fire protection services is not an environmental impact that the Project would be required to mitigate. Therefore, no significant cumulative impacts will result from the Project, related projects, and other growth, and the Project's contribution to cumulative impacts will not be cumulatively considerable. (Draft EIR at page 5.11-20).

5.15 Transportation

Less Than Significant Effects: TRA-1

The Project is not expected to conflict with a program, plan, ordinance or policy addressing the circulation system, including transit, roadway, bicycle and pedestrian facilities. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

Truck trips during construction would comply with truck route requirements identified within the Ontario Plan. Construction of the proposed Project would not conflict with any program, plan, ordinance, or policy related to roadway facilities The addition of proposed Project traffic is forecast to add delay to one intersection, Archibald Avenue at Mission Boulevard, under the Opening Year (2025) Plus Phase 1 Project Conditions scenario. Improvements, consisting of adding a dedicated left-turn pocket for the southbound approach with protected left-turn phasing for the northbound and southbound left-turn phases are identified that will improve operating conditions at this intersection to better than pre-Project conditions. Additionally, Archibald Avenue is programmed in the SCAG RTP to be widened to six lanes in each direction which is greater than the improvements identified as needed to improve this intersection to better than pre-Project conditions. As such, the proposed Project would not conflict with any standard related to roadway facilities or services under Phase 1 Opening Year (2025) Conditions with the implementation of recommended roadway improvements. Roadway facilities improvements to Intersection 1, Euclid Ave/SR-83 at Mission Boulevard and Archibald Avenue at Mission Boulevard, would occur as part of the proposed Project to be completed by Phase 2 Opening Year (2029). The improvements would optimize signal timing, improving intersection operations to better than pre-Project conditions, consistent with the Ontario Plan and CMP requirements related to LOS at Intersection 1, Euclid Ave/SR-83. The improvements for the intersection of Archibald Avenue at Mission Boulevard recommended for Opening Year (2025) Plus Phase 1 Project Conditions would also improve Opening Year (2029) Plus Phase 1 and Phase 2 Project Conditions to better than pre-Project conditions during both peak hours. Additionally, Cumulative Year (2040) roadway improvements, as part of the proposed Project, include intersection realignments and widening. Intersection 1, Euclid Avenue/SR-83 at Mission Boulevard, Intersection 5, Grove Avenue at Mission Boulevard, and Intersection 26, Airport Drive at Haven Avenue, would include lane configurations that would improve intersection operations to acceptable conditions. Storage capacities for all SR-60, I-10, and I-15 off ramps in the Study Area do not exceed the storage capacity defined by Caltrans (see Appendix 5.12-1, of the Draft EIR). As such, the proposed Project would not conflict with any program, plan, ordinance, or policy related to roadway facilities or services.

Construction of the proposed Project may result in temporary effects on adjacent streets, including effects from any temporary lane closures needed. Transit facilities would not substantially change during construction of the proposed Project. Construction of the proposed Project would not conflict with a conflict program, plan, ordinance, or policy related to transit facilities. The proposed Project would not substantially change or eliminate bus facilities or transit routes, nor would it conflict with a policy or program related to transit access.

Bicycle facilities are not proposed and would not change as part of the proposed Project. Temporary construction impacts to bicycle facilities may occur during construction as a result of potential lane closures for roadway improvements. Construction of the proposed Project would not conflict with any program, plan, ordinance, or policy related to bicycle facilities. The proposed Project does not include any changes to proposed or existing bicycle facilities. The proposed Project would not conflict with any existing or planned bicycle facilities. The proposed Project is consistent with the adopted plans regarding bicycle facilities and is not expected to decrease the performance or safety of these facilities.

Pedestrian facilities would have temporary construction impacts during construction as a result of potential sidewalk closures for roadway improvements. Construction of the proposed Project would not conflict with any program, plan, ordinance, or policy related to pedestrian facilities. There are no proposed pedestrian facilities on Avion Street or Avion Drive outside the proposed Project area. The proposed Project would not conflict with any existing or planned pedestrian and bicycle facilities. The proposed Project is consistent with the adopted plans regarding pedestrian facilities and is not expected to decrease the performance or safety of these facilities. (Draft EIR at pages 5.12-33 - 5.12-60).

Less Than Significant Effects: TRA-3

The Project is not expected to substantially increase hazards due to a geometric design feature or incompatible uses. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

The proposed Project includes the realignment of and widening of arterial roadways and intersections. The existing roadway network consists of industrial-scaled, block-defining thoroughfares that enable goods movements to and from the Project site and functions well for pedestrians, bicyclists, drivers, transit users, and those operating emergency vehicles. The proposed roadway network identifies access points on the surrounding streets at appropriate locations that would not create any hazards. This includes new driveways to access the proposed Project along East Avion Street. All roadway and driveway improvements would comply with federal, State, and local design and safety standards. All driveway access points are perpendicular to the public right-of-way and adequately spaced from existing signalized intersections. No pedestrian or bicycle facilities are proposed for East Avion Street. All sidewalk and crosswalk improvements as a result of roadway improvements would comply with federal, State, and local design and safety standards. Further, the proposed air cargo facility uses are consistent with surrounding uses. Phase 2 Opening Year (2029) would include roadway improvements to Intersection 1, Euclid Avenue/SR-83 at Mission Boulevard. The improvements for the intersection of Archibald Avenue at Mission Boulevard recommended for Opening Year

(2025) Plus Phase 1 Project Conditions would also improve Opening Year (2029) Plus Phase 1 and Phase 2 Project Conditions to better than pre-Project conditions during both peak hours.

Cumulative Year (2040) roadway improvements, as part of the proposed Project, include intersection realignments and widening. Intersection 1, Euclid Avenue/SR-83 at Mission Boulevard, Intersection 5, Grove Avenue at Mission Boulevard, and Intersection 26, Airport Drive at Haven Avenue, would include lane configurations that would improve intersection operations to acceptable conditions. Accordingly, the Proposed Project would not create or substantially increase safety hazards due to a design feature or incompatible use. The proposed Project does not increase hazards due to a geometric design feature. (Draft EIR at pages 5.12-66 - 5.12-67).

Less Than Significant Effects: TRA-4

The Project is not expected to result in inadequate emergency access. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

No hazards would be associated with construction of the proposed Project. All proposed Projectrelated construction traffic would be required to comply with a temporary traffic control plan that meets the applicable requirements of the California Manual on Uniform Traffic Control Devices. The proposed Project would maintain adequate emergency access during construction. Primary access to the proposed Project area is proposed from East Avion Street. The proposed Project would provide emergency access on East Avion Street to major arterials Archibald Avenue, Jurupa Street, and Vineyard Avenue. The location and design of these access points would be adequate for emergency access. The proposed roadway network improvements would not result in inadequate emergency access to the site and would not impede existing emergency access to the existing surrounding uses. (Draft EIR at page 5.12-68).

5.16 Utilities

Less Than Significant Effects: U-1

The Project is not expected to require or result in the relocation or construction of new or expanded water, wastewater treatment or storm water drainage, electric power, natural gas, or telecommunications facilities, the construction or relocation of which could cause significant environmental effects. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

Short-term construction activities would require minimal water and are not expected to have any adverse impacts on the existing water system or available water supplies. During operation, the Air Cargo Sort Building would be connected to the existing 16-inch water main along East Avion Street. Water would be supplied to the Air Cargo Sort Building, parking garage, and aircraft apron, for consumption as well as fire suppression. The projected water demand for the Project site in the 2020 UWMP is sufficient to account for the water needed for the Project. The proposed Project would not require the construction of new or expanded water conveyance, treatment, or collection facilities. The impacts on water facilities during construction and operation would be less than significant, and no mitigation is required.

Additionally, based on the available sewer line and wastewater treatment capacity, the proposed Project would not require the construction of new or expanded water conveyance, treatment, or collection facilities, and impacts would be less than significant. the Project would implement LID features and stormwater effluent from the Project site during construction and operation, which would be stored and discharged at a controlled rate (not greater than 24 cfs for the main portion of the Project site and 9 cfs for the portion of the Project site for the proposed parking garage), the proposed Project would not require the construction of new or expanded water conveyance, treatment, or collection facilities and impacts would be less than significant. Phase 1 of the proposed Project would require approximately 8.5 MW of power. Phase 2 of the proposed Project would require approximately 2.85 MW of power at buildout. An additional 10 percent of other miscellaneous loads is needed for the proposed Project. At full development, the proposed Project would require approximately 12.4 MW of power. A new substation is being planned by SCE, as a part of the proposed Project, to meet the need for additional power for the proposed Project. This 135- foot by 160-foot proposed substation will be located on previously disturbed areas within the Project site. The Air Cargo Sort Building would not utilize natural gas. Therefore, the proposed Project would not require the construction of new or expanded natural gas facilities and impacts would be less than significant. Construction and operation of the proposed Project would not necessitate the construction of off-site telecommunication facilities that would have the potential to cause significant environmental impacts. The proposed Project would not require the construction of new or expanded telecommunications facilities and impacts would be less than significant. (Draft EIR at pages 5.14-21 - 5.14-27).

Less Than Significant Effects: U-2

The Project is not expected to have sufficient water supplies available to serve the Project and reasonably foreseeable future development during normal, dry and multiple dry years. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

Based on the information provided in the 2020 UWMP and Project-specific water demand, the OMUC's projected water supplies will be sufficient to satisfy the demands of the proposed Project, in addition to existing and planned future uses under normal, dry, and multiple dry water years. Therefore, impacts would be less than significant. (Draft EIR at pages 5.14-27 - 5.14-30).

Less Than Significant Effects: U-3

The Project is not expected to result in a determination by the wastewater treatment provider which serves or may serve the Project that it has adequate capacity to serve the Project's Projected demand in addition to the provider's existing commitments. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

The proposed Project would constitute approximately 0.28 percent of the total daily wastewater capacity for Regional Plant 1. Considering this facility is already operating below its maximum capacity, the proposed Project would not have significant effect on the processing capacity. Therefore, the proposed Project would not require the construction of new wastewater treatment

facilities or expansion of facilities, the construction of which could cause significant environmental effects and impacts would be less than significant. (Draft EIR at page 5.14-30).

Less Than Significant Effects: U-4

The Project is not expected to generate solid waste in excess of State or local standards, or in excess of the capacity of local infrastructure, or otherwise impair the attainment of solid waste reduction goals. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

Phase 1 demolition would generate approximately 192,484 square feet of building debris and 2,047,320 square feet of concrete and asphalt paving. Phase 2 demolition would generate approximately 432,295 square feet of building debris and approximately 1,045,440 square feet of concrete and asphalt paving. The building debris would need to be removed and disposed of offsite. The concrete and asphalt paving debris would be recycled for use on the site. It is expected that all pavement found suitable for recycling and reuse would be recycled on-site.

Demolition and disposal of demolition debris would be conducted in accordance with applicable laws and regulations, including Ontario Municipal Code Section 6-3.602, Construction & Demolition Recycling Plans, and the 2019 California Green Building Standards Code with regard to the diversion of recyclable material away from landfills, as well as South Coast Air Quality Management District Rule 403 regarding the generation of fugitive dust at construction sites. The proposed Project will also meet the City's current and future recycling goals during operation and meet the City's waste management ordinance to divert at least 65 percent of potential waste disposal. As such, the proposed Project would not create a significant impact on solid waste generation. (Draft EIR at pages 5.14-30 - 5.14-32).

Less Than Significant Effects: U-5

The Project is expected to comply with federal, state, and local management and reduction statutes and regulations related to solid waste. Therefore, impacts would be less than significant, and no mitigation is required.

Facts in Support of Findings:

The proposed Project would be consistent with the applicable regulations associated with solid waste. The proposed Project would also comply with AB 939, AB 341, AB 1826, SB 1383, and City waste diversion goals as presented in the Ontario Municipal Code, as applicable. Since the proposed Project would comply with federal, State, and local statutes and regulations related to solid waste, impacts would be less than significant. (Draft EIR at pages 5.14-32 – 5.14-33).

5.17 Utilities (Cumulative)

Less Than Significant Effects:

The Project's cumulative contribution to utilities would not be cumulatively considerable.

Facts in Support of Findings:

Compliance of the proposed Project and related projects with regulatory requirements that promote water conservation such as the Ontario Municipal Code, the California Green Building Code, as well as AB 32, would assist in ensuring that adequate water supply is available on a cumulative basis. According to the City's 2020 UWMP, the City has adequate supplies to serve 100 percent of its customers during normal, dry year, and multiple dry year demand through 2045 accounting for projected population increases and corresponding increases in water demand. Projected water demand for the Project was included in the 2020 UWMP projections based on the General Plan land use designation for "Industrial" uses. The Project would consist of 857,762 square feet of warehouse and office space in the Air Cargo Sort Building. The remainder of the site acreage would consist of aircraft uses and truckyard and visitor parking. The projected water demand for the Project is 0.48 percent of the water demand for the land use that was accounted for in the 2020 UWMP.

Wastewater generated by the proposed Project would be processed at the RP-1 treatment plant. RP-1 treats an average influent of wastewater flow of approximately 28 mgd. The hydraulic design can treat domestic sewage of up to 44 mgd and 60 mgd of solids. Moreover, this facility is currently operating below its maximum capacity. The proposed Project's wastewater would represent 0.28 percent of the total daily capacity for RP-1.

The proposed Project would meet applicable LID requirements and would meet these standards by retaining and treating all stormwater on the site prior to discharge. As a result, the amount of peak stormwater flows from the Project site will decrease from existing conditions. All other new development would decrease as compared to older sites that did not include recent LID requirements.

Solid waste disposal is addressed by the County of San Bernardino. The County promotes the efforts of individual jurisdictions to maximize waste reduction and recycling, expand existing landfills, and promote alternative technologies to reduce waste. In response to State-mandated waste reduction goals set forth in CalGreen, and as part of the City's commitment to sustainable development, the City adopted an ordinance that requires certain demolition and/or construction projects to divert at least 65 percent of waste either through recycling, salvage, or deconstruction. The El Sobrante Landfill would serve the proposed Project's solid waste generation with a maximum capacity of 10,000 tons per day. SB 1383 establishes statewide greenhouse gas emission reduction goals consisting of reducing the amount of organic material disposed in landfills by 50% from the 2014 level by the year 2020 and reducing the amount of organic material disposed in landfills by 75% from the 2014 level by the year 2025. Detailed components regarding waste reduction and recycling would be finalized for each related project on a project-by-project basis at the time of plan submittal to the City for the necessary building permits and reviews conducted pursuant to the California Green Building Code, as applicable.

During construction and operation, other future related projects would be required to incorporate energy conservation features, comply with applicable regulations including anti-idling construction vehicle regulations, the 2019 Title 24 standards and CALGreen code, and incorporate mitigation measures, as necessary. Natural gas infrastructure is typically expanded in response to increasing demand, and system expansion and improvements by SoCalGas occur as needed. Related projects within its service area would also be anticipated to incorporate site-specific

Ontario International Airport South Airport Cargo Center Project
infrastructure improvements, as appropriate, and incorporate mitigation measures, as necessary. The Air Cargo Sort Building would not utilize natural gas.

During construction and operation, related projects would be required to incorporate energy conservation features, comply with applicable regulations including anti-idling construction vehicle regulations, the 2019 Title 24 standards and CALGreen code, and incorporate mitigation measures, as necessary. A new substation is being planned by SCE to meet the need for additional power for the proposed Project. This 135-foot by 160-foot proposed substation will be located within the Project site to the west of the proposed parking structure. The new substation would connect to existing infrastructure along Mission Boulevard directly south of the Project site. It is not anticipated that development of this new substation will result in any significant environmental effects as it would be sited on previously disturbed areas within the Project site and within the development footprint and profile of other Project components.

Telecommunications are regulated by the Federal Communications Commission and the California Public Utilities Commission. No significant cumulative impacts will result from the Project, cumulatively considered projects and other growth, and the Project's contribution to cumulative impacts will not be cumulatively considerable. (Draft EIR at pages 5.14-33 - 5.14-35).

5.18 Issues Deemed No Impact or Less than Significant Impact in the Initial Study/Notice of Preparation

In accordance with section 15063 of the CEQA Guidelines, OIAA prepared an Initial Study/Environmental Checklist for the Project and distributed it along with the Notice of Preparation (NOP) for the Draft EIR to responsible and interested agencies, and key interest groups. In preparing the Initial Study/Notice of Preparation (IS/NOP), which is **Appendix 1.0** to the Draft EIR, OIAA determined the following issues not to be significant; and, in accordance with section 15128 of the Guidelines, they did not receive further evaluation in the Draft EIR:

- Agricultural and Forestry Resources: The Project site is not designated farmland or under a Williamson Act contract. The California Department of Conservation designates the site Urban and Built-Up Land. The site is currently developed with, and surrounded by, airport related and industrial uses. Due to its location within the Airport and its developed condition, the Project site is not suitable for agricultural and timberland production. Based on these characteristics, the proposed Project would not impact agricultural and timberland resources.
- Land Use/Planning: The proposed Project has been designed and would operate in accordance with OIAA rules and regulations, and as an aeronautical development and use under OIAA's jurisdiction. Also, the City of Ontario General Plan land use designation for the Project site is Airport, and the site is zoned ONT for Ontario International Airport. The ONT zoning district allows airport terminals (including commercial and service uses related to the terminals), car rental agencies, and airport-related industrial and delivery uses, at a maximum intensity of 0.55 floor to area ratio (FAR). The Project proposes 1,261,712 square feet of buildings and structures on the 97-acre site, which results in a FAR of approximately 0.34, under the maximum allowed intensity of 0.55 FAR.

The proposed Project is also required to comply with the FAA-approved Ontario International Airport Layout Plan (ALP). The ALP serves as a guide for the Airport's future development and identifies the Project site as "Airport Development Area," which is the designation for future development.6 The proposed Project would be consistent with adjacent Airport and industrial uses, as well as applicable OIAA, City of Ontario, and FAA-adopted plans, policies, and regulations. The proposed Project would not physically divide an established community or result in off-site land use changes. Accordingly, the proposed Project would have less-than-significant effects related to Land Use and Planning.

- **Mineral Resources**: According to the Ontario General Plan Final EIR, the City contains no mineral resources of Statewide significance. There are, however, a few sites in the City, the closest of which is approximately one mile to the north, which contain regionally significant mineral resources deposited by the Deer and Day Creek alluvial fan with potential aggregate resources, commonly known as gravel. Project implementation would not impact these sites or result in the loss of regionally and locally important mineral resources. Based on this information, the proposed Project would have a less than significant effect on mineral resources.
- **Population/Housing**: The Project site contains airport office buildings, hangars, and support facilities. There are no residences on the Project site. Project implementation would not displace people or result in the demolition of existing housing that would require the construction of replacement housing elsewhere. The proposed Project would not impact housing stock. The proposed Project includes utility improvements; however, these would be designed to serve Project operations and would not directly or indirectly result in unplanned population growth.

The proposed Project would increase employment opportunities in the region. The proposed Project would create approximately 1,315 jobs. According to the US Bureau of Labor Statistics, in June 2021, there was an unemployment rate of 7.9 percent (approximately 165,600 people were unemployed) in the Riverside-San Bernardino-Ontario area. Accordingly, the 1,315 jobs generated by the proposed Project can employ existing residents in the Riverside-San Bernardino-Ontario area; thus, the proposed Project would not trigger the need for new housing. The proposed Project would result in less-than-significant effects related to population and housing.

- **Parks/Recreation**: The City of Ontario contains a variety of recreational opportunities, including regional and City parks, school recreation facilities, private parks and golf courses, and recreational trails for bicycles, horses, and hiking. Park and recreation facilities closest to the Project site include a bicycle corridor along Mission Boulevard and the Cucamonga Creek Multipurpose Trail. Project construction and operation would not directly affect these or other recreation facilities. Therefore, the proposed Project would not result in significant effects on parks and recreation facilities.
- **Public Services** (Schools and Other Public Facilities. Potential impacts to Fire and Police Public Services are discussed in Section 5.11 of the Draft EIR): The Project site is within the boundaries of the Ontario-Montclair Elementary School District and Chaffey Joint Union High School District. The proposed Project does not include residential

development and would not generate students that would need to be housed at public school facilities. Nevertheless, the Project would comply with applicable laws and regulations, including the payment of school impact fees for the proposed commercial/industrial development that would reduce potential impacts to school facilities to less than significant. The Project would not require any other government services, such as library and public health services; therefore, potential effects related to other public facilities would be less than significant.

• Wildfire: The Project site is in a Local Responsibility Area and classified by CAL FIRE as non-VHFHSZ (non-very high fire hazard severity zone). The site and surrounding areas are flat and developed with urban uses that would not contribute to the uncontrolled spread of wildfire or exacerbate potential wildfire risks, including downslope flooding and landslides caused by runoff, slope instability, or drainage changes from wildfire. Furthermore, as further discussed in HAZ-6, Section 5.8: Hazards and Hazardous Materials, of the Draft EIR, the proposed Project would not impair adopted emergency response and evaluation plans. Therefore, the proposed Project would not result in, or be subject to, significant effects related to wildfire risk.

Reference:

Additional detail on Effects Found Not to be Significant can be found in the Draft EIR at pages 7.0-1 through 7.0-4 and in Appendix 1.0).

6.0 FINDINGS REGARDING PROJECT ALTERNATIVES

CEQA requires that an EIR analyze a reasonable range of feasible alternatives that could substantially reduce or avoid the significant impacts of a project while also meeting the project's basic objectives. An EIR must identify ways to substantially reduce or avoid the significant effects that a project may have on the environment (PRC Section 21002.1). Section 15126.6(a) of the CEQA Guidelines requires the discussion of "a reasonable range of alternatives to a project, or the location of a project, which would feasibly attain most of the basic objectives of the project but would avoid or substantially lessen any of the significant effects of the project and evaluate the comparative merits of the alternatives." Accordingly, the discussion of alternatives shall focus on alternatives to a project or its location which are capable of avoiding or substantially reducing any significant effects of the project, even if these alternatives would impede to some degree the attainment of the project objectives, or would be more costly. The alternative analysis included in the Draft EIR, therefore, identified a reasonable range of Project alternatives focused on avoiding or substantially reducing the Project's significant impacts.

Three alternatives, including the No Project Alternative, were analyzed in detail in the Draft EIR, Chapter 8, Project Alternatives: (1) No Project/No Development Alternative; (2) Reduced Project Size Alternative; and (3) Different Location on Airport Alternative.

These alternatives have been evaluated for their ability to (i) substantially lessen the significant impacts of the Project identified in the EIR, and (ii) substantially meet most of the basic objectives of the Project as described in the EIR. OIAA has determined that the alternatives listed above and evaluated in the EIR constitute a reasonable range of alternatives consistent with CEQA Guidelines Section 15126.6. In addition, the EIR considered but dismissed one alternative: (1) Alternative

Airport Locations. Due to the failure to meet airfield infrastructure operational criteria, the location of the proposed Project at other airports in the region was determined to be infeasible and this alternative was eliminated.

These Findings contrast and compare the Alternatives, where appropriate, to show that the selection of Project, while still resulting in significant environmental impacts, has substantial overriding economic, legal, social, technological, and other benefits. In rejecting the Project alternatives, OIAA has reviewed the environmental impacts and the Project objectives and weighed the ability of the various alternatives to meet the objectives. OIAA finds, after due consideration of a reasonable range of alternatives as set forth in the EIR and below, and based on substantial evidence in the record, that the Project best attains a balance between protecting against local environmental impacts and best meets the approved Project objectives with the least environmental impact.

Section 15126.6(a) of the CEQA Guidelines requires the discussion of "a reasonable range of alternatives to a project, or the location of a project, which would feasibly attain most of the basic objectives of the project but would avoid or substantially lessen any of the significant effects of the project and evaluate the comparative merits of the alternatives."

6.1 Alternative 1: No Project/No Development Alternative

Description of Alternative:

Section 15126.6(e) of the CEQA Guidelines state: "the No Project/No Build Alternative means 'no build' wherein the existing environmental setting is maintained." Accordingly, for purposes of this analysis, Alternative 1, the No Project/No Development Alternative (Alternative 1), assumes the proposed Project is not built and the existing airport-related buildings located on the Project site, which includes hangars, ancillary structures, and related parking facilities and site improvements, would remain. Existing leases and non-OIAA tenant operations would continue to operate on the Project site and no relocation of these existing uses would occur.

Impact Summary:

The No Project/No Build Alternative would avoid all significant impacts identified for the proposed Project. The No Project Alternative would not, however, achieve any of the objectives of the proposed Project.

Finding:

Pursuant to Public Resources Code Section 21081(a)(3), the OIAA finds that specific economic, legal, social, technological, other considerations, including considerations for the provision of employment opportunities for highly trained workers, make Alternative 1 infeasible.

Facts in Support of Finding:

Alternative 1 would avoid the significant impacts for the Project. However, it would not meet any of the Project's objectives or the underlying purpose to develop and operate an air cargo facility at

the Airport to meet increased regional air cargo volumes and Project proponent facility requirements. Therefore, for these reasons, Alternative 1 is less desirable than the Project.

Reference:

For a complete discussion of impacts associated with Alternative 1, see Section 6.5 of the Draft EIR at pages 6.0-8 - 6.0-13.

6.2 Alternative 2: Reduced Project Size Alternative

Description of Alternative:

Alternative 2 considers reducing the size of the proposed Project to reduce the significant and unavoidable transportation (VMT), operational air quality, and greenhouse gas emission (GHG) impacts identified for the Project as proposed. The proposed Project would result in an increase in the number of annual aviation operations at the Airport. In 2029, with completion of Phase 2, the proposed Project would include up to 33 daily departures and arrivals (66 total aircraft operations) with up to 17 daytime (7:00 AM–6:59 PM) departures and 20 daytime arrivals, and 3 evening (7:00 PM–9:59 PM) departures. In addition, the proposed Project would accommodate 3 evening arrivals, 13 nighttime (10:00 PM–6:59 AM) departures, and 10 nighttime arrivals. Truck operations would occur daily, primarily coinciding with the arrival and departure times of the scheduled flights. At proposed Project buildout, the proposed facility would operate with 1,315 employees. As discussed in 6.2 Significant Impacts of the Ontario Airport South Airport Cargo Center Project, above, with the total aircraft operations, truck operations, and employee trips under both Phase 1 and Phase 2, the proposed Project would result in significant unavoidable VMT, operational air quality, and GHG impacts.

The proposed Project is an air cargo facility serving a large region, and the operational and economic viability of the proposed Project relies on truck and aviation operations. For these reasons, it is not feasible to modify the proposed Project by reducing the size of the proposed Project and the associated activities could sufficiently reduce the proposed Project's VMT, operational Air Quality emissions and GHG emissions, to a less than significant level. A reduction in the size of the proposed Project could, however, result in a meaningful reduction in these impacts.

Alternative 2 considers construction and operation of only Phase 1 of the proposed Project. This would include the demolition of existing structures and site improvements in the Phase 1 area, site preparation, and construction of all proposed improvements on the eastern 60 acres of the Project site, including the Air Cargo Sort Building (611,158 square feet on six (6) acres), aircraft apron improvements and GSE support (47 acres), truckyard and visitor parking (five (5) acres), and employee parking garage (four (4) acres).

Aircraft operations would include up to 22 daily arrivals and departures with a maximum of 44 total daily aircraft operations. In 2025, it is anticipated aircraft operations would occur seven days per week, with up to 8 daytime (7:00 AM–6:59 PM) departures and 9 daytime arrivals, 1 evening (7:00 PM–9:59 PM) departure and 3 evening arrivals, and 13 nighttime (10:00 PM–6:59 AM) departures and 10 nighttime arrivals.

Construction of Alternative 2 would start in the third quarter of 2023 and be completed by the third quarter of 2025 when the proposed air cargo flight operations at the Airport would begin. Construction would include the demolition of existing structures and site improvements in the Phase 1 area, site preparation and grading, and construction of all proposed improvements under Phase I.

Impact Summary:

This Alternative would substantially lessen the unavoidable significant air quality and greenhouse gas impacts, and incrementally reduce the VMT impacts identified for the proposed Project. While reduced, these impacts would remain significant after implementation of all feasible mitigation.

Finding:

Pursuant to Public Resources Code Section 21081(a)(3), the OIAA finds that specific economic, legal, social, technological, other considerations, including considerations for the provision of employment opportunities for highly trained workers, make Alternative 2 infeasible.

Facts in Support of Finding:

Development of only Phase 1 of the proposed Project would also not meet the objectives of the proposed Project to accommodate current and projected air cargo volume growth, and would only partially meet the objectives of redeveloping and maximizing revenue for the OIAA from underutilized Airport property.

Additionally, mitigation measures are identified for the proposed Project to avoid the potential for significant noise impacts generated by operation of the proposed facility that would also apply to this alternative. As the proposed Project would result in less than significant impacts with mitigation, and Alternative 2 would not. Therefore, avoiding or substantially lessening any significant noise impacts that would result from the proposed Project could be done with the Project and not with Alternative 2.

With the VMT impacts of Phase 1 of the proposed Project, as Alternative 2 would be incrementally less than those associated with Phase 1 and Phase 2 combined, this impact would remain significant with mitigation. Alternative 2 would therefore not avoid or substantially lessen the significant transportation impacts identified for the proposed Project.

Mitigation measures identified for the proposed Project to avoid the potential for significant impacts to any subsurface tribal cultural resources that may be encountered during construction would apply to Phase 1 and to Alternative 2. Alternative 2 would therefore not avoid or substantially lessen any significant impacts to TCRs.

Mitigation measures are identified for the proposed Project, and would be similarly utilized in Alternative 2, to avoid significant impacts related to hazards and hazardous materials during construction and operational activities. As the proposed Project would result in less than significant impacts with mitigation, Alternative 2 would not, therefore, avoid or substantially lessen any significant hazards and hazardous materials impacts that would occur with the proposed Project.

Reducing Phase 1 operational emissions from aviation operations to a less than significant level would not be feasible and Alternative 2 would result in the same operational emission from aviation operations. While Alternative 2 would substantially lessen emissions and avoid the significant operational SO2 impact under Phase 2 of the proposed Project, operational air quality impacts would remain significant.

Reference

For a complete discussion of impacts associated with Alternative 2, see Section 6.5 of the Draft EIR at pages 6.0-13 - 6.0-26.

6.3 Alternative 3: Different Location on Airport Alternative

Description of Alternative:

Under this alternative, the proposed Project would be constructed and operate on a site located on the northwest edge of the Airport. This site provides a contiguous land area of approximately 90 acres in size. The site would provide direct airfield access to support the international and domestic cargo aircraft for the proposed Project. The location of Alternative 3 would provide the airfield infrastructure to support the operational needs of the proposed Project, including access to two runways, one at least 12,000 feet in length and one no less than 10,000 feet in length, with at least one runway with CAT III approach capability to accommodate air cargo aircraft fleet mix. This location at the Airport also has connections via the surrounding street network to the I-10, SR-60, and I-15 Freeways.

Impact Summary:

This Alternative would not avoid or substantially lessen any significant impacts when compared to the proposed Project.

Finding:

Pursuant to Public Resources Code Section 21081(a)(3), the OIAA finds that specific economic, legal, social, technological, other considerations, including considerations for the provision of employment opportunities for highly trained workers, make Alternative 3 infeasible.

Facts in Support of Finding:

Alternative 3 therefore would not avoid or substantially lessen any significant impacts to aesthetics, biological resources, cultural resources, energy, geology and soils, GHG emissions, hazards and hazardous materials, noise, hydrology and water quality, public services, tribal cultural resources, transportation, and utilities

The Alternative 3 site is located closer to residential receptors to the north of the Airport. As such, health risk impacts may increase due to this alternative. It is anticipated that Alternative 3 would implement similar PDFs and MMs to reduce construction and operation emissions. However, no feasible mitigation measures are available to mitigate the emissions generated by the aircraft operations associated with the proposed Project to a less than significant impact. Alternative 3 would not, therefore, avoid or substantially lessen the significant air quality impacts identified for the proposed Project.

Reference

For a complete discussion of impacts associated with Alternative 3, see Section 6.5 of the Draft EIR at pages 6.0-27 - 6.0-35.

7.0 OTHER CEQA CONSIDERATIONS

7.1 Reasons Why the Project is Being Proposed, Notwithstanding Significant Unavoidable Impacts

CEQA Guidelines Section 15126.2(c) requires that an EIR describe any significant impacts that cannot be avoided, even with the implementation of feasible mitigation measures. Based on the analysis conducted within this Draft EIR document, operation of the proposed Project would result in significant air quality, greenhouse gas emission, and transportation impacts that cannot be mitigated to a less-than-significant level. While a lead agency must contemplate the implications of adverse environmental impacts and mitigate those impacts to the extent feasible, here, the benefits of the Project (as outlined in Section 10.0, below) will outweigh the significant and unavoidable environmental impacts.

7.2 Growth-Inducing Impacts

Section 15126.2(e) of the CEQA Guidelines requires that a Draft EIR include discussion of the potential growth-inducing impacts of a project. This Draft EIR addresses the ways in which the proposed Project "could foster economic or population growth, or the construction of additional housing, either directly or indirectly, in the surrounding environment." This information can be an important factor in a decision to approve a project. As stated in CEQA Guidelines, "It must not be assumed that growth in any area is necessarily beneficial, detrimental, or of little significance to the environment."

The proposed Project would not facilitate any unplanned growth and is located entirely on developed and active Airport property within an urbanized area of San Bernardino County. The proposed Project is an air cargo center, and its construction would not open additional areas to development. The proposed Project is required to comply with the FAA-approved Ontario International Airport Layout Plan (ALP), which designates the Project site for future aeronautical development. The proposed Project would be consistent with adjacent Airport and industrial uses, as well as applicable OIAA, City of Ontario, and FAA-adopted plans, policies, and regulations. Land uses surrounding the Project site include airport-related and industrial uses. No changes to existing or planned land uses on or off Airport property would result from the proposed Project. The proposed Project is consistent with plans, goals, policies, zoning, and local controls that have been adopted and govern over the Project site.

The proposed Project would increase employment opportunities in the region by creating approximately 1,315 jobs and it will employ existing residents in the Riverside-San Bernardino-Ontario area. Therefore, the proposed Project would not trigger the need for new housing. Development of the proposed Project would increase runway use and flight patterns; however, there would not be an increase in the number of passengers expected to use the Airport as the proposed Project is an air cargo facility. It is not expected that the proposed Project would affect population growth or tourism in Ontario and the surrounding region. The proposed Project would

not induce growth at the Airport beyond that which would occur without the improvements and therefore would not result in a significant growth-inducing impact.

7.3 Significant Irreversible Environmental Changes

Section 15126.2(d) of the CEQA Guidelines requires discussion of irreversible environmental change. The Guidelines indicate that "uses of nonrenewable resources during the initial and continued phases of the project may be irreversible since a large commitment of such resources makes removal or nonuse thereafter unlikely" and "irretrievable commitments of resources should be evaluated to assure that such current consumption is justified."

The Project site is already developed and dedicated to airport uses and would not result in a new commitment of land. Construction of the proposed Project would require the consumption of resources that do not replenish themselves or which may renew so slowly as to be considered nonrenewable. This would include resources such as asphalt and concrete, metals, and petrochemical construction materials. Construction and operation of the proposed Project would also require energy resources such as electricity and various fuels. This would represent the loss of non-renewable resources, which are generally not retrievable; however, there are sufficient resources to serve the proposed Project.

Nonrecoverable materials and energy would be used during construction and operation activities; however, the amounts needed would be accommodated by existing supplies. Further, OIAA is committed to constructing the proposed facilities to meet high standards for efficiency and environmental design. Implementation of best practices and standards that emphasize strategies for sustainable site development, water savings, energy efficiency, materials selection, and environmental quality would reduce the use of renewable and nonrenewable resources that would continue over time through construction and long-term operation of the proposed Project.

Although the implementation of best practices and standards that emphasize strategies for sustainable site development would reduce the use of materials and energy during construction and operation of the proposed Project, they would nevertheless be unavailable for other uses. The resources utilized for the proposed Project would be permanently committed to the Airport and, therefore, be considered irreversible.

7.4 Potentially Significant Effects from Implementation of Mitigation Measures

Pursuant to CEQA Guidelines Section 15126.4(a)(1)(D), the EIR evaluates whether implementation of any of the recommended mitigation measures would result in significant impacts (see Section 8.4: Potential Secondary Effects, of the Draft EIR). Except where such impacts are specifically noted, OIAA finds no significant impacts will occur as a result of implementation of mitigation measures. (Draft EIR at pages 8.0-6 – 8.0-9).

8.0 GENERAL CEQA FINDINGS

8.1 Mitigation Monitoring and Reporting Program

As required by CEQA Section 21081.6, OIAA, in adopting these Findings, also adopts the Project Mitigation Monitoring and Reporting Program (MMRP). The MMRP is incorporated as conditions

of Project approval, and is designed to ensure that, during implementation of the Project, OIAA and other responsible parties will comply with the adopted mitigation measures.

8.2 CEQA Guidelines Sections 15091 and 15092 Findings

Based on the foregoing findings and the information contained in the administrative record, OIAA has made one or more of the following findings with respect to each of the significant effects of the Project:

- 1. Changes or alterations have been required in, or incorporated into, the project which mitigate, avoid, or substantially lessen the significant effects on the environment as identified in the Final EIR.
- 2. Those changes or alterations are within the responsibility and jurisdiction of another public agency and such changes have been adopted by such other agency or can and should be adopted by such other agency.
- 3. Specific economic, legal, social, technological, or other considerations, including considerations for the provision of employment opportunities for highly-trained workers, make infeasible the mitigation measures or alternatives identified in the Final EIR.

Based on the foregoing findings and the information contained in the administrative record, and as conditioned by the foregoing:

- 1. All significant effects on the environment due to the project have been eliminated or substantially lessened where feasible.
- 2. Any remaining significant effects that have been found to be unavoidable are acceptable due to the overriding considerations set forth in Section 10 of these Findings.

8.3 OIAA's Preparation of the EIR Pursuant to CEQA Guidelines Section 15084(d)

CEQA Guidelines Section 15084(d) provides a Lead Agency may choose one of the following arrangements or a combination of them for preparing a draft EIR:

- 1. Preparing the Draft EIR directly with its own staff.
- 2. Contracting with another entity, public or private, to prepare the draft EIR.
- 3. Accepting a draft prepared by the Applicant, a consultant retained by the Applicant, or any other person.
- 4. Executing a third-party contract or memorandum of understanding with the Applicant to govern the preparation of a draft EIR by an independent contractor.
- 5. Using a previously prepared EIR.

OIAA has relied on Sections 15084(d)(1) and (3), and 15084(e), of the CEQA Guidelines, which allows the OIAA to prepare and/or review the Draft and Final EIR with its own staff, which EIR materials have been prepared by consultants retained by the applicant.

8.4 OIAA's Independent Judgment

Pursuant to CEQA Section 21082.1(c), OIAA finds it has independently reviewed and analyzed the Draft and Final EIR, and that the Daft and Final EIR reflects the independent judgment of OIAA as the Lead Agency for the Project.

8.5 Nature of Findings

Any finding made by OIAA shall be deemed made, regardless of where it appears in this document. All of the language included in this document constitutes findings by OIAA, whether or not any particular sentence or clause includes a statement to that effect. OIAA intends that these findings be considered as an integrated whole and, whether or not any part of these findings fail to cross reference or incorporate by reference any other part of these findings, that any finding required or committed to be made by OIAA with respect to any particular subject matter of the EIR, shall be deemed to be made if it appears in any portion of these findings.

8.6 Reliance on Record

Each and all of the findings and determinations contained herein are based on the competent and substantial evidence, both oral and written, contained in the entire administrative record relating to the Project.

The findings and determinations constitute the independent findings and determinations of OIAA in all respects and are fully and completely supported by substantial evidence in the record as a whole.

8.7 Custodian of Records

The custodian of the documents or other material which constitute the record of proceedings upon which the OIAA's decision is based is identified as follows:

Kevin Keith Ontario International Airport Authority 1923 East Avion Street Ontario, CA 91761

8.8 Relationship of Findings to EIR

These findings are based on the most current information available. Accordingly, to the extent there are any apparent conflicts or inconsistencies between the EIR and these Findings, these Findings shall control, and the EIR is hereby amended as set forth in these findings.

8.9 Recirculation Not Required

CEQA Guidelines Section 15088.5 provides the criteria that a lead agency is to consider when deciding whether it is required to recirculate an EIR. Recirculation is required when "significant new information" is added to the EIR after public notice of the availability of the Draft EIR is given, but before certification. (CEQA Guidelines, § 15088.5(a).) "Significant new information," as defined in CEQA Guidelines Section 15088.5(a), means information added to an EIR that changes the EIR so as to deprive the public of a meaningful opportunity to comment on a "substantial adverse environmental effect" or a "feasible way to mitigate or avoid such an effect

(including a feasible Project alternative) that the Project's proponents have declined to implement."

An example of significant new information provided by the CEQA Guidelines is a disclosure showing that a "new significant environmental impact would result from the project or from a new mitigation measure proposed to be implemented;" that a "substantial increase in the severity of an environmental impact would result unless mitigation measures are adopted to reduce the impact to a level of insignificance;" or that a "feasible project alternative or mitigation measure considerably different from others previously analyzed would clearly lessen the significant environmental impacts of the project, but the project's proponents decline to adopt it." (CEQA Guidelines, § 15088.5(a)(1)-(3).)

Recirculation is not required where "the new information added to the EIR merely clarifies or amplifies or makes insignificant modifications in an adequate EIR." (CEQA Guidelines, § 15088.5(b).) Recirculation also is not required simply because new information is added to the EIR - indeed, new information is oftentimes added given CEQA's public/agency comment and response process and CEQA's post-Draft EIR circulation requirement of proposed responses to comments submitted by public agencies. Instead, recirculation is "intended to be an exception rather than the general rule." *(Laurel Heights Improvement Assn. v. Regents of University of California* (1993) 6 Cal.4th 1112, 1132.)

In this legal context, OIAA finds that recirculation of the Draft EIR prior to certification is not required. In addition to providing responses to comments, the Final EIR includes revisions to expand upon information presented in the Draft EIR; explain or enhance the evidentiary basis for the Draft EIR's findings; update information; and make clerical revisions to the Draft EIR. The Final EIR's revisions, clarifications and/or updates do not result in any new significant impacts or increase the severity of a previously identified significant impact.

In sum, the Final EIR demonstrates that the Project will not result in any new significant impacts or increase the severity of a significant impact, as compared to the analysis presented in the Draft EIR. The changes reflected in the Final EIR also do not indicate that the meaningful public review of the Draft EIR was precluded in the first instance. Accordingly, recirculation of the EIR is not required as revisions to the EIR are not significant as defined in Section 15088.5 of the CEQA Guidelines.

9.0 CERTIFICATION OF THE FINAL ENVIRONMENTAL IMPACT REPORT, CEQA GUIDELINES, § 15090

OIAA certifies that the Final EIR has been completed in compliance with CEQA and the State CEQA guidelines, that the Final EIR was presented to OIAA, and that OIAA reviewed and considered the information contained therein before approving the Project, and that the Final EIR reflects the independent judgment and analysis of OIAA.

10.0 STATEMENT OF OVERRIDING CONSIDERATIONS

10.1 Background

Section 15093 of the Guidelines provides as follows:

Ontario International Airport South Airport Cargo Center Project

- "(a) CEQA requires the decision making agency to balance, as applicable, the economic, legal, social, technological, or other benefits, including region-wide or statewide environmental benefits, of a proposed project against its unavoidable environmental risks when determining whether to approve the project. If the specific economic, legal, social, technological, or other benefits, including region-wide or statewide environmental benefits, of a proposed project outweigh the unavoidable adverse environmental effects, the adverse environmental effects may be considered "acceptable."
- (b) When the lead agency approves a project which will result in the occurrence of significant effects which are identified in the final EIR but are not avoided or substantially lessened, the agency shall state in writing the specific reasons to support its action based on the final EIR and/or other information in the record. This statement of overriding considerations shall be supported by substantial evidence in the record.
- (c) If an agency makes a statement of overriding considerations, the statement should be included in the record of the project approval and should be mentioned in the notice of determination. This statement does not substitute for, and shall be in addition to, findings required pursuant to § 15091."

In accordance with the provisions of CEQA and the Guidelines, and as part of its certification of the adequacy of Final EIR for approval of the Project, OIAA finds that the mitigation measures discussed in these Findings and the Mitigation Monitoring and Reporting Plan, when implemented, avoid, mitigate or substantially lessen the significant effects of the Project identified in the Final EIR. Nonetheless, certain significant effects of the Project are unavoidable even after incorporation of all feasible mitigation measures. In summary, even with implementation of the mitigation measures described in Final EIR and the MMRP, the following effects of the Project are considered to be significant and unavoidable at this time:

Air Quality

Impact AQ-2 – The proposed Project's operational emissions during Phase 1 and Phase 2 would exceed regional SCAQMD significance thresholds for CO, VOC, and NOx, primarily due to aircraft emissions, followed by employee vehicles, delivery trucks, and emergency generators. Therefore, the Project would result in a significant, unavoidable impact due to operation related air quality.

The Project's cumulative contribution to air quality would be cumulatively considerable.

Greenhouse Gas Emissions

Impact GHG-1 – The Project would result in a net increase in GHG emissions during Phase 1 and Phase 2 operation over baseline conditions, which is considered to be a significant impact on the environment.

Impact GHG-2 – The Project would conflict with an applicable plan, policy or regulation adopted for the purpose of reducing the emissions of greenhouse gases.

The Project's cumulative contribution to greenhouse gas emissions would be cumulatively considerable.

Transportation

Impact TRA-2 – The truck, employee and other trips generated by the proposed Project during operation would result in the Project Total VMT per service population (employees for this proposed Project) being 22 percent above the City's VMT significance threshold of 29.76 VMT per service population. Therefore, the Project would result in a significant, unavoidable impact due to operation related VMT.

The Project's cumulative contribution to transportation would be cumulatively considerable.

OIAA finds that (i) all feasible mitigation measures identified in the Final EIR that are within the purview of OIAA will be implemented with the Project; (ii) that those changes or alterations found within the responsibility and jurisdiction of another public agency have been adopted by such other agency, or can and should be adopted by such other agency; and (iii) the remaining significant unavoidable effects are outweighed and are found to be acceptable due to the following specific overriding economic, legal, social, technological, or other benefits, based upon the facts set forth in these Findings, the Final EIR, and the record.

10.2 Overriding Considerations

OIAA finds that the Project would have economic, legal, social, technological, or other overriding benefits, including region-wide or statewide environmental benefits, listed below. Each of the benefits cited below constitutes a separate and independent basis that justifies approval of the Project and outweighs the unavoidable adverse environmental effects of approving the Project and thus makes adverse environmental effects acceptable. Thus, even in the absence of one or more of the reasons set forth below, OIAA has determined that each remaining reason, or any combinations of reasons, is a sufficient basis for approving the Project, notwithstanding any significant and unavoidable impacts that may occur.

Accordingly, the OIAA adopts the following Statement of Overriding Considerations. The City recognizes that significant and unavoidable impacts would result from implementation of the Project. Having (i) adopted all feasible mitigation measures, (ii) rejected as infeasible the alternatives to the Project discussed above, (iii) recognized all significant, unavoidable impacts, and (iv) balanced the benefits of the Project against the Project's significant and unavoidable impacts, the OIAA hereby finds that each of the Project's benefits, as listed below, outweigh and override the significant unavoidable impacts relating to the impacts of the Ontario International Airport South Caro Center Project for air quality, greenhouse gas emissions and transportation.

The below stated reasons summarize the benefits, goals and objectives of the Project, and provide the detailed rationale for the benefits of the Project. These overriding considerations of economic, social, and environmental benefits for the Project justify adoption of the Project and certification of the completed EIR. Each of the listed Project benefits set forth in this Statement of Overriding Considerations provides a separate and independent ground for the OIAA's decision to approve the Project despite the Project's identified significant and unavoidable environmental impacts. Each of the following overriding considerations separately and independently (i) outweighs the adverse environmental impacts of the Project, and (ii) justifies adoption of the Project and certification of the completed EIR. In particular, achieving the underlying purpose for the Project would be sufficient to override the significant environmental impacts of the Project.

- 1. The Project Would Develop Improved Air Cargo System Management. The FAA and Caltrans forecast air cargo volume to increase in the region. The proposed Project would redevelop an underutilized portion of the Airport with an air cargo center that would accommodate a portion of the regional growth in air cargo operations forecasted by the FAA and Caltrans. The new air cargo center would provide an efficient facility for the proposed Project proponent's airside, landside, and sorting operations. Specifically, the Air Cargo Sort Building for the proposed Project would include state of the art technology to support the efficient processing of cargo. The proposed new air cargo center would increase air cargo capacity at the Airport.
- 2. The Project Would Improve Existing Transportation Infrastructure at and around the Airport. Transportation infrastructure includes air transportation facilities in a location with access to major transportation corridors. The proposed Project would include demolition of existing buildings consisting of hangars, ancillary structures, and parking facilities, as well as existing landscaping and trees on the Project site. The proposed Project includes an Air Cargo Sort Building, truckyard, parking facilities, aircraft parking apron improvements, GSE parking, and aviation support facilities. OIAA would terminate existing leases, and non-OIAA tenant operations would vacate the facilities prior to construction of the proposed Project. OIAA occupies facilities on the Project site and their operations would be relocated to existing facilities both on- and off-airport. The proposed Project would redevelop underutilized Airport property, accommodate regional growth of air cargo operations, and integrate proposed Project air transportation facilities in a location with access to major transportation corridors. In addition to protecting air transportation, the proposed Project would not conflict with a program, plan, ordinance, or policy related to the roadway, transit, bicycle, and pedestrian facilities. Therefore, the proposed Project would improve existing transportation infrastructure at and around the Airport.
- 3. The Project Would Increase Goods Movement Capacity. While the proposed air cargo center would not provide more transportation choices for passengers, the proposed Project would increase the air cargo capacity at the Airport and increase transportation choices to meet the increased demand for air cargo services in the region, consistent with this theme.
- 4. The Project Would Facilitate The Financial Self-Sustainability Of The Airport As Required By FAA Regulations. Under federal codes, FAA Orders and rules (FAA Order 5190.6B, Airport Compliance Manual), and FAA grant assurances for Airport projects (49 USC section 47101(a)(13)), the OIAA is required to maximize its revenue generation at the Airport to ensure financial self-sustainability. The Project operations would help to fulfill this requirement by providing significant annual rent revenues and cargo freight landing fees/charges, totaling in the many millions of dollars per year, for the term of a decades-long later lease.
- 5. The Project Would Represent Sustainable Airport Development. The proposed Project would use and operate electric-powered equipment, electric cargo planes, and electric charging stations in the employee and visitor parking lots and truckyard. This technology

would be leveraged to ensure the Air Cargo Sort Building would meet LEED certification standards. The proposed Project incorporates sustainable project design features and technology in both design and operation. The Project would incorporate a variety of technologies into the building design to reduce energy use, track energy consumption to support identification of further improvements, generate renewable energy on site, and utilize clean energy sources. The Air Cargo Sort Building would meet LEED certification standards. A 3.8-Megawatt Solar PV Panel system would be installed on the rooftop of the Air Cargo Sort Building with an option to construct an additional 0.75-megawatt rooftop system on the parking garage. The proposed Project would include the use of zero-emission or near zero-emission trucks as part of business operations beginning in 2025 (within at least 25% of the Project fleet) the use of zero-emission or near zero-emission trucks as part of business operations beginning in 2029 (within at least 50% of the Project fleet). The proposed Project would include the use and operation of electric-powered equipment, including forklifts, loaders, tugs, ground power units, and ramp support (vans/carts) that would be stored and charged in designated areas in the cargo building and aircraft apron. Moreover, the proposed Project proposes the operation of electric cargo planes, for which charging stations would be provided in the southeast corner of the Project site. Electric charging stations would also be provided in the employee and visitor parking lots, as well as the truckyard.

6. The Project Would Support Commerce, Economic Development, and Employment Opportunities. The proposed Project would accommodate a portion of the projected regional growth in air cargo operations as forecast by the FAA and Caltrans. By redeveloping an underutilized portion of the Airport, the proposed Project would maximize revenue generation from Airport property. The proposed Project would also increase employment opportunities in the region by creating approximately 1,315 jobs. According to the US Bureau of Labor Statistics, in June 2021 there was an unemployment rate of 7.9 percent (or 165,600 people were unemployed) in the Riverside-San Bernardino-Ontario area.31 The 1,315 jobs generated by the proposed Project would be available to existing residents in the Riverside-San Bernardino-Ontario area. For these reasons, the proposed Project would support commerce, economic development, and employment opportunities.

10.3 Conclusion

In light of the foregoing, and the information contained within the Final EIR and other portions of Project record, OIAA concludes that implementation of the Project will result in the development of a beneficial Project as outlined above. OIAA also finds that the benefits identified above outweigh and make acceptable the significant, unavoidable environmental impacts associated with the Project and, accordingly, adopts this Statement of Overriding Considerations.

A RESOLUTION OF THE ONTARIO INTERNATIONAL AIRPORT AUTHORITY TO APPROVE THE SOUTH AIRPORT CARGO CENTER PROJECT AT THE ONTARIO INTERNATIONAL AIRPORT AND TO ADOPT FINDINGS REGARDING THE PROJECT

WHEREAS, the Ontario International Airport Authority (OIAA or Authority) was established under an August 2012 Joint Exercise of Powers Agreement between the City of Ontario and the County of San Bernardino pursuant to the Joint Exercise of Powers Act of the State of California (Government Code Sections 6500.1 *et seq.*), for the purpose of operating, maintaining, managing, developing, and marketing the Ontario International Airport (Airport) and related Airport facilities;

WHEREAS, the Authority owns approximately 97 acres of land and related existing improvements on the southern end of the Airport where the proposed South Airport Cargo Center Project (Project) would be located. The Project site includes portions of Assessor Parcel Numbers (APNs) 11326106, 11326107, 11326108, 11327101, and 11327102 (collectively, the Property), located in the southern half of the Airport, immediately west of the Cucamonga Channel and north of Mission Boulevard. Most of the Project site is located north of East Avion Street with the remainder located between East Avion Street and Mission Boulevard west of South Hellman Avenue. The Property is surrounded by aeronautical structures and land uses, including taxiways and runways to the north;

WHEREAS, on October 27, 2022, the OIAA and Project applicant entered into that Exclusive Negotiating Agreement, as may be amended from time to time (ENA), regarding the Property and the Project. The purpose of the ENA is to establish procedures for the potential lease of the Property for the Project and for the parties to pursue and obtain environmental, land use, and other governmental approvals regarding the Project;

WHEREAS, the proposed Project is an aeronautical development and use that is within the Airport boundaries and is consistent with the Ontario International Airport Layout Plan. The proposed Project would replace existing, underutilized airport related buildings and site improvements with an air cargo center. The proposed Project includes demolition of the existing buildings, site improvements, and the development of a new air cargo center in two phases, as described further below. The proposed Project includes an Air Cargo Sort Building, truckyard, parking facilities, two aviation support buildings (ground service equipment and aircraft line maintenance buildings), and aircraft apron improvements. The Air Cargo Sort Building, which would be located north of East Avion Street, would contain a sorting facility and office spaces. The aircraft parking apron would surround the building to the west, north, and east. A ground-level visitor parking lot and truckyard are proposed on the south side of the Air Cargo Sort Building, with access from East Avion Street. A parking structure for employees is proposed south of East Avion

Street with a pedestrian bridge connecting the parking structure to the Air Cargo Sort Building. The proposed Project would be implemented in two phases. Phase 1 includes the project improvements to be constructed and operated on the easternmost 62 acres of the Project site and Phase 2 includes the project improvements to be constructed and operated on the remaining western 35 acres;

WHEREAS, the Property has structures that have been vacant and unused for decades. The proposed Project involves the redevelopment of the Property with aeronautical uses consistent with the purposes of the OIAA and with the Airport Layout Plan for the Airport approved by the Federal Aviation Administration, which designates the Property for aeronautical uses;

WHEREAS, a Final Environmental Impact Report (the FEIR) (SCH#2021100226), including a Draft Environmental Impact Report that was circulated for public comment on March 13, 2023 (the DEIR), has been prepared by the OIAA, as the Lead Agency, pursuant to the requirements of the California Environmental Quality Act (CEQA) (Public Resources Code Sections 21000 *et seq.*) and the State CEQA Guidelines (14 Cal Code Regs. Sections 15000 *et seq.*) to address the potential environmental impacts associated with the proposed Project at the Airport;

WHEREAS, the Project applicant has also proposed a long-term ground lease of the Property between the applicant and the OIAA (the Ground Lease) for the proposed Project uses, which Ground Lease will be considered by the OIAA at a later public meeting;

WHEREAS, the development and operation of the proposed Project pursuant to the Ground Lease will be also consistent with supply chain improvements that are encouraged in Governor Newsom's Executive Order N-19-21 regarding supply chain issues, which Order recognized that "the movement of goods and health of supply and distribution chains across California is a matter of vital statewide importance";

WHEREAS OIAA staff recommends approval of the Project substantially as described, reflected and depicted in (1) Section 3.4 (Project Description) of the DEIR (including, without limitation, the Figures therein), as modified in the FEIR, and (2) the civil architectural plans and drawings for the Project submitted to the OIAA on June 12, 2023 (the civil plan set includes 27 sheets dated April 28, 2023 and the architectural plan set includes 28 sheets dated June 1, 2023);

WHEREAS, the OIAA Commission has reviewed and considered all materials identified and referred to in this Resolution, as well as all materials and documents discussed in and attached to the Agenda Report for this matter; and

WHEREAS, on June 19, 2023, the Authority provided notice of the Regular OIAA Commission meeting, which includes consideration of this matter, scheduled for June 22, 2023, in accordance with the Ralph M. Brown Act, Government Code § 54950, *et seq.*, and provided notice to persons who specifically requested notice of this meeting.

NOW, THEREFORE, BE IT RESOLVED by the OIAA Commission as follows:

SECTION 1. FINDINGS.

All Whereas clauses or Recitals in this Resolution above are incorporated by reference into these Findings.

Further, the OIAA Commission, in light of the whole record before it, including, but not limited to (1) the Agenda Report, including all of the attachments thereto, for this matter, (2) the FEIR and (3) other substantial evidence within the record and/or provided at this June 22, 2023, Regular public meeting (all of which administrative record materials for this matter are incorporated by reference into this Resolution No. 2023-11), finds and determines as follows:

- A. The Project and intended use of the Property are consistent with (1) the OIAA's purpose of operating, maintaining, managing, developing, and marketing the Airport and related Airport facilities and (2) the Airport Layout Plan for the Airport approved by the Federal Aviation Administration, which Plan designates the Property for aeronautical uses.
- B. The OIAA has the authority under Government Code Sections 6500.1 *et seq.* and the August 2012 Joint Exercise of Powers Agreement to operate, maintain, manage, develop, and market the Airport and related Airport facilities and aeronautical uses for the benefit of the citizens of the City of Ontario and the County of San Bernardino, and the economic development of the Inland Region of Southern California generally, and Ontario and San Bernardino specifically, and the Project and intended use of the Property are consistent with the OIAA's purpose and function under Government Code Sections 6500.1 *et seq* and the August 2012 Joint Exercise of Powers Agreement.
- C. The Project and intended use of the Property are consistent with vital State objectives regarding the movement of goods and health of supply and distribution chains across California. The Project consists of an air cargo development related to the timely distribution of goods and products. The operation of the Project will therefore be consistent with supply chain improvements that are encouraged in Governor Newsom's Executive Order N-19-21 regarding supply chain issues, which Order recognized that "the movement of goods and health of supply and distribution chains across California is a matter of vital statewide importance …."

SECTION 2. OIAA COMMISSION ACTIONS:

A. <u>Approval of Findings</u>: The OIAA Commission approves and adopts the Findings set forth in Section 1, above, of this Resolution.

- B. <u>Approval of the Project:</u> The OIAA Commission approves the Project substantially as described, reflected and depicted in (1) Section 3.4 (Project Description) of the DEIR (including, without limitation, the Figures therein), as modified in the FEIR, and (2) the civil and architectural plans and drawings for the Project submitted to the OIAA on June 12, 2023 (the civil plan set includes 27 sheets dated April 28, 2023 and the architectural plan set includes 28 sheets dated June 1, 2023), as may be refined and modified in a substantially similar form.
- C. <u>Conditions of Approval:</u> The OIAA Commission's approval of the Project as described in Section 2.A, above, is subject to the following conditions of approval:
 - 1. The OIAA Commission adopts and incorporates as conditions to this approval of the Project all of the mitigation measures recommended in the FEIR and contained in the MMRP Section 4.0 of the FEIR, and in companion Resolution No. 2023-10, and the related CEQA Findings of Fact and Statement of Overriding Considerations attached thereto, and directs that all such mitigation measures be implemented at a time and in a manner consistent with the approved MMRP.
 - 2. Approval of the Project is subject to the Indemnity provisions in Section 11 of the ENA.
 - 3. The approval of the Project pursuant to this Resolution No. 2023-11 shall expire in the event that the Ground Lease has not been executed by the applicant and the OIAA on or before the later of (a) the expiration date of the ENA, as the same may be extended by the applicant pursuant to the terms thereof, or (b) ten (10) business days following the date on which the OIAA Commission takes action on the Ground Lease, unless in either case the parties agree in writing to a different date.

<u>SECTION 3.</u> Effective Date. This Resolution will take effect immediately upon its adoption.

<u>SECTION 4.</u> Certification. The Secretary/Assistant Secretary shall certify as to the adoption of this Resolution.

PASSED, APPROVED, AND ADOPTED at a Regular Meeting this 22nd day of June, 2023.

ALAN D. WAPNER, OIAA PRESIDENT

ATTEST:

NORMA I. ALLEY, MMC, ASSISTANT SECRETARY

APPROVED AS TO LEGAL FORM:

LORI D. BALLANCE, GENERAL COUNSEL

STATE OF CALIFORNIA COUNTY OF SAN BERNARDINO CITY OF ONTARIO

I, Norma I. Alley, MMC, Board Clerk/Assistant Secretary of the Ontario International Airport Authority, DO HEREBY CERTIFY the foregoing Resolution No. 2023-11 is the original and was duly passed and adopted by the Commission of the Ontario International Airport Authority at their Regular Meeting held on June 22, 2023, by the following roll call vote, to wit:

)

AYES: COMMISSIONERS:

NOES: COMMISSIONERS:

ABSENT: COMMISSIONERS:

NORMA I. ALLEY, MMC BOARD CLERK/ASSISTANT SECRETARY

(SEAL)

ONTARIO INTERNATIONAL AIRPORT AUTHORITY



MEETING DATE: JUNE 22, 2023

AGENDA SECTION: ADMINISTRATIVE REPORTS/DISCUSSION/ACTION

SUBJECT: RESOLUTION TO ADOPT THE FYE 2024 OPERATING BUDGET

RELEVANT STRATEGIC OBJECTIVE: Invest in ONT Invest in ONT

RECOMMENDED ACTION(S): Approve a resolution adopting the FYE 2024 Operating Budget.

FISCAL IMPACT SUMMARY: The proposed FYE 2024 budget results in Net Income from Operations of \$11.6M a \$2.0M (14.8%) decrease compared to FYE 2023. Interest costs will be funded from airline rates and charges in accordance with the Airline Operating Use and Lease Agreement (ULA).

BACKGROUND: Fiscal Year Financial Performance: The proposed operating budget for the fiscal year ending June 30, 2024, is driven by strong aviation activity and financial performance realized by OIAA through March 31, 2023. Net income from operations totaled \$18.7M with a favorable variance to budget of \$8.3M. Operating revenue of \$75.7M exceeded budget by \$4.5M (6.3%) and Operating expenses of \$57.4M were favorable to budget by \$8.4M (12.9%). The proposed budget was developed from OIAA Goals and Objectives and includes significant increases in resources to meet current and expected near term growth.

OPERATING REVENUES: Total FYE 2024 budgeted operating revenues of \$113.5M exceed the budget FYE 2023 by \$10.9M (10.7%), with aeronautical revenue increasing by \$2.7M (5.6%) and nonaeronautical revenues by \$8.2M (15.1%). The Authority has budgeted \$6.6M of ARPA grant funds to maintain a relatively level rate structure. Budget revenue forecasts do not reflect proposed changes in any airline rates and charges, except for landing fees and terminal rents that are calculated under the terms of the Operating Use and Lease Agreement. In addition, to these changes, an increase to parking rates is being proposed and is included in the budget.

Aeronautical revenue increases are primarily driven by increases in landing fees and airline handling service fees. Landing fees increased by \$1.8M (12.4%) and are associated with greater landed weights and a higher landing fee rate. Terminal rents decrease by \$573K (3.5%) are driven by higher per square foot rental rates. In addition, ground handling revenue increases reflect higher than expected revenues from third-party ground handling service providers to signatory airlines.

Nonaeronautical revenues budgeted FYE 2024 of \$63.0M are greater than FYE 2023 by \$8.2M (15.1%). This increase is driven by higher concession revenues. The most significant operating revenue budget increases include parking revenues of \$7.0M (24.9%) and food & beverage of \$1.6M (105.6%). Parking revenue increases are associated with forecasted increases in aviation activity and proposed increases parking lot rates, depending

on the lot. Rental car revenues are estimated to be lower FYE 2024 compared to FYE 2023 after several years of favorable growth. Operating grant revenues budgeted FYE 2024 of \$6.6M is the same as budgeted FYE 2023. This leaves unspent operating grant funding of \$11.4M which will be rolled over to FYE 2025. The grant funding must be spent by December 2025.

OPERATING EXPENSES: Total budgeted Operating Expenses FYE 2024 of \$101.9M, exceed the budget FYE 2023 of \$89.0M, by \$ \$12.9M (14.5%). Major changes are for personnel, public safety, contractual services, and other operating expenses. No uniform budget increases have been provided for inflation, as contracted service and material expenses are largely controlled by established agreements.

Personnel expenses FYE 2024 of \$17.6M, are greater than FYE 2023 by \$4.5M, (34.8%). This includes 17.5 new OIAA staff positions, totaling \$1.1M. Also, the FYE 2024 personnel budget assumes a 3% COLA increase, as well as a 5% merit pool for performance-based increases, which totals \$775K in the FYE 2024 budget.

Nonpersonnel expenses FYE 2024 of \$84.3M increase by \$8.4M (11.1%), compared to FYE 2023 budgeted expenses of \$76.0M. Public Safety expenses of \$23.3M FYE 2024 are higher than the FYE 2023 budget by \$2.1M. Public safety, police, fire, and dispatch services are provided under a Municipal Services agreement with the City of Ontario. The budgeted increase is 10% based on an estimated contractually required increases as well as police personnel. The contractual expense budget FYE 2024 of \$39.7M is greater than the FYE 2023 budget by \$4.1M (11.7%). The greatest increases are for security and technology support services to complete information technology initiatives as well as the implementation and operation of the new Common Use system in Terminal 2.

Nonoperating Revenues are budgeted to increase by \$1.4M (12.4%). Most of this net increase is driven by activity-based facility charges and interest income. These facility charges include Passenger Facility Charges and Customer Facility Charges and are directly related to passenger activity. There are no changes in the rate structure for these facility charges, PFCs remain at \$4.50 per enplaned passenger and CFCs stay the same at \$10 per rental car contract. Interest income is expected to be greater as interest rates on short-term cash equivalent investments increase.

Other Sources and (Uses), net, are from debt service, required reserves, depreciation, and fund transfers. Total Other Sources and (Uses) budgeted at \$22.0M FYE 2024 are greater than the FYE 2023 budget by \$777K (3.7%).

Debt service is associated with required principal and interest payments for the series 2016 and 2021 revenue bonds. Debt service is expected to decrease by \$782K. The decrease is associated with early retirement of non-revenue bond debt in FYE 2022.

Reserve balances and adjustments are established under the Operating Use and Lease Agreement, Revenue Bond Covenants, and regulatory authority. The Discretionary reserve is to fund Capital projects without airline approval, that would otherwise require approval under the Majority in Interest provisions of the ULA. The Discretionary reserve increase is for required reserve adjustments based on CPI. The Maintenance and Operations (M&O) reserve requires that 25% of operating expenses be included in the M&O reserve on a rolling funding basis, whereby adjustments in a fiscal year are provided to adjust the reserve balance by an amount necessary to meet the 25% requirement. The Debt Service reserve is also on a rolling fund basis, to provide a reserve balance equal to 25% of the revenue bond debt service for the budgeted fiscal year.

The unrestricted fund transfer is for estimated airline incentive program savings for qualified airlines and routes. These fees result in a loss of revenues and an increase in rates and charges to other airlines. This loss may not be recovered through rates and charges to the airlines and must be funded through transfers from the unrestricted net position balance.

The budget establishes a net residual requirement, that is covered by landing fees and terminal rental rates. Landing fees are established by dividing the net requirement in the Airfield Cost Center by the estimated landed weight for cargo and passenger air carriers. The terminal rental rate is established by dividing the net requirement in the Terminal Cost Center by the total of terminal space leased by the airlines, including joint use space in public areas.

The calculated landing fee rate for FYE 2024 is \$1.83 per 1,000 lbs. of landed weight generating \$16.0M in landing fees and the terminal rental rate FYE 2024 is \$87.1, which results in \$15.7M of terminal rent revenue.

PROCUREMENT: N/A

CEQA COMPLIANCE AND LAND USE APPROVALS: N/A

STAFFING IMPACT (# OF POSITIONS): N/A

IMPACT ON OPERATIONS: The financial results provide additional resources for the OIAA to respond to increased passenger carrier activity and to meet OIAA goals and objectives.

SCHEDULE: N/A

ATTACHMENTS:

- 1. Resolution No. 2023-12
- 2. Operating Budget
- 3. Organizational Chart

STAFF MEMBER PRESENTING: Chief Executive Officer Atif Elkadi

Originator Name:	John M. Schubert, Chief Financial Officer
Originating Dept.:	Finance
Director Review:	N/A O OA O
Chief Review:	John M. Schubert, Chief Financial Officer
CEO Approval:	A===

This Agenda Report has been reviewed by OIAA General Counsel.

The Agenda Report references the terms and conditions of the recommended actions and request for approval. Any document(s) referred to herein, which are not attached or posted online, may be reviewed prior to, or following scheduled Commission meetings in the Office of the Clerk of the Board. Office hours are 8:30 a.m. to 5:00 p.m., Monday through Friday, although these hours and review procedures may be modified. In that case, the documents may be requested by email at <u>clerk@flyontario.com</u>.

BOARD DISPOSITION: Approved Denied Continued to

A RESOLUTION OF THE ONTARIO INTERNATIONAL AIRPORT AUTHORITY ADOPTING THE ANNUAL OPERATING BUDGET FOR FISCAL YEAR 2023-24

WHEREAS, the Ontario International Airport Authority ("the Authority" or "OIAA") was established under a Joint Exercise of Powers Agreement between the City of Ontario and the County of San Bernardino (the "Joint Powers Agreement") pursuant to the purpose of operating, maintaining, developing, and marketing the Ontario International Airport (the "Airport"); and

WHEREAS, under the Joint Powers Agreement, Section 10, Funds, the Treasurer of the Authority shall assume the duties described in California Government Code Section 6505.5; and

WHEREAS, the Authority staff, including the Chief Executive Officer ("CEO") and the Chief Financial Officer ("CFO") of the Authority, have determined the Authority's Capital Budget for Fiscal Year 2023-24 in accordance with the provisions of the Joint Powers Agreement; and

WHEREAS, the Authority staff have consulted with the Airport and Airline Affairs Committee ("AAAC") in regards to this proposed operating budget for Fiscal Year 2023-24 and the AAAC has not expressed any objections to its adoption or implementation.

NOW, THEREFORE, BE IT RESOLVED by the Ontario International Airport Authority Commission as follows:

SECTION 1. The Ontario International Airport Authority Commission hereby adopts the proposed Fiscal Year 2023-24 Operating Budget, which is attached as Exhibit "A", and by this reference incorporated.

SECTION 2. Effective Date. This Resolution will take effect immediately upon its adoption.

SECTION 3. Certification. The Secretary/Assistant Secretary shall certify as to the adoption of this Resolution.

PASSED, APPROVED, AND ADOPTED at a Regular Meeting this 22nd day of June, 2023.

ALAN D. WAPNER, OIAA PRESIDENT

ATTEST:

APPROVED AS TO LEGAL FORM:

NORMA I. ALLEY, MMC, ASSISTANT SECRETARY

STATE OF CALIFORNIA COUNTY OF SAN BERNARDINO CITY OF ONTARIO

I, Norma I. Alley, MMC, Board Clerk/Assistant Secretary of the Ontario International Airport Authority, DO HEREBY CERTIFY the foregoing Resolution No. 2023-12 is the original and was duly passed and adopted by the Commission of the Ontario International Airport Authority at their Regular Meeting held June 22, 2023, by the following roll call vote, to wit:

)

)

AYES: COMMISSIONERS:

NOES: COMMISSIONERS:

ABSENT: COMMISSIONERS:

NORMA I. ALLEY, MMC BOARD CLERK/ASSISTANT SECRETARY

(SEAL)

OPERATING BUDGET

FOR THE YEAR ENDED JUNE 30, 2024

ONTARIO INTERNATIONAL AIRPORT AUTHORITY





PROPOSED BUDGET

ONTARIO INTERNATIONAL AIRPORT AUTHORITY OPERATING BUDGET FOR THE YEAR ENDING JUNE 30, 2024

Table of Contents:

ONTARIO INTERNATIONAL AIRPORT AUTHORITY
Operating Budget1
Summary by Division and Department 3
Summary by Division and Department Percent of Total Operating Expenses 4
Summary by Division and Department by Expense Category5
OPERATING BUDGETS DETAILS BY DIVISION AND DEPARTMENT
OPERATIONS DIVISION
REVENUE MANAGEMENT DIVISION7
EXECUTIVE DIVISION
ADMINISTRATIVE DIVISION
INFORMATION TECHNOLOGY DIVISION
FINANCE DIVISION11
CAPITAL DEVELOPMENT DIVISION
MARKETING AND COMMUNICATIONS13

ONTARIO INTERNATIONAL AIRPORT AUTHORITY BUDGET - FYE 2024 Proposed vs FYE 2023 Adopted OSED BUDGET

	Adopted Budget	Proposed Budget	Increase (Decrease) FY 2 vs FY 2023 Add	2024 Proposed opted
ŕ	FY 2023	FY 2024	\$	%
Aeronautical				
Landing fees	\$ 14,239,780	\$ 16,006,964	\$ 1,767,185	12.41%
Facility rentals	16,261,999	15,688,558	(573,441)	-3.53%
Land rentals	12,857,518	12,455,895	(401,623)	-3.12%
Gate use and jet bridge fees	1,526,080	806,912	(719,168)	-47.13%
Plane parking	539,255	643,257	104,002	19.29%
Airline handling service fees	1,676,571	4,032,833	2,356,262	140.54%
Operating grants	297,600	321,000	23,400	7.86%
Other aeronautical revenues	401,143	527,237	126,094	31.43%
Total aeronautical	47,799,946	50,482,656	2,682,710	5.61%
Nonaeronautical				
Auto parking	28,156,137	35,150,798	6,994,661	24.84%
Rental cars	10.477.107	9.996.171	(480,936)	-4.59%
Food and beverage	1,473,447	3,029,527	1,556,080	105.61%
News and gifts	2.041.517	2,124,170	82.652	4.05%
Ground transportation	1.740.302	2,144,696	404.394	23.24%
Advertising and Other Concessions	1,122,000	1,024,316	(97,684)	-8.71%
Eacility & Land rentals -	, ,			
nonaeronautical	2 593 109	2 728 792	135 683	5 23%
Other	550 114	199 971	(350 143)	-63 65%
Operating grants - ARPA	6.575.000	6.575.000	-	0.00%
Total nonaeronautical	54,728,735	62.973.441	8.244.707	15.06%
	0 1,1 20,1 00		c,,. c.	
Total operating revenues	102,528,680	113,456,097	10,927,417	10.66%
Personnel				
Salaries, wages and overtime	10,394,503	13,690,071	3,295,569	31.70%
Employee benefits and taxes	2,664,896	3,911,943	1,247,048	46.80%
Total personnel	13,059,398	17,602,014	4,542,616	34.78%
Nonpersonnel				
Public safety	21.175.717	23.293.289	2.117.572	10.00%
Contractual services	35,552,476	39,694,741	4,142,265	11.65%
Insurance and administration	2,147,461	1,957,391	(190,070)	-8.85%
Materials and Supplies	1,919,787	2,699,817	780,030	40.63%
Telecommunications and utilities	7,293,100	7,331,304	38,204	0.52%
Other Operating Expenses	7,814,140	9,320,324	1,506,184	19.28%
Total nonpersonnel	75,902,681	84,296,866	8,394,185	11.06%
Total operating expenses	88,962,079	101,898,880	12,936,801	14.54%
Net Income from operations	13,566,601	11,557,217	(2,009,385)	-14.81%

ONTARIO INTERNATIONAL AIRPORT AUTHORITY BUDGET - FYE 2024 Proposed vs FYE 2023 Adopted OSED BUDGET

	Adopted Budget	Proposed Budget	Increase (Decrease) FY 2024 Proposed vs FY 2023 Adopted					
	FY 2023	FY 2024	\$	%				
Nonoperating Revenues								
Interest income	\$ 77,143	\$ 505,546	\$ 428,403	555.34%				
Passenger facility charges	11,582,266	12,589,028	1,006,762	8.69%				
Customer facility charges	3,462,334	3,454,045	(8,289)	-0.24%				
Other	-	-	-	0.00%				
Total nonoperating revenues	15,121,742	16,548,620	1,426,877	12.37%				
Net income	28,688,344	28,105,837	(582,507)	-2.03%				
Other Sources and (Uses)								
Debt Service	12.010.020	11.228.344	(781.676)	-6.51%				
Reserve Balance (Increase)	,,	, -,-	(-))					
Decrease	3,129,698	3,539,663	409,965	13.10%				
Depreciation	6,055,996	7,954,879	1,898,883	31.36%				
Unrestricted Fund Transfer	-	(750,000)	(750,000)	0.00%				
Total other sources and (uses)	21,195,714	21,972,887	777,173	3.67%				
	¢ 7.400.600	¢ 6 4 2 2 0 5 0	¢ (4.250.690)	10.450/				
Net increase (decrease)	۶ <i>1</i> ,492,630	۶ 6,132,950	\$ (1,359,680)	-18.15%				

Landing Fee (1,000 lb. units) Terminal Rental Rate (square foot)

\$ 1.64	\$ 1.83	\$ 0.23	14.4%
\$ 91.65	\$ 87.07	\$ 7.07	8.8%

ONTARIO INTERNATIONAL AIRPORT AUTHORITY BUDGET - TOTAL OPERATING EXPENSES BY DIVISION AND DEPARTMENT BUDGET FOR THE YEAR ENDING 2023-24 VS 2022-23

			Increase (Decrease) FY 2	2024 Proposed vs
	Adopted	Proposed	FY 2023 Add	opted
	FY 2023	FY 2024	\$	%
Operations:				
Operations	\$ 782,874	\$ 706,476	\$ (76,398)	-9.76%
Public Safety	21,607,162	23,806,869	2,199,706	10.18%
Airfield Operations	5,667,507	6,646,674	979,167	17.28%
Security	6,099,893	6,397,407	297,514	4.88%
Emergency Management	629,253	623,675	(5,578)	-0.89%
Vehicle and Equipment Maintenance	2,073,717	2,384,157	310,440	14.97%
Landside Operations	7,740,121	8,380,733	640,612	8.28%
Customer Experience	1,480,901	1,8/5,//3	394,873	20.00%
Total Operations	40,001,420	50,021,704	4,740,335	10.29%
Capital Development:				
Planning	6,217,062	6,428,603	211,541	3.40%
Project Management	2,218,998	2,502,690	283,692	12.78%
Total Capital Development	8,436,060	8,931,293	495,233	5.87%
Revenue Management:				
Revenue Management Department	2 641 501	2 635 075	(6 4 2 6)	-0 24%
Commercials Real Estate	825.552	840.520	14.968	1.81%
Ground Transportation	7,076,469	7,563,328	486,859	6.88%
Film Services	157,500	42,500	(115,000)	-73.02%
Total Revenue Management	10,701,022	11,081,423	380,402	3.55%
Marketing and Communications:				
Marketing and Communications.	6 837 231	8 721 324	1 884 093	27 56%
Digital	2 010 786	2 506 064	495 277	24.63%
Total Marketing and Communications	8,848,018	11,227,388	2,379,370	26.89%
Executive				
Executive:	E 4E0 470	4 005 999	(454 501)	0.240/
Office Administrator	5,450,479	4,995,000	(454,591)	-0.34%
	676 290	204, 149	204,149	29 72%
Government Relations	463 787	524 529	60 742	13 10%
Total Executive	6,590,557	6,661,819	71,262	1.08%
				
	700.070	1 000 016	202 227	44.250/
Risk Management	1 284 838	1,002,310	293,237	41.35%
Procurement	1,204,030	1,370,007	95,049 866 / 10	208.09%
Total Administrative	2,410,289	3,663,794	1,253,505	52.01%
Information Technology:	0.000.000	0 707 570	0.004.050	05 400/
Information Technology	3,623,220	6,707,573	3,084,353	85.13%
Total information Technology	3,023,220	6,707,573	3,064,353	05.13%
Finance:				
Financial Accounting and Reporting	1,797,123	2,241,218	444,096	24.71%
Budget and Finance	474,363	562,608	88,244	18.60%
Total Finance	2,271,486	2,803,826	532,340	23.44%
Total Operating Expenses	\$ 88,962,079	\$ 101.898.880	\$ 12,936,801	14.54%
i otai opointing Experises	,,,,,,,,,,,	,,,,,,	,•••,••	

ONTARIO INTERNATIONAL AIRPORT AUTHORITY BUDGET - TOTAL OPERATING EXPENSES BY DIVISION AND DEPARTMENT BUDGET FOR THE YEAR ENDING 2023-24 VS 2022-23

	Adopted	% of Total	Proposed	% of Total
	FY 2023	Operating Budget	FY 2024	Operating Budget
Operations:				
Operations	\$ 782.874	0.9%	\$ 706.476	0.7%
Public Safety	φ 702,074 21.607.162	24.3%	φ 700,470 23 806 860	23.4%
Airfield Operations	5 667 507	6.4%	25,000,009 6 6/6 67/	6.5%
Security	6,000,307	0.4 % 6 0%	6 307 407	6.3%
Emergency Management	620 253	0.5%	623 675	0.5%
Vohiolo and Equipment Maintonanco	2 072 717	0.7 /0	2 284 157	0.070
Lendeide Operations	2,073,717	2.370	2,304,137	2.370
	7,740,121	0.7%	0,300,733	0.2%
Customer Experience	1,480,901	1.7%	1,875,773	1.8%
Total Operations	46,081,428	51.9%	50,821,764	49.8%
Capital Development:				
Planning	6,217,062	7.0%	6,428,603	6.3%
Project Management	2,218,998	2.5%	2,502,690	2.5%
Total Capital Development	8,436,060	9.5%	8,931,293	8.8%
Revenue Management				
Revenue Management Donartmont	2 6/1 501	2 00/	2 625 075	2 60/
	2,041,301	3.0%	2,035,075	2.070
Commercials Real Estate	020,002	0.9%	040,020	0.0%
	7,070,409	0.0%	7,003,328	7.4%
Film Services	157,500	0.2%	42,500	0.0%
lotal Revenue Management	10,701,022	12.1%	11,081,423	10.8%
Marketing and Communications:				
Marketing and Communication	6,837,231	7.7%	8,721,324	8.6%
Digital	2,010,786	2.3%	2,506,064	2.5%
Total Marketing and Communications	8,848,018	10.0%	11,227,388	11.1%
Executive:				
Executive Office	5 450 479	6.1%	1 005 888	1 9%
Office Administrator	5,450,475	0.170	4,995,000	4.970
Air Service Development	676 200	0.90/	204,149	0.3 %
All Service Development	070,290	0.6%	677,200	0.9%
	403,707 6 590 557	0.3%	524,529 6 661 819	0.3%
	6,590,557	7.4/0	0,001,019	0.0 %
Administrative:				
Human Resources	709,079	0.8%	1,002,316	1.0%
Risk Management	1,284,838	1.4%	1,378,687	1.4%
Procurement	416,372	0.5%	1,282,791	1.3%
Total Administrative	2,410,289	2.7%	3,663,794	3.7%
Information Technology:				
Information Technology	3 623 220	1 1%	6 707 573	6.6%
Total Information Technology	5,025,220	4.170	0,101,515	0.070
rotal mormation recimology				
Finance:				
Financial Accounting and Reporting	1,797,123	2.0%	2,241,218	2.2%
Budget and Finance	474,363	0.5%	562,608	0.6%
Total Finance	2,271,486	2.5%	2,803,826	2.8%
Total Operating Expenses	\$ 88,962,079	100.0%	\$ 101,898,880	100.0%

							P	opos	sed Budget F	YE :	2024						
				Revenue				lı	nformation				Capital	Ν	Marketing and		
	C	Operations	M	anagement	Executive	Ad	Iministrative	Т	echnology		Finance	De	evelopment	Co	ommunications		Total
Personnel																	
Salaries, wages and overtime	\$	3,688,164	\$	1,173,173	\$ 1,732,934	\$	1,481,372	\$	1,121,189	\$	1,406,745	\$	1,392,728	\$	1,693,768	\$	13,690,071
Employee benefits and taxes		1,087,298		278,486	457,385		434,023		339,438		406,813		404,993		503,506		3,911,943
Total personnel		4,775,462		1,451,659	2,190,319		1,915,394		1,460,627		1,813,558		1,797,720		2,197,275	-	17,602,014
Nonpersonnel																	
Public safety		23,293,289		-	-		-		-		-		-		-		23,293,289
Contractual services		13,299,673		8,580,002	3,563,485		330,870		3,537,946		785,792		7,007,873		2,589,100		39,694,741
Insurance and administration		13,400		411,500	197,740		1,129,900		-		111,351		2,500		91,000		1,957,391
Materials and supplies		1,309,817		25,000	29,500		17,500		1,306,000		6,000		6,000		-		2,699,817
Telecommunications and utilities		6,963,304		-	-		-		368,000		-		-		-		7,331,304
Other operating expenses		1,166,819		613,262	680,775		270,130		35,000		87,125		117,200		6,350,013		9,320,324
Total nonpersonnel		46,046,302		9,629,764	4,471,500		1,748,400		5,246,946		990,268		7,133,573		9,030,113		84,296,866
Total operating expenses	\$	50,821,764	\$	11,081,423	\$ 6,661,819	\$	3,663,794	\$	6,707,573	\$	2,803,826	\$	8,931,293	\$	11,227,388	\$	101,898,880
Headcount Proposed		40.50		10.00	13.50		14.00		13.00		14.00		10.00		15.50		130.50

							Α	dopt	ted Budget F	YE 2	023					
				Revenue				lr	nformation				Capital		Marketing and	
	C	Operations	Ма	anagement	Executive	Ad	ministrative	T	echnology		Finance	De	evelopment	Co	Communications	Total
Personnel																
Salaries, wages and overtime	\$	3,085,505	\$	1,130,095	\$ 1,908,074	\$	633,933	\$	105,900	\$	1,016,298	\$	1,295,921	\$	1,218,777	\$ 10,394,503
Employee benefits and taxes		834,648		271,423	465,171		176,031		27,584		273,728		337,491		278,819	2,664,896
Total personnel		3,920,153		1,401,518	2,373,246		809,964		133,484		1,290,026		1,633,412		1,497,596	13,059,398
Nonpersonnel																
Public safety		21,175,717		-	-		-		-		-		-		-	21,175,717
Contractual services		11,876,466		8,112,938	3,303,670		314,750		2,458,736		795,696		6,700,248		1,989,972	35,552,476
Insurance and administration		25,600		751,825	150,000		1,101,000		-		98,536		-		20,500	2,147,461
Materials and supplies		1,207,887		30,500	2,000		10,900		657,500		5,000		6,000		-	1,919,787
Telecommunications and utilities		6,959,600		-	-		-		333,500		-		-		-	7,293,100
Other operating expenses		916,005		404,241	761,641		173,675		40,000		82,228		96,400		5,339,950	7,814,140
Total nonpersonnel		42,161,275		9,299,504	4,217,311		1,600,325		3,489,736		981,460		6,802,648		7,350,422	75,902,681
Total operating expenses	\$	46,081,428	\$	10,701,022	\$ 6,590,557	\$	2,410,289	\$	3,623,220	\$	2,271,486	\$	8,436,060	\$	8,848,018	\$ 88,962,079
Headcount Approved		35.00		10.00	13.50		11.00		7.00		12.00		10.00		14.50	113.00

ONTARIO INTERNATIONAL AIRPORT AUTHORITY BUDGET - OPERATIONS DIVISION FOR THE YEAR ENDING JUNE 30, 2024

Proposed FY 2024 Budget

	Operations	Public Safety	Airfield Operations	Security	Emergency Management	Vehicle & Equipment Maintenance	Landside Operations	Customer Experience	Total
	2.5	3.0	-	11.0	1.0	9.5	4.5	9.0	40.50
es and overtime	\$ 527,080	\$ 379,278	\$-	\$ 756,287	\$ 146,639	\$ 774,302	\$ 433,860	\$ 670,718	\$ 3,688,164
nefits and taxes	118,396	82,285	-	276,420	43,592	217,255	107,496	241,855	1,087,298
rsonnel	645,476	461,563	-	1,032,707	190,231	991,557	541,355	912,573	4,775,462
	-	23,293,289	-	-	-	-	-	-	23,293,289
ervices	-	-	6,634,099	5,079,000	101,500	75,000	747,374	662,700	13,299,673
administration	2,500	-	-	-	600	5,000	-	5,300	13,400
supplies	5,000	52,017	-	240,300	225,000	577,000	135,000	75,500	1,309,817
cations and utilities	-	-	-	-	10,800	-	6,952,504	-	6,963,304
ng expenses	53,500	-	12,575	45,400	95,544	735,600	4,500	219,700	1,166,819
npersonnel	61,000	23,345,306	6,646,674	5,364,700	433,444	1,392,600	7,839,378	963,200	46,046,302
erating expenses	\$ 706,476	\$ 23,806,869	\$ 6,646,674	\$ 6,397,407	\$ 623,675	\$ 2,384,157	\$ 8,380,733	\$ 1,875,773	\$ 50,821,764
ease) vs FY 2023 Adopted	\$ (76,398)	\$ 2,199,706	\$ 979,167	\$ 297,514	\$ (5,578)	\$ 310,440	\$ 640,612	\$ 394,873	\$ 4,740,335
rease) vs FY 2023 Adopted	-9.76%	10.18%	17.28%	4.88%	-0.89%	14.97%	8.28%	26.66%	10.29%

Headcount

Personnel Salaries, wage Employee ber

Total per

Nonpersonnel Public safety Contractual se Insurance and Materials and Telecommunio Other operatir Total no

Total op

\$ Increase (Decr % Increase (Decrease) vs FY 2023 Adopted

				Adopte	d FY 2023 Budg	jet			
Operations	S	Public Safety	Airfield Operations	Security	Emergency Management	Vehicle & Equipment Maintenance	Landside Operations	Customer Experience	Total
	2.5	2.0	-	9.0	1.0	8.0	3.5	9.0	35.00
502 96	,025 ,099	315,864 63,564	-	588,678 198,195	138,276 39,822	665,236 183,881	228,592 51,441	646,834 201,646	\$ 3,085,505 834,648
598	8,124	379,428	-	786,873	178,098	849,117	280,032	848,481	3,920,15
	-	21,175,717	-	-	-	-	-	-	21,175,717
99	,000	-	5,654,932	5,062,750	140,500	50,000	405,589	463,695	11,876,466
15	,000	-	-	-	600	3,000	-	7,000	25,600
25	,000	52,017	-	204,870	225,000	601,000	100,000	-	1,207,887
	-	-	-	-	9,600	-	6,950,000	-	6,959,600
45	,750	-	12,575	45,400	75,455	570,600	4,500	161,725	916,005
184	,750	21,227,734	5,667,507	5,313,020	451,155	1,224,600	7,460,089	632,420	42,161,275
\$ 782	,874	\$ 21,607,162	\$ 5,667,507	\$ 6,099,893	\$ 629,253	\$ 2,073,717	\$ 7,740,121	\$ 1,480,901	\$ 46,081,428

Headcount Approved

Personnel Salaries, wages and overtime Employee benefits and taxes

Total personnel

Nonpersonnel Public safety

Contractual services Insurance and administration Materials and supplies Telecommunications and utilities Other operating expenses

Total nonpersonnel

Total operating expenses

26.66%

10.29%

ONTARIO INTERNATIONAL AIRPORT AUTHORITY BUDGET - REVENUE MANAGEMENT DIVISION FOR THE YEAR ENDING JUNE 30, 2024

r

		FI	Sposed FT 2024 Buu	yei	
	Revenue Management Department	Film Services	Ground Transportation	Commercials Real Estate	Total
Headcount	7.0	-	2.0	1.0	10.00
Personnel					
Salaries, wages and overtime	826,865	-	212,400	133,908	\$ 1,173,173
Employee benefits and taxes	205,723	-	36,751	36,012	278,486
Total personnel	1,032,588	-	249,151	169,920	1,451,659
Nonpersonnel Public safety Contractual services Insurance and administration Materials and supplies Other operating expenses Total nonpersonnel	994,620 11,500 15,000 581,367 1,602,487	42,500 - - - 4 2,500	7,287,882 - 10,000 16,295 7,314,177	255,000 400,000 - 15,600 670,600	8,580,002 411,500 25,000 613,262 9,629,764
Total operating expenses	\$ 2,635,075	\$ 42,500	\$ 7,563,328	\$ 840,520	\$ 11,081,423
\$ Increase (Decrease) vs FY 2023 Adopted	\$ (6,426)	\$ (115,000)	\$ 486,859	\$ 14,968	\$ 380,402
% Increase (Decrease) vs FY 2023 Adopted	-0.24%	-73.02%	6.88%	1.81%	3.55%

	Adopted FY 2023 Budget				
	Revenue Management Department	Film Services	Ground Transportation	Commercials Real Estate	Total
Headcount Original Adopted FY2023 Budget Headcount Approved	5.0 7.0	-	2.0 2.0	2.0 1.0	9.00 10.00
Personnel					
Salaries, wages and overtime	708,586	-	186,061	235,449	\$ 1,130,095
Employee benefits and taxes	149,224	-	67,695	54,503	271,423
Total personnel	857,810	-	253,756	289,952	1,401,518
Nonpersonnel Public safety	-	-	_	-	
Contractual services	1,043,020	157,500	6,792,418	120,000	8,112,938
Insurance and administration	351,825	-	-	400,000	751,825
Materials and supplies	15,000	-	15,500	-	30,500
Other operating expenses	373,846	-	14,795	15,600	404,241
Total nonpersonnel	1,783,691	157,500	6,822,713	535,600	9,299,504
Total operating expenses	\$ 2,641,501	\$ 157,500	\$ 7,076,469	\$ 825,552	\$ 10,701,022
ONTARIO INTERNATIONAL AIRPORT AUTHORITY BUDGET - EXECUTIVE DIVISION FOR THE YEAR ENDING JUNE 30, 2024

PROPOSED BUDGET

		Pro	posed FY 2024 Bu	aget	
	Executive Office	Air Service	Government	Office	Total
	Executive Office	Development	Relations	Authinistrator	TOLAI
Headcount	8.5	3.0	1.0	1.0	13.50
Personnel					
Salaries, wages and overtime	990.408	431.510	187.643	123.372	\$ 1.732.934
Employee benefits and taxes	271,280	117,943	42,386	25,777	457,385
Total personnel	1,261,688	549,453	230,029	149,149	2,190,319
Nonpersonnel Public safety Contractual services Insurance and administration Materials and supplies Telecommunications and utilities Other operating expenses Total nonpersonnel	3,189,700 137,500 - 407,000 3,734,200	- 89,285 240 2,000 - 236,275 327,800	- 284,500 - - 10,000 294,500	- 60,000 27,500 - 27,500 115,000	3,563,485 197,740 29,500 - 680,775 4,471,500
Total operating expenses	\$ 4,995,888	\$ 877,253	\$ 524,529	\$ 264,149	\$ 6,661,819
\$ Increase (Decrease) vs FY 2023 Adopted	\$ (454,591)	\$ 200,963	\$ 60,742	\$ 264,149	\$ 71,262
% Increase (Decrease) vs FY 2023 Adopted	-8.34%	29.72%	13,10%	100.00%	1.08%

Adopted FY 2023 Budget									
Executive Office	Air Service Development	Government Relations	Office Administrator	Total					
9.5	3.0	1.0	-	13.50					
1,490,874	277,200	140,000	-	\$ 1,908,074					
369,605	63,779	31,787	-	465,171					
1,860,479	340,979	171,787	-	2,373,246					
-	-	-	-	-					
2,960,000	59,170	284,500	-	3,303,670					
150,000	-	-	-	150,000					
-	2,000	-	-	2,000					
-	-	-	-	-					
480,000	274,141	7,500	-	761,641					
3,590,000	335,311	292,000	-	4,217,311					
¢ 5 450 470	¢ 676 200	¢ 463 797	¢	¢ 6 600 667					
	ə 676,290	ə 463,787	р -	¢,590,557 چ					

Headcount	Approved
nouaooune	, .pp: 010a

Personnel					
Salaries, wages and overtime					
Employee benefits and taxes					
Total personnel					

Nonpersonnel

Public safety Contractual services Insurance and administration Materials and supplies Telecommunications and utilities Other operating expenses

Total nonpersonnel

Total operating expenses

ONTARIO INTERNATIONAL AIRPORT AUTHORITY BUDGET - ADMINISTRATIVE DIVISION FOR THE YEAR ENDING JUNE 30, 2024

BUDGET SFD Proposed FY2024 Budget

	Hui Reso	man urces	Risk Management	Pi	rocurement	Total
Headcount		4.0	2.0		8.0	14.00
Personnel	Į					
Salaries, wages and overtime	\$	435,395	\$ 159,531	\$	886,446	\$ 1,481,372
Employee benefits and taxes		137,981	55,172		240,869	434,023
Total personnel		573,376	214,702	2	1,127,316	1,915,394
Nonpersonnel Public safety Contractual services Insurance and administration Materials and supplies Telecommunications and utilities Other operating expenses Total nonpersonnel		- 247,540 - 15,000 - 166,400 428,940	25,330 1,129,300 - - 9,355 1,163,985		58,000 600 2,500 - 94,375 155,475	330,870 1,129,900 17,500 - 270,130 1,748,400
Total operating expenses	\$ 1	,002,316	\$ 1,378,687	\$	1,282,791	\$ 3,663,794
\$ Increase (Decrease) vs FY 2023 Adopted	\$	293,237	\$ 93,849	\$	866,419	\$ 1,253,505
% Increase (Decrease) vs FY 2023 Adopted		41.35%	7.30%	,	208.09%	52.01%

	Adopted FY2023 Budget							
	Re	Human esources	N	Risk Ianagement	P	rocurement		Total
Headcount Approved		4.0		1.0		6.0		11.00
Personnel								
Salaries, wages and overtime	\$	213,752	\$	119,691	\$	300,489	\$	633,933
Employee benefits and taxes		61,677		37,397		76,957		176,031
Total personnel		275,429		157,088		377,447		809,964
Nonpersonnel Public safety Contractual services		- 280,250		- 18,500		- 16,000		- 314,750
Insurance and administration		-		1,100,000		1,000		1,101,000
Materials and supplies Telecommunications and utilities Other operating expenses		8,400 - 145,000		- - 9,250		2,500 - 19,425		173,675
i otal nonpersonnel Total operating expenses	\$	433,650	\$	1,127,750	\$	38,925 416,372	\$	2,410,289

ONTARIO INTERNATIONAL AIRPORT AUTHORITY BUDGET - INFORMATION TECHNOLOGY DIVISION FOR THE YEAR ENDING JUNE 30, 2024

Proposed FY2024 Budget

	Proposed FY2024 Budget					
	Information Technology			Total		
Headcount		13.0		13.00		
Personnel						
Salaries, wages and overtime	\$	1,121,189	\$	1,121,189		
Employee benefits and taxes		339,438	·	339,438		
Total personnel		1,460,627		1,460,627		
Nonpersonnel Public safety Contractual services Insurance and administration Materials and supplies Telecommunications and utilities Other operating expenses Total nonpersonnel		3,537,946 - 1,306,000 368,000 35,000 5,246,946		3,537,946 1,306,000 368,000 35,000 5,246,946		
Total operating expenses	\$	6,707,573	\$	6,707,573		
\$ Increase (Decrease) vs FY 2023 Adopted	\$	3,084,353	\$	3,084,353		
% Increase (Decrease) vs FY 2023 Adopted		85.13%		85.13%		

	Adopted FY 2023 Budget					
	I T	nformation echnology		Total		
Headcount Approved		7.0		7.00		
Personnel						
Salaries, wages and overtime	\$	105,900	\$	105,900		
Employee benefits and taxes		27,584		27,584		
Total personnel		133,484		133,484		
Nonpersonnel						
Public safety		-		-		
Contractual services		2,458,736		2,458,736		
Insurance and administration		-		-		
Materials and supplies		657,500		657,500		
Telecommunications and utilities		333,500		333,500		
Other operating expenses		40,000		40,000		
Total nonpersonnel		3,489,736		3,489,736		
Total operating expenses	\$	3,623,220	\$	3,623,220		

PROPOSED BUDGET

Proposed FY 2024 Budget								
	F Acc Re	inancial counting & eporting		Budget & Finance		Total		
Headcount		11.0		3.0		14.00		
Personnel								
Salaries, wages and overtime	\$	1,141,565	\$	265,180	\$	1,406,745		
Employee benefits and taxes		336,988		69,825		406,813		
Total personnel		1,478,552		335,005		1,813,558		
Nonpersonnel Public safety Contractual services Insurance and administration Materials and supplies Telecommunications and utilities Other operating expenses Total nonpersonnel		- 655,126 19,415 6,000 - 82,125 762,666		- 130,666 91,936 - 5,000 227,602		785,792 111,351 6,000 - 87,125 990,268		
Total operating expenses	\$	2,241,218	\$	562,608	\$	2,803,826		
\$ Increase (Decrease) vs FY 2023 Adopted	\$	444,096	\$	88,244	\$	532,340		
% Increase (Decrease) vs FY 2023 Adopted		24.71%		18.60%		23.44%		

Adopted FY 2023 Budget								
	Financial Accounting & Reporting			Budget & Finance	Total			
Headcount Approved		10.0		2.0		12.00		
Personnel	¢	810 620	¢	106 667	¢	1 016 208		
Employee benefits and taxes	φ	223,634	φ	50,094	φ	273,728		
Total personnel		1,043,265		246,761		1,290,026		
Nonpersonnel								
Public safety Contractual services		- 665.030		- 130 666		- 795 696		
Insurance and administration		6,600		91,936		98,536		
Materials and supplies		5,000		-		5,000		
Telecommunications and utilities		-		-		-		
Other operating expenses		77,228		5,000		82,228		
Total nonpersonnel		753,858		227,602		981,460		
Total operating expenses	\$	1,797,123	\$	474,363	\$	2,271,486		

ONTARIO INTERNATIONAL AIRPORT AUTHORITY BUDGET - CAPITAL DEVELOPMENT DIVISION FOR THE YEAR ENDING JUNE 30, 2024

SED BUDGET Proposed FY 2024 Budget

	Project	Diamina	Total
	wanagement	Planning	TOTAL
Headcount	4.0	6.0	10.00
Personnel			
Salaries, wages and overtime	\$ 552,333	\$ 840,395	\$ 1,392,728
Employee benefits and taxes	185,357	219,635	404,993
Total personnel	737,690	1,060,030	1,797,720
Nonpersonnel Public safety Contractual services Insurance and administration Materials and supplies Telecommunications and utilities Other operating expenses Total nonpersonnel	1,719,000 - 6,000 - 40,000 1,765,000	5,288,873 2,500 - - 77,200 5,368,573	7,007,873 2,500 6,000 - 117,200 7,133,573
Total operating expenses	\$ 2,502,690	\$ 6,428,603	\$ 8,931,293
\$ Increase (Decrease) vs FY 2023 Adopted	\$ 283,692	\$ 211,541	\$ 495,233
% Increase (Decrease) vs FY 2023 Adopted	12.78%	3.40%	5.87%

	Adopted FY 2023 Budget							
	Program Management	Planning	Total					
Headcount Approved	4.0	6.0	10.0					
Personnel Salaries, wages and overtime Employee benefits and taxes	\$ 501,235 152,763	\$ 794,686 184,728	\$ 1,295,921 337,491					
Nonpersonnel	653,996	979,414	1,633,412					
Public safety Contractual services Insurance and administration	- 1,519,000 -	- 5,181,248 -	- 6,700,248 -					
Materials and supplies Telecommunications and utilities	6,000 -	-	6,000 -					
Other operating expenses Total nonpersonnel	40,000 1,565,000	56,400 5,237,648	96,400 6,802,648					
Total operating expenses	\$ 2,218,998	\$ 6,217,062	\$ 8,436,060					

ONTARIO INTERNATIONAL AIRPORT AUTHORITY

BUDGET - MARKETING AND COMMUNICATIONS DIVISION ROPOSED BUDGET

	Proposed FY 2024 Budget				
	Marketing and Communication	Digital	Total		
Headcount	11.5	4.0	15.50		
Personnel					
Salaries, wages and overtime	\$ 1,220,224	\$ 473,544	\$ 1,693,768		
Employee benefits and taxes	357,187	146,319	503,506		
Total personnel	1,577,411	619,864	2,197,275		
Nonpersonnel Public safety Contractual services Insurance and administration Materials and supplies Other operating expenses Total nonpersonnel	745,000 85,500 - 6,313,413 7,143,913	- 1,844,100 5,500 - 36,600 1,886,200	2,589,100 91,000 6,350,013 9,030,113		
Total operating expenses	\$ 8,721,324	\$ 2,506,064	\$ 11,227,388		
\$ Increase (Decrease) vs FY 2023 Adopted	\$ 1,884,093	\$ 495,277	\$ 2,379,370		
% Increase (Decrease) vs FY 2023 Adopted	27.56%	24.63%	26.89%		

	Adopted FY 2023 Budget				
	Marketing and Communication	Digital	Total		
Headcount Approved	10.5	4.0	14.5		
Personnel					
Salaries, wages and overtime	\$ 852.564	\$ 366.213	\$ 1.218.777		
Employee benefits and taxes	196,117	82,701	278,819		
Total personnel	1,048,681	448,914	1,497,596		
Nonpersonnel					
Public safety	-	-	-		
Contractual services	450,000	1,539,972	1,989,972		
Insurance and administration	15,000	5,500	20,500		
Materials and supplies	-	-	-		
Other operating expenses	5,323,550	16,400	5,339,950		
Total nonpersonnel	5,788,550	1,561,872	7,350,422		
Total operating expenses	\$ 6,837,231	\$ 2,010,786	\$ 8,848,018		



Public Safety Supervisor

Public Safety Specialist

Security Contractor

NEW FYE24

Airport Badging Specialist I

NEW FYE24

ARFF/OPD



Page 227

THIS PAGE INTENTIONALLY LEFT BLANK

ONTARIO INTERNATIONAL AIRPORT AUTHORITY



MEETING DATE: JUNE 22, 2023

AGENDA SECTION: ADMINISTRATIVE REPORTS/DISCUSSION/ACTION

SUBJECT: RESOLUTION TO ADOPT THE FYE 2024 CAPITAL IMPROVEMENT PROGRAM BUDGET

RELEVANT STRATEGIC OBJECTIVE: Invest in ONT Invest in ONT

RECOMMENDED ACTION(S): Approve a resolution adopting the FYE 2024 Capital Improvement Program and Budget.

FISCAL IMPACT SUMMARY: Total cost of the FYE 2024 Capital Improvement Program Budget is \$35,480,000.

BACKGROUND: The Ontario International Airport is a commercial service airport, and its capital requirements are substantial with a continuous need to maintain, acquire and construct infrastructure, facilities, equipment, and technology. Capital requirements are driven by several factors including safety and security, operational needs, customer demands and expectations, air service and commercial development. These factors are impacted by Board of Commissioners goals and objectives, FAA and other governmental laws, regulations and policies, and the Airline Operating Use and Lease Agreement (ULA).

The Capital Budget is comprised of Projects that are developed individually and within the scope of a comprehensive near and long-term Capital Improvement Program (CIP). They require significant financial resources that must be balanced through airline and tenant rates and charges, business development and financing.

The Ontario International Airport Authority utilizes a Project Development process. Annually OIAA staff submit Project Requests. These projects undergo a high-level review and approval process.

Projects that successfully complete this process are incorporated into the OIAA Capital Improvement Program. The CIP does not authorize the expenditure of funds on Projects. Authorization to incur expenditures is completed through the development and Board approval of a Capital Budget. Projects are budgeted in the fiscal year in which expenditures on those Projects are first expected to be incurred, although Project expenditures may span several fiscal years. Expenditures on these Projects are subject to OIAA purchasing policies, which may include competitive bidding for products and services necessary to complete the project as well as Board approval of contracts awarded. In addition to Board authorization and OIAA policies airport projects are subject to certain limitations as part of the Airline Operating Use and Lease (ULA) agreement between the OIAA and signatory airlines. Table 1 FY2024 Capital Improvement Program Projects identifies these projects, cost estimates and budget authorization request. Following the table is a report which describes each project in detail.

Projects that exceed the dollar limit thresholds in the ULA, net of grants and passenger facility charges and as adjusted by CPI annually, require approval by a majority of signatory airlines that are party to the ULA, which are referred to as Majority-In-Interest (MII) Projects. For FY 2024 those MII limits are \$507,136 per Project and \$2,028,545 for all Projects below the MII maximum.

In accordance with the ULA certain Projects are exempt from these requirements, including assets destroyed through fire, natural disaster or accidental destruction, assets that require repair or replacement to maintain continuity of operations and are of an emergency nature and those Projects or improvements that ensure compliance with a rule, regulation, or order of any federal, state, or other governmental body.

Capital project costs are financed through passenger facility charges, grants, airport rates and chargers, OIAA reserves and debt. They are recovered through depreciation and debt service corresponding to the life of the asset and or the terms of the underlying debt.

The FY 2024 Capital Improvement Program includes 16 Projects. The Budget Authorization request for these projects is \$35,480,000, which includes an additional budget request of \$4,730,000 for the Administration Office Reconfiguration and Security Screening Checkpoint Expansion projects. Six of these Projects totaling \$26,165,000 require MII approval.

The FY 2024 Capital Budget is anticipated to be financed through the revolving credit facility, a FY 2025 revenue bond issue and OIAA reserves. Grant funding has not been secured for any of these projects, however OIAA continues to explore grant opportunities. Project costs are allocated to either the Airside or Terminal cost centers and will be recovered on a residual basis according to the ULA in the form of landing fees and terminal rents.

PROCUREMENT: Project products, professional and construction services will be procured through Requests for Proposals. Contracts will be negotiated with the selected Proposers and brought to the Board of Commissioners for approval.

CEQA COMPLIANCE AND LAND USE APPROVALS: CEQA and land use approval requirements are assessed and completed prior to awarding contracts.

STAFFING IMPACT (# OF POSITIONS): N/A.

IMPACT ON OPERATIONS: Varies by Project.

SCHEDULE: N/A

ATTACHMENTS:

- 1. Resolution No. 2023-13
- 2. Ontario International Airport FY 2024 Capital Improvement Program and Budget

STAFF MEMBER PRESENTING: Chief Executive Officer Atif Elkadi

Originator Name:	John M. Schubert, Chief Financial Officer
Originating Dept.:	Finance
Director Review:	N/A Allos
Chief Review:	John M. Schubert, Chief Financial Officer
CEO Approval:	A^{2}

This Agenda Report has been reviewed by OIAA General Counsel.

The Agenda Report references the terms and conditions of the recommended actions and request for approval. Any document(s) referred to herein, which are not attached or posted online, may be reviewed prior to, or following scheduled Commission meetings in the Office of the Clerk of the Board. Office hours are 8:30 a.m. to 5:00 p.m., Monday through Friday, although these hours and review procedures may be modified. In that case, the documents may be requested by email at <u>clerk@flyontario.com</u>.

BOARD DISPOSITION: Approved Denied Continued to

A RESOLUTION OF THE ONTARIO INTERNATIONAL AIRPORT AUTHORITY ADOPTING THE ANNUAL CAPITAL BUDGET FOR FISCAL YEAR 2023-24

WHEREAS, the Ontario International Airport Authority ("the Authority" or "OIAA") was established under a Joint Exercise of Powers Agreement between the City of Ontario and the County of San Bernardino (the "Joint Powers Agreement") pursuant to the purpose of operating, maintaining, developing, and marketing the Ontario International Airport (the "Airport"); and

WHEREAS, under the Joint Powers Agreement, Section 10, Funds, the Treasurer of the Authority shall assume the duties described in California Government Code Section 6505.5; and

WHEREAS, the Authority staff, including the Chief Executive Officer ("CEO") and the Chief Financial Officer ("CFO") of the Authority, have determined the Authority's Capital Budget for Fiscal Year 2023-24 in accordance with the provisions of the Joint Powers Agreement; and

WHEREAS, the Authority staff have consulted with the Airport and Airline Affairs Committee ("AAAC") in regards to this proposed Capital Budget for Fiscal Year 2023-24 and the AAAC has not expressed any objections to its adoption or implementation.

NOW, THEREFORE, BE IT RESOLVED by the Ontario International Airport Authority Commission as follows:

SECTION 1. The Ontario International Airport Authority Commission hereby adopts the proposed Fiscal Year 2023-24 Capital Budget, which is attached as Exhibit "A", and by this reference incorporated.

SECTION 2. Effective Date. This Resolution will take effect immediately upon its adoption.

SECTION 3. Certification. The Secretary/Assistant Secretary shall certify as to the adoption of this Resolution.

PASSED, APPROVED, AND ADOPTED at a Regular Meeting this 22nd day of June, 2023.

ALAN D. WAPNER, OIAA PRESIDENT

ATTEST:

APPROVED AS TO LEGAL FORM:

NORMA I. ALLEY, MMC, ASSISTANT SECRETARY

STATE OF CALIFORNIA COUNTY OF SAN BERNARDINO CITY OF ONTARIO

I, Norma I. Alley, MMC, Board Clerk/Assistant Secretary of the Ontario International Airport Authority, DO HEREBY CERTIFY the foregoing Resolution No. 2023-13 is the original and was duly passed and adopted by the Commission of the Ontario International Airport Authority at their Regular Meeting held June 22, 2023, by the following roll call vote, to wit:

)

)

AYES: COMMISSIONERS:

NOES: COMMISSIONERS:

ABSENT: COMMISSIONERS:

NORMA I. ALLEY, MMC BOARD CLERK/ASSISTANT SECRETARY

(SEAL)

2024 Proposed Capital Improvements Proposal to the OIAA Board of Commissioners



INTERNATIONAL AIRPOR
—— So Cal. So Easy. ——

TABLE OF CONTENTS

1.	Proje	cts Ove	rview	1-1
2.	Proje	ct Desci	riptions	2-1
	2.1	Airside/	ACIP Projects	2-1
		2.1.1	New Airport Beacon	2-1
	2.2	Landsid	e Projects	2-2
		2.2.1	Terminal 2 and 4 Curb Improvements	2-2
		2.2.2	Terminal 2 and 4 Curbside Signage	2-2
	2.3	Termina	al Projects	2-4
		2.3.1	Existing FIS Improvements	2-4
		2.3.2	Advertising Display Replacment	2-4
		2.3.3	Global Entry Enrollment Center	2-4
	2.4	Facilitie	s Projects	2-6
		2.4.1	Tenant Metering within Terminal Building	2-6
		2.4.2	Permanent Bus Charging Station and Parking Operator Offices	2-6
		2.4.3	Police Terminal Offices	2-6
		2.4.4	Terminals 2 and 4 Fire Alarm System Replacement	2-7
	2.5	Security	/ Projects	2-8
		2.5.1	South Security Airport Access Point (SSAAP)	2-8
	2.6	Equipm	ent Purchases	2-9
		2.6.1	OIAA Fleet Vehicles	2-9
	2.7	Informa	ition Technology Projects	2-10
		2.7.1	Server Room Relocation	2-10
	2.8	Environ	mental Projects	2-11
		2.8.1	Water Consumption Reduction	2-11
3.	Addit	tional Fu	unding Requests	3-1
		3.1.1	Administration Office Reconfiguration	3-1
		3.1.2	Security Screening Checkpoint Expansion Project	3-1

LIST OF TABLES

1. PROJECTS OVERVIEW

The projects requested for the FY 2024 Capital Improvement Program (CIP) at ONT are summarized in the following table. Brief project descriptions can be found in Section 2 of this document.

TABLE 1: FY 2024 CAPITAL IMPROVEMENT PROGRAM PROJECTS

Section	Project	Duration	Cost Magnitude (Rounded to Nearest 1,000)
2.1	Airside/ACIP Projects		
2.1.1	New Airport Beacon	3 months	\$250,000
	Sub-total		\$250,000
2.2	Landside Projects		
2.2.1	Terminal 2 and 4 Curb Improvements	1 year	\$2,000,000
2.2.2	Terminal 2 and 4 Curbside Signage	1 year	\$2,500,000
	Sub-total		\$4,500,000
2.3	Terminal Projects		
2.3.1	Existing FIS Improvements	2 years	\$10,000,000
2.3.2	Advertising Display Replacement	3 months	\$500,000
2.3.3	Global Entry Enrollment Center	3 months	\$200,000
	Sub-total		\$10,700,000
2.4	Facilities Projects		
2.4.1	Tenant Metering within Terminal Building	6 months	\$300,000
2.4.2	Permanent Bus Charging Station and Parking Operator Offices	2 years	\$4,500,000
2.4.3	Police Terminal Offices	1 year	\$250,000
2.4.4	Terminals 2 and 4 Fire Alarm System Replacement	1 year, 6 months	\$2,700,000
	Sub-total		\$7,750,000
2.5	Security Projects		
2.5.1	South Security Airport Access Point (SSAAP)	2 years	\$3,500,000
	Sub-total		\$3,500,000
2.6	Equipment Purchases		
2.6.1	OIAA Fleet Vehicles	1 month	\$300,000
	Sub-total		\$300,000
2.7	Information Technology Projects		
2.7.1	Server Room Relocation	3 months	\$500,000
	Sub-total		\$500,000
2.8	Environmental Projects		
2.8.1	Water Consumption Reduction	2 years, 6 months	\$3,250,000
	Sub-total		\$3,250,000
3	Additional Funding Requests		
3.1.1	Administration Office Reconfiguration	1 year	\$815,000
3.1.2	Security Screening Checkpoint Expansion Project	1 year	\$8,255,000
	Sub-total		\$9,070,000
	Total FY 2024 Recommended Capital Imp	provement Program	\$35,480,000*

2. **PROJECT DESCRIPTIONS**

2.1 AIRSIDE/ACIP PROJECTS

2.1.1 NEW AIRPORT BEACON

Ontario International Airport (ONT or the Airport) is currently operating with an old airport beacon that has exceeded its useful life and often requires parts replacement. Due to the age of the beacon maintenance staff are challenged to find parts, sometimes being completely unavailable requiring custom fabrication of a single part. Ontario International Airport Authority (OIAA) staff recommend the purchase and installation of a new Airport beacon to ensure continued operation of the visual navigation aid.

Duration: 3 months

Cost Magnitude: \$250,000

2.2 LANDSIDE PROJECTS

2.2.1 TERMINAL 2 AND 4 CURB IMPROVEMENTS

This project involves a redesign of Terminals 2 and 4 curbs to allow for an extended/longer area for TNC pick-ups. The three lanes of the terminal curbs would be extended to the east creating additional lane frontage for TNCs to park while picking-up passengers. Additional vehicle through-put and storage capacity would be provided in the extended sections of the center and outer lanes. This project would serve as a short/mid-term solution until the airport ground transportation center is completed.

Terminals 2 and 4 curbsides experience congestion during peak periods due to the lack of available curb space and the singlelevel roadway mixing enplaning and deplaning curbside operations. To meet the forecasted growth in passenger traffic at the Airport with existing facilities, additional curb frontage will be required.

Duration: 1 year

Cost Magnitude: \$2,000,000



2.2.2 TERMINAL 2 AND 4 CURBSIDE SIGNAGE

The ONT Planning Department has developed an improved terminal curbside signage package. This signage follows the same color scheme and design intent of the proposed interior signage and wayfinding program. The primary signage color will match branding colors spanning the width of the bottom of the sign. Lettering will be high visibility white on blue to meet or exceed ADA requirements.

The terminal curbside signage package includes enhanced terminal identification signage as well as larger terminal entry/exit signage for arrivals and departures to facilitate passenger wayfinding. Signage under the terminal curb canopy will include signage noting terminal identification, check-in, and baggage claim. Airlines will be given their own sign in front of the terminal in which it operates.

The signage along the commercial curb will also be replaced. The commercial space designations will remain the same but the signage identifying the commercial entity/amenity for a particular location will be improved. The signage will span the width of the commercial canopy with messaging left justified.

Duration: 1 year

Cost Magnitude: \$2,500,000



2.3 TERMINAL PROJECTS

2.3.1 EXISTING FIS IMPROVEMENTS

This project will involve various federal inspection station (FIS) improvements for the benefit of FIS employees and the traveling public. While the scope has not been fully defined, improvements will include updating processor areas to meet U.S. Customs and Border Protection (CBP) design guidelines, and industry passenger level of service (LOS) goals. OIAA will explore replacing the bag claim device to allow for more bag claim frontage, as well as improvements to the overall layout of the facility. There are a number of enhancements that are being mandated by local CBP since the timing of a new international arrivals facility is now extended and undetermined.

Duration: 2 years

Cost Magnitude: \$10,000,000

NOTE: This project would span multiple fiscal years. The full cost magnitude would not be incurred in FY 2024.

2.3.2 ADVERTISING DISPLAY REPLACMENT

This project will replace aging LED display screens and TV display screens in the baggage claim and concourse areas of both passenger terminals. The average lifespan of LED displays for this use is from 4 – 7 years and these displays are over 6 years old. Many are showing degradation (bright circles or dark shadows) or complete loss of the display. These screens are a critical component of ONT's Partnership and Advertising Program which generates over \$1M in revenue per year. Any interruption to the ability to push content to these screens would result in lost revenue for the airport.

Duration: 3 months

Cost Magnitude: \$500,000



2.3.3 GLOBAL ENTRY ENROLLMENT CENTER

Offering an Enrollment Center at ONT at the FIS Facility allows OIAA to provide convenient and world-class amenities to our customers while utilizing the current afternoon and early evening slower periods at our International Arrivals terminal. This

will allow the Global Entry interview appointments to be scheduled and occur without adding additional CBP personnel costs to OIAA.

Duration: 3 months

Cost Magnitude: \$500,000

2.4 FACILITIES PROJECTS

2.4.1 TENANT METERING WITHIN TERMINAL BUILDING

OIAA staff are requesting the purchase and installation of a utility metering system in each of the two terminals. The proposed metering system would allow OIAA to accurately track utility usage by tenant and would allow OIAA to bill tenants for their utility usage in Terminal 2 and Terminal 4.

Duration: 6 months

Cost Magnitude: \$300,000

2.4.2 PERMANENT BUS CHARGING STATION AND PARKING OPERATOR OFFICES

This project consists of building electric bus charging stations and a parking operations office central to the Terminal complex area. In 2022 OIAA replaced the parking and rental car shuttle buses with electric buses. To ensure adequate charging of the electric bus fleet a permanent bus charging station is required. Currently there are only two temporary charging stations at the Terminal 1 parking lot near the existing parking operations office. Relocation of the offices and building the charging stations near the terminals reduces the on airport shuttle miles.

Duration: 2 years

Cost Magnitude: \$4,500,000

NOTE: This project would span multiple fiscal years. The full cost magnitude would not be incurred in FY 2024.

2.4.3 POLICE TERMINAL OFFICES

This project includes modernization to the overall appearance, ergonomics, and useability of the police offices within Terminals 2 and 4. The existing office locations will be upgraded because they are centrally located within each terminal, and it is more cost effective than moving spaces.

Improvements will include upgrading the monitors for viewing video and ACAM activations, improved storage for access to forms, and improved security of safety equipment.

The monitors will be larger, wall-mounted, and web-accessible to facilitate viewing surveillance cameras and the OnGuard system. Due to insufficient workspace and limited storage, there is no acceptable place to store or organize routine paperwork, or supplies. The overall footprint of the workstations will be remapped to better optimize the space, as well as provide space for employees to eat away from computer equipment. This will reduce the likelihood of liquid spills damaging equipment.

Upgrades to the offices will limit the need for officers to respond to the police modular substation, allowing them to remain closer to the terminals. Improved video camera surveillance capabilities will allow officers to mitigate problems faster, resulting in fewer delays to other Airport operations.

Duration: 1 year

Cost Magnitude: \$250,000

2.4.4 TERMINALS 2 AND 4 FIRE ALARM SYSTEM REPLACEMENT

This project involves the complete replacement of the fire alarm system in Terminals 2 and 4. The existing fire alarm system is a multiplexed Spectronics product solution, which has exceeded its normal, expected life cycle.

OIAA staff recommend the replacement of the fire alarm system and have identified a proposed solution. The proposed, new fire alarm system includes a fully integrated digital voice evacuation system using the latest Simplex 4120 voice technology that presently exists and would be useable at the Airport (Simplex 4100ES in a 4120-network node and token ring network topology concept). The new system will implement the addressable self-testing speakers, speaker/visuals, and visual only devices where required for a fully code compliant new system.

The new fire alarm system will also include a True Site Workstation (TSW) that will be in the Airport's command/control center in Terminal 2. A connection to the Ontario Fire Department is also required, to allow for a remote supervisory point of annunciation at the Fire Department. A TSW graphical command center with Digital Alarm Communicating Receiver (DACR) with an IP Communicator module will be installed to meet this requirement.

Duration: 1 year, 6 months

Cost Magnitude: \$2,700,000

NOTE: This project would span multiple fiscal years. The full cost magnitude would not be incurred in FY 2024.

2.5 SECURITY PROJECTS

2.5.1 SOUTH SECURITY AIRPORT ACCESS POINT (SSAAP)

This project includes the construction of a new security airport access point on the south side of the Airport. The current south security airport access point (SSAAP) will be impacted by tenant development on the south side of the Airport at and around the existing SSAAP site. Once constructed the existing SSAAP would be decommissioned and the new SSAAP would provide the only means of ingress and egress from the airside.

Duration: 2 years

Cost Magnitude: \$3,500,000

NOTE: This project would span multiple fiscal years. The full cost magnitude would not be incurred in FY 2024.

2.6 EQUIPMENT PURCHASES

2.6.1 OIAA FLEET VEHICLES

The project is a capital purchase allowance for new OIAA vehicles to meet needs required by increased airport operations and to replace aging fleet. Planned vehicles to purchase: Service van, two work trucks, additional replacement vehicles for aging fleet.

Duration: 1 month

Cost Magnitude: \$300,000

2.7 INFORMATION TECHNOLOGY PROJECTS

2.7.1 SERVER ROOM RELOCATION

This project includes the relocation of the airport's Main Distribution Frame (MDF) to increase the size, add sufficient climate control systems, monitoring systems, allow landside access, and other features and systems according to best practices. Currently the server room is located on the south side of Terminal 2 with limited access adjacent to the apron, which may be better utilized as a different space for operators. Proposed relocated server room locations include existing spaces which are currently being explored but one has not yet been selected.

Duration: 6 months

Cost Magnitude: \$500,000

2.8 ENVIRONMENTAL PROJECTS

2.8.1 WATER CONSUMPTION REDUCTION

The airport intends to extend a reclaimed water line (also known as gray water) connection in Archibald north of the railroad tracks down to the airport. The airport intends on aligning the completion of this project at the same time as the Airport Dr. Rehabilitation project. After bringing the connection to the airport campus, gray water could be used for landscape irrigation and extended into the terminals for potential uses for gray water could be the chiller and/or plumbing systems in the terminals.

Use of gray water in the terminals would require additional studies to assess capabilities with existing infrastructure and possible additional treatments to ensure the water is up to acceptable standards. This project would initially consist of a planning effort to examine the current landscaping around the ONT campus and to recommend a campus landscape plan for implementation. Xeriscaping will also be considered as a component of this project.

As regional water resources become more limited, the OIAA should initiate efforts to reduce the airport's potable water consumption. This project will reduce water consumption for the region, assist financially as gray water has lower rates, and add to the environmental sustainability and reduction of carbon footprint.

Duration: 2 years, 6 months

Cost Magnitude: \$3,250,000

NOTE: This project would span multiple fiscal years. The full cost magnitude would not be incurred in FY 2024.

3. ADDITIONAL FUNDING REQUESTS

Previously Approved Projects through MII.

3.1.1 ADMINISTRATION OFFICE RECONFIGURATION

This request is for an additional funding allowance for the OIAA Administration Office Reconfiguration. Previously approved for \$485,000 in the FYE 2023 capital budget, additional staff requirements, complicated phasing, and discovery of unknown conditions during design have driven an estimated additional \$815,000 in potential cost for this project. The current total project cost is estimated to be \$1.3 million. This project will maximize existing space in the administrative building to accommodate staff.

Duration: 9 months

<u>Cost Magnitude:</u> \$815,000 increase for total project cost of \$1.3 million.

3.1.2 SECURITY SCREENING CHECKPOINT EXPANSION PROJECT

This request is for an additional funding allowance for the Security Screening Checkpoint Expansion Project. Previously approved for \$3,085,000 under the FYE 2023 Capital Budget, higher than expected inflation and supply chain issues in construction costs, unknown issues found during design, new requirements identified by TSA, an improved finish level for the exit corridor and the additional of advertising infrastructure to generate revenue, haves drive cost estimate increases for this project in the amount of \$8,255,000 for a total estimated project budget of \$11,340,000. The addition of advertising opportunities in the exit corridors is expected to generate approximately \$495,000 annually.

- Add two screening lanes in Terminal 2 and an additional screening lane in Terminal 4 checkpoints by relocating the exit corridor.
- Relocate the TSA Manager's Station and Screening Room closer to the windows at both Terminal 2 and Terminal 4 to allow almost 20-feet of clear space in the circulation corridor.
- Add another screening lane at the Terminal 2 checkpoint by replacing the Manager's Station and modifying existing screening equipment.
- Replace document check podiums with CAT scanners for a total of 11 units at each checkpoint.
- Add to and/or adjust existing CCTV, electrical systems, communications, and lighting to accommodate new and modified screening lanes at both checkpoints.
- Renovate adjacent TSA breakrooms in both terminals.
- Include new CCTV and Backflow Alarm System at the exit corridor in both terminals.

The passenger security checkpoints for the existing passenger terminals at ONT experience longer than acceptable wait times during the peak periods and are not sufficient to meet increasing passenger volumes. The checkpoints require additional screening lanes to process passengers as demand levels continue to increase and to create a better experience for airport users. Expansion of the checkpoints will provide a total of six screening lanes at both Terminal 2 and Terminal 4. Additionally, the checkpoint expansion project presents an opportunity to improve the circulation corridor immediately after security screening.

Duration: 1 year, 6 months

<u>Cost Magnitude:</u> \$8,255,000 increase for total project cost of \$11,340,000.

NOTE: This project would span multiple fiscal years. The full cost magnitude would not be incurred in FY 2024





ONTARIO INTERNATIONAL AIRPORT AUTHORITY



MEETING DATE: JUNE 22, 2023

AGENDA SECTION: ADMINISTRATIVE REPORTS/DISCUSSION/ACTION

SUBJECT: RESOLUTION APPROVING THE ONTARIO INTERNATIONAL AIRPORT SYSTEM RATES AND CHARGES FOR FISCAL YEAR 2023-2024

RELEVANT STRATEGIC OBJECTIVE: 🛛 Invest in ONT 🖄 Master the Basics 🖄 Plan for the Future

RECOMMENDED ACTION(S): Approve a resolution approving the Airport System Rates and Charges for Fiscal Year 2023-2024 at Ontario International Airport (ONT).

FISCAL IMPACT SUMMARY: Upon OIAA Commission approval, the newly calculated landing fee rate for fiscal year 2023-2024 shall be \$1.83 per 1,000 lbs. of landed weight for signatory carriers and \$2.29 per 1,000 lbs. of landed weight for non-signatory carriers, generating a total of \$16.0M in landing fees. The newly calculated terminal rental rate for fiscal year 2023-2024 shall be \$7.26 per square foot per month for signatory carriers and \$9.08 per square foot per month for non-signatory carriers, generating a total of \$15.7M in terminal rent revenue. The Non-Preferential Terminal Gate Use Fee will be \$344.87, a 9% decrease from the current Non-Preferential Terminal Gate Use Fee, which is estimated to generate \$807K in Gate Use and Jet Bridge Fees. Aircraft parking charges for diversions shall be increased from \$0.40 per 1,000 lbs. of landed weight to \$0.80 per 1,000 lbs. of landed weight.

Parking rates will also be increased between \$3.00 - \$5.00 per lot. The estimated fiscal impact will be an increase of nearly \$5M in annual revenue. Additionally, the minimum General Ground and Facility Rental Rates are increased by the Consumer Price Index (CPI) which will increase our Facility & Land rent revenue by an average of 3.65%. Finally, ancillary fees for property/facilities have been added to recover costs related to lock installation and additional key requests for tenant spaces.

BACKGROUND: The FY2022-2023 Airport System Rates and Charges were last updated and approved on January 26th, 2023 by the Board of Commissioners. For the new fiscal year, the newly calculated landing fee and terminal rental rate are based on the FY2023-2024 Airport System Requirement from the FY2023-2024 Operating Budget. Other rates have been updated based on forecasted revenue as shown in the Operating Budget and/or added to ensure costs are recovered by the airport. Ground and Facility Rental Rates are also updated regularly based on CPI and/or Fair Market Value.

A parking rate analysis was done to determine what the average daily parking rates were for all California Airports (small, medium, and large.) The analysis showed that our daily parking rates were slightly below the averages of our peers. The potential fiscal impact we project to be nearly \$5M per year. This figure takes the previous year's occupancy levels and factors in a 3% increase in lot utilization. Staff proposes that the Authority

approve the appropriate increases based upon the results of this analysis. To respond to changing market conditions, staff is also requesting authority for the CEO to amend parking rates up to a 15% difference from time to time, without necessity of an amendment to the Airport System Rates & Charges. Therefore, parking rates shown in the Airport System Rates & Charges shall be listed as the then-approved rates and the most updated and current parking rates shall always be listed at https://www.flyontario.com/parking/lots.

PROCUREMENT: N/A

CEQA COMPLIANCE AND LAND USE APPROVALS: Exclusion from the definition of "project": The creation of government funding mechanisms or other government fiscal activities which do not involve any commitment to any specific project which may result in a potentially significant physical impact on the environment. (CEQA Guidelines §15378(b)(4).)

STAFFING IMPACT (# OF POSITIONS): N/A

IMPACT ON OPERATIONS: N/A

SCHEDULE: N/A

ATTACHMENTS:

1. Resolution No. 2023-14

STAFF MEMBER PRESENTING: Chief Executive Officer Atif Elkadi

Originator Name:	Annie Lin
Originating Dept.:	Revenue Management
Director Review:	N/A
Chief Review:	Exlerg
CEO Approval:	$A^{2=}$

This Agenda Report has been reviewed by OIAA General Counsel.

The Agenda Report references the terms and conditions of the recommended actions and request for approval. Any document(s) referred to herein, which are not attached or posted online, may be reviewed prior to or following scheduled Commission meetings in the Office of the Clerk of the Board. Office hours are 8:30 a.m. to 5:00 p.m., Monday through Friday, although these hours and review procedures may be modified. In that case, the documents may be requested by email at <u>clerk@flyontario.com</u>.

BOARD DISPOSITION: Approved Denied Continued to _____

A RESOLUTION OF THE ONTARIO INTERNATIONAL AIRPORT AUTHORITY APPROVING THE AIRPORT SYSTEM RATES AND CHARGES FOR FISCAL YEAR 2023-24

WHEREAS, the Ontario International Airport Authority ("the Authority" or "OIAA") was established under a Joint Exercise of Powers Agreement between the City of Ontario and the County of San Bernardino (the "Joint Powers Agreement") pursuant to the purpose of operating, maintaining, developing, and marketing the Ontario International Airport (the "Airport"); and

WHEREAS, the Authority staff, including the Chief Executive Officer ("CEO") and the Chief Financial Officer ("CFO") of the Authority, have determined the Authority's Operating Budget for Fiscal Year 2023-24 in accordance with the provisions of the Joint Powers Agreement; and

WHEREAS, the Authority staff have consulted with the Airport and Airline Affairs Committee ("AAAC") in regards to the proposed operating budget and rates and charges schedule for Fiscal Year 2023-24, and the AAAC has not expressed any objections to its adoption or implementation; and

WHEREAS, the Authority Commission has reviewed the proposed updated Airport System Rates and Charges for Fiscal Year 2023-24 and has held a public meeting relative to its adoption.

NOW, THEREFORE, BE IT RESOLVED by the Ontario International Airport Authority Commission as follows:

SECTION 1. The Authority hereby approves the Airport System Rates and Charges, effective July 1, 2023, which is attached as Exhibit "B" and by this reference incorporated.

SECTION 2. Effective Date. This Resolution shall take effect immediately upon its adoption.

SECTION 3. Certification. The Secretary/Assistant Secretary shall certify as to the adoption of this Resolution.

PASSED, APPROVED, AND ADOPTED at a Regular Meeting this 22nd day of June, 2023.

ALAN D. WAPNER, OIAA PRESIDENT

ATTEST:

APPROVED AS TO LEGAL FORM:

NORMA I. ALLEY, MMC, ASSISTANT SECRETARY

LORI D. BALLANCE, GENERAL COUNSEL

STATE OF CALIFORNIA COUNTY OF SAN BERNARDINO CITY OF ONTARIO

I, Norma I. Alley, MMC, Board Clerk/Assistant Secretary of the Ontario International Airport Authority, DO HEREBY CERTIFY the foregoing Resolution No. 2023-14 is the original and was duly passed and adopted by the Commission of the Ontario International Airport Authority at their Regular Meeting held June 22, 2023, by the following roll call vote, to wit:

)

)

AYES: COMMISSIONERS:

NOES: COMMISSIONERS:

ABSENT: COMMISSIONERS:

NORMA I. ALLEY, MMC BOARD CLERK/ASSISTANT SECRETARY

(SEAL)

ONTARIO INTERNATIONAL AIRPORT AUTHORITY AIRPORT SYSTEM RATES AND CHARGES FY2023-2024 EXHIBIT "B"

ONTARIS

Туре	Unit	Sig	natory	Non	-Signatory
Landing Fees:					
Aircraft over 25,000 pounds	Per 1,000 lbs. MGLW	\$	1.83	\$	2.29
Aircraft over 12,500 pounds up to 25,000 pounds	Per Landing	\$	59.00	\$	74.00
Aircraft less than 12,500 pounds	Per Landing	\$	31.00	\$	38.00
Terminal Rental Rates:					
Terminal Space	PSFPM	\$	7.26	\$	9.08
Terminal Use Fees (Gate Use Charges):		-			
Non-Preferential Terminal Gate Use Fee	Per Operation	\$	344.87	\$	344.87
Ticket Counter Use	Counter per Hour	\$	1.30	\$	1.62
Joint Use Fees:					
Joint Use Fee	Per Deplaned		Based on Jo	oint U	se Formula
FIS USA FARS'	l assenger				
FIS Use Fee	Per Deplaned Passenger	\$	11.00	\$	11.00
Ramp and Apron Charges					
Each 15-minute period or fraction thereof after airline	Per 15 Minute Period	\$	100.00	\$	100.00
has been given notice by the airport that the aircraft					
leave the area					
Each 10-minute period or fraction thereof when	Per 10 Minute Period	\$	100.00	\$	100.00
aircraft is double-parked or in a position other than a					
regular gate position and the airline has been given					
notice to leave the area		-		•	
Each 15-minute period or fraction thereof in excess	Per 15 Minute Period	\$	100.00	\$	100.00
of 30 minutes for the cleanup of fuel spills					
Aircraft Parking Charges*		•		•	
Parking Rate Per Airplane (Diversions) - For each 24-	Per 1,000 lbs.	\$	0.80	\$	0.80
nr period after the first three nours the plane is on the	MGLW				
poriod					
Parking Rate Per Airplane (Non-Diversions) - For	Per 1 000 lbs	\$	0.40	¢	0.40
each 24-hr period the plane is on the ground. Any	MGI W	Ψ	0.40	Ψ	0.40
fraction thereof shall be a full 24-hr period	moen				
Ground Support Equipment Fee				1	
For every wide-body passenger aircraft operation	Per Use Per	\$	450.00	\$	450.00
diverted to ONT. This will go to a special fund to	Equipment	,		ŗ	
recover the cost of the AVIRAMPs.					
Passenger Facility Charge - (PFC)	-				
Passenger Facility Charge - (before \$0.11	Per Enplaned	\$	4.50	\$	4.50
administrative fee)	Passenger				

ONTARIO INTERNATIONAL AIRPORT AUTHORITY AIRPORT SYSTEM RATES AND CHARGES FY2023-2024 EXHIBIT B

Туре	Unit	Rates / Fees / Charges
Clerk's Office Fees		
Subpoena for Appearance	Per Day	\$ 280.00
Subpoena for Records	Per Subpoena	\$ 15.00
Public Records Request	Per Page	\$ 0.15
Public Records Request	Per USB Drive	\$ 10.00
Badging Fees		
New Badging/Fingerprinting	Each	\$ 70.00
Badge Renewal	Each	\$ 70.00
Badge Re-Issuance***	Each	\$ 30.00
First Lost or Stolen Badge Replacement	Each	\$ 100.00
Second Lost or Stolen Badge Replacement	Each	\$ 150.00
Unrecovered Badge Fee	Each	\$ 250.00
Parking Rates		
Lot 2		
Premium Parking Daily / Immediately in front	Per Dav	\$ 30.00
General Parking	Per Day	\$ 24.00
Lot 3	l ol Day	
Daily Parking	Per Day	\$ 20.00
Lot 4	1 of Duy	÷ 20.00
Premium Parking Daily / Immediately in front	Per Day	\$ 30.00
General Parking	Per Day	\$ 24.00
	I of Duy	ψ 27.00
Daily Parking	Per Day	\$ 18.00
	1 of Buy	•
Daily Parking	Per Day	\$ 18.00
Parking Rates shown above are listed as is during the	he time of approval of the	his Rates & Charges Exhibit Rates
may be subject to change from time to time without	amendment of this exh	ibit. Current rates are always listed
at https://www.flyo	ontario.com/parking/lots	S.
Emplovee Parking	<u> </u>	
Employee with SIDA Badge	Per Month	\$ 50.00
Employee without SIDA Badge	Per Month	\$ 175.00
Ground Transportation Permits/Agreements		
· · · · · · · · · · · · · · · · · · ·		
TNCs	Per Drop-Off /	\$ 4.00
	Pickup	
Charter Limousines/Sedan	Per Drop-Off /	\$ 3.00
	Pickup	
Charter Buses	Per Drop-Off /	\$ 3.00
Charter Vans	Per Drop-Off /	\$ 3.00
	Pickup	
Hotel/Motel Courtesy Van Annual Permit	Per Company	\$ 8,449.00
Off-Airport Parking Shuttles	Gross Revenues	12.0%
Taxi Cabs	Per Pick-Up	\$ 3.00
Customer Facility Charge - (CFC)		φ (a. c.
Customer Facility Charge	Per Transaction	\$ 10.00

Updated June 2023
ONTARIO INTERNATIONAL AIRPORT AUTHORITY AIRPORT SYSTEM RATES AND CHARGES FY2023-2024 EXHIBIT B

Туре	Unit	Rates / Fees / Charges
Annual Permits and Licenses		
Motor Vehicle Operating Permit	Annually Per Vehicle	\$ 100.00
Non-Exclusive License Agreement	Annually Per	\$ 500.00
Fuel Delivery Permit	Annually Per	\$ 1,000.00
Film-Related Fees		
General Fees		
Holding Fee (Amount goes towards deposit)	Up to 7 Days	\$ 250.00
Cancellation Fee (If cancelled 72 hrs prior to start)	Per Cancellation	\$ 500.00
Tech/Director Scouting Fee (after the first hour)	Per Hour	\$ 100.00
OIAA Facilities and Vehicles		
Facility Use	Per Hour	\$ 375.00
Facility Prep/Strike	Per Hour	\$ 312.00
Production Vehicles		
≥5 Ton Trucks/Trailers	Per Day	\$ 32.00
<5 Ton Trucks/Trailers	Per Day	\$ 19.00
Cast/Crew Personal Auto	Per Day	\$ 24.00
B727 Use		
View	Per Day	\$ 1,000.00
Exterior up to plane	Per Day	\$ 1,750.00
Interior/Exterior on plane	Per Day	\$ 3,900.00
Airport Personnel		
Film Desk Support Staff	Per Hour	\$ 100.00
Security Officer	Per Hour	\$ 50.00
Airport Police Sergeant	Per Hour	\$ 176.00
Law Enforcement Officer (LEO)	Per Hour	\$ 145.00
Electrician	Per Hour	\$ 70.00
Air Conditioning Mechanic	Per Hour	\$ 66.00
Instrument Mechanic	Per Hour	\$ 80.00
Communications Electrician	Per Hour	\$ 83.00
Operations Specialist	Per Hour	\$ 82.00
General Ground and Facility Rental Rates - Minimum**		
Unpaved Areas	PSFPM	\$ 0.16
Paved for Autos/Vehicles	PSFPM	\$ 0.20
Paved for Aircraft	PSFPM	\$ 0.25
General Building/Hangar Rate	PSFPM	\$ 0.33
Seasonal/Short-Term Use	PSFPM	\$ 0.23
Fuel Delivery Fees		
Aviation Gasoline	Per Gallon	\$ 0.05
Turbine Fuel	Per Gallon	\$ 0.05
Lubricants	Per Gallon	\$ 0.15
Property/Facilities Fees		
Cypher Lock Installation	Per Lock	\$ 500.00
Additional Tenant Keys	Per Key	\$ 20.00
Temporary Space Rental, Additional Fee	Per Day	\$ 100.00

Additional Clarifications:

- (a) <u>"FIS"</u> Federal Inspection Services facility, also known as the International Arrivals Terminal where Customs and Border Protection inspect arriving passengers from a foreign country.
- (b) <u>"Joint Use Fee"</u> The fee charged to an air carrier for Joint Use Space, based on the Joint Use Formula as defined in the Operating Use and Terminal Lease Agreement and Air Carrier Operating Permit.
- (c) <u>"Joint Use Formula"</u> The formula used to prorate eighty percent (80%) of the specified charge for Joint Use Space according to the ratio of the number of each airline's (both Signatory Airlines and Non-Signatory Airlines) deplaning passengers at the Airport during the most recent completed quarter for which such information is available to the total number of deplaning passengers of all airline users of the service or space during that same quarter. The remaining twenty percent (20%) is to be prorated equally among all airline users (both Signatory Airlines and Non-Signatory Airlines) of the space.
- (d) <u>"MGLW"</u> Maximum Gross Landing Weight; the FAA Certificated Maximum Gross Landing Weight or actual gross landing weight of an Aircraft if no such specification exists. In computing fees prescribed herein, except for Aircraft weighing less than 25,000 pounds, 500 pounds or any larger part of 1,000 pounds shall be counted as if a whole 1,000 pounds, and any smaller part shall be disregarded.
- (e) <u>"Non-Preferential Gate"</u> A gate that is not preferentially leased by a Signatory Air Carrier.
- (f) <u>"Non-Signatory Air Carrier"</u> A cargo or passenger carrier that has not entered into the Operating Use and Terminal Lease Agreement with the OIAA.
- (g) <u>"Operation"</u> Shall be the equivalent of one aircraft turn (one arrival and one departure).
- (h) <u>"Preferential Gate"</u> A gate that is preferentially leased by a Signatory Air Carrier.
- (i) <u>"PSFPM"</u> Per Square Foot Per Month.
- (j) <u>"Signatory Air Carrier"</u> A cargo or passenger carrier that has entered into the Terminal Operating Use and Terminal Lease Agreement with the OIAA.
- (k) <u>"Terminal"</u> Includes Terminal 1, Terminal 2, and Terminal 4, and the Federal Inspection Services Facility (International Arrivals Terminal).
- (I) <u>"Ticket Counter Use"</u> Use of a ticket counter with common use equipment. The air carrier leasing such counter shall be provided a credit based on the time used by the secondary air carrier, and the secondary air carrier shall be charged the fee based on the time used. The usage of such counter shall be rounded up to the nearest hour.
- (m) *<u>"Aircraft Parking Charges"</u> Includes all airport-controlled/non-preferential aircraft parking positions at Airport. If a Jet Bridge is used on gates 411-414, then the additional appropriate charges shall incur. If a Preferential Gate is utilized, then additional fees may be levied.

- (n) **<u>"Ground and Facility Rental Rates"</u> These are subject to CPI or Fair Market Value adjustments based on the language written in each Facility Use Agreement or Lease.
- (o) ***<u>"Badge Re-Issuance"</u> Adding badge icons, replacing a badge, and reprinting of a badge is considered a Badge Re-Issuance.